Susanne Grässel · Attila Aszódi Editors

Cartilage Volume 3: Repair Strategies and Regeneration



Cartilage

Susanne Grässel • Attila Aszódi Editors

Cartilage

Volume 3: Repair Strategies and Regeneration



Editors Susanne Grässel Orthopädische Klinik für die Universität Regensburg Universitätsklinikum Regensburg Regensburg Germany

Attila Aszódi Department of General, Trauma, and Reconstruction Surgery Ludwig-Maximilians-University Munich Germany

ISBN 978-3-319-53314-8 DOI 10.1007/978-3-319-53316-2

ISBN 978-3-319-53316-2 (eBook)

Library of Congress Control Number: 2017931314

© Springer International Publishing AG 2017

This work is subject to copyright. All rights are reserved by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

The publisher, the authors and the editors are safe to assume that the advice and information in this book are believed to be true and accurate at the date of publication. Neither the publisher nor the authors or the editors give a warranty, express or implied, with respect to the material contained herein or for any errors or omissions that may have been made. The publisher remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Printed on acid-free paper

This Springer imprint is published by Springer Nature The registered company is Springer International Publishing AG

The registered company address is: Gewerbestrasse 11, 6330 Cham, Switzerland

Preface

The closing, third volume of the Cartilage book series is dedicated to provide an overview about the current procedures of articular cartilage repair. The text is designed to be of use to multiple medical and basic science disciplines as orthopedics, rheumatology, and trauma surgery and all basic and clinical investigators working in the field of cartilage regeneration. This volume covers various cartilage repair strategies including cell-based, biomolecule-modulated, gene transfer, and tissue engineering approaches.

Chapter 1 provides an overview of the current status and future perspectives of cartilage repair strategies. The authors, after giving a short introduction to the clinical relevance and pathology of cartilage injuries, summarize the principles of currently practiced cartilage repair modalities including palliative approaches, strategies based on bone marrow stimulation, whole-tissue transplantation, and tissue engineering strategy. Recent advances of stem cell-based therapies and biomaterial scaffold designs in cartilage repair strategies are also presented. The chapter opens a discussion of the remaining scientific and clinical challenges in cartilage repair, specifically highlighting the need of enhancement of tissue integration, maintenance of cell phenotype, prevention of OA progress, and simplification of surgical and rehabilitation procedures.

The next four chapters give details for cell-based cartilage regeneration strategies. Chapter 2 focuses on adult mesenchymal stem cells (MSCs) and their growth factor-modulated surface marker expression. The authors clarify the characteristics and the embryonic origin of cartilage progenitor cells and summarize the contribution of MSCs from different origins to cartilage repair. Finally, a few examples of promoting articular cartilage phenotype by growth factor administration, in relation to the modulation of surface marker expression, are given. Chapter 3 introduces the chondrogenic progenitor cell (CPC), a specific cell type bearing stem cell characteristics such as migratory activity, clonogenicity, and multi-potency. These cells, which are present in osteoarthritic cartilage tissue and involved in regeneration processes, provide a promising alternative approach for cartilage repair. Various factors modulating the chondrogenic potential of CPCs including transcription factors, cytokines, growth factors, extracellular matrix molecules, and calcium homeostasis are presented. Chapter 4 demonstrates the attractiveness of induced pluripotent stem (iPS) cells for cartilage regeneration, which originates from their immense expandability and their intrinsic ability to give rise to stable hyaline cartilage. The

application of iPS cells can overcome most problems of classical cell-based regeneration strategies such as the extremely limited supply of human articular chondrocytes and the restricted differentiation capacity of mesenchymal stem cells from bone marrow or adipose tissue. Beyond being a potential alternative cell source for articular chondrocyte implantation, iPS cells are particularly promising for in vitro modeling of genetic diseases and for drug testing. Reprogramming patient-specific cells with a genetic predisposition and engineering disease-specific genetic variations into healthy control iPS cells promise to recapitulate "diseases in a dish" more realistically than immortalized human cell lines and will be an invaluable complementation for animal models. Whether iPS cells will satisfy these tremendous expectations will depend on our ability to upscale iPS cell culture, to derive sufficient amounts of relevant cell types like chondrocytes from iPS cells with acceptable efforts, and to find clinically safe reprogramming techniques for iPS cell-based therapies. Chapter 5 provides a short overview about current procedures for cellbased treatment strategies like bone marrow stimulation techniques, osteochondral transplantation, and autologous chondrocyte transplantation. Requirements and outcome parameters for a successful treatment and future directions in cartilage regeneration are discussed. Finally treatment recommendations according to cartilage defect size and depth are given.

The following two chapters deal with the role of biologic agents for the regenerative process of cartilage injury. Chapter 6 describes the growth factors with the most promising in vitro and in vivo data in cartilage repair, namely, bone morphogenetic protein-7, transforming growth factor- β , fibroblast growth factor-18, connective tissue growth factor, insulin-like growth factor-1, and recent advancements with autologous solutions of growth factors, such as platelet-rich plasma. Each section provides a background on mechanism of action, summarizes pivotal basic science research, and describes the results of clinical application in animal and human models of chondral disease. In chapter 7, platelet-rich plasma (PRP), an autologous blood-derived concentrate rich in growth factors, is introduced. Currently PRP is the most exploited biological approach for conservative management (simple intraarticular injections) and as an augmentation during surgical procedures. It has been applied both to treat osteoarthritis and chondral/osteochondral lesions in different joints, with the primary aim of providing symptomatic relief and functional recovery, and to induce a positive modulation of the entire articular microenvironment. The authors summarize the clinical evidence available on the role of PRP to treat cartilage pathology, focusing in particular on the data coming from randomized controlled trials.

Subchondral sclerosis is one of the hallmark findings of osteoarthritis (OA) and has long been discussed as one of its causes. Chapter 7 focuses on the changes in the *subchondral bone*, which often precede cartilage destruction in the development of the disease. Integration of the so far published data including in vitro, in vivo, and mathematical work suggests a critical role for this tissue in nutrition and oxygen supply to the articular cartilage, which may become even more critical in energy-demanding processes of healing and regeneration. Indeed, the success of current predictive diagnostics like specialized MRI techniques and scintigraphy as well as

successful regenerative clinical therapies like microfracturing, AMIC, or NAMIC can be better explained if the subchondral bone is taken into account as a supply route for the cartilage. Consequently the subchondral bone has to be included into the diagnostic and therapeutic concepts aiming to regenerate lost or damaged cartilage for advanced diagnosis and treatment of OA.

The next two chapters provide the concepts of gene therapy and tissue engineering for cartilage repair. Gene therapy protocols are well suited to deliver genes coding for therapeutic factors over time in a spatially defined manner within sites of cartilage injury resulting from acute trauma or during osteoarthritis. The focus of Chapter 9 is to examine the benefits of gene therapy to improve cartilage repair in such lesions, based on promising experimental and clinical evidence in relevant models in vivo using growth, transcription, and signaling factors capable of stimulating the chondrogenic and chondro-reparative processes locally. A continuous, combined effort between scientists and orthopedic surgeons may allow to bring gene therapy from encouraging data at the bench to a successful, safe translation in the broadly affected human population. Chapter 10 summarizes several promising options to engineer articular cartilage-like constructs, ranging from applying biological factors to mechanical, magnetic, or even electrical stimuli. The paradigm of cartilage tissue engineering classically comprises three pillars: cells, scaffolds, and signals. As cell sources for cartilage repair are addressed by other chapters in this volume, the author focuses on the two remaining pillars. First, due to their importance for the subsequent tissue engineering path, scaffold-free and scaffold-based applications are distinguished. Although most classical techniques in the field are scaffold-based, relatively more attention is now paid to emerging scaffold-free methods as articular cartilage repair constructs. Only proper tissue organization will permit long-term functional durability, and mimicking tissue growth without artificial support structures holds a lot of potential. While the extracellular matrix is an integral aspect of the tissue properties, it also impedes the integration of the repair construct into the surrounding host tissue. Several approaches to tackle this dilemma are depicted. The importance to develop bioreactors is also emphasized as they are inevitable for the reproducible application of sophisticated mechanobiological stimulation regimes. In this context, the contribution of selected growth factors is described. At the end of the chapter, the importance of integrating multiple of these parameters into multimodal concepts for achieving phenotypic stability of the engineered cartilage-like constructs is addressed.

Finally, Chapter 11 focuses on different *animal models* which play an important role to test novel experimental strategies and reconstructive surgical treatments of focal articular cartilage defects. Such animal models need to reflect the different appearances and etiologies of cartilage defects, e.g., caused by trauma or osteoar-thritis. Depth of articular cartilage defects plays an important role. Full-thickness chondral defects do not extend into the subchondral bone, while osteochondral defects penetrate the cement line and extend to the subchondral bone, thereby changing its structural integrity. Mice, rats, rabbits, goat, sheep, minipigs, and horses are representing good models, bridging the gap between in vitro studies and clinical experiments in human. Each of them has benefits and limitations. Evaluation

of cartilage repair may be performed using a large variety of methods, among which nondestructive evaluations and histological scoring, the latter being considered as the gold standard. As the available reconstructive surgical approaches for articular cartilage repair become increasingly complex, precise animal models to test and to translate new surgical techniques into appropriate clinical treatments are required.

Bringing together international experts from diverse fields of musculoskeletal research was a demanding task requiring patience and persistence. For that we are very grateful to the authors of this volume who managed to complete their chapters on time and who dedicated their spare time to writing their reviews.

Regensburg, Germany Munich, Germany Susanne Grässel Attila Aszódi

Contents

| 1 | Overview: State of the Art and Future Prospectives | 1 |
|----|--|-----|
| | Yangzi Jiang, Hang Lin, and Rocky S. Tuan | 1 |
| 2 | MSC Populations for Cartilage Regeneration | 35 |
| 3 | Chondrogenic Progenitor Cells and Cartilage Repair Diana Oellerich and Nicolai Miosge | 59 |
| 4 | Induced Pluripotent Stem Cells and Cartilage Regeneration Solvig Diederichs and Wiltrud Richter | 73 |
| 5 | Cell-Based Cartilage Regeneration Johannes Zellner and Peter Angele | 95 |
| 6 | Growth Factors in Cartilage Repair Maximilian A. Meyer, Atsushi Urita, Brian J. Cole, and Susanna Chubinskaya | 109 |
| 7 | The Role of Platelet-Rich Plasma in Cartilage Repair Elizaveta Kon, Giuseppe Filardo, Berardo Di Matteo, and Maurilio Marcacci | 127 |
| 8 | Subchondral Bone in Articular Cartilage Regeneration | 139 |
| 9 | Gene Transfer Strategies for Articular Cartilage Repair Magali Cucchiarini | 151 |
| 10 | Tissue Engineering Strategies for Cartilage RepairHolger Jahr | 169 |
| 11 | Animal Models in Cartilage Repair Lars Goebel, Liang Gao, and Henning Madry | 189 |

List of Contributors

Peter Angele University Medical Center Regensburg, Department of Trauma Surgery, Regensburg, Germany

Sporthopaedicum Regensburg, Regensburg, Germany

Pieter A.J. Brama Section of Veterinary Clinical Sciences, School of Veterinary Medicine, University College Dublin, Belfied, Dublin 4, Ireland

Susanna Chubinskaya Midwest Orthopaedics at Rush, Rush University Medical Center, Chicago, IL, USA

Mairéad A. Cleary Department of Orthopaedics, Erasmus MC, University Medical Center, Rotterdam, The Netherlands

Section of Veterinary Clinical Sciences, School of Veterinary Medicine, University College Dublin, Belfied, Dublin 4, Ireland

Brian J. Cole Midwest Orthopaedics at Rush, Rush University Medical Center, Chicago, IL, USA

Magali Cucchiarini Center of Experimental Orthopaedics, Saarland University Medical Center, Homburg, Saar, Germany

Solvig Diederichs Research Centre for Experimental Orthopaedics, Orthopaedic University Hospital Heidelberg, Heidelberg, Germany

Giuseppe Filardo Nano-Biotechnology Lab, Rizzoli Orthopaedic Institute, Bologna, Italy

Liang Gao Center of Experimental Orthopaedics, Saarland University Medical Center and Saarland University, Homburg, Saar, Germany

Cartilage Net of the Greater Region, Homburg, Saar, Germany

Lars Goebel Center of Experimental Orthopaedics, Saarland University Medical Center and Saarland University, Homburg, Saar, Germany

Cartilage Net of the Greater Region, Homburg, Saar, Germany

Department of Orthopaedic Surgery, Saarland University Medical Center and Saarland University, Homburg, Saar, Germany

Holger Jahr RWTH University Hospital Aachen, Department of Orthopaedic Surgery, Aachen, Germany

Yangzi Jiang Center for Cellular and Molecular Engineering, Department of Orthopaedic Surgery, University of Pittsburgh School of Medicine, Pittsburgh, PA, USA

Elizaveta Kon III Orthopaedic Division, Knee Joint Reconstruction Center, Humanitas Clinical Institute, Rozzano, Milan, Italy

Hang Lin Center for Cellular and Molecular Engineering, Department of Orthopaedic Surgery, University of Pittsburgh School of Medicine, Pittsburgh, PA, USA

Henning Madry Center of Experimental Orthopaedics, Saarland University Medical Center and Saarland University, Homburg, Saar, Germany

Cartilage Net of the Greater Region, Homburg, Saar, Germany

Department of Orthopaedic Surgery, Saarland University Medical Center and Saarland University, Homburg, Saar, Germany

Maurilio Marcacci III Orthopaedic Division, Knee Joint Reconstruction Center, Humanitas Clinical Institute, Rozzano, Milan, Italy

Berardo Di Matteo III Orthopaedic Division, Knee Joint Reconstruction Center, Humanitas Clinical Institute, Rozzano, Milan, Italy

Biomechanics Laboratory, Rizzoli Orthopaedic Institute, Bologna, Italy

Maximilian A. Meyer Midwest Orthopaedics at Rush, Rush University Medical Center, Chicago, IL, USA

Nicolai Miosge Georg August University, Tissue Regeneration Work Group, Department of Prosthodontics, Göttingen, Germany

Roberto Narcisi Department of Orthopaedics, Erasmus MC, University Medical Center, Rotterdam, The Netherlands

Diana Oellerich Georg August University, Tissue Regeneration Work Group, Department of Prosthodontics, Göttingen, Germany

Gerjo J.V.M. van Osch Department of Orthopaedics, Erasmus MC, University Medical Center, Rotterdam, The Netherlands

Department of Otorhinolaryngology, Erasmus MC, University Medical Center, Rotterdam, The Netherlands

Wiltrud Richter Research Centre for Experimental Orthopaedics, Orthopaedic University Hospital Heidelberg, Heidelberg, Germany

Arndt F. Schilling Klinik für Unfallchirurgie, Orthopädie und Plastische Chirurgie, Universitätsmedizin Göttingen, Göttingen, Germany

Kavitha Sivasubramaniyan Department of Orthopaedics, Erasmus MC, University Medical Center, Rotterdam, The Netherlands

Rocky S. Tuan Center for Cellular and Molecular Engineering, Department of Orthopaedic Surgery, University of Pittsburgh School of Medicine, Pittsburgh, PA, USA

Atsushi Urita Midwest Orthopaedics at Rush, Rush University Medical Center, Chicago, IL, USA

Johannes Zellner University Medical Center Regensburg, Department of Trauma Surgery, Regensburg, Germany

Overview: State of the Art and Future Prospectives for Cartilage Repair

1

Yangzi Jiang, Hang Lin, and Rocky S. Tuan

Abstract

Articular cartilage degeneration, for example, resulting from joint injury and trauma, has remained a major clinical challenge as cartilage does not have selfhealing capability, and osteoarthritis (OA) often ensues. OA affects over 15% of the population, including 65% of those above 65 years of age, and is a major cause of physical disabilities. There is thus a need to develop treatment strategies that can effectively target prevention and/or blockage of early stage disease progress, rather than prosthetic replacement of the joint at the end stage. This chapter provides an overview of the state of the art and future prospectives of cartilage repair strategies. The clinical relevance and tissue pathology of cartilage injury are first introduced, covering the structure and function of cartilage tissue and evaluation and clinical management of cartilage injuries. Next, the principles and strategies of currently practiced cartilage repair are summarized, including palliative approaches (e.g., arthroscopic debridement/lavage), intrinsic repair (e.g., bone marrow stimulation technique-abrasion, drilling, and microfracture), whole tissue transplantation (e.g., osteochondral graft transplantation), and tissue engineering strategy (e.g., autologous chondrocyte implantation/transplantation). An overview of recent advances in cartilage repair strategies is presented, particularly the progress made in stem cell-based therapies and biomaterial scaffold designs. The chapter concludes with a discussion of the remaining scientific and clinical challenges in cartilage repair, specifically highlighting the need of enhancement of tissue integration, maintenance of cell phenotype, prevention of OA progress, and simplification of surgical and rehabilitation procedures.

Y. Jiang • H. Lin • R.S. Tuan (\boxtimes)

Center for Cellular and Molecular Engineering, Department of Orthopaedic Surgery, University of Pittsburgh School of Medicine, Pittsburgh, PA, USA e-mail: yaj6@pitt.edu; hal46@pitt.edu; rst13@pitt.edu

[©] Springer International Publishing AG 2017

S. Grässel, A. Aszodi (eds.), Cartilage, DOI 10.1007/978-3-319-53316-2_1

1.1 Introduction

1.1.1 Clinical Relevance and Needs of Cartilage Repair

Cartilage is the load-bearing surface of the synovial joint, or diarthrosis, which is generally considered a tissue with simple structure, as it is avascular and hypocellular and consists primarily of extracellular matrix (ECM) and a small amount of chondrocytes. The principal function of cartilage is to provide weightbearing and mechanical support. However, cartilage has limited self-repair ability. Injury in cartilage often represents the initiation of joint degeneration and eventually leads to degenerative joint diseases, such as osteoarthritis (OA). OA is the most common cause of loss of mobility in elderly adults, and it has profound social, physical, psychological, and economical consequences (Callaghan 2003). Taking the United States as an example, OA affects 27 million adult Americans, including 65% of those above 65 years of age (Goldring 2006; Lethbridge-Cejku et al. 2003), and directly contributes to 9-10% of the disabilities (Felson 2004). There is no cure for OA, and osteoarthritic patients suffer from chronic pain and limited joint movement, distress and depression, and lost productivity. With the increasing aging demographics, OA is thus recognized as a significant global burden with clearly unmet clinical needs (World Health Organization 2002).

OA of the knee and hip joints are major causes of mobility impairment. The risk factors of OA include both intrinsic or progressive systemic factors, as well as factors that affect joint local mechanical environments (Fig. 1.1). Gender, family history, developmental joint growth, and shape abnormality are considered the intrinsic systemic factors, while progressive systemic factors, such as aging, hormonal status, nutrition, and lifestyle, vary among individuals. These systemic risk factors are related to the susceptibility to OA, but insults to the joint local mechanical environment, caused by overload (e.g., as a result of obesity), repetitive joint loading (e.g., in elite athletes), injury and trauma (e.g., from accident), or instability (e.g., resulting from joint surgeries), directly harm articular cartilage.

Current clinical OA management is mainly concerned with symptom reduction, e.g., pain, swelling, and stiffness, with oral nonsteroidal anti-inflammatory drugs (NSAIDs) being the most commonly used pharmacological treatment at mid-stage of the disease, and arthroplasty, an irreversible procedure, as the final solution to maintain joint function (Fig. 1.2). Consequently, it is highly desirable to develop treatment strategies that target the prevention and/or blockage of the progress of diseases in the early stage, rather than replacement of the joint at the end stage. Therefore, treatments that aim to repair cartilage defects have been under active investigation and are expected to provide more treatment choices to OA patients.



Fig. 1.1 Risk factors of osteoarthritis (OA) and the relationship between cartilage injury and OA



Fig. 1.2 Current OA clinical management and cartilage repair interventions. ACI autologous chondrocyte implantation, MACI matrix-associated autologous chondrocyte implantation

1.2 Cartilage Structure, Function, and Injuries

1.2.1 Cartilage Structure

Articular cartilage is the matrix-rich tissue that covers the surface of joints. With a thickness of 2–8 mm, cartilage is maintained by chondrocytes, which make up less than 10% of the tissue volume, whereas over 90% of the tissue is constituted by a unique cartilaginous extracellular matrix (ECM). In the articular cartilage, chondrocytes are distributed into four zones: (1) superficial zone, ~10–20% of the thickness, consisting of a thin layer of small, flattened chondrocytes arranged parallel to the surface; (2) middle or transitional zone, ~ 40–60% of the thickness, with chondrocytes that are spherical and separated; (3) deep or radial zone, ~30% of the thickness, ness, consisting of large chondrocytes that form columns perpendicular to the surface; and (4) calcified zone, in which chondrocytes are hypertrophic and the matrix is calcified, representing a transition to subchondral bone (Fig. 1.3).



Fig. 1.3 Schematic diagram for longitudinal section of articular cartilage. *Left*, cell distribution; *right*, ECM component and distribution. *CILP* cartilage intermediated layer protein, *ALP* alkaline phosphatase

The major components of the ECM are collagen and proteoglycans (PGs). Collagen forms cross-banded fibrils, fibrils form meshes, and PGs are trapped in the mesh to hold water, responsible for the compressive strength of the cartilage ECM. In addition to collagen and PGs, other minor ECM components are localized to specific zones of the articular cartilage: (1) in the superficial zone, PGs that reduce friction are found, e.g., lubricin, which protects chondrocytes and cartilage surfaces, as well as inhibits synovial cell overgrowth (Rhee et al. 2005; Waller et al. 2013); decorin, biglycan, and tenascin C are also found in the superficial zone; (2) ECM proteins that are associated within the middle zone consist of collagen type II (Col2), aggrecan (ACAN), and cartilage intermediated layer protein (CILP); (3) in the deep zone, a zonal-specific protein, osteopontin, has been identified; and (4) in the calcified zone, collagen type X (Col10) and alkaline phosphatase (ALP) are found associated with chondrocyte hypertrophy and in the calcified matrix environment (Fig. 1.3) (Grogan et al. 2013).

The ECM within the articular cartilage may also be designated into three regions based on proximity to the chondrocyte—the pericellular, territorial, and interterritorial regions (Heinegard and Saxne 2011). Matrix content and the diameter of collagen fibrils differ in these regions. Collagen type VI (Col6) is deposited specifically in the chondrocyte pericellular region (Zhang 2015), whereas large collagen fibrils and the majority of PGs reside in the territorial and interterritorial regions. In terms of orientation, collagen fibrils are oriented mostly parallel to the surface in the superficial zone, obliquely in the middle zone, and perpendicular to the joint surface in the deep zone, which is suited to load transmission (Johnson and Pedowitz 2007).

Cartilage is classified into hyaline cartilage, fibrocartilage, and elastic cartilage, each with a unique ECM composition. Hyaline cartilage, which contains high amounts of Col2 and PGs, is formed during embryonic skeletogenesis, as an integral part of the endochondral skeleton in the form of the growth plate and articular cartilage, as well as in other tissues, such as respiratory tract cartilage, temporomandibular disc, etc. Fibrocartilage contains collagen type I (Col1), Col2, and PGs and is found largely in the intervertebral disc, pubic symphysis, and meniscus. Elastic cartilage contains elastic fibers (elastin), collagens, and PGs (Williams 2007) and is found mostly in the outer ear, Eustachian tube, and epiglottis. In this chapter, emphasis will be on articular cartilage which, with its poor self-healing ability, is a tissue of significant clinical relevance.

1.2.2 Cartilage Injury

Injury of cartilage may happen in different joints, usually initiated by direct microtrauma or trauma, or concomitant with other injuries to joint tissues, such as meniscus or anterior crucial ligament, leading to joint instability and axial malalignments. The location and severity of injury and the size of the defect are crucial for the selection of potential treatments. For example, defects in a relatively non-load-bearing area may not need immediate treatment, while treatments of concomitant injuries should co-address the damages to the meniscus, subchondral bone, and ligament/ tendon and correct the axial malalignments to achieve optimal repair cartilage. Cartilage defects can be detected by a number of methods. For example, the integrity of cartilage and subchondral bone can be observed by noninvasive magnetic resonance imaging (MRI) (Williams 2007), as well as via arthroscopic inspection. The severity of cartilage defects has been codified by the International Cartilage Repair Society (ICRS), which has provided a cartilage injury evaluation package to inform clinical decision-making (Mats Brittberg et al. 2000). In addition to the location and the size of the defect (Fig. 1.4), five levels of articular cartilage defects are defined: (*Level 1*) normal and intact articular cartilage superficial lesions; (*Level 2*) softening of cartilage surface and/or superficial fissures and cracks of cartilage surface; (*Level 3*) abnormal defects occupying <50% of cartilage depth; (*Level 4*) severely abnormal defects occupying >50% of cartilage depth, down to calcified zone, or down to the subchondral bone, or with blisters; and (*Level 5*) severely abnormal defects penetrating the subchondral lamella or subchondral bone (Fig. 1.4).

1.2.3 Clinical Outcomes of Treatments

Clinical outcomes of treatment of cartilage degenerative conditions are influenced by multiple factors, such as the general health status of the patient, the severity of the lesion, the specific treatment procedure, as well as the rehabilitation regimen. Some of the commonly used evaluations of joint health include physical examination, monitoring activities of daily living, kinematic assessment of joint functions, medical imaging such as magnetic resonance imaging (MRI) and computed tomography (CT), arthroscopic examinations, and tissue biopsy. Information collected from these procedures is used to form the basis of evaluation, such as described in Table 1.1.

1.3 Current Cartilage Repair Strategies

Current clinically available cartilage repair treatments may be classified into five different levels, and the decision-making is generally based on the severity of the injury, lesion size, patient health status, history, and preference and whether the purpose is to address the outcomes of a previous surgical intervention.

1.3.1 Palliative Approach: Arthroscopic Debridement/Lavage

The goal of the palliative approach is removal of the mechanically offensive tissue sources and the dead, damaged, and inflammation-affected tissues by debridement and lavage, to achieve pain relief. This symptomatic therapy can be effective for several months and occasionally a few years; however, it does not result in filling of cartilage lesions.





| Tissue repair/regeneration ^a | 1. Palliative | 2. Bone marror techniques | w ^a stimulation | 3. Whole tissue transf | olantation | 4. Tissue engineering ^a |
|---|--------------------------------|---|--|--|--|--|
| Interventions | Debridement/lavage | Drilling | Microfracture | Osteochondral autograft | Osteochondral allograft | Autologous chondrocyte implantation (ACI/ MACI) |
| Demand | Low | $\mathrm{High} \to \mathrm{Low}$ | | High | High | High |
| Lesion size (cm^2) | 0-4 | (0.5-2)-(4-6) | | 1-5 | 3-10+ | 2-12+ |
| Known durability | <2 years | 2-6 years | | 5-20 years | 2-10 years | >20 years |
| Filling tissue | I | Fibrocartilage | | Hyaline cartilage | Hyaline cartilage | Hyaline-like |
| Filling scale | Ι | Partial | | Near total | Near total | Near total |
| Limitation | Temporary effect, no repair | Incomplete del of subchondral quality of repa | fect filling, injury I bone plate, low ir tissue | Limited availability of healthy tissue; secondary injury | Limited availability; immune response; potential necrosis | Limited cell number; two-stage procedure |

 Table 1.1
 Summary of current treatment strategies for cartilage injuries

^aInvolving endogenous tissue repair/regeneration; ^trefers to other sizes beyond as indicated (Williams 2007; Chahal et al. 2013)

1.3.2 Intrinsic Repair: Bone Marrow Stimulation Technique— Abrasion, Drilling, and Microfracture

Cartilage defect has been shown to exhibit a healing response, particularly when associated with bone marrow or other neighboring tissues. The initiation of such a healing process is dependent on the depth of the lesion site and whether the defect is in communication with the bone marrow. When the lesion of the articular cartilage is located only in the superficial zone to middle/deep zone, termed as partialand full-thickness cartilage defects, the defects generally do not exhibit any intrinsic healing. On the other hand, osteochondral defects, which violate the tidemark and penetrate into the subchondral bone and thus connect the defect to the bone marrow cavity, exhibit limited but definitive reparative capacity, most likely because the tissue defect permits bone marrow and marrow stromal cells to infiltrate into the lesion site to form a stem cell-rich fibrin clot to stimulate intrinsic repair (Jiang et al. 2010, 2011). In clinical practice, this intrinsic reparative property is exploited in the microfracture technique by drilling into the subchondral bone, to treat small size cartilage defect (usually 0.5-2 cm²) (Mithoefer et al. 2005; Williams 2007). This approach, referred to as marrow stimulation, provides partial fill to the defects with fibrocartilage, and the known durability is 2-6 years. The marrow stimulation technique is thus safe and readily applicable in clinical practice and cost-effective as a first-line treatment for grade 3 or 4 lesions (Pestka et al. 2012; Knutsen et al. 2007). However, it is not applicable for larger defects with higher mobility demands, such as in young athletes, and the quality of the repaired tissue is unpredictable and variable, thus affecting the effectiveness of the second-line treatments, such as autologous chondrocyte implantation (see below), after the first microfracture treatment has failed (Pestka et al. 2012) (for details, see Chap. 5).

1.3.3 Whole Tissue Transplantation: Osteochondral Graft Transplantation

For those full-thickness defects larger than 2 cm^2 or associated with osteochondritis dissecans, transplantation of healthy cartilage and subchondral bone is one of the options. The result has been proven useful in clinical practice, and the source of the transplant tissue may be from the same person (autologous) or from other donors (allogeneic). This whole tissue transplantation approach is the biologic reconstructive method to salvage the failed first-line less invasive treatments.

In autologous osteochondral transplantation, or mosaicplasty, one or more osteochondral cylinders are collected from less weight-bearing areas of the joint surface or, less frequently, from the contralateral knee joint and then inserted singly or as multiple plugs into the defect area. This method thus creates a mosaic hyaline cartilage surface and may be considered as first-line treatment for patients of high mobility demand and with smaller lesions, but is generally limited by the availability of structurally and functionally adequate donor graft tissues (for details, see Chap. 5).

In practice, both fresh and cryopreserved frozen allografts are used. Fresh allografts provide higher cell viability, but host-graft immune response and possible disease transmission are possible concerns, whereas frozen allografts have lower viable cell yield. In comparison, as there is no donor site morbidity, allograft-based cartilage resurfacing technique has the longest history with good long-term results (Williams 2007; Sherman et al. 2014). However, the availability of allograft specimens is also limited, and the cost is significantly higher than autografts.

1.3.4 Tissue-Engineered Cartilage

To develop alternative methods of tissue transplantation, tissue engineering/regenerative medicine approaches have been explored and developed rapidly since the 1970s. The aim of tissue engineering is to replace diseased or lost tissue/organ with an engineered tissue, derived from cells, biomaterial scaffolds, and bioactive factors, for the purpose of restoring or establishing normal function (Mason and Dunnill 2008). The three key elements of tissue engineering are tissue-forming cells, structural scaffolds, and signaling molecules, the combination and application of which result in a functional tissue construct to promote tissue healing and regeneration (Kuo et al. 2006).

1.3.4.1 Autologous Chondrocyte Implantation/Transplantation (ACI/ ACT)

The concept of using autologous periosteal grafting to treat cartilage defects was first proposed for the repair of full-thickness defect in a rabbit model (O'Driscoll et al. 1986). Cultured autologous chondrocytes were later used in addition to the periosteum (Grande et al. 1987, 1989). In the early 1990s, ACI was applied for the first time in clinical practice to treat full-thickness chondral defects in the knee joint by the Peterson group (Brittberg et al. 1994), and a subsequent 10–20-year follow-up showed that 74% in 224 patients reported their status being as good or better compared to pre-ACI years (Peterson et al. 2010). At present, the autologous chondrocyte-based therapies are by far one of the most successful examples in the clinical application of the cell-based tissue engineering principle to achieve tissue regeneration.

The ACI technique is more complicated than microfracture or osteochondral transplantation, as it requires two separate surgeries and an intermediate step of in vitro cell culture. In brief, in the first surgery, a very small amount (~250 mg) of healthy cartilage is collected under arthroscopy from a presumably non-load-bearing area of the articular joint surface, and chondrocytes are isolated and culture expanded in the laboratory for ~3–6 weeks. The culture-expanded cells are then sent to the operating room and introduced into the cartilage defect area and sealed with a sutured autologous periosteal flap. With ACI, hyaline-like cartilage tissue is usually observed in the defects within a year, and this procedure can be used for the repair of larger defects ($\geq 10 \text{ cm}^2$). The overall therapeutic efficacy is 70–90% in 0–5 years, as evidenced by relief of symptoms and improvement of joint function

(Jiang et al. 2011). The therapeutic effects have been reported to last longer than the marrow stimulation technique (Zaslav et al. 2009) and also can be used for osteochondritis dissecans (OCD) knee lesions (Cole et al. 2012) (for details, see Chap. 5).

With advances in biomaterial development, the ACI technique has also evolved accordingly. Biomaterials have been used instead of autologous periosteal flap covers, thus eliminating the extra open injury sites of tissue harvesting and shortening the operation time. The biomaterials used are usually with high biocompatibility (e.g., membranes of collagen types I and III, and the cells are seeded on the biomaterials and delivered as a single piece of tissue-engineered construct. This technique is now referred to as matrix-associated autologous chondrocyte implantation (MACI) or tissue-engineered cartilage (TEC) transplantation (Jiang et al. 2011).

1.3.5 Total Joint Replacement

When the injury and disease are judged to have reached end stage, the final solution is to replace the damaged joint with a prosthesis that is often constructed from metal (e.g., cobalt-chromium or titanium-based alloys), plastic (e.g., ultrahigh molecular weight polyethylene), and/or ceramics. Total joint arthroplasty (TJA), generally considered as one of the most successful medical procedures, has a long clinical history, and the prostheses employed, e.g., for the knee and hip, are designed to replicate the movement of the normal joint. The most common complications to these otherwise highly effective but irreversible procedures include septic or aseptic periprosthetic osteolysis, the latter most likely caused by the accumulation of wear debris from the prosthesis, thus limiting the functional life expectancy of the prosthesis to less than 10–15 years (Tuan et al. 2008). For younger TJA patients (<40–50 years of age), a revision procedure is generally expected, thus presenting additional surgical challenges.

In Table 1.1, a summary of the current treatment strategies for cartilage injuries and concerns for decision-making in cartilage repair procedures is presented.

1.4 New Strategies Under Investigation

Although the outcomes of ACI/MACI have been mostly favorable, there are also significant limitations (Jiang et al. 2011), including the low number of harvested cells and potential leakage and uneven distribution of cells in the defect (Kim et al. 2010). ACI/MACI is also a long and complicated procedure with multiple steps, and the repair tissue is only hyaline-like but not completely hyaline cartilage, as collagen type I, a hallmark of fibrocartilage, can be found in the repair tissue. In addition, many patients with chronic OA may be unable to get the ACI treatment because of the lack of healthy donor tissue. To overcome these limitations, contemporary research efforts have actively focused on the optimization of cell sources,

identification and application of growth factors, and development of scaffolds with improved biomimetic properties. The following is an overview of the progress in these endeavors.

1.4.1 Cell-Based Therapies

1.4.1.1 Stem Cells

Stem cells present the most promising candidate cell type for cartilage tissue repair because of their self-renewal ability and chondrogenic potential-the ability to form cartilage. Stem cells in mammals may be divided into two broad types: embryonic stem cells (ESCs) and adult stem cells. ESCs are pluripotent stem cells derived experimentally by extraction from the inner cell mass of a blastocyst, while adult stem/progenitor cells are found in different adult organs and tissues. Pluripotency refers to the potential of a stem cell to differentiate into cells of all three germ layers-endoderm, mesoderm, and ectoderm. In addition to ESCs, induced pluripotent stem cells (iPS cells or iPSCs) have recently been generated from adult somatic cells by transcription factor gene-based reprogramming and exhibit similar pluripotency. In comparison, adult stem cells are multipotent, i.e., they can be induced to differentiate into the cell types that are closely related in developmental origin (e.g., from the same embryonic germ layer). Finally, progenitor cells are developmentally committed cells, with the ability to differentiate into only one or few cell types, referred to as exhibiting uni- or oligopotency (for details, see Chaps. 2 and 4).

Stem Cell Types Relevant to Cartilage Repair

Embryonic Stem Cells (ESCs) and Induced Pluripotent Stem Cells (iPSCs) Because of their highly uncommitted state, the pluripotent ESCs and iPSCs present more challenges in terms of being induced to differentiate into cartilage cells, compared to multipotent stem cells, such as adult stem cells. Generally, an additional step, for example, first turning pluripotent stem cells into a mesodermal lineage, is required (Diederichs and Tuan 2014), whereas mesoderm-like multipotent stem cells, such as bone marrow-derived mesenchymal stem cells, may be directly induced to undergo chondrogenic differentiation. Both ESCs and iPSCs have been shown to have chondrogenic capability in vitro (Yamashita et al. 2010; Hiramatsu et al. 2011) and in animals (Wakitani et al. 2004a; Toh et al. 2010; Dattena et al. 2009; Yamashita et al. 2013). However, some unsolved problems remain, e.g., differentiation efficacy, because not all of the cells contribute to hyaline cartilage formation (Dattena et al. 2009), and only selected clones/cell lines have been reported with cartilage matrix-forming ability (Yamashita et al. 2013). Another issue of concern is safety, in view of potential tumor formation and immunological rejection of allogeneic cells when allogeneic ESCs or iPSCs are introduced in vivo. More standardization and safer methods are clearly needed to bring these pluripotent stem cells into clinical use (for details, see Chap. 4).

Mesenchymal Stem Cells (MSCs) MSCs are currently the most promising therapeutic cells for cartilage repair, in view of their more defined and ready chondrogenic potential compared to ESCs/iPSCs and their increasingly apparent beneficial trophic bioactivities, such as anti-inflammatory activity (Tuan et al. 2003). MSCs can be isolated from bone marrow, adipose tissue, muscle, bone, synovium, and other adult tissues. Among cells derived from diverse tissue sources, autologous bone marrow-derived MSCs (BMSCs) hold the best potential of application because of their ready availability and high chondrogenic efficiency. BMSCs produce higher amount of cartilage matrix and promote higher cartilage recovery than MSCs isolated from periosteum, synovium, adipose tissue, and muscle (Li et al. 2011b). Synovium-derived MSCs (Vinardell et al. 2012), adipose MSCs, and muscle MSCs (Li et al. 2011b) exhibited chondrogenesis potential but also fibrous tissue formation. Clinical application of BMSCs for cartilage repair was reported by several groups (Teo et al. 2013; Kasemkijwattana et al. 2011; Wakitani et al. 2004b), and comparable results using BMSCs and ACI were reported at 2 years post-surgery (Nejadnik et al. 2010). Longer follow-up is needed to verify the level of efficacy.

Adipose-derived stem cells (ADSCs) are another promising, candidate MSC type, especially in terms of their availability. Adipose tissues contain over 100-fold of mononuclear, stem-like cells per volume than bone marrow aspirates (Pendleton et al. 2013). However, their chondrogenic potential appears to be lower compared to BMSCs (Hildner et al. 2011) and requires different chondroinductive factors, e.g., BMP6 (Estes et al. 2006; Diekman et al. 2010). Improving the cartilage-forming ability of these cells is thus required and is one of the challenges for their application for articular cartilage repair (for details, see Chap. 2).

MSCs isolated from other tissues surrounding the joint, such as synovium and periosteum, are also being investigated for articular cartilage repair (Koga et al. 2008). A summary of the in vitro and animal research models used for stem cell-based cartilage repair studies is presented in Table 1.2.

Tissue-Specific Stem Cells In many adult tissues (e.g., muscle), endogenous stem/ progenitor cells are the cells that function to maintain tissue homeostasis, an inherent activity of tissue repair mechanism. Articular cartilage has traditionally been considered a physiologically self-renewing tissue. Interestingly, cartilage stem/progenitor cells (CSPCs) have recently been found within human articular cartilage (Williams et al. 2010; Quintin et al. 2010; Fickert et al. 2004; Alsalameh et al. 2004; Jiang and Tuan 2015). These cells are being characterized as stem/progenitor cells in vitro by virtue of their self-renewal ability, multipotent differentiation capability, and expression of stem cell markers. However, there is currently no single marker that could define CSPCs in vivo, and their derivation, fate, and lineage information are still largely unknown.

Chondrocytes in long-term culture are known to lose their cartilage matrix-forming ability, a process known as dedifferentiation, a process that limits ACI application; however, the discovery of the phenotypic plasticity of human articular chondrocytes (Tallheden et al. 2003) suggested a possible origin of tissue-specific stem/progenitor

| Usage | Evaluation | Characteristics | Tools | Reference |
|----------------------|--|---|---|---------------------------------------|
| Regular | Short Form (SF)-36 | Patient-reported survey, 36 items | - | Towbin et al. (1989) |
| OA | Kellgren and Lawrence Grading system (K&L) | OA pathology, indicated by radiographic changes (e.g., joint space narrowing, osteophytes, and bony sclerosis) | X-ray | Kellgren and Lawrence (1957) |
| | Criteria of American College of Rheumatology (ACR) | Includes laboratory tests such as synovial fluid observation, white blood cell count, and erythrocyte sedimentation rate | Lab tests | Moskowitz (2007) |
| | The Western Ontario and McMaster Universities Osteoarthritis Index (WOMAC) | Survey assesses pain, stiffness, and physical function in patients with hip and/or knee OA | _ | Kornfeld and Kornfeld (1985) |
| Joint function | Tegner Lysholm Knee Scoring Scale (Lysholm) | Knee function | - | Tegner and Lysholm (1985) |
| | International Knee Documentation Committee (IKDC) Score | Patient-reported score | _ | Akagi et al. (2002) |
| | Activities of Daily Living (ADL) Score | Daily self-care activity and life quality | - | Irrgang et al. (1998) |
| | Modified Cincinnati Score (Cincinnati) | Function, pain, symptoms, (function in | _ | Noyes et al. (1983) |
| | Knee Injury and Osteoarthritis Outcome Score (KOOS) | daily living, function in sport and recreation, and knee-related quality of life) | | Roos et al. (1998) |
| Cartilage defects | International Cartilage Repair Society (ICRS) Cartilage Injury Evaluation Package | Includes doctor score and patient score. Minimally invasive surgical procedure by arthroscopy, locating defects and intra- articular cartilage surface position and repair at the same time | Arthroscopy, histology, etc. | Gibson et al. (1998) |
| | Magnetic Resonance Observation of Cartilage Repair Tissue (MOCART) System | Noninvasive location of defects in cartilage and subchondral bone | Magnetic resonance imaging (MRI) | Marlovits et al. (2004) |

Table 1.2 Cartilage and joint health evaluations

cells in adult cartilage. Recently, a subgroup of stem/progenitor cells is found in chondrocytes isolated from adult articular cartilage, termed chondrocyte-derived progenitor cells (CDPCs), identified by their demonstrated dynamic change of pheno-type from that of mature chondrocytes toward that of MSCs (Jiang et al. 2016) (for details, see Chap. 3). Upon low density and low glucose in vitro expansion, CDPC number was enriched in a shorter period of time, and these cells showed enhanced chondrogenic capability compared to human BMSCs. The CDPCs were tested in a MACI procedure in place of chondrocytes for the repair of large-sized cartilage defects (6–13 cm²) in young patients, and 1-year follow-up showed pain reduction, joint function improvement, and hyaline cartilage filling (Jiang et al. 2016). The existence of CDPCs in articular cartilage suggests a novel approach to cartilage repair and provides a new view of the phenomenon of chondrocyte dedifferentiation that illustrates the importance of extrinsic factors that regulate cell fate in cartilage.

Signaling Factors: Stem Cell Differentiation and Cell Phenotype Maintenance

Knowledge gained from studies of embryonic development has revealed that skeletogenic differentiation of mesenchymal cells in vivo is an intricate process regulated by a variety of signals that act coordinately and sequentially at different stages of development to guide the formation of the hyaline cartilage and to maintain its differentiated state and function of the chondrocytes. Specifically, signaling factors, such as members of the transforming growth factor β (TGF β) superfamily, are required. During embryonic development, chondrogenesis in the primordial limb bud begins with mesenchymal cell recruitment, proliferation, and condensation. Several growth factors and signaling molecules act in concert in the initiation of cell condensations, including TGF β family members, such as the bone morphogenetic proteins (BMPs), fibroblast growth factor (FGF), and Wnt signaling. Specific extracellular matrix molecules are also involved in defining and enabling the transition from chondroprogenitor to chondrocytes; for example, fibronectin, hyaluronan, and collagen type I are involved in modulating cell-cell interaction during mesenchymal condensation, mediated via N-cadherin (Goldring et al. 2006; Shum et al. 2003; Wezeman 1998; DeLise et al. 2000; Gehris et al. 1996; Oberlender and Tuan 1994a, b) by regulating a key chondrogenesis-specific transcription factor, Sox9 (Akiyama and Lefebvre 2011).

As presented earlier, compared to the multipotent MSCs, ESCs/iPSCs are pluripotent cells; thus, deriving cartilage tissue from ESCs/iPSCs is more complicated than from MSCs, because they need to undergo additional commitment and differentiation steps. In general, differentiating ESC/iPSCs into cartilage involves a twostep differentiation strategy, combining culture conditioning and growth factor induction. ESCs/iPSCs are first differentiated into a multipotent state, followed by differentiation along the chondrocytic lineage (Hwang et al. 2008). This differentiation scheme is guided by the knowledge on embryonic mesodermal commitment, with the second step derived from known pathway of MSC chondrogenic differentiation. A number of growth factors have been shown to regulate MSC proliferation and chondrogenic differentiation, including members of the TGF β superfamily, insulin-like growth factors (IGFs), FGFs, and platelet-derived growth factors (PDGFs) (Heng et al. 2004; Bobick et al. 2009; Danisovic et al. 2012; Freyria and Mallein-Gerin 2012), and are being applied to promote chondrogenic differentiation of MSC-like cells derived from ESCs and iPSCs.

Among these growth factors, TGF β s and BMPs are the most important inducers of chondrogenic differentiation. In particular, BMP-2 and BMP-7 (also known as OP-1) have been approved by the FDA for clinical application (Alaoui-Ismaili and Falb 2009). Interestingly, MSCs from different tissue origins exhibit different growth factor requirements for chondrogenesis. For example, TGF β 1 and TGF β 3 are used to promote chondrogenesis of bone marrow-derived MSCs (Barry et al. 2001; Byers et al. 2008; Huang et al. 2009), whereas BMP6 is required for efficient stimulation of chondrogenesis with TGF β s for adipose-derived MSCs (Estes et al. 2006; Hennig et al. 2007), and BMP4, in combination with TGF β s, enhances the cartilage matrix-forming ability of bone marrow or cartilage-derived cells (Jiang et al. 2010, 2016). Different growth factor receptor repertoires likely contribute to the different biological characteristics of the MSCs derived from different adult tissue sources (for details, see Chap. 6).

1.4.1.2 Allogeneic Chondrocytes

Similar to allogeneic osteochondral transplantation, the use of allogenic chondrocytes is of potential for cartilage repair, in particular if they may be derived from ESCs and iPSCs. Previous studies have shown that pure allogeneic chondrocyte/ tissue implantation evoked a systemic immunological reaction followed by rejection by the host (Moskalewski et al. 2002; Revell and Athanasiou 2009). A potential solution is the use of biomaterials as a vehicle for allogeneic chondrocyte delivery for cartilage repair, which may blunt the immune reaction (Fragonas et al. 2000). A clinical trial of 21 patients reported that patients implanted with allogenic chondrocytes cultured in alginate beads and delivered into osteochondral lesions showed apparent functional improvement, with a 19.5% failure rate at mean follow-up of 6.1 years, but without any signs of clinical deterioration or adverse reactions to the alginate beads/allogenic chondrocyte implantation (Almqvist et al. 2009; Dhollander et al. 2012). Such an approach may present practical possibilities for the introduction of different types of allogeneic cells for the repair of articular cartilage defects.

1.4.2 Biomaterial Scaffolds for Cartilage Repair

1.4.2.1 Current Biomaterials and 3D Scaffolds

In tissue engineering/regenerative medicine, biomaterial scaffolds are employed to supply mechanical support and cell growth niche, as well as deliver cells/signaling factors into the tissue defect sites to enhance repair. The scaffold design must meet the combined demands of material biocompatibility, mechanical property, capability of integration with target tissue, and maintenance of the bioactivities of implants (i.e., cell and growth factors). Because articular cartilage is bathed in synovial fluid and there is constant joint movement, cells for cartilage repair or regeneration must be delivered to the tissue site encased in a three-dimensional (3D) scaffold with adequate mechanical support. Table 1.3 presents a summary of the materials that

| | Models | Cell types | Reference | | |
|--------------------------|--|--|---|--|--|
| Animal models | Nude mice | iPSCs or iPSC-derived cells | Hiramatsu et al. (2011) | | |
| | Rat | Adipose-derived stem cells | Xie et al. (2012) | | |
| | | ESCs or ESC-derived cells | Wakitani et al. (2004a) and Toh et al. (2010) | | |
| | | Muscle-derived stem cells | Mifune et al. (2013) and Matsumoto et al. (2009) | | |
| | | Bone marrow-derived stem cells | Park et al. (2006) and Yoshimura et al. (2007) | | |
| | | iPSCs or iPSC-derived cells | Jiang et al. (2012) | | |
| | Rabbit | Bone marrow-derived stem cells | Koga et al. (2008), Murphy et al. (2003), Kayakabe et al. (2006), Guo et al. (2010), Li et al. (2011a), Yan and Yu (2007), Im et al. (2001), and Park et al. (2009) | | |
| | | Adipose-derived stem cells | Fernandez et al. (2012), Koga et al. (2008), and Li et al. (2011a) | | |
| | | Synovium-derived stem cells | Koga et al. (2008), Lee et al. (2013), Lee et al. (2012), Suzuki et al. (2012), and Pei et al. (2009) | | |
| | | Periosteum-derived stem cells | Li et al. (2011a) | | |
| | | Muscle-derived stem cells | Li et al. (2011a) and Koga et al. (2008) | | |
| | | Umbilical cord-derived stem cells | Yan and Yu (2007) | | |
| | Goat | Bone marrow-derived stem cells | Sharma et al. (2007) | | |
| | Sheep | ESCs or ESC-derived cells | Dattena et al. (2009) | | |
| | Mini pig | Bone marrow-derived stem cells | Zhou et al. (2006) | | |
| | | Synovium-derived cells | Pei et al. (2013) | | |
| Clinical studies | Clinical study/ case report | Autologous chondrocytes | Brittberg et al. (1994) and Peterson et al. (2010) | | |
| | | Cartilage-derived progenitor cells | Jiang et al. (2016) | | |
| | | Bone marrow-derived stromal/stem cells | Wakitani et al. (2002), Kuroda et al. (2007), Wakitani et al. (2004b), and Wakitani et al. (2007) | | |
| Non- animal models | Engineered Chondrocytes, bone tissue marrow-derived stromal/stem cells | | Rackwitz et al. (2014) | | |
| | 3D cultures | Adult and embryonic- derived cells | Cheng et al. (2013), Tuli et al. (2004), Jiang et al. (2016), and Toh et al. (2010) | | |

 Table 1.3
 Research models used in cell-based cartilage repair studies

have been used for cartilage tissue engineering. Additional information is provided in a systematic review on scaffold-based repair for cartilage healing by Filardo et al. (2013a) and in a review on product development and clinical trial by Ahmed and Hincke (2010) (for further details, see Chap. 10).

1.4.2.2 Major Concerns in Scaffold Design

Biocompatibility Biocompatibility, referring to the property of a scaffold not harming cells or tissues or evoking severe immune reaction, is the fundamental requirement for the application of biomaterial scaffold application. A unique characteristic of articular cartilage is its avascularity, rendering it less immune reactive than most other tissues (Revell and Athanasiou 2009). The majority of biomaterials currently used in cartilage repair studies and applications are biocompatible, although undesirable reactions are sometimes observed depending on the type of polymer, crosslinking methods, and the nature of the by-products.

Mechanical Property The compressive modulus of native human articular cartilage ranges from 4.3 to 11.6 MPa (Shepherd and Seedhom 1999). Ideally, the grafted scaffolds should possess similar mechanical strength in order to withstand compressive loads immediately after surgery. As shown in Table 1.4, synthetic polymers generally have higher compressive modulus than native biomaterials. However, cell-binding ligands are absent in synthetic polymers; in addition, higher mechanical strength often is a consequence of a structurally dense composition, which is likely to limit nutrient penetration, cell growth, and neocartilage formation. Therefore, hybrid scaffolds that combine the properties and structures of both synthetic and native materials must be considered in the design and development of biomaterial scaffold for future cartilage repair applications (for details, see Chap. 10).

In Situ Fabrication One approach to improve the efficacy of cartilage repair is the formation of cartilage in situ, instead of implantation of engineered cartilage constructs produced from cells and scaffold cultured ex vivo. To allow in situ delivery of cells and growth factors with minimal surgical invasion, the use of injectable scaffold such as cross-linkable hydrogels represents another approach. Currently, there are several strategies to initiate and control the in situ cross-linking process:

- 1. Photo cross-linking. The basic principle of this method is to first derivatize vinyl groups onto the polymer and then use free radicals, produced from photoinitiator by light exposure (most often using UV light), to polymerize the monomers. Due to the potential cytotoxic effects of UV exposure on cells, visible light-activated cross-linking has recently received attention (Lin et al. 2013; Fairbanks et al. 2009).
- Thermal cross-linking. This process involves the incorporation of thermosensitive polymers, such as poly(N-isopropylacrylamide-co-acrylic acid). The precursor solution is liquid at ambient temperature and undergoes gelation at around 37 °C (Klouda and Mikos 2008).

| Product name | BioSeed-C (PLA, PGA copolymer) | ChonDux (with chondroitin sulfate) | SaluCartilage | | Cartipatch (mixture of agarose and alginate) | Cartipatch (mixture of agarose and alginate) | BST-CarGel | Chondro-Gide (with collagen III), CaReS, PureCol/VitroCol/ Nutragen |
|------------------------------|--|---|---|---|--|--|---|--|
| Clinical trial | Yes | Yes | Yes | I | Yes | Yes | Yes | Yes |
| Biodegradable | Yes | 1 | 1 | Yes | 1 | 1 | Yes | Yes |
| Cell- binding ligands | I | 1 | 1 | Yes | I | I | 1 | Yes |
| Compression modulus | 20 kPa at 68% PLA in PGA mesh (Moran et al. 2003) | 320 kPa at 100% (Riley et al. 2001) | 250 kPa at 20% (Holloway et al. 2010) | <2 kPa (Kisiday et al. 2002) | 5–10 kPa at 2% (Huang et al. 2009) | 20–30 kPa at 2% (Wong et al. 2001) | 20 kPa at 2% (Subramanian and Lin 2005) | 7 kPa at 0.7% (Li et al. 2012) |
| Sources | | | | | Marine algae | Marine algae | Mushroom, arthropod exoskeletons | Tendon or skin from human or animals |
| Monomer form or precursor | Lactic acid/glycolic acid | Ethylene glycol | Vinyl alcohol | RADA (Kopesky et al. 2010), RGDS (Miller et al. 2010), KLDL (Kopesky et al. 2011) | Galactose and 3,6-anhydrogalactose | β-D-mannuronic acid and R-L-guluronic acid | D-glucosamine, N-acetylglucosamine | Collagen 1α1, 1α2; collagen 3α1 |
| Material | Polylactic acid (PLA)/ polyglycolic acid (PGA) | Poly(ethylene glycol) | Poly(vinyl alcohol) | Self-assembling peptide | Agarose | Alginate | Chitosan | Collagen types I/III |
| | Synthesized | | | | | Natural derived | | |

 Table 1.4
 Materials and products that have been used for cartilage repair

(continued)

| Product name | | Tissucol | Hyalograft C, HYAFF11 | | | |
|------------------------------|---|-----------------------------------|--|---|---|--|
| Clinical trial | 1 | Yes | Yes | I | I | I |
| Biodegradable | Yes | Yes | Yes | Yes | Yes | Yes |
| Cell- binding ligands | Yes | Yes | Yes | Yes | Yes | Yes |
| Compression modulus | 0.88 kPa at 1% (Pfeiffer et al. 2008) | 0.029 kPa | 8–30 kPa (Bian et al. 2012; Toh et al. 2012) | 13–34 kPa at 8% (Yan et al. 2012) | 1.92 MPa (Schwarz et al. 2012) | <5 KPa (Ando et al. 2008; Mitani et al. 2009) |
| Sources | Cartilage from human or animals | Blood from human or animals | Cartilage from human or animals | Silkworms | Cartilage from human or animals (Schwarz et al. 2012) | Stem cells (Ando et al. 2008) or chondrocyte (Mitani et al. 2009) |
| Monomer form or precursor | Collagen 2α1 | Fibrinogen | D-Glucuronic acid, D- <i>N</i> - acetylglucosamine | Fibroin | Native ECM from cartilage | ECM secreted by cells |
| Material | Collagen type II | Fibrin | Hyaluronan | Silk fibroin | Decellularized matrix | Cell sheet |
| | | | | | | |

Table 1.4 (continued)

- Enzymatic cross-linking. Peroxidases, such as horseradish peroxidase, catalyze oxidative polymerization of phenol groups in the monomers in the presence of H₂O₂ (Zavada et al. 2016).
- 4. pH-sensitive cross-linking.
- 5. Ionic interaction- and hydrophobic interaction-initiated cross-linking.
- 6. Chemical cross-linking. In general, scaffolds capable of in situ fabrication are preferred in terms of a better fit for the tissue defects prefabricated scaffolds.

1.5 Unsolved Problems and Challenges in Cartilage Repair

1.5.1 Tissue Integration

Integration of implant tissue is critical for cartilage repair, particularly in view of the presence of mechanical loading which will otherwise dislodge the implanted construct. Current methods to promote tissue integration include: (1) application of biocompatible glue such as fibrin (Bekkers et al. 2010), (2) suturing (Bryant et al. 2007), (3) pin or transosseous fixation (Zelle et al. 2007), (4) regenerative interaction, and (5) chemical cross-linking between the biomaterial scaffold and host tissue (Sharma et al. 2013). New strategies should encourage both inherent cell-based repair and external implants, for instance, the application of the implanted cells to host tissue or recruit cells from host tissue (Zhang et al. 2013) and stimulating production of cartilage ECM to seal the gap between graft and tissue.

1.5.2 Maintenance of Cell Phenotype

A major current challenge is the long-term maintenance of the hyaline chondrocyte phenotype of the transplanted cells in vivo, particularly the differentiated stem cells. For MSC populations that have undergone chondrogenic differentiation, fibrous tissue formation or a hypertrophic phenotype is frequently seen. Chondrocyte dedifferentiation is generally thought to be the reason for fibrocartilage formation, which affects the quality and mechanical property of the repair tissue, while hypertrophy represents unregulated "maturation" of the chondrocytes along the endochondral ossification pathway as seen in long bone development. The host environment also greatly influences the phenotypes of the transplanted cells and the repaired tissue. Thus, host matrix, growth factor composition, as well as mechanobiological signals all play important roles in phenotype maintenance and production of the new cartilage ECM.

Three-Dimensional Environment A three-dimensional (3D) production of the environment is believed to be critical for the maintenance of the chondrocyte phenotype. Thus, experimentally, 3D cultures such as pellets (Anderer and

Libera 2002) or alginate hydrogel encapsulated cultures (Henrionnet et al. 2012) are commonly used. The current trend of stem cell-based strategies is to deliver chondrogenic inductive factors together with cell seeding or to deliver the chondrogenic inductive growth factors to the local stem cells, in order to guide proper tissue repair. These factors can be directly injected or embedded in the scaffolds; however, due to the short half-lives of bioactive factors (mostly proteins), and their rapid diffusion or clearance from the host environment, high doses and/or repeated administrations are often required. New methods of encapsulation of biomolecules are being developed to achieve spatiotemporally controlled, sustained release of biofactors to recapitulate the cell-cell paracrine signaling responsible for promoting and directing tissue repair (Cleaver and Melton 2003; Werner and Grose 2003). For example, micro- and nanoparticle-based control-release systems can provide protection against drug degradation/inactivation and achieve a localized, sustained release (Tayalia and Mooney 2009; Alvarez et al. 2015).

Bioreactor Technology Advances in bioreactor technology provide another approach to assess the proper microenvironment conditions, including oxygen concentration, medium composition, and mechanical stimulation (Mabvuure et al. 2012), that are required to enhance tissue neogenesis. In particular, mechanical stimulation at an appropriate range and mode could enhance cartilage matrix deposition, e.g., dynamic cyclic loading results in the increase of ECM synthesis in engineered cartilage (Spiller et al. 2011; Saini and Wick 2003) (for details, see Chap. 10).

Gene-Based Delivery of Growth Factors Another developing approach is to achieve growth factor delivery using plasmid or viral gene vectors incorporated in cells or biomaterials. The genetically modified cells could maintain production of the signaling factors transiently or permanently after transplantation (Koria 2012). For example, a cell-mediated gene therapy system for the repair of knee arthritis, in which allogenic chondrocytes express TGF- β 1, is in Phase II clinical trial (TissueGene) (Ha et al. 2012). However, in general, the safety of viral vector usage and controlled gene expression requires further investigation (for details, see Chap. 9).

1.5.3 Simplifying Surgical and Rehabilitation Procedures

ACI and MACI both involve two-stage surgeries with inherent complexity. A singlestage point-of-care strategy is thus preferred and ideally involves a single application of growth factors, biomaterial scaffolds, and cells ready to be used at the same time of harvest (Stanish et al. 2013). Such a technique would require "smart" scaffolds that can provide sufficient mechanical support to allow immediate mechanical loading after surgery, in order to create an environment that enhances posttransplantation tissue repair. Growth factors also need to be incorporated efficiently and at optimized levels. For cells, higher yield harvesting, efficacious sourcing, and optimal chondrogenic capacity are needed.

1.5.4 Prevention of OA Progress

Beside the local repair event in the cartilage defect, the ultimate goal of cartilage repair is to reestablish joint health and prevent OA progress. For this purpose, it is envisioned that an appropriate anti-inflammatory therapy is needed either concurrent with or subsequent to the regenerative procedure, as well as proper mechanical conditions and tissue environment.

1.5.4.1 Establishing Correct Joint Mechanical Environment

Overloading or an abnormal mechanical environment could be harmful to a healthy joint, not to mention the injured tissue. It is noteworthy that ACI/MACI cannot improve joints that have undergone previous or combined meniscectomies or ACL surgeries (Filardo et al. 2013b), and a high body mass index (BMI) had a deleterious effect on the clinical outcome of ACI/MACI (Jaiswal et al. 2012). A stable and healthy mechanical environment is thus the first consideration before joint resurfacing. For example, varus deformities of 5° and more were considered for high tibial osteotomy in patients with cartilage defects (Bode et al. 2013). Another example is that a meniscus transplantation is needed in combination with ACI/MACI in meniscus-deficient knees (Bhosale et al. 2007).

1.5.4.2 Rebuilding Subchondral Bone

Cartilage defects are often accompanied by subchondral bone lesions, such as in osteochondritis dissecans (OCD). Subchondral bone pathology is involved in the progression of OA both biochemically and mechanically (Cox et al. 2011; Pan et al. 2009). In large osteochondral defects, the lack of subchondral bone self-repair can result in the absence of surface cartilage repair (Madry et al. 2010). Subchondral bone remodeling is a primary cofactor to consider during cartilage repair in joint function restoration; for example, some MACI procedures for OCD are performed after bone grafting (Ochs et al. 2011; Vijayan et al. 2012; Filardo et al. 2012). The implantation of biomaterial scaffolds to facilitate subchondral bone regeneration (Jiang et al. 2013).

1.5.4.3 Anti-Inflammatory Actions

The host tissue microenvironment is critical in regulating the biological activities of the transplanted cells. Thus, an inflammatory host environment could compromise cartilage ECM production by the transplanted cells during repair due to the presence and action of pro-inflammatory factors. For example, interleukin-1 β (IL-1 β) and tumor necrosis factor- α (TNF- α) are well-known OA mediators, and pro-inflammatory factors are elaborated in the OA joint that inhibit cartilage ECM generation and promote chondrocyte apoptosis (Tetlow et al. 2001). The application of

neutralizing antibodies and inhibitors of pro-inflammatory factors for OA is under active investigation, such as intra-articular injection of TNF- α neutralizing antibody (Urech et al. 2010) or interleukin-1 receptor antagonist (IL-1Ra) (Whitmire et al. 2012). It is believed that these agents can act by altering and neutralizing the pro-inflammatory environment to promote the reparative activities of the transplanted cells/tissue. Interestingly, BMSCs or ADSCs have been clearly shown to exhibit immunoregulatory characteristics (Tuan et al. 2003; Bartholomew et al. 2002; Aggarwal and Pittenger 2005; Bailey et al. 2010), rendering them as promising anti-inflammatory cell candidates for both OA and tissue repair. However, the exact approaches to balance their differentiation ability and immunoregulatory activities and how long the effect will last still require further investigation.

Conclusion

The significant disease burden of OA, which affects up to 15% of the populace, underscores the need for novel and effective therapies. The emerging discipline of tissue engineering and regenerative medicine has expanded the therapeutic vista by offering the potential of "bio-resurfacing" that may restore both structure and function to the diseased joint. To realize the promise of this exciting approach, challenges related to cells (e.g., sourcing, expansion, differentiation, and phenotype stability), scaffold (e.g., biocompatibility, mechanical properties), and chondro-supportive biofactors (e.g., composition, delivery), as well as long-term safety, must be overcome by developing effective and efficacious technologies, which depends critically on active interaction and collaboration among scientists, engineers, and clinicians.

Acknowledgments The authors thank the International Cartilage Repair Society (ICRS) for their kind approval of using illustrations from the ICRS Cartilage Injury Evaluation Package (Fig. 1.4). Research funding is supported by the US Department of Defense (W81XWH-14-2-0003, W81XWH-14-1-0217), NIH (R01 EB019430, P30 AG024827), and the Arthur J. Rooney, Sr. Endowment.

References

- Aggarwal S, Pittenger MF (2005) Human mesenchymal stem cells modulate allogeneic immune cell responses. Blood 105(4):1815–1822. doi:10.1182/blood-2004-04-1559
- Ahmed TA, Hincke MT (2010) Strategies for articular cartilage lesion repair and functional restoration. Tissue Eng Part B Rev 16(3):305–329. doi:10.1089/ten.TEB.2009.0590
- Akagi A, Tajima S, Ishibashi A, Matsubara Y, Takehana M, Kobayashi S, Yamaguchi N (2002) Type XVI collagen is expressed in factor XIIIa+ monocyte-derived dermal dendrocytes and constitutes a potential substrate for factor XIIIa. J Invest Dermatol 118:267–274
- Akiyama H, Lefebvre V (2011) Unraveling the transcriptional regulatory machinery in chondrogenesis. J Bone Miner Metab 29(4):390–395. doi:10.1007/s00774-011-0273-9
- Alaoui-Ismaili MH, Falb D (2009) Design of second generation therapeutic recombinant bone morphogenetic proteins. Cytokine Growth Factor Rev 20(5–6):501–507. doi:10.1016/j. cytogfr.2009.10.001
- Almqvist KF, Dhollander AA, Verdonk PC, Forsyth R, Verdonk R, Verbruggen G (2009) Treatment of cartilage defects in the knee using alginate beads containing human mature allogenic chondrocytes. Am J Sports Med 37(10):1920–1929. doi:10.1177/0363546509335463
- Alsalameh S, Amin R, Gemba T, Lotz M (2004) Identification of mesenchymal progenitor cells in normal and osteoarthritic human articular cartilage. Arthritis Rheum 50(5):1522–1532. doi:10.1002/art.20269
- Alvarez LM, Rivera JJ, Stockdale L, Saini S, Lee RT, Griffith LG (2015) Tethering of epidermal growth factor (EGF) to beta tricalcium phosphate (betaTCP) via fusion to a high affinity, multimeric betaTCP-binding peptide: effects on human multipotent stromal cells/connective tissue progenitors. PLoS One 10(6):e0129600. doi:10.1371/journal.pone.0129600
- Anderer U, Libera J (2002) In vitro engineering of human autogenous cartilage. J Bone Miner Res 17(8):1420–1429. doi:10.1359/jbmr.2002.17.8.1420
- Ando W, Tateishi K, Katakai D, Hart DA, Higuchi C, Nakata K, Hashimoto J, Fujie H, Shino K, Yoshikawa H, Nakamura N (2008) In vitro generation of a scaffold-free tissue-engineered construct (TEC) derived from human synovial mesenchymal stem cells: biological and mechanical properties and further chondrogenic potential. Tissue Eng Part A 14(12):2041–2049. doi:10.1089/ten.tea.2008.0015
- Bailey AM, Kapur S, Katz AJ (2010) Characterization of adipose-derived stem cells: an update. Curr Stem Cell Res Ther 5(2):95–102
- Barry F, Boynton RE, Liu B, Murphy JM (2001) Chondrogenic differentiation of mesenchymal stem cells from bone marrow: differentiation-dependent gene expression of matrix components. Exp Cell Res 268(2):189–200. doi:10.1006/excr.2001.5278
- Bartholomew A, Sturgeon C, Siatskas M, Ferrer K, McIntosh K, Patil S, Hardy W, Devine S, Ucker D, Deans R, Moseley A, Hoffman R (2002) Mesenchymal stem cells suppress lymphocyte proliferation in vitro and prolong skin graft survival in vivo. Exp Hematol 30(1):42–48. doi:10.1016/S0301-472x(01)00769-X
- Bekkers JE, Tsuchida AI, Malda J, Creemers LB, Castelein RJ, Saris DB, Dhert WJ (2010) Quality of scaffold fixation in a human cadaver knee model. Osteoarthritis Cartilage 18(2):266–272. doi:10.1016/j.joca.2009.09.001
- Bhosale AM, Myint P, Roberts S, Menage J, Harrison P, Ashton B, Smith T, McCall I, Richardson JB (2007) Combined autologous chondrocyte implantation and allogenic meniscus transplantation: a biological knee replacement. Knee 14(5):361–368
- Bian L, Zhai DY, Zhang EC, Mauck RL, Burdick JA (2012) Dynamic compressive loading enhances cartilage matrix synthesis and distribution and suppresses hypertrophy in hMSC-laden hyaluronic acid hydrogels. Tissue Eng Part A 18(7–8):715–724. doi:10.1089/ten.TEA.2011.0455
- Bobick BE, Chen FH, Le AM, Tuan RS (2009) Regulation of the chondrogenic phenotype in culture. Birth Defects Res C Embryo Today 87(4):351–371. doi:10.1002/bdrc.20167
- Bode G, Schmal H, Pestka JM, Ogon P, Sudkamp NP, Niemeyer P (2013) A non-randomized controlled clinical trial on autologous chondrocyte implantation (ACI) in cartilage defects of the medial femoral condyle with or without high tibial osteotomy in patients with varus deformity of less than 5 degrees. Arch Orthop Trauma Surg 133(1):43–49. doi:10.1007/s00402-012-1637-x
- Brittberg M, Lindahl A, Nilsson A, Ohlsson C, Isaksson O, Peterson L (1994) Treatment of deep cartilage defects in the knee with autologous chondrocyte transplantation. N Engl J Med 331(14):889–895. doi:10.1056/NEJM199410063311401
- Bryant D, Dill J, Litchfield R, Amendola A, Giffin R, Fowler P, Kirkley A (2007) Effectiveness of bioabsorbable arrows compared with inside-out suturing for vertical, reparable meniscal lesions – a randomized clinical trial. Am J Sport Med 35(6):889–896. doi:10.1177/0363546506298582
- Byers BA, Mauck RL, Chiang IE, Tuan RS (2008) Transient exposure to transforming growth factor beta 3 under serum-free conditions enhances the biomechanical and biochemical maturation of tissue-engineered cartilage. Tissue Eng Part A 14(11):1821–1834. doi:10.1089/ten. tea.2007.0222
- Callaghan JJ (2003) The adult knee. 1. Lippincott Williams & Wilkins
- Chahal J, Gross AE, Gross C, Mall N, Dwyer T, Chahal A, Whelan DB, Cole BJ (2013) Outcomes of osteochondral allograft transplantation in the knee. Arthroscopy 29(3):575–588. doi:10.1016/j.arthro.2012.12.002
- Cheng NC, Estes BT, Young TH, Guilak F (2013) Genipin-crosslinked cartilage-derived matrix as a scaffold for human adipose-derived stem cell chondrogenesis. Tissue Eng Part A 19(3–4):484–496. doi:10.1089/ten.TEA.2012.0384

- Cleaver O, Melton DA (2003) Endothelial signaling during development. Nat Med 9(6):661–668. doi:10.1038/nm0603-661nm0603-661
- Cole BJ, DeBerardino T, Brewster R, Farr J, Levine DW, Nissen C, Roaf P, Zaslav K (2012) Outcomes of autologous chondrocyte implantation in study of the treatment of articular repair (STAR) patients with osteochondritis dissecans. Am J Sports Med 40(9):2015–2022. doi:10.1177/0363546512453292
- Cox LG, van Rietbergen B, van Donkelaar CC, Ito K (2011) Bone structural changes in osteoarthritis as a result of mechanoregulated bone adaptation: a modeling approach. Osteoarthritis Cartilage 19(6):676–682. doi:10.1016/j.joca.2011.02.007
- Danisovic L, Varga I, Polak S (2012) Growth factors and chondrogenic differentiation of mesenchymal stem cells. Tissue Cell 44(2):69–73. doi:10.1016/j.tice.2011.11.005
- Dattena M, Pilichi S, Rocca S, Mara L, Casu S, Masala G, Manunta L, Manunta A, Passino ES, Pool RR, Cappai P (2009) Sheep embryonic stem-like cells transplanted in full-thickness cartilage defects. J Tissue Eng Regen Med 3(3):175–187. doi:10.1002/term.151
- DeLise AM, Fischer L, Tuan RS (2000) Cellular interactions and signaling in cartilage development. Osteoarthritis Cartilage 8(5):309–334. doi:10.1053/joca.1999.0306
- Dhollander AA, Verdonk PC, Lambrecht S, Verdonk R, Elewaut D, Verbruggen G, Almqvist KF (2012) Midterm results of the treatment of cartilage defects in the knee using alginate beads containing human mature allogenic chondrocytes. Am J Sports Med 40(1):75–82. doi:10.1177/ 0363546511423013
- Diederichs S, Tuan RS (2014) Functional comparison of human-induced pluripotent stem cellderived mesenchymal cells and bone marrow-derived mesenchymal stromal cells from the same donor. Stem Cells Dev 23(14):1594–1610. doi:10.1089/scd.2013.0477
- Diekman BO, Estes BT, Guilak F (2010) The effects of BMP6 overexpression on adipose stem cell chondrogenesis: interactions with dexamethasone and exogenous growth factors. J Biomed Mater Res A 93(3):994–1003. doi:10.1002/jbm.a.32589
- Estes BT, Wu AW, Guilak F (2006) Potent induction of chondrocytic differentiation of human adipose-derived adult stem cells by bone morphogenetic protein 6. Arthritis Rheum 54(4):1222–1232. doi:10.1002/art.21779
- Fairbanks BD, Schwartz MP, Bowman CN, Anseth KS (2009) Photoinitiated polymerization of PEGdiacrylate with lithium phenyl-2,4,6-trimethylbenzoylphosphinate: polymerization rate and cytocompatibility. Biomaterials 30(35):6702–6707. doi:10.1016/j.biomaterials.2009.08.055
- Felson DT (2004) An update on the pathogenesis and epidemiology of osteoarthritis. Radiol Clin North Am 42(1):1–9. doi:10.1016/S0033-8389(03)00161-1S0033838903001611
- Fernandez FB, Shenoy S, Babu SS, Varma HK, John A (2012) Short-term studies using ceramic scaffolds in lapine model for osteochondral defect amelioration. Biomed Mater 7(3). doi:10.1088/1748-6041/7/3/035005
- Fickert S, Fiedler J, Brenner RE (2004) Identification of subpopulations with characteristics of mesenchymal progenitor cells from human osteoarthritic cartilage using triple staining for cell surface markers. Arthritis Res Ther 6(5):R422–R432. doi:10.1186/ar1210
- Filardo G, Kon E, Berruto M, Di Martino A, Patella S, Marcheggiani Muccioli GM, Zaffagnini S, Marcacci M (2012) Arthroscopic second generation autologous chondrocytes implantation associated with bone grafting for the treatment of knee osteochondritis dissecans: results at 6 years. Knee 19(5):658–663. doi:10.1016/j.knee.2011.08.007
- Filardo G, Kon E, Roffi A, Di Martino A, Marcacci M (2013a) Scaffold-based repair for cartilage healing: a systematic review and technical note. Arthroscopy 29(1):174–186. doi:10.1016/j. arthro.2012.05.891
- Filardo G, Vannini F, Marcacci M, Andriolo L, Ferruzzi A, Giannini S, Kon E (2013b) Matrixassisted autologous chondrocyte transplantation for cartilage regeneration in osteoarthritic knees: results and failures at midterm follow-up. Am J Sports Med 41(1):95–100. doi:10.1177/0363546512463675
- Fragonas E, Valente M, Pozzi-Mucelli M, Toffanin R, Rizzo R, Silvestri F, Vittur F (2000) Articular cartilage repair in rabbits by using suspensions of allogenic chondrocytes in alginate. Biomaterials 21(8):795–801

- Freyria AM, Mallein-Gerin F (2012) Chondrocytes or adult stem cells for cartilage repair: the indisputable role of growth factors. Injury 43(3):259–265. doi:10.1016/j.injury.2011.05.035
- Gehris AL, Oberlender SA, Shepley KJ, Tuan RS, Bennett VD (1996) Fibronectin mRNA alternative splicing is temporally and spatially regulated during chondrogenesis in vivo and in vitro. Dev Dyn 206(2):219–230. doi:10.1002/(sici)1097-0177(199606)206:2<219::aid-aja11>3.0.co;2-y
- Gibson MA, Finnis ML, Kumaratilake JS, Cleary EG (1998) Microfibril-associated glycoprotein-2 (MAGP-2) is specifically associated with fibrillin-containing microfibrils but exhibits more restricted patterns of tissue localization and developmental expression than its structural relative MAGP-1. J Histochem Cytochem 46(8):871–886
- Goldring MB (2006) Update on the biology of the chondrocyte and new approaches to treating cartilage diseases. Best Pract Res Clin Rheumatol 20(5):1003–1025. doi:10.1016/j.berh.2006.06.003
- Goldring MB, Tsuchimochi K, Ijiri K (2006) The control of chondrogenesis. J Cell Biochem 97(1):33–44. doi:10.1002/jcb.20652
- Grande DA, Singh IJ, Pugh J (1987) Healing of experimentally produced lesions in articular cartilage following chondrocyte transplantation. Anat Rec 218(2):142–148
- Grande DA, Pitman MI, Peterson L, Menche D, Klein M (1989) The repair of experimentally produced defects in rabbit articular cartilage by autologous chondrocyte transplantation. J Orthop Res 7(2):208–218
- Grogan SP, Chen X, Sovani S, Taniguchi N, Colwell CW Jr, Lotz MK, D'Lima DD (2013) Influence of cartilage extracellular matrix molecules on cell phenotype and neocartilage formation. Tissue Eng Part A. doi:10.1089/ten.TEA.2012.0618
- Guo X, Park H, Young S, Kretlow JD, van den Beucken JJ, Baggett LS, Tabata Y, Kasper FK, Mikos AG, Jansen JA (2010) Repair of osteochondral defects with biodegradable hydrogel composites encapsulating marrow mesenchymal stem cells in a rabbit model. Acta Biomater 6(1):39–47. doi:10.1016/j.actbio.2009.07.041S1742-7061(09)00326-2
- Ha CW, Noh MJ, Choi KB, Lee KH (2012) Initial phase I safety of retrovirally transduced human chondrocytes expressing transforming growth factor-beta-1 in degenerative arthritis patients. Cytotherapy 14(2):247–256. doi:10.3109/14653249.2011.629645
- Heinegard D, Saxne T (2011) The role of the cartilage matrix in osteoarthritis. Nat Rev Rheumatol 7(1):50–56. doi:10.1038/nrrheum.2010.198
- Heng BC, Cao T, Lee EH (2004) Directing stem cell differentiation into the chondrogenic lineage in vitro. Stem Cells 22(7):1152–1167. doi:10.1634/stemcells.2004–0062
- Hennig T, Lorenz H, Thiel A, Goetzke K, Dickhut A, Geiger F, Richter W (2007) Reduced chondrogenic potential of adipose tissue derived stromal cells correlates with an altered TGFbeta receptor and BMP profile and is overcome by BMP-6. J Cell Physiol 211(3):682–691. doi:10.1002/jcp.20977
- Henrionnet C, Wang Y, Roeder E, Gambier N, Galois L, Mainard D, Bensoussan D, Gillet P, Pinzano A (2012) Effect of dynamic loading on MSCs chondrogenic differentiation in 3-D alginate culture. Biomed Mater Eng 22(4):209–218. doi:10.3233/BME-2012-0710
- Hildner F, Albrecht C, Gabriel C, Redl H, van Griensven M (2011) State of the art and future perspectives of articular cartilage regeneration: a focus on adipose-derived stem cells and plateletderived products. J Tissue Eng Regen Med 5(4):E36–E51. doi:10.1002/Term.386
- Hiramatsu K, Sasagawa S, Outani H, Nakagawa K, Yoshikawa H, Tsumaki N (2011) Generation of hyaline cartilaginous tissue from mouse adult dermal fibroblast culture by defined factors. J Clin Invest 121(2):640–657. doi:10.1172/Jci44605
- Holloway JL, Lowman AM, Palmese GR (2010) Mechanical evaluation of poly(vinyl alcohol)based fibrous composites as biomaterials for meniscal tissue replacement. Acta Biomater 6(12):4716–4724. doi:10.1016/j.actbio.2010.06.025
- Huang AH, Stein A, Tuan RS, Mauck RL (2009) Transient exposure to transforming growth factor beta 3 improves the mechanical properties of mesenchymal stem cell-laden cartilage constructs in a density-dependent manner. Tissue Eng Part A 15(11):3461–3472. doi:10.1089/ten. TEA.2009.0198
- Hwang NS, Varghese S, Lee HJ, Zhang Z, Ye Z, Bae J, Cheng L, Elisseeff J (2008) In vivo commitment and functional tissue regeneration using human embryonic stem cell-derived mesenchymal cells. Proc Natl Acad Sci U S A 105(52):20641–20646. doi:10.1073/pnas.0809680106

- Im GI, Kim DY, Shin JH, Hyun CW, Cho WH (2001) Repair of cartilage defect in the rabbit with cultured mesenchymal stem cells from bone marrow. J Bone Joint Surg Br 83(2):289–294
- Irrgang JJ, Snyder-Mackler L, Wainner RS, Fu FH, Harner CD (1998) Development of a patientreported measure of function of the knee. J Bone Joint Surg Am 80(8):1132–1145
- Jaiswal PK, Bentley G, Carrington RW, Skinner JA, Briggs TW (2012) The adverse effect of elevated body mass index on outcome after autologous chondrocyte implantation. J Bone Joint Surg Br 94(10):1377–1381. doi:10.1302/0301-620X.94B10.29388
- Jiang Y, Tuan RS (2015) Origin and function of cartilage stem/progenitor cells in osteoarthritis. Nat Rev Rheumatol 11(4):206–212. doi:10.1038/nrrheum.2014.200
- Jiang Y, Chen LK, Zhu DC, Zhang GR, Guo C, Qi YY, Ouyang HW (2010) The inductive effect of bone morphogenetic protein-4 on chondral-lineage differentiation and in situ cartilage repair. Tissue Eng Part A 16(5):1621–1632. doi:10.1089/ten.TEA.2009.0681
- Jiang YZ, Zhang SF, Qi YY, Wang LL, Ouyang HW (2011) Cell transplantation for articular cartilage defects: principles of past, present, and future practice. Cell Transplant 20(5):593–607. doi :10.3727/096368910X532738
- Jiang Y, Tong T, Heng B, Ouyang H (2012) Cartilage injuries: role of implantation of human stem/ progenitor cells. In: Hayat MA (ed) Stem cells and cancer stem cells, vol 3. Springer Netherlands, pp 327–333. doi:10.1007/978-94-007-2415-0_33
- Jiang Y, Chen L, Zhang S, Tong T, Zhang W, Liu W, Xu G, Tuan RS, Heng BC, Crawford R, Xiao Y, Ouyang HW (2013) Incorporation of bioactive polyvinylpyrrolidone-iodine within bilayered collagen scaffolds enhances the differentiation and subchondral osteogenesis of mesenchymal stem cells. Acta Biomater. doi:10.1016/j.actbio.2013.05.014
- Jiang Y, Cai Y, Zhang W, Yin Z, Hu C, Tong T, Lu P, Zhang S, Neculai D, Tuan RS, Ouyang HW (2016) Human cartilage-derived progenitor cells from committed chondrocytes for efficient cartilage repair and regeneration. Stem Cells Transl Med. doi:10.5966/sctm.2015-0192
- Johnson DH, Pedowitz RA (2007) Practical orthopaedic sports medicine & arthroscopy [electronic resource]. Wolters Kluwer Health/Lippincott Williams & Wilkins
- Kasemkijwattana C, Hongeng S, Kesprayura S, Rungsinaporn V, Chaipinyo K, Chansiri K (2011) Autologous bone marrow mesenchymal stem cells implantation for cartilage defects: two cases report. J Med Assoc Thai 94(3):395–400
- Kayakabe M, Tsutsumi S, Watanabe H, Kato Y, Takagishi K (2006) Transplantation of autologous rabbit BM-derived mesenchymal stromal cells embedded in hyaluronic acid gel sponge into osteochondral defects of the knee. Cytotherapy 8(4):343–353. doi:10.1080/14653240600845070
- Kellgren JH, Lawrence JS (1957) Radiological assessment of osteo-arthrosis. Ann Rheum Dis 16(4):494–502
- Kim MK, Choi SW, Kim SR, Oh IS, Won MH (2010) Autologous chondrocyte implantation in the knee using fibrin. Knee Surg Sports Traumatol Arthrosc 18(4):528–534. doi:10.1007/ s00167-009-0905-y
- Kisiday J, Jin M, Kurz B, Hung H, Semino C, Zhang S, Grodzinsky AJ (2002) Self-assembling peptide hydrogel fosters chondrocyte extracellular matrix production and cell division: implications for cartilage tissue repair. Proc Natl Acad Sci U S A 99(15):9996–10001. doi:10.1073/ pnas.142309999142309999
- Klouda L, Mikos AG (2008) Thermoresponsive hydrogels in biomedical applications. Eur J Pharm Biopharm 68(1):34–45. doi:10.1016/j.ejpb.2007.02.025
- Knutsen G, Drogset JO, Engebretsen L, Grontvedt T, Isaksen V, Ludvigsen TC, Roberts S, Solheim E, Strand T, Johansen O (2007) A randomized trial comparing autologous chondrocyte implantation with microfracture. Findings at five years. J Bone Joint Surg Am 89(10):2105–2112
- Koga H, Muneta T, Nagase T, Nimura A, Ju YJ, Mochizuki T, Sekiya I (2008) Comparison of mesenchymal tissues-derived stem cells for in vivo chondrogenesis: suitable conditions for cell therapy of cartilage defects in rabbit. Cell Tissue Res 333(2):207–215. doi:10.1007/ s00441-008-0633-5
- Kopesky PW, Vanderploeg EJ, Sandy JS, Kurz B, Grodzinsky AJ (2010) Self-assembling peptide hydrogels modulate in vitro chondrogenesis of bovine bone marrow stromal cells. Tissue Eng Part A 16(2):465–477. doi:10.1089/ten.TEA.2009.0158

- Kopesky PW, Vanderploeg EJ, Kisiday JD, Frisbie DD, Sandy JD, Grodzinsky AJ (2011) Controlled delivery of transforming growth factor beta1 by self-assembling peptide hydrogels induces chondrogenesis of bone marrow stromal cells and modulates Smad2/3 signaling. Tissue Eng Part A 17(1–2):83–92. doi:10.1089/ten.TEA.2010.0198
- Koria P (2012) Delivery of growth factors for tissue regeneration and wound healing. BioDrugs 26(3):163–175. doi:10.2165/11631850-00000000-00000
- Kornfeld R, Kornfeld S (1985) Assembly of asparagine-linked oligosaccharides. Annu Rev Biochem 54:631–664
- Kuo CK, Li WJ, Mauck RL, Tuan RS (2006) Cartilage tissue engineering: its potential and uses. Curr Opin Rheumatol 18 (1):64–73. pii:00002281-200601000-00011
- Kuroda R, Ishida K, Matsumoto T, Akisue T, Fujioka H, Mizuno K, Ohgushi H, Wakitani S, Kurosaka M (2007) Treatment of a full-thickness articular cartilage defect in the femoral condyle of an athlete with autologous bone-marrow stromal cells. Osteoarthritis Cartilage 15(2):226–231. doi:10.1016/j.joca.2006.08.008
- Lee JC, Lee SY, Min HJ, Han SA, Jang J, Lee S, Seong SC, Lee MC (2012) Synovium-derived mesenchymal stem cells encapsulated in a novel injectable gel can repair osteochondral defects in a rabbit model. Tissue Eng Part A 18(19–20):2173–2186. doi:10.1089/ten.tea.2011.0643
- Lee JC, Min HJ, Park HJ, Lee S, Seong SC, Lee MC (2013) Synovial membrane-derived mesenchymal stem cells supported by platelet-rich plasma can repair osteochondral defects in a rabbit model. Arthroscopy 29(6):1034–1046. doi:10.1016/j.arthro.2013.02.026S0749-8063(13)00181-3
- Lethbridge-Cejku M, Helmick CG, Popovic JR (2003) Hospitalizations for arthritis and other rheumatic conditions – data from the 1997 National Hospital Discharge Survey. Med Care 41(12):1367–1373. doi:10.1097/01.Mlr.0000100582.52451.Ac
- Li Q, Tang J, Wang R, Bei C, Xin L, Zeng Y, Tang X (2011a) Comparing the chondrogenic potential in vivo of autogeneic mesenchymal stem cells derived from different tissues. Artif Cells Blood Substit Immobil Biotechnol 39(1):31–38. doi:10.3109/10731191003776769
- Li QA, Tang JC, Wang RY, Bei CY, Xin LW, Zeng YJ, Tang XY (2011b) Comparing the chondrogenic potential in vivo of autogeneic mesenchymal stem cells derived from different tissues. Artif Cell Blood Sub 39(1):31–38. doi:10.3109/10731191003776769
- Li K, Guo L, Fan Y, Zhang X (2012) Experimental study on collagen hydrogel scaffolds for cartilage tissue engineering. Zhongguo Xiu Fu Chong Jian Wai Ke Za Zhi 26(11):1356–1361
- Lin H, Zhang D, Alexander PG, Yang G, Tan J, Cheng AW, Tuan RS (2013) Application of visible light-based projection stereolithography for live cell-scaffold fabrication with designed architecture. Biomaterials 34(2):331–339. doi:10.1016/j.biomaterials.2012.09.048
- Mabvuure N, Hindocha S, Khan WS (2012) The role of bioreactors in cartilage tissue engineering. Curr Stem Cell Res Ther 7 (4):287–292. pii:CSCRT-EPUB-20120508-008
- Madry H, van Dijk CN, Mueller-Gerbl M (2010) The basic science of the subchondral bone. Knee Surg Sports Traumatol Arthrosc 18(4):419–433. doi:10.1007/s00167-010-1054-z
- Marlovits S, Striessnig G, Resinger CT, Aldrian SM, Vecsei V, Imhof H, Trattnig S (2004) Definition of pertinent parameters for the evaluation of articular cartilage repair tissue with high-resolution magnetic resonance imaging. Eur J Radiol 52(3):310–319. doi:10.1016/j.ejrad.2004.03.014
- Mason C, Dunnill P (2008) A brief definition of regenerative medicine. Regen Med 3(1):1–5. doi:10.2217/17460751.3.1.1
- Mats Brittberg PA, Gambardella R, Hangody L, Haruselmann HJ, Jakob RP, Levine D, Lohmander S, Mandelbaum BR, Peterson L, Staubli H-U (2000) ICRS cartilage injury evaluation package. www.cartilage.org
- Matsumoto T, Cooper GM, Gharaibeh B, Meszaros LB, Li G, Usas A, Fu FH, Huard J (2009) Cartilage repair in a rat model of osteoarthritis through intraarticular transplantation of muscle-derived stem cells expressing bone morphogenetic protein 4 and soluble Flt-1. Arthritis Rheum 60(5):1390–1405. doi:10.1002/art.24443
- Mifune Y, Matsumoto T, Takayama K, Ota S, Li H, Meszaros LB, Usas A, Nagamune K, Gharaibeh B, Fu FH, Huard J (2013) The effect of platelet-rich plasma on the regenerative therapy of muscle derived stem cells for articular cartilage repair. Osteoarthritis Cartilage 21(1):175–185. doi:10.1016/j.joca.2012.09.018

- Miller RE, Grodzinsky AJ, Vanderploeg EJ, Lee C, Ferris DJ, Barrett MF, Kisiday JD, Frisbie DD (2010) Effect of self-assembling peptide, chondrogenic factors, and bone marrow-derived stromal cells on osteochondral repair. Osteoarthritis Cartilage 18(12):1608–1619. doi:10.1016/j. joca.2010.09.004
- Mitani G, Sato M, Lee JIK, Kaneshiro N, Ishihara M, Ota N, Kokubo M, Sakai H, Kikuchi T, Mochida J (2009) The properties of bioengineered chondrocyte sheets for cartilage regeneration. BMC Biotechnol 9:17. doi:10.1186/1472–6750–9-17
- Mithoefer K, Williams RJ 3rd, Warren RF, Potter HG, Spock CR, Jones EC, Wickiewicz TL, Marx RG (2005) The microfracture technique for the treatment of articular cartilage lesions in the knee. A prospective cohort study. J Bone Joint Surg Am 87(9):1911–1920. doi:10.2106/ JBJS.D.02846
- Moran JM, Pazzano D, Bonassar LJ (2003) Characterization of polylactic acid-polyglycolic acid composites for cartilage tissue engineering. Tissue Eng 9(1):63–70. doi:10.1089/107632703762687546
- Moskalewski S, Hyc A, Osiecka-Iwan A (2002) Immune response by host after allogeneic chondrocyte transplant to the cartilage. Microsc Res Tech 58(1):3–13. doi:10.1002/jemt.10110
- Moskowitz RW (2007) Osteoarthritis: diagnosis and medical/surgical management. Wolters Kluwer Health/Lippincott Williams & Wilkins
- Murphy JM, Fink DJ, Hunziker EB, Barry FP (2003) Stem cell therapy in a caprine model of osteoarthritis. Arthritis Rheum 48(12):3464–3474. doi:10.1002/art.11365
- Nejadnik H, Hui JH, Feng Choong EP, Tai BC, Lee EH (2010) Autologous bone marrow-derived mesenchymal stem cells versus autologous chondrocyte implantation: an observational cohort study. Am J Sports Med 38(6):1110–1116. doi:10.1177/0363546509359067
- Noyes FR, Matthews DS, Mooar PA, Grood ES (1983) The symptomatic anterior cruciate-deficient knee. Part II: the results of rehabilitation, activity modification, and counseling on functional disability. J Bone Joint Surg Am 65(2):163–174
- Oberlender SA, Tuan RS (1994a) Expression and functional involvement of N-cadherin in embryonic limb chondrogenesis. Development 120(1):177–187
- Oberlender SA, Tuan RS (1994b) Spatiotemporal profile of N-cadherin expression in the developing limb mesenchyme. Cell Adhes Commun 2(6):521–537
- Ochs BG, Muller-Horvat C, Albrecht D, Schewe B, Weise K, Aicher WK, Rolauffs B (2011) Remodeling of articular cartilage and subchondral bone after bone grafting and matrixassociated autologous chondrocyte implantation for osteochondritis dissecans of the knee. Am J Sports Med 39(4):764–773. doi:10.1177/0363546510388896
- O'Driscoll SW, Keeley FW, Salter RB (1986) The chondrogenic potential of free autogenous periosteal grafts for biological resurfacing of major full-thickness defects in joint surfaces under the influence of continuous passive motion. An experimental investigation in the rabbit. J Bone Joint Surg Am 68(7):1017–1035
- Pan J, Zhou X, Li W, Novotny JE, Doty SB, Wang L (2009) In situ measurement of transport between subchondral bone and articular cartilage. J Orthop Res 27(10):1347–1352. doi:10.1002/ jor.20883
- Park J, Gelse K, Frank S, von der Mark K, Aigner T, Schneider H (2006) Transgene-activated mesenchymal cells for articular cartilage repair: a comparison of primary bone marrow-, perichondrium/periosteum- and fat-derived cells. J Gene Med 8(1):112–125. doi:10.1002/jgm.826
- Park JS, Yang HN, Woo DG, Chung HM, Park KH (2009) In vitro and in vivo chondrogenesis of rabbit bone marrow-derived stromal cells in fibrin matrix mixed with growth factor loaded in nanoparticles. Tissue Eng Part A 15(8):2163–2175. doi:10.1089/ten.tea.2008.0532
- Pei M, He F, Boyce BM, Kish VL (2009) Repair of full-thickness femoral condyle cartilage defects using allogeneic synovial cell-engineered tissue constructs. Osteoarthritis Cartilage 17(6):714– 722. doi:10.1016/j.joca.2008.11.017
- Pei M, He F, Li J, Tidwell JE, Jones AC, McDonough EB (2013) Repair of large animal partialthickness cartilage defects through intraarticular injection of matrix-rejuvenated synoviumderived stem cells. Tissue Eng Part A 19(9–10):1144–1154. doi:10.1089/ten.TEA.2012.0351
- Pendleton C, Li Q, Chesler DA, Yuan K, Guerrero-Cazares H, Quinones-Hinojosa A (2013) Mesenchymal stem cells derived from adipose tissue vs bone marrow: in vitro

comparison of their tropism towards gliomas. PLoS One 8(3):e58198. doi:10.1371/journal. pone.0058198PONE-D-12-24856

- Pestka JM, Bode G, Salzmann G, Sudkamp NP, Niemeyer P (2012) Clinical outcome of autologous chondrocyte implantation for failed microfracture treatment of full-thickness cartilage defects of the knee joint. Am J Sports Med 40(2):325–331. doi:10.1177/0363546511425651
- Peterson L, Vasiliadis HS, Brittberg M, Lindahl A (2010) Autologous chondrocyte implantation: a long-term follow-up. Am J Sports Med 38(6):1117–1124. doi:10.1177/0363546509357915
- Pfeiffer E, Vickers SM, Frank E, Grodzinsky AJ, Spector M (2008) The effects of glycosaminoglycan content on the compressive modulus of cartilage engineered in type II collagen scaffolds. Osteoarthritis Cartilage 16(10):1237–1244. doi:10.1016/j.joca.2008.02.014
- Quintin A, Schizas C, Scaletta C, Jaccoud S, Applegate LA, Pioletti DP (2010) Plasticity of fetal cartilaginous cells. Cell Transplant 19(10):1349–1357. doi:10.3727/096368910X506854
- Rackwitz L, Djouad F, Janjanin S, Noth U, Tuan RS (2014) Functional cartilage repair capacity of de-differentiated, chondrocyte- and mesenchymal stem cell-laden hydrogels in vitro. Osteoarthritis Cartilage 22(8):1148–1157. doi:10.1016/j.joca.2014.05.019
- Revell CM, Athanasiou KA (2009) Success rates and immunologic responses of autogenic, allogenic, and xenogenic treatments to repair articular cartilage defects. Tissue Eng Part B Rev 15(1):1–15. doi:10.1089/ten.teb.2008.0189
- Rhee DK, Marcelino J, Baker M, Gong Y, Smits P, Lefebvre V, Jay GD, Stewart M, Wang H, Warman ML, Carpten JD (2005) The secreted glycoprotein lubricin protects cartilage surfaces and inhibits synovial cell overgrowth. J Clin Invest 115(3):622–631. doi:10.1172/JCI22263
- Riley SL, Dutt S, De La Torre R, Chen AC, Sah RL, Ratcliffe A (2001) Formulation of PEG-based hydrogels affects tissue-engineered cartilage construct characteristics. J Mater Sci Mater Med 12(10–12):983–990
- Roos EM, Roos HP, Lohmander LS, Ekdahl C, Beynnon BD (1998) Knee Injury and Osteoarthritis Outcome Score (KOOS)--development of a self-administered outcome measure. J Orthop Sports Phys Ther 28(2):88–96
- Saini S, Wick TM (2003) Concentric cylinder bioreactor for production of tissue engineered cartilage: effect of seeding density and hydrodynamic loading on construct development. Biotechnol Prog 19(2):510–521. doi:10.1021/Bp0256519
- Schwarz S, Koerber L, Elsaesser AF, Goldberg-Bockhorn E, Seitz AM, Durselen L, Ignatius A, Walther P, Breiter R, Rotter N (2012) Decellularized cartilage matrix as a novel biomatrix for cartilage tissue-engineering applications. Tissue Eng Part A 18(21–22):2195–2209. doi:10.1089/ten.tea.2011.0705
- Sharma B, Williams CG, Khan M, Manson P, Elisseeff JH (2007) In vivo chondrogenesis of mesenchymal stem cells in a photopolymerized hydrogel. Plast Reconstr Surg 119(1):112–120. doi:10.1097/01.prs.0000236896.22479.52
- Sharma B, Fermanian S, Gibson M, Unterman S, Herzka DA, Cascio B, Coburn J, Hui AY, Marcus N, Gold GE, Elisseeff JH (2013) Human cartilage repair with a photoreactive adhesivehydrogel composite. Sci Transl Med 5(167). doi:10.1126/scitranslmed.3004838
- Shepherd DE, Seedhom BB (1999) The 'instantaneous' compressive modulus of human articular cartilage in joints of the lower limb. Rheumatology (Oxford) 38(2):124–132
- Sherman SL, Garrity J, Bauer K, Cook J, Stannard J, Bugbee W (2014) Fresh osteochondral allograft transplantation for the knee: current concepts. J Am Acad Orthop Surg 22(2):121– 133. doi:10.5435/JAAOS-22-02-121
- Shum L, Coleman CM, Hatakeyama Y, Tuan RS (2003) Morphogenesis and dysmorphogenesis of the appendicular skeleton. Birth Defects Res C Embryo Today 69(2):102–122. doi:10.1002/ bdrc.10012
- Spiller KL, Maher SA, Lowman AM (2011) Hydrogels for the repair of articular cartilage defects. Tissue Eng Part B Rev 17(4):281–299. doi:10.1089/ten.teb.2011.0077
- Stanish WD, McCormack R, Forriol F, Mohtadi N, Pelet S, Desnoyers J, Restrepo A, Shive MS (2013) Novel scaffold-based BST-CarGel treatment results in superior cartilage repair compared with microfracture in a randomized controlled trial. J Bone Joint Surg Am 95(18):1640– 1650. doi:10.2106/JBJS.L.01345

- Subramanian A, Lin HY (2005) Crosslinked chitosan: its physical properties and the effects of matrix stiffness on chondrocyte cell morphology and proliferation. J Biomed Mater Res A 75(3):742–753. doi:10.1002/jbm.a.30489
- Suzuki S, Muneta T, Tsuji K, Ichinose S, Makino H, Umezawa A, Sekiya I (2012) Properties and usefulness of aggregates of synovial mesenchymal stem cells as a source for cartilage regeneration. Arthritis Res Ther 14(3):R136. doi:10.1186/ar3869ar3869
- Tallheden T, Dennis JE, Lennon DP, Sjogren-Jansson E, Caplan AI, Lindahl A (2003) Phenotypic plasticity of human articular chondrocytes. J Bone Joint Surg Am 85-A(Suppl 2):93–100
- Tayalia P, Mooney DJ (2009) Controlled growth factor delivery for tissue engineering. Adv Mater 21(32–33):3269–3285. doi:10.1002/adma.200900241
- Tegner Y, Lysholm J (1985) Rating systems in the evaluation of knee ligament injuries. Clin Orthop Relat Res 198:43–49
- Teo BJ, Buhary K, Tai BC, Hui JH (2013) Cell-based therapy improves function in adolescents and young adults with patellar osteochondritis dissecans. Clin Orthop Relat Res 471(4):1152–1158. doi:10.1007/s11999-012-2338-z
- Tetlow LC, Adlam DJ, Woolley DE (2001) Matrix metalloproteinase and proinflammatory cytokine production by chondrocytes of human osteoarthritic cartilage: associations with degenerative changes. Arthritis Rheum 44(3):585–594. doi:10.1002/1529-0131(200103)44:3<585::AID-ANR107>3.0.CO;2-C
- Toh WS, Lee EH, Guo XM, Chan JK, Yeow CH, Choo AB, Cao T (2010) Cartilage repair using hyaluronan hydrogel-encapsulated human embryonic stem cell-derived chondrogenic cells. Biomaterials 31(27):6968–6980. doi:10.1016/j.biomaterials.2010.05.064S0142-9612(10)00717-9
- Toh WS, Lim TC, Kurisawa M, Spector M (2012) Modulation of mesenchymal stem cell chondrogenesis in a tunable hyaluronic acid hydrogel microenvironment. Biomaterials 33(15):3835– 3845. doi:10.1016/j.biomaterials.2012.01.065. pii:S0142-9612(12)00178-0
- Towbin H, Staehelin T, Gordon J (1989) Immunoblotting in the clinical laboratory. J Clin Chem Clin Biochem 27(8):495–501
- Tuan RS, Boland G, Tuli R (2003) Adult mesenchymal stem cells and cell-based tissue engineering. Arthritis Res Ther 5(1):32–45
- Tuan RS, Lee FY, Konttinen YT, Wilkinson JM, Smith RL (2008) What are the local and systemic biologic reactions and mediators to wear debris, and what host factors determine or modulate the biologic response to wear particles? J Am Acad Orthop Surg 16(Suppl 1):S42–S48
- Tuli R, Nandi S, Li WJ, Tuli S, Huang X, Manner PA, Laquerriere P, Noth U, Hall DJ, Tuan RS (2004) Human mesenchymal progenitor cell-based tissue engineering of a single-unit osteochondral construct. Tissue Eng 10(7–8):1169–1179. doi:10.1089/ten.2004.10.1169
- Urech DM, Feige U, Ewert S, Schlosser V, Ottiger M, Polzer K, Schett G, Lichtlen P (2010) Antiinflammatory and cartilage-protecting effects of an intra-articularly injected anti-TNF{alpha} single-chain Fv antibody (ESBA105) designed for local therapeutic use. Ann Rheum Dis 69(2):443–449. doi:10.1136/ard.2008.105775
- Vijayan S, Bartlett W, Bentley G, Carrington RW, Skinner JA, Pollock RC, Alorjani M, Briggs TW (2012) Autologous chondrocyte implantation for osteochondral lesions in the knee using a bilayer collagen membrane and bone graft: a two- to eight-year follow-up study. J Bone Joint Surg Br 94(4):488–492. doi:10.1302/0301-620X.94B4.27117
- Vinardell T, Sheehy EJ, Buckley CT, Kelly DJ (2012) A comparison of the functionality and in vivo phenotypic stability of cartilaginous tissues engineered from different stem cell sources. Tissue Eng Part A 18(11–12):1161–1170. doi:10.1089/ten.tea.2011.0544
- Wakitani S, Imoto K, Yamamoto T, Saito M, Murata N, Yoneda M (2002) Human autologous culture expanded bone marrow mesenchymal cell transplantation for repair of cartilage defects in osteoarthritic knees. Osteoarthritis Cartilage 10(3):199–206. doi:10.1053/joca.2001.0504
- Wakitani S, Aoki H, Harada Y, Sonobe M, Morita Y, Mu Y, Tomita N, Nakamura Y, Takeda S, Watanabe TK, Tanigami A (2004a) Embryonic stem cells form articular cartilage, not teratomas, in osteochondral defects of rat joints. Cell Transplant 13(4):331–336

- Wakitani S, Mitsuoka T, Nakamura N, Toritsuka Y, Nakamura Y, Horibe S (2004b) Autologous bone marrow stromal cell transplantation for repair of full-thickness articular cartilage defects in human patellae: two case reports. Cell Transplant 13(5):595–600
- Wakitani S, Nawata M, Tensho K, Okabe T, Machida H, Ohgushi H (2007) Repair of articular cartilage defects in the patello-femoral joint with autologous bone marrow mesenchymal cell transplantation: three case reports involving nine defects in five knees. J Tissue Eng Regen Med 1(1):74–79. doi:10.1002/Term.8
- Waller KA, Zhang LX, Elsaid KA, Fleming BC, Warman ML, Jay GD (2013) Role of lubricin and boundary lubrication in the prevention of chondrocyte apoptosis. Proc Natl Acad Sci U S A 110(15):5852–5857. doi:10.1073/pnas.1219289110
- Werner S, Grose R (2003) Regulation of wound healing by growth factors and cytokines. Physiol Rev 83(3):835–870. doi:10.1152/physrev.00031.200283/3/835
- Wezeman FH (1998) Morphological foundations of precartilage development in mesenchyme. Microsc Res Tech 43(2):91–101
- Whitmire RE, Wilson DS, Singh A, Levenston ME, Murthy N, Garcia AJ (2012) Self-assembling nanoparticles for intra-articular delivery of anti-inflammatory proteins. Biomaterials 33(30):7665–7675. doi:10.1016/j.biomaterials.2012.06.101. pii:S0142-9612(12)00766-1
- Williams RJ (2007) Cartilage repair strategies. Humana Press, Totowa
- Williams R, Khan IM, Richardson K, Nelson L, McCarthy HE, Analbelsi T, Singhrao SK, Dowthwaite GP, Jones RE, Baird DM, Lewis H, Roberts S, Shaw HM, Dudhia J, Fairclough J, Briggs T, Archer CW (2010) Identification and clonal characterisation of a progenitor cell subpopulation in normal human articular cartilage. PLoS One 5(10):e13246. doi:10.1371/journal. pone.0013246
- Wong M, Siegrist M, Wang X, Hunziker E (2001) Development of mechanically stable alginate/ chondrocyte constructs: effects of guluronic acid content and matrix synthesis. J Orthop Res 19(3):493–499. doi:10.1016/S0736-0266(00)90023-8. pii:S0736-0266(00)90023-8
- World Health Organization (2002) The World Health Report 2002: reducing risks, promoting healthy life. World Health Organization
- Xie X, Wang Y, Zhao C, Guo S, Liu S, Jia W, Tuan RS, Zhang C (2012) Comparative evaluation of MSCs from bone marrow and adipose tissue seeded in PRP-derived scaffold for cartilage regeneration. Biomaterials 33(29):7008–7018. doi:10.1016/j.biomaterials.2012.06.058S0142-9612(12)00708-9
- Yamashita A, Nishikawa S, Rancourt DE (2010) Identification of five developmental processes during chondrogenic differentiation of embryonic stem cells. PLoS One 5(6):e10998. doi:10.1371/journal.pone.0010998
- Yamashita A, Liu S, Woltjen K, Thomas B, Meng G, Hotta A, Takahashi K, Ellis J, Yamanaka S, Rancourt DE (2013) Cartilage tissue engineering identifies abnormal human induced pluripotent stem cells. Sci Rep 3:1978. doi:10.1038/srep01978
- Yan H, Yu CL (2007) Repair of full-thickness cartilage defects with cells of different origin in a rabbit model. Arthroscopy 23(2):178–187. doi:10.1016/j.arthro.2006.09.005
- Yan LP, Oliveira JM, Oliveira AL, Caridade SG, Mano JF, Reis RL (2012) Macro/microporous silk fibroin scaffolds with potential for articular cartilage and meniscus tissue engineering applications. Acta Biomater 8(1):289–301. doi:10.1016/j.actbio.2011.09.037. pii:S1742-7061(11)00430-2
- Yoshimura H, Muneta T, Nimura A, Yokoyama A, Koga H, Sekiya I (2007) Comparison of rat mesenchymal stem cells derived from bone marrow, synovium, periosteum, adipose tissue, and muscle. Cell Tissue Res 327(3):449–462. doi:10.1007/s00441-006-0308-z
- Zaslav K, Cole B, Brewster R, DeBerardino T, Farr J, Fowler P, Nissen C (2009) A prospective study of autologous chondrocyte implantation in patients with failed prior treatment for articular cartilage defect of the knee: results of the Study of the Treatment of Articular Repair (STAR) clinical trial. Am J Sports Med 37(1):42–55. doi:10.1177/0363546508322897
- Zavada SR, Battsengel T, Scott TF (2016) Radical-mediated enzymatic polymerizations. Int J Mol Sci 17(2). doi:10.3390/ijms17020195

- Zelle S, Zantop T, Schanz S, Petersen W (2007) Arthroscopic techniques for the fixation of a threedimensional scaffold for autologous chondrocyte transplantation: Structural properties in an in vitro model. Arthroscopy 23(10):1073–1078. doi:10.1016/j.arthro.2007.05.015
- Zhang Z (2015) Chondrons and the pericellular matrix of chondrocytes. Tissue Eng Part B Rev 21(3):267–277. doi:10.1089/ten.TEB.2014.0286
- Zhang W, Chen J, Tao J, Jiang Y, Hu C, Huang L, Ji J, Ouyang HW (2013) The use of type 1 collagen scaffold containing stromal cell-derived factor-1 to create a matrix environment conducive to partial-thickness cartilage defects repair. Biomaterials 34(3):713–723. doi:10.1016/j. biomaterials.2012.10.027. pii:S0142-9612(12)01148-9
- Zhou G, Liu W, Cui L, Wang X, Liu T, Cao Y (2006) Repair of porcine articular osteochondral defects in non-weightbearing areas with autologous bone marrow stromal cells. Tissue Eng 12(11):3209–3221. doi:10.1089/ten.2006.12.3209

MSC Populations for Cartilage Regeneration

Roberto Narcisi, Mairéad A. Cleary, Kavitha Sivasubramaniyan, Pieter A.J. Brama, and Gerjo J.V.M. van Osch

Abstract

Adult mesenchymal stem cells (MSCs) have an excellent capacity to repair tissues since they can proliferate and differentiate to form various tissues, cartilage included. Moreover, MSCs are potentially accessible in high quantities with low donor site morbidity and reasonable cartilage-forming capacity. In 1998, Johnstone et al. (Exp Cell Res 238(1):265–272) were the first that proposed an effective protocol to chondrogenically differentiate MSCs by using transforming growth factor- β (TGF- β), now used by many groups in the world and since then hardly changed. However, MSCs are a heterogeneous population, and the amount

R. Narcisi • K. Sivasubramaniyan

Department of Orthopaedics, Erasmus MC, University Medical Center, Rotterdam, The Netherlands

e-mail: r.narcisi@erasmusmc.nl; k.sivasubramaniyan@erasmusmc.nl

M.A. Cleary Department of Orthopaedics, Erasmus MC, University Medical Center, Rotterdam, The Netherlands

Section of Veterinary Clinical Sciences, School of Veterinary Medicine, University College Dublin, Belfied, Dublin 4, Ireland e-mail: mairead.cleary@ucdconnect.ie

P.A.J. Brama Section of Veterinary Clinical Sciences, School of Veterinary Medicine, University College Dublin, Belfied, Dublin 4, Ireland e-mail: pieter.brama@ucd.ie

G.J.V.M. van Osch (⊠) Department of Orthopaedics, Erasmus MC, University Medical Center, Rotterdam, The Netherlands

Department of Otorhinolaryngology, Erasmus MC, University Medical Center, Rotterdam, The Netherlands e-mail: g.vanosch@erasmusmc.nl and type of cartilage formed are strongly influenced by intra- and inter-donor variation. In this chapter, we mainly focused on surface markers and their modulation by growth factors. We aim to first clarify the characteristics and the embry-onic origin of cartilage progenitor cells (chondroprogenitor), then to summarize the characteristics and the contribution to cartilage repair by MSCs from different origins both in vivo and in vitro, and finally, to show a few examples of promoting articular cartilage phenotype by growth factor administration, in relation to the modulation of surface marker expression. With the exception of the next section focused on embryology, our interest was posed specifically on MSCs from human origin.

2.1 Embryonic Origin of Chondroprogenitor Cells (CPCs)

Adult joints contain progenitor cells (Im 2016). Some insight into the origin, fate, and function of such progenitor cells, including those associated with articular cartilage, has been gained through multiple studies utilizing various conditional reporter mice. Whether these adult progenitor cells are derived from interzone cells, the compact layer of mesenchymal cells from which synovial joints and their specialized tissues arise (Holder 1977), remains to be clarified.

Interzone cells emerge at sites previously occupied by chondrocytes (Craig et al. 1987; Nalin et al. 1995). This led to the hypothesis that interzone cells represent dedifferentiated chondrocytes. Sox9 and Dcx lineage-tracking studies using mouse embryos, however, demonstrate that while articular and growth plate chondrocytes arise from a common population of mesenchymal progenitors, articular chondrocytes arise from distinct populations of these mesenchymal progenitor cells (Soeda et al. 2010; Zhang et al. 2011). This hypothesis is supported by the observed migration of joint progenitor cells into the prospective joint site from flanking regions (Hyde et al. 2008; Pacifici et al. 2006; Koyama et al. 2007; Li et al. 2013).

Indian hedgehog null (Ihh^{-/-}) mouse embryos lack joints in their skeletal elements, which remain completely cartilaginous (Koyama et al. 2007). In Ihh^{-/-};Gdf5 mouse embryos, in which Gdf5 is selectively expressed by interzone cells, Gdf5expressing cells were observed at respective joint sites, but not within the future joint site itself; instead they flanked and surrounded the uninterrupted joint site (Koyama et al. 2007; Storm and Kingsley 1996). The absence of a joint in Ihh^{-/-} mouse embryos may arise from the inability of the Gdf5-expressing cells to migrate into the prospective joint area.

Recruitment and immigration of surrounding cells into the interzone are also supported by studies involving the TGF- β type II receptor that is essential for joint formation, especially in the hands/feet (Spagnoli et al. 2007; Li et al. 2013). At E13.5, Tgfbr2-positive cells are present in the dorsal and ventral regions of the joint but completely absent from the central region of the interzone. By E16.5 and postnatally, positive cells are present in various joint tissues, although not in articular cartilage. Tgfbr2-deficient mouse embryos display fusion of digit joints, a



Fig. 2.1 Model of joint development. Unknown upstream mediators determine the location of the future joint site along *Sox9/Col2a1/Dcx*-expressing cells within the uninterrupted cartilaginous template. Soon after, the interzone mesenchymal population is defined and morphologically observable among *Sox9/Col2a1/Matn1*-expressing cells. *Gdf5/Dcx*-expressing cells differentiate to form the articular chondrocytes of the synovial joint. Remaining interzone cells are involved in forming other joint tissues and structures (Adapted from Decker et al. 2014)

phenotype which can be rescued by blocking the MCP-5 receptor CCR2, further supporting the need for low expression of Mcp5 for interzone and joint formation (Longobardi et al. 2012). Moreover, Tgfbr2-expressing cells present in the adult joint are observed, through BrdU-labeling experiments, to constitute slow-cycling stem/progenitor cells (Li et al. 2013).

Thus, along the initial uninterrupted Sox9/Col2/Dcx-expressing cartilaginous anlagen, the future joint site develops, the site itself identified by as yet unknown upstream morphogenetic and determination mechanisms. Soon after, the interzone mesenchymal population is specified through Gdf5 expression with concomitant cell recruitment and maintenance of Dcx expression. Dorsal and ventral flanking cells activate Tgfbr2 expression and articular chondrocytes and then arise from Gdf5/Dcx-expressing cells with a Sox9/Col2a1+/Matn1 history (Fig. 2.1).

2.2 In Vivo Presence of MSCs in Adult Joints

Endogenous stem or progenitor cells contribute to maintenance of healthy tissues by acting as reservoirs of repair cells or as immunomodulatory sentinels to reduce inflammation. In a joint environment, these cell populations exist in tissues adjacent to articular cartilage, including the articular cartilage itself, bone marrow, synovium, synovial fluid, and infrapatellar fat pad (Im 2016). A great deal has been learnt in recent years about the isolation and characterization of endogenous MSCs from these tissues. The characteristics of endogenous stem or progenitor cell populations from these tissues are discussed below.

2.2.1 Characteristics of MSCs In Vivo

2.2.1.1 Articular Cartilage-Derived Chondroprogenitor cells (CPCs)

Chondrocytes are the most abundant cells within articular cartilage. A small population of these chondrocytes constitutes the chondroprogenitors (Dowthwaite et al. 2004; Khan et al. 2009). CD166 is the most broadly used biomarker for the identification and localization of these progenitor cells in human articular cartilage with high chondrogenic potential (Pretzel et al. 2011; Swart 2002). One study investigated the zonal distribution of CD166⁺ CPCs in articular cartilage, where it was verified that these cells were almost exclusively located in the superficial and middle zones (Pretzel et al. 2011). Notably, 99% of the mesenchymal progenitor cells in that study co-expressed CD166 with another surface marker: the TGF- β co-receptor CD105. Indeed, the superficial zone seems to be of central importance for the growth of articular cartilage in young animals (Hunziker et al. 2007) and presumably relies on a progenitor cell population located in this zone (Hayes et al. 2001). Apart from CD166, also expression of Notch-1 (Dowthwaite et al. 2004) or triple positivity for CD44/CD151/CD49c (Grogan et al. 2007) or CD9/CD90/CD166 (Fickert et al. 2004) has been shown to isolate chondroprogenitors from articular cartilage.

2.2.1.2 Bone Marrow-Derived MSCs (BMSCs)

Bone marrow (BM)-derived MSCs are most extensively studied and best characterized. Several markers were identified that are suitable to isolate MSCs directly from fresh bone marrow samples including antibodies specific for a variety of cell surface molecules, CD49a, CD63, CD73 (SH3/SH4), CD90, CD105 (SH2), CD106, CD140b, CD146, CD200, CD271, CD349, TNAP, Hsp90, GD2, TM4SF1, and NG2 as well as orphan antigens defined by antibodies STRO-1 and 3G5 (Lv et al. 2014; Harichandan et al. 2013; Harichandan and Buhring 2011). In other approaches, BMSCs were enriched using negative selection, by depletion of hematopoietic cells, employing markers such as CD14, CD34, CD45, and/or CD235 (glycophorin A) and other lineage-negative markers (Harichandan and Buhring 2011).

CD271 is the most widely used marker for the characterization and purification of primary human bone marrow MSCs (Alvarez-Viejo et al. 2015). MSCs reside in CD271^{bright} cells, while CD271^{dim} cells comprise CD45⁺ hematopoietic cells (Buhring et al. 2007). More selective markers for MSC isolation than CD271 are SUSD2 (W5C5) and CD140b, as they are expressed on CD271^{bright}, but not CD271^{dim} cells (Sivasubramaniyan et al. 2012, 2013). Another well-known marker for BMSCs is Stro-1. But it is unsuitable as a sole marker to separate MSCs from its harboring tissue, as greater than 95% of Stro-1⁺ cells in the human BM are glycophorin A expressing nucleated erythroid cells (Simmons and Torok-Storb 1991). Delorme

et al. (2008) reported CD73, CD130, CD146, CD200, and integrin $a5/\beta5$ as markers to enrich colony-forming unit fibroblasts (CFU-Fs) from bone marrow-derived mononuclear cells (MNCs), while other known MSC markers CD49b, CD90, and CD105 showed less enrichment. The neural ganglioside GD2 was introduced by Martinez et al. (2007) as a stand-alone marker to isolate BMSCs. However, in another study, GD2 was found to be expressed only in cultured MSCs and not in primary MSCs (Sivasubramaniyan et al. 2013).

Several groups have reported that BMSCs are heterogeneous with respect to their growth and differentiation potential (Lv et al. 2014). However, little information exists about markers that discriminate between anatomically and functionally distinct MSC subsets. A few studies have defined CD146 and SSEA-3 as markers of perivascular BMSCs while CD56 and CD166 as markers of endosteal BMSCs (Sivasubramaniyan et al. 2012; Tormin et al. 2011). In addition, CD56⁺ BMSCs give rise to osteoblasts and chondrocytes but not to adipocytes (Battula et al. 2009), while CD56-BMSCs give rise to osteoblasts and adipocytes but poorly to chondrocytes. This suggests that apart from the distinct surface antigen expression profile and differentiation potential, bone-lining and perivascular MSCs may also have distinct, yet to be identified, functional characteristics/properties.

2.2.1.3 Synovium-Derived MSCs (SM-MSCs)

In 2001, De Bari et al. introduced synovium as a source of MSCs that possess chondrogenic potential (De Bari et al. 2001). Recent work has shown subpopulations of MSCs to express different surface markers, such as CD105, CD166, CD90, CD9, and CD271 (Chang et al. 2013; Ogata et al. 2015; Van Landuyt et al. 2010). Chang et al. (2013) reported that CD105- and CD166-enriched cells derived from human synovium may be valuable sources for cartilage regeneration due to their enhanced chondrogenic potential. The most widely used BMSC marker CD271 is shown to be present in a specific subpopulation of inflamed synovium but not at all or barely present in expanded cell populations from the same patients (Van Landuyt et al. 2010; Ogata et al. 2015). In another study, Van Landuyt et al. (2010) reported the detection of CD34 expression on a subpopulation of CD271⁺ MSCs, but the chondrogenic differentiation ability of this population is not known. So far, CD90 is the most commonly used marker for isolation of SM-MSCs. It has been used in combination with CD9/CD166 or CD271 to isolate a subpopulation of highly chondrogenic SM-MSCs (Ogata et al. 2015; Fickert et al. 2003). But these studies did not characterize the chondrogenic ability of all the MSC subpopulations from synovium, thus questioning the reliability of CD90 or CD271 in isolating highly chondrogenic SM-MSCs.

2.2.1.4 Synovial Fluid-Derived MSCs (SF-MSCs)

SF-MSCs form a pool of highly clonogenic cells with chondrogenic potential and are present both in healthy and OA joints (de Sousa et al. 2014; Jones et al. 2008). The exact origin of SF-MSCs is unclear. Morito et al. showed that they originate neither from BM nor from circulating MSCs but probably from the synovium or cartilage (Morito et al. 2008; Lee et al. 2012). The phenotype of freshly isolated human SF-MSCs is not yet known.

2.2.1.5 Infrapatellar Fat Pad-Derived MSCs (IF-MSCs)

Subcutaneous human adipose tissue is an interesting source of multipotent progenitors (Zuk et al. 2002). In recent days, also human infrapatellar fat pad is newly gathering attention as a source of MSCs (Khan et al. 2012). Jurgens et al. (2009) characterized stem cells in the freshly isolated stromal fraction based on the presence of the early marker CD34 and the absence of the endothelial marker CD31. These CD34⁺CD31⁻ cells were characterized further and shown to be additionally positive for the stem cell-associated markers CD29, CD54, CD90, CD105, and CD166. Others have also characterized the phenotype of IF-MSCs but only looked into culture-passaged cells. Thereby, certain markers such as CD34 were found to be negative, whereas in fact they are present on the surface of freshly isolated cells and lost upon culturing (Wickham et al. 2003). In another study, a small subpopulation of human IF-MSCs has been shown to express the pericyte marker 3G5, and anatomically these cells are localized to the perivascular region (Khan et al. 2008). In a recent study, CD44 was used to isolate IF-MSCs, and these freshly isolated IF-MSCs have been shown to have enhanced chondrogenic potential in vivo when seeded on a cartilage-extracellular matrix-derived scaffold (Almeida et al. 2015). These findings indicate that infrapatellar fat pad may be a promising cell source for endogenous MSCs.

2.2.2 Contribution to Repair by MSCs In Vivo

Until now, there is no clear blueprint of the host cell sources that participate in in situ cartilage regeneration. But more new knowledge is being gained in this field starting from the work of Hunziker and Rosenberg (1996) where for the first time a combination of fibrin, cells, and growth factors was used to repair a cartilage defect. For instance, Lee et al. (2010) showed that the articular surface of the synovial joint can be regenerated with TGF- β 3 infused in a bioscaffold composite of poly- ϵ -caprolactone and hydroxyapatite. Though the source of endogenous stem cells bringing about the repair is not known, they speculate that the endogenous cells are derived from stem or progenitor cells of synovium, bone marrow, infrapatellar fat pad, and perhaps vasculature.

First approaches to heal cartilage by in situ regeneration date back as early as 1959 (Muller and Kohn 1999). The Pridie technique was directed at recruitment of BMSCs to cartilage defects by drilling small holes into the subchondral bone that underlies regions of damaged cartilage. It was refined later on, by reducing the size of the perforations, and was called microfracture technique which is now a frequently performed and well-studied procedure (Richter 2009; Steadman et al. 2001). While microfracture is restricted to lesions of 1–4 cm² size, autologous matrix scaffolds that induce chondrogenesis were developed and used to treat larger defects in combination with microfracture (Kramer et al. 2006) (for details see Chap. 5). Development of scaffolds with different materials and biophysical properties containing bioactive factors aiming to attract endogenous MSCs, especially BMSCs, is a growing field of research. In a study, Koelling et al. (2009) found some

evidence that chondrogenic cells migrate from the bone marrow to the cartilage defect through breaks in the tidemark. Hence, for full-thickness defects, the contribution of BMSCs is most likely, but for partial thickness defects, where direct access of BMSCs to the joint cavity is restricted, MSCs from synovium (Hunziker and Rosenberg 1996) and joint fat pads could be involved.

Cells derived from human synovium were shown to have the highest chondrogenic potential among the various mesenchymal tissue-derived cells, indicating a possible source for cartilage repair (Sakaguchi et al. 2005). Nishimura et al. (1999) showed that explants of synovium embedded in agarose undergo chondrogenesis when cultured in the presence of TGF- β 1. The data indicate a possible synovial origin for the chondrocytic cells. Furthermore, these data are consistent with the clinical findings of synovial chondrogenesis leading to synovial chondromatosis (Nishimura et al. 1999). It was recently demonstrated by Kurth et al. (2011) that there is an in vivo presence of slow-cycling SMSCs in synovium of murine knee joints. Through iododeoxyuridine labeling and surface staining, these cells were shown to increase in number after cartilage injury, showing their activation in response to cartilage damage. Hence, homing of stem cells from synovium to sites of cartilage injury is an appealing concept that warrants further investigation.

Cartilage-derived progenitors have been observed in human, equine, and bovine articular cartilage (Jiang and Tuan 2015) and have been identified as slow-cycling cells by pulse-chase experiments in a murine model (Candela et al. 2014). In a recent study, Tong et al. (2015) confirmed the existence of CPCs in situ for the first time in murine model, both in normal and OA articular cartilage, and showed that these CPCs were activated from resting state in OA. Regarding the ability of articular cartilage to self-repair, they speculate that this ability probably depends on the presence of CPCs, as the number of transient-proliferating CPCs is synchronous with the OA progression in the early stage.

In the infrapatellar fat pad, MSCs are localized in the perivascular region, and hence, they can infiltrate into the synovial fluid through the synovium and participate in cartilage repair. The direct involvement of IF-MSCs in cartilage repair in situ is yet to be explored.

MSCs are present in the SF from knee joints of healthy individuals as well as from individuals with OA, RA, ligament injury, and meniscal injury (Jones et al. 2004, 2008; Matsukura et al. 2014; Morito et al. 2008). Diseased or injured joint SF has greater MSC numbers than healthy joint SF (Jones et al. 2004; Matsukura et al. 2014; Morito et al. 2008), with total number positively correlating with the period of injury (Morito et al. 2008) or severity of OA (Jones et al. 2004; Lee et al. 2012; Sekiya et al. 2012). These SF-MSCs are possibly derived from dislodged synovial fragments. This is particularly noteworthy since synovial stem cells have been proposed to contribute to spontaneous cartilage repair (Hunziker and Rosenberg 1996; Kurth et al. 2011), and synovial fluid could therefore be viewed as a possible conduit for their passage (Jones et al. 2008). Taken together, these findings provide a platform to explore the role of SF-MSCs in superficial cartilage homeostasis and repair.

2.3 MSCs In Vitro

It has been proposed that the best markers for identification of progenitor cells may be different between freshly isolated and culture-expanded cells. The first definitive markers of MSCs in vitro were proposed in the pioneering study of Pittenger et al. (1999) and included CD105 (SH2) and CD73 (SH3). These two markers alongside CD90 remain the primary molecules used to identify MSCs by the International Society for Cellular Therapy (ISCT) which also advices that MSCs should be negative for the expression of CD11b or CD14, CD19 or CD79a, CD34, CD45, and HLA-DR (Dominici et al. 2006). This is primarily to allow the exclusion of hematopoietic cells which may contaminate MSC cultures.

Cultured MSCs, however, are uniformly and strongly positive/negative for many markers regardless of their passage or time in culture. That many markers are expressed at similar levels in early- and late-passage MSCs indicates their value may be limited to basic MSC characterization only. The following is a concise review of in vitro markers of MSCs from different tissue sources (Table 2.1) and their correlation with chondrogenic potential.

2.3.1 Articular Cartilage-Derived MSCs (AC-MSCs)

Barbero et al. (2003) were the first to describe the existence of a subpopulation of cells within the dedifferentiated adult human articular chondrocyte population that, at clonal level, possess properties of MSCs, including tri-lineage differentiation potential. Hoechst 33342 dye uptake (Grogan et al. 2009) and adhesion to fibronectin (Williams et al. 2010) are utilized in an attempt to isolate this population of cells from the chondrocyte population. Equally, these cells can be isolated by their ability to migrate out of explants in vitro, although only from damaged cartilage (Koelling et al. 2009), in which they are present in higher numbers (Alsalameh et al. 2004).

AC-MSCs migrate from diseased cartilage in response to nerve growth factor (NGF) (Jiang et al. 2015) raising the possibility that these NGF-responsive cells are native CD271⁺ cells, as CD271 expression is not expressed during in vitro expansion (Jiang et al. 2015; Koelling et al. 2009). In contrast, almost all clonal cells obtained by adhesion to fibronectin (Williams et al. 2010) and that have high chondrogenic capacity express CD49e (Grogan et al. 2007). Clonal cells with high chondrogenic capacity also express CD49c, CD49f, CD166, CD44, CD90, and CD151 higher than clonal cells with low chondrogenic capacity (Grogan et al. 2007); singly sorted CD49c^{Bright} and CD44^{Bright} cells have greater chondrogenic capacity than unsorted chondrocyte populations (Grogan et al. 2007).

Notch-1, Stro-1, and CD106 are highly expressed on clonal cells compared to articular chondrocytes (Ustunel et al. 2008; Williams et al. 2010); however, they are too widely expressed on mature chondrocytes in vivo to be useful (Grogan et al. 2009). Similarly, despite a CD105/CD166-enriched adherent population having chondrogenic capacity comparable to BMSCs, these markers are also expressed in vivo (Alsalameh et al. 2004). The combination of CD9/CD90/CD166 to obtain cells with chondrogenic differentiation capacity after culture may prove useful; however, further investigation is warranted (Fickert et al. 2004).

| Marker | Percentage of positive cells and reference |
|-----------------|---|
| B2microglobulin | BM: +++ [7] |
| BMPR1A | SF: – [16] |
| CXCR4 | SF: - [16] |
| | FP: - [1] |
| D7-FIB | BM: +++ [7–8] |
| | SM: ++ [12] |
| | FP: +++ [2] |
| FLK-1 | $BM: -[15] \pm [9]$ |
| | SM: ± [9] |
| | SF: – [16] |
| Glycophorin A | BM: – [7] |
| | SM: - [11] |
| HLA-I | BM: +++ [7] |
| HLA-II | BM: – [7] |
| HLA-DR | BM: - [4] |
| | SM [11] SF' - [4] |
| | $FP: \pm [3-4]$ |
| KDR | SM: - [15] |
| | FP: - [1] |
| STRO-1 | AC: ± [19] |
| | BM: $-$ to $+$ [15], \pm to $+++$ [9] |
| | $SM: - [11], -t0 + [15], \pm t0 + + + [9]$ SF: + t0 ++ [16] |
| TGE6R11 | BM:- [8] |
| TOT pictr | FP: - [2] |
| CD3 | BM: - [7] |
| CD9 | BM: +++ [7] |
| | FP: + [3] |
| CD10 | BM: $-$ to $+$ [15], \pm to $+++$ [9] |
| | SM: $-[15]$, $+/1$ to $++[9]$, \pm to $+[11]$, $+$ to $++[15]$ |
| | SF: - to + [16] ED: [2] |
| CD11a | SM: - [11] |
| CDIIa | FP: -[3] |
| CD11b | BM: - [7] |
| | FP: - [3] |
| CD11c | FP: - [3] |
| CD13 | BM: \pm to + [7], +++ [4,8] |
| | SM: + to +++ [11], ++ to +++ [12] |
| | SF: +++ [4] EP: +++ [2 5] |
| CD1/ | 11. ттт [2,3] ВМ: _ [7] |
| CDIT | SM: -[11] |
| | FP: - [3] |
| | A |

Table 2.1 Summary of the markers expressed by human MSCs in vitro

(continued)

| Marker | Percentage of positive cells and reference |
|--------|---|
| CD15 | BM: – [7] |
| CD16 | BM: – [7] |
| CD18 | BM: - [7] |
| | FP: - [3] |
| CD19 | BM: – [7] |
| CD28 | BM: – [7] |
| CD29 | BM: +++ [7] |
| | SM: ++ to +++ [15] |
| | FP: ++ [3,5] |
| CD31 | BM: $-[7,10]$, $-$ to $\pm [4,9]$ |
| | $SM: -[11], -to \pm [9]$ SE: [4.16] |
| | FP: -[3,4] + [5] |
| CD34 | $BM: -[4, 6-10], \pm [9]$ |
| | $SM: -[13], -to \pm [9], -to + [11]$ |
| | SF: $-[16,18], -to \pm [4], +[17]$ |
| | $FP: - [2], \pm [1,3,4,6], + [5]$ |
| CD36 | BM: – [4,7] |
| | SF: -[4] |
| CD20 | FP: ± [4] |
| CD38 | BM: - [/] |
| CD44 | $BM: +++ [6, 7, 8, 9, 10]$ $SM: ++ [10, 14] ++ t_{0} +++ [11, 15] +++ [0, 12]$ |
| | SW. $++ [10, 14], ++ (0 +++ [11, 15], +++ [9, 15]$ SF: $++ t_0 +++ [16, 18] +++ [4, 17]$ |
| | FP: +++ [2,3,4,5,6] |
| CD45 | BM: - [6,7,8,9,10] |
| | SM: - [10,11,13], - to + [12], ± [9] |
| | SF: - [16,18] |
| | $FP: -[1,2,3,5]. \pm [6]$ |
| CD49a | BM: $++$ to $+++$ [7] |
| CD40b | $SM. = 10 + + [15], \pm 10 + [11]$ |
| CD490 | $DM_{1} \pm 10 \pm [7]$ |
| CD49c | BM: ++ to +++ $[7]$, +++ $[4]$ SF: ++ to +++ $[4]$ |
| | FP: +++ [4] |
| CD49d | BM: + to ++ [4] |
| | SM: - [11] |
| | SF: ++ [4] |
| | FP: +++ [4] |
| CD49e | BM: ++ to +++ [7] |
| | FP: ++ [3] |
| CD49t | BM: -[/], +[4] |
| | $51^{\circ}, \pm [4]$ FP· + [4] |
| CD50 | BM [.] – [7] |
| 0000 | $FP: \pm [3]$ |
| | |

 Table 2.1 (continued)

| Marker | Percentage of positive cells and reference |
|--------|---|
| CD51 | BM: +++ [7] |
| CD54 | BM: ± to ++ [7], ± to +++ [9], + [4,10] |
| | SM: ± to +++ [9], + [15] |
| | SF: + [4], ++ [16] |
| | FP: \pm [3], + to ++ [4] |
| CD55 | BM: +++ [7] |
| | SM: – [11] |
| CD56 | BM: $-$ [7], \pm [6] |
| | SM: - [11] |
| | FP: $-[6], \pm [3]$ |
| CD58 | BM: ++ to +++ [7] |
| CD59 | BM: +++ [4,7] |
| | SF: +++ [4] |
| | FP: +++ [3,4] |
| CD61 | BM: – [7] |
| CD62 | SM: – [11] |
| CD62e | BM: – [7] |
| | FP: + [3] |
| CD62L | BM: – [7] |
| CD62P | BM: – [7] |
| CD68 | SM: - [11] |
| CD71 | BM: - [7], + to ++ [4] |
| | SF: ± [4] |
| | FP: [4] |
| CD73 | BM: +++ [4,8] |
| | SM: + to ++ [11,13], ++ to +++ [12,14,15] |
| | SF: ++ to +++ [18], +++ [4] |
| | FP +++ [1,2,4] |
| CD90 | AC: +++ [19] |
| | BM: +++ [4,6,7,9,10] |
| | SIVI: $++ [10, 13, 14, 13], +++ [9]$ SF: $\pm t_0 \pm \pm [16] \pm \pm [18]$ |
| | FP: +++ [1, 4, 5, 6] |
| CD95 | BM: - to + [7] |
| CD)5 | SF: +++ [4.17.18] |
| CD102 | BM: - [7] |
| CD104 | BM: - [7] |
| CD105 | $\Delta C \cdot \pm [19]$ |
| CD105 | BM: $++$ to $+++$ [7.10], $+++$ [4.6.8.9] |
| | SM: + to +++ [13], ++ [12], ++ to +++ [10,14,15], +++ [9,11] |
| | SF: – to + [18], + to ++ [16], +++ [4] |
| | FP: ++ [5], +++ [1,2,3,4,6] |
| CD106 | BM: - to + [15], ± to +++ [9], ++ [4], ++ to +++ [7] |
| | SM: - [11], - to + [10, 12,15], ± to +++ [9], + [13] |
| | $SF: -[16], -to \pm [4]$ |
| | FP: \pm [4], + [2,3,5] |

 Table 2.1 (continued)

(continued)

| Marker | Percentage of positive cells and reference |
|---------|---|
| CD117 | BM: $-[7,8,10], -to \pm [4,9]$ SM: $-[10,11], -to \pm [9]$ SF: $-[4,16]$ FP: $-to \pm [4]$ |
| CD133 | BM: - [7,8] SM: - [11] FP: - [1,2] |
| CD133/1 | BM: - [4] SF: - [4] FP: - [4] |
| CD140b | BM: +++ [4] SF: +++ [4] FP: +++ [4] |
| CD146 | BM: ++ [8] SM: - to + [12] FP: + [2] |
| CD147 | BM: ++ to +++ [15] SM: ++ to +++ [15], +++ [9] SF: ++ [16] |
| CD151 | BM: +++ [4,8] SM: ++ to +++ [14] SF: +++ [4] FP: ++ [5], +++ [2,4] |
| CD166 | AC: ++ [19] BM: ± to + [7], ± to +++ [9], ++ [15], +++ [4,8] SM: ± [11], ± to +++ [9], + to ++ [10,13], ++ to +++ [12,15], ++ [14] SF: - to + [16], +++ [4] FP: ++ [3,5], +++ [2,4] |
| CD271 | BM: $-[8]$, $-to \pm [4,7,9,10]$, $+[15]$ SM: $-to \pm [9]$, $-to + [10,12]$, $+[13]$ SF: $-[4,16]$ FP: $-to \pm [4]$, $+[2]$ |

Table 2.1 (continued)

Percentage of positive cells: (+++) = 80-100%, (++) = 40-80%, (+) = 15-40%, (±) = 5-15%, (-) = 0-5%. When the authors reported different percentage of expression following the MSC expansion in vitro, a range of expression is indicated in the table. Due to the large number of literature available, for some of the listed markers only a representative selection of the literature is reported. References (see the reference paragraph for more details): [1] Lopez-Ruiz E. et al., *Osteoarthritis and Cartilage* (2013), [2] Jones E.A. et al., *Arthritis Rheum* (2004) and *Arthritis Rheum* (2008), [3] Wickham M.Q. et al., *Clin Orthop Relat Res* (2003), [4] Alegre-Aguaron E. et al., *Cells Tissues Organs* (2012), [5] Lopa S. et al., *Eur Cell Mater* (2014), [6] Ding D.C. et al., *Cell Transplant* (2015), [7] de la Fuente R. et al., *Exp Cell Res* (2004), [8] English A. et al., *Rheumatology* (2007), [9] Shirasawa S. et al., *J Cell Biochem* (2006), [10] Sakaguchi Y. et al., *Arthritis Rheum* (2005), [11] Jo C.H. et al., *Cytotherapy* (2007), [12] Karystinou A. et al., *Rheumatology* (2009), [13] Arufe M.C. et al., *J Cell Biochem* (2010), [14] Jones E. et al., *Ann Rheum Dis* (2010), [15] Han H.S. et al., *J Orthop Res* (2014), [16] Morito T. et al., *Rheumatology* (2008), [17] Lee D.H. et al., *Osteoarthritis and Cartilage* (2012), [18] Matsukura Y. et al., *Clin Orthop Relat Res* (2014), [19] Ozbey O. et al., *Acta Histochem* (2014).

AC articular cartilage-derived MSCs, *BM* bone marrow-derived MSCs, *SM* synovium membranederived MSCs, *SF* synovial fluid-derived MSCs, *FP* infrapatellar fat pad-derived MSCs

2.3.2 Bone Marrow-Derived MSCs (BMSCs)

The studies of Johnstone and Pittenger were the first to demonstrate and characterize individual clonal BMSCs and their chondrogenic or multi-lineage potential (Johnstone et al. 1998; Pittenger et al. 1999), establishing that not all clonal cells are capable of chondrogenesis. Colony cells were uniformly positive for CD105 (SH2), CD73 (SH3), CD29, CD44, CD71, CD90, CD106, CD120a, and CD124 and negative for CD14, CD34, and CD45, similar to their parent culture. From the beginning, therefore, these surface markers appeared not suitable for exclusive identification of BMSCs with high chondrogenic potential. Indeed, many of the surface markers characterized for BMSCs are acquired during culture and remain consistently expressed (Alegre-Aguaron et al. 2012). This is in contrast to the chondrogenic potential of these cells, which decreases with culture (Banfi et al. 2000; Bonab et al. 2006). Furthermore, CD105-sorted populations have been shown to possess similar chondrogenic potential and are comparable to the total, unsorted populations (Majumdar et al. 2000; Cleary et al. 2016).

Conversely, other markers, such as CD106, have been identified which appear more sensitive to culture conditions, including passage number, with declining expression in later passages (Fukiage et al. 2008). Moreover, CD106 may be a marker of MSC differentiation potential as it is strongly downregulated after chondrogenic differentiation (Gronthos et al. 2003). CD271 expression, present on a small percentage of freshly isolated BMSCs, also decreases with culture. However, this decrease occurs almost immediately during the first passage (Jones et al. 2006; Quirici et al. 2002; Mifune et al. 2013). Both CD271⁺ and CD271⁻ BMSCs, though, are capable of healing chondral defects in vivo, although CD271⁻ repair ECM is less rich in proteoglycans (Hermida-Gomez et al. 2011; Mifune et al. 2013). Similarly, CD146⁺ MSCs derived from bone marrow have higher CFU-F ability and proteoglycan content when compared to the total or CD146⁻ population (Hagmann et al. 2014; Sacchetti et al. 2007; Kaltz et al. 2010). These observations require further investigation, however, as no difference in CFU-F potential and differentiation potential has been reported (Espagnolle et al. 2014).

2.3.3 Synovium-Derived MSCs (SM-MSCs)

Although similar to MSCs derived from other sources, synovium-derived MSCs often seem to possess superior chondrogenic potential and higher CFU-F numbers (Jo et al. 2007; Sakaguchi et al. 2005; Shirasawa et al. 2006). In vitro, chondrogenically differentiated synovium-derived MSCs acquire the expression of markers associated with the stable cartilage phenotype of articular cartilage-derived cells. This phenotype, however, is only transient, and chondrogenically differentiated SM-MSCs fail to form stable cartilage in vivo (De Bari et al. 2004).

Surface marker expression is characteristic of MSCs and is relatively stable during culture, with the exception of the first passage; CD14, CD34, CD45, and HLA-DR disappear, CD105 and CD166 appear, and CD10, CD13, CD44, CD49a, and CD73 increase. Thereafter, only mild fluctuations in expression occur (Jo et al. 2007; Nagase et al. 2008). A CD9⁺/CD90⁺/CD166⁺ population of synovium-derived cells, the total frequency of which increases with culture, represents an osteochondral progenitor population. There is, however, no difference between this and the unsorted population (Fickert et al. 2003). The CD34⁻/CD44⁺/CD90⁺ subpopulation too possesses multipotent differentiation potential (Lee et al. 2012).

CD44 and CD90 expression alone correlates positively with chondrogenic potential (Jo et al. 2007; Jones et al. 2010), with CD90 expression correlating with pellet weight (Jo et al. 2007). CD90 is also expressed on >80% of CD271⁺ MSCs from synovium (20–30% of the total population) which, along with CD73⁺ cells, have better chondrogenic potential (Arufe et al. 2010; Harvanova et al. 2011). Indeed, CD73⁺ cells also displayed the best osteogenic phenotype indicating that this surface protein is likely a marker of osteochondral cells. CD105⁺ subpopulations also show chondrogenic potential, with expression also on >80% of CD271⁺ cells (Arufe et al. 2009, 2010). A population of CD271⁺/CD90⁺ in combination with CD73, CD44, or CD105, therefore, may prove the most chondrogenic. Moreover, CD14⁺ cells dampen the chondrogenic capacity of SM-MSCs. Depletion of this subpopulation is, therefore, likely to improve the chondrogenic potential of the starting population (Han et al. 2014).

2.3.4 Synovial Fluid-Derived MSCs (SF-MSCs)

Several studies have demonstrated no difference in chondrogenic potential for MSCs derived from SF, BM, and SM (Kurose et al. 2010; Lee et al. 2012). Indeed, SF-MSCs are assumed to be derived from either BM or SM, with current evidence weighing in the favor of an SM origin (Jones et al. 2004; Matsukura et al. 2014; Morito et al. 2008; Sekiya et al. 2012). However, others indicate that differences in chondrogenic potential may exist between SF-derived MSCs and MSCs derived from other sources, including having weaker chondrogenic potential than BM- or IF-derived MSCs (Alegre-Aguaron et al. 2012).

Akin to MSCs derived from other tissue sources, a proportion of cultured SF-MSCs can be CD271⁺ (Jones et al. 2004), although at levels lower than MSCs from BM (Alegre-Aguaron et al. 2012; Jones et al. 2004). Similarly, cultured SF-MSCs express different levels of CD117, CD106, CD71, CD54, CD49c, HLA-DR, CD34, CD166, and CD133/1 to BMSCs (Alegre-Aguaron et al. 2012); SF-MSCs increase their expression of CD34 and CD49d, while decreasing their expression of CD71, CD106, and CD271 (Alegre-Aguaron et al. 2012).

While no difference is observed in the ability of CD90⁺ and CD90⁻ SF-MSCs to undergo osteogenesis or adipogenesis, greater chondrogenic potential exists in the CD90⁺ fraction. This potential, however, is not superior to that of the total, unsorted population (Krawetz et al. 2012). CD105⁺ SF-MSCs are also capable of tri-lineage differentiation (Harvanova et al. 2011), but further studies are required to understand whether this fraction is superior to the total or negative fraction.

2.3.5 Infrapatellar Fat Pad-Derived MSCs (IF-MSCs)

The presence of MSC-like cells within a cellular population derived from the infrapatellar fat pad (IPFP) was first described by Wickham et al. (2003). Such cells can be isolated in higher quantities from the IPFP than from other tissue sources (Khan et al. 2007) and are phenotypically comparable to MSCs, including their differentiation and CFU-F ability (English et al. 2007). IF-MSCs, however, have been described to maintain their chondrogenic differentiation capacity longer in culture than MSCs derived from other tissues (English et al. 2007). Moreover, several studies have identified higher chondrogenic capacity in this MSC source (Ding et al. 2015; English et al. 2007; Khan et al. 2008; Lopa et al. 2014). IF-MSCs seem to lack age-related declines in proliferative and differentiation potential (Khan et al. 2008), and similar to other MSC populations, when expanded in the presence of fibroblast growth factor-2 (FGF2), proliferation and chondrogenic potential is enhanced (Khan et al. 2008).

Cell surface marker expression of these cells indicates a profile similar, although not identical, to other MSC populations; IF-MSCs express relatively higher levels of CD34, CD45 (Ding et al. 2015), CD271 (English et al. 2007), and CD106 (Mochizuki et al. 2006; Lopa et al. 2014), with lower expression of CD146 (English et al. 2007) and CD10 (Mochizuki et al. 2006). The maintenance of CD271 expression on a subpopulation of cultured IF-MSCs raises the possibility that in vivo, MSCs present in the infrapatellar fat pad also express CD271 (English et al. 2007). This observation remains unclear, however, as CD271, CD45, and CD34 expression on cultured IF-MSCs is often described as negative (Khan et al. 2007, 2012; English et al. 2007). 3G5 is also observed on a subpopulation of these cultured IF-MSCs (Khan et al. 2007, 2008, 2012) showing that pericytes are present in the infrapatellar fat pad.

2.4 Manipulation of Joint Stem Cell Phenotype In Vitro by Growth Factors

Growth factors are often used as media supplement for MSC expansion. The first growth factor tested on human MSCs was FGF2 in 1997 (Martin et al. 1997), and few years later, its role in enhancing chondrogenic capacity was established (Mastrogiacomo et al. 2001; Tsutsumi et al. 2001). Although the capacity of FGF2 to improve proliferation and chondrogenic capacity of MSCs is evident and acknowledged by the scientific community, very little is known about the role of FGF2 and other growth factors in influencing surface marker expression.

Addition of FGF2 decreases the expression of CD146 and alkaline phosphatase (ALP) on BMSCs, and the effect seems reversible after removal of the growth factor, possibly indicating a direct effect of FGF2 on CD146 expression. The effect of FGF2 on CD146 and ALP expression was observed on both the whole population of BMSCs (Gharibi and Hughes 2012; Hagmann et al. 2013b) and on the CD146+ subpopulation (Sacchetti et al. 2007). The group of Paolo Bianco further studied the

effect of FGF2 and other growth factors on preselected CD146+ BMSCs, observing that FGF2 and platelet-derived growth factor-BB (PDGF-BB) also reduce the expression of CD105 and CD49a, and TGF-β enhances CD63 and α-SMA expression (Sacchetti et al. 2007). Epidermal growth factor (EGF) slightly increases ALP expression, and WNT3a treatment did not significantly influence cell surface marker expression (Gharibi and Hughes 2012). Surprisingly, none of those studies related the effect of growth factors on surface marker expression with the chondrogenic capacity of the cells. This was evaluated by Hagmann et al. (2013a) in a study comparing the use of different expansion media on BMSCs. However, the media used were supplemented with several different factors, and therefore, it is not possible to conclude on the effect of a single growth factor on surface marker expression or chondrogenesis. Only more recently, we observed that the expansion of BMSCs in the presence of WNT3a influenced the chondrogenic capacity of the cells and also their surface marker expression (Narcisi et al. 2015). However, the only direct effect observed was an enhanced number of CD271+ cells found in the cells treated with WNT3a. All the other markers tested were either unchanged by the treatment (CD73 and CD146) or maintained longer during the expansion in vitro (CD90, CD105, CD166), but they were not regulated directly. The ability of the BMSCs treated with WNT3a to enhance or retain the expression of certain surface markers in culture was linked by the authors to the capacity of the WNT3a-stimulated BMSCs to maintain, over multiple passages in vitro, a robust chondrogenic capacity.

Considering the other cell sources, to our knowledge only one article underlined the effect of growth factors on surface marker expression, demonstrating that FGF2 does not alter the expression of CD13, CD29, CD44, CD90, CD105, 3G5, STRO-1, CD34, and CD56 in IF-MSCs (Khan et al. 2008). However, again, no correlation with the chondrogenic capacity was reported.

Conclusion

In this chapter we provided an overview of the relation between surface marker expression and chondrogenic capacity of human adult MSCs from different sources. Since a large number of studies are available, in particular for BMSCs, we had to make a selection of the literature. However, two clear messages should be evident: (1) surface marker expression of cells in vivo strongly depends on tissue and cell localization, and (2) in vitro surface marker expression of expanded MSCs with high chondrogenic capacity is different between MSCs isolated from different sources. Several attempts to isolate, characterize, and purify populations of MSCs either directly from the tissue or after expansion in culture have not led to a clear description of MSCs with enhanced chondrogenic capacity. Moreover, especially for the culture-expanded MSCs, the use of different isolation techniques or expansion media could strongly influence the outcome. It is also interesting to note that MSCs derived from cartilage (CPCs), infrapatellar fat pad (IF-MSCs), and synovium (SD-MSCs) are still poorly characterized compared to BMSCs, and, moreover, only a very limited number of studies have directly compared the chondrogenic capacity of selected populations from different sources. Despite the use of surface markers for cell selection, the resulting populations are still heterogeneous, which might explain the lack of success in finding the most suitable marker to select the best chondrogenic population of cells directly from the native tissue or after expansion in vitro. Therefore, researchers are continuously looking for new surface markers or new combinations of markers. However, we are convinced that additional efforts in exploring the use of alternative systems to purify or select chondrogenic cells with methods not necessarily based on surface marker expression might be required.

Acknowledgment Our work and the writing of this chapter were supported by the Netherlands Institute for Regenerative Medicine (FES0908), the Science Foundation Ireland (11/RFP/BMT/3150), the Translational Adult Stem Cell Research by ZonMw (116005009), a VENI grant from STW (13659), and SmartStep (MRC-MR-L022893).

References

- Alegre-Aguaron E, Desportes P, Garcia-Alvarez F, Castiella T, Larrad L, Martinez-Lorenzo MJ (2012) Differences in surface marker expression and chondrogenic potential among various tissue-derived mesenchymal cells from elderly patients with osteoarthritis. Cells Tissues Organs 196(3):231–240. doi:10.1159/000334400
- Almeida HV, Cunniffe GM, Vinardell T, Buckley CT, O'Brien FJ, Kelly DJ (2015) Coupling freshly isolated CD44(+) infrapatellar fat pad-derived stromal cells with a TGF-beta3 eluting cartilage ECM-derived scaffold as a single-stage strategy for promoting chondrogenesis. Adv Healthc Mater 4(7):1043–1053
- Alsalameh S, Amin R, Gemba T, Lotz M (2004) Identification of mesenchymal progenitor cells in normal and osteoarthritic human articular cartilage. Arthritis Rheum 50(5):1522–1532. doi:10.1002/art.20269
- Alvarez-Viejo M, Menendez-Menendez Y, Otero-Hernandez J (2015) CD271 as a marker to identify mesenchymal stem cells from diverse sources before culture. World J Stem Cells 7(2):470–476
- Arufe MC, De la Fuente A, Fuentes-Boquete I, De Toro FJ, Blanco FJ (2009) Differentiation of synovial CD-105(+) human mesenchymal stem cells into chondrocyte-like cells through spheroid formation. J Cell Biochem 108(1):145–155. doi:10.1002/jcb.22238
- Arufe MC, De la Fuente A, Fuentes I, de Toro FJ, Blanco FJ (2010) Chondrogenic potential of subpopulations of cells expressing mesenchymal stem cell markers derived from human synovial membranes. J Cell Biochem 111(4):834–845. doi:10.1002/jcb.22768
- Banfi A, Muraglia A, Dozin B, Mastrogiacomo M, Cancedda R, Quarto R (2000) Proliferation kinetics and differentiation potential of ex vivo expanded human bone marrow stromal cells: implications for their use in cell therapy. Exp Hematol 28(6):707–715
- Barbero A, Ploegert S, Heberer M, Martin I (2003) Plasticity of clonal populations of dedifferentiated adult human articular chondrocytes. Arthritis Rheum 48(5):1315–1325. doi:10.1002/ art.10950
- Battula VL, Treml S, Bareiss PM, Gieseke F, Roelofs H, de Zwart P, Muller I, Schewe B, Skutella T, Fibbe WE, Kanz L, Buhring HJ (2009) Isolation of functionally distinct mesenchymal stem cell subsets using antibodies against CD56, CD271, and mesenchymal stem cell antigen-1. Haematologica 94(2):173–184
- Bonab MM, Alimoghaddam K, Talebian F, Ghaffari SH, Ghavamzadeh A, Nikbin B (2006) Aging of mesenchymal stem cell in vitro. BMC Cell Biol 7:14. doi:10.1186/1471-2121-7-14
- Buhring HJ, Battula VL, Treml S, Schewe B, Kanz L, Vogel W (2007) Novel markers for the prospective isolation of human MSC. Ann NY Acad Sci 1106:262–271

- Candela ME, Cantley L, Yasuaha R, Iwamoto M, Pacifici M, Enomoto-Iwamoto M (2014) Distribution of slow-cycling cells in epiphyseal cartilage and requirement of beta-catenin signaling for their maintenance in growth plate. J Orthop Res 32(5):661–668
- Chang CB, Han SA, Kim EM, Lee S, Seong SC, Lee MC (2013) Chondrogenic potentials of human synovium-derived cells sorted by specific surface markers. Osteoarthritis Cartilage 21(1):190–199
- Cleary MA, Narcisi R, Focke K, van der Linden R, Brama PA, van Osch GJ (2016) Expression of CD105 on expanded mesenchymal stem cells does not predict their chondrogenic potential. Osteoarthritis Cartilage 24(5):868–872. doi:10.1016/j.joca.2015.11.018
- Craig FM, Bentley G, Archer CW (1987) The spatial and temporal pattern of collagens I and II and keratan sulphate in the developing chick metatarsophalangeal joint. Development 99(3):383–391
- De Bari C, Dell'Accio F, Tylzanowski P, Luyten FP (2001) Multipotent mesenchymal stem cells from adult human synovial membrane. Arthritis Rheum 44(8):1928–1942
- De Bari C, Dell'Accio F, Luyten FP (2004) Failure of in vitro-differentiated mesenchymal stem cells from the synovial membrane to form ectopic stable cartilage in vivo. Arthritis Rheum 50(1):142–150. doi:10.1002/art.11450
- Decker RS, Koyama E, Pacifici M (2014) Genesis and morphogenesis of limb synovial joints and articular cartilage. Matrix Biol 39:5–10. doi:10.1016/j.matbio.2014.08.006
- Delorme B, Ringe J, Gallay N, Le Vern Y, Kerboeuf D, Jorgensen C, Rosset P, Sensebe L, Layrolle P, Haupl T, Charbord P (2008) Specific plasma membrane protein phenotype of culture-amplified and native human bone marrow mesenchymal stem cells. Blood 111(5):2631–2635
- De la Fuente R et al. (2004) Dedifferentiated adult articular chondrocytes: a population of human multipotent primitive cells. Exp Cell Res 297(2):313–28
- Ding DC, Wu KC, Chou HL, Hung WT, Liu HW, Chu TY (2015) Human infrapatellar fat padderived stromal cells have more potent differentiation capacity than other mesenchymal cells and can be enhanced by Hyaluronan. Cell Transplant 24(7):1221–1232. doi:10.3727/0963689 14X681937
- Dominici M, Le Blanc K, Mueller I, Slaper-Cortenbach I, Marini F, Krause D, Deans R, Keating A, Prockop D, Horwitz E (2006) Minimal criteria for defining multipotent mesenchymal stromal cells. The International Society for Cellular Therapy position statement. Cytotherapy 8(4):315–317
- Dowthwaite GP, Bishop JC, Redman SN, Khan IM, Rooney P, Evans DJ, Haughton L, Bayram Z, Boyer S, Thomson B, Wolfe MS, Archer CW (2004) The surface of articular cartilage contains a progenitor cell population. J Cell Sci 117(Pt 6):889–897
- English A, Jones EA, Corscadden D, Henshaw K, Chapman T, Emery P, McGonagle D (2007) A comparative assessment of cartilage and joint fat pad as a potential source of cells for autologous therapy development in knee osteoarthritis. Rheumatology (Oxford) 46(11):1676–1683. doi:10.1093/rheumatology/kem217
- Espagnolle N, Guilloton F, Deschaseaux F, Gadelorge M, Sensebe L, Bourin P (2014) CD146 expression on mesenchymal stem cells is associated with their vascular smooth muscle commitment. J Cell Mol Med 18(1):104–114. doi:10.1111/jcmm.12168
- Fickert S, Fiedler J, Brenner RE (2003) Identification, quantification and isolation of mesenchymal progenitor cells from osteoarthritic synovium by fluorescence automated cell sorting. Osteoarthritis Cartilage 11(11):790–800
- Fickert S, Fiedler J, Brenner RE (2004) Identification of subpopulations with characteristics of mesenchymal progenitor cells from human osteoarthritic cartilage using triple staining for cell surface markers. Arthritis Res Ther 6(5):R422–R432. doi:10.1186/ar1210
- Fukiage K, Aoyama T, Shibata KR, Otsuka S, Furu M, Kohno Y, Ito K, Jin Y, Fujita S, Fujibayashi S, Neo M, Nakayama T, Nakamura T, Toguchida J (2008) Expression of vascular cell adhesion molecule-1 indicates the differentiation potential of human bone marrow stromal cells. Biochem Biophys Res Commun 365(3):406–412. doi:10.1016/j.bbrc.2007.10.149
- Gharibi B, Hughes FJ (2012) Effects of medium supplements on proliferation, differentiation potential, and in vitro expansion of mesenchymal stem cells. Stem Cells Transl Med 1(11):771– 782. doi:10.5966/sctm.2010-0031

- Grogan SP, Barbero A, Diaz-Romero J, Cleton-Jansen AM, Soeder S, Whiteside R, Hogendoorn PC, Farhadi J, Aigner T, Martin I, Mainil-Varlet P (2007) Identification of markers to characterize and sort human articular chondrocytes with enhanced in vitro chondrogenic capacity. Arthritis Rheum 56(2):586–595
- Grogan SP, Miyaki S, Asahara H, D'Lima DD, Lotz MK (2009) Mesenchymal progenitor cell markers in human articular cartilage: normal distribution and changes in osteoarthritis. Arthritis Res Ther 11(3):R85. doi:10.1186/ar2719
- Gronthos S, Zannettino AC, Hay SJ, Shi S, Graves SE, Kortesidis A, Simmons PJ (2003) Molecular and cellular characterisation of highly purified stromal stem cells derived from human bone marrow. J Cell Sci 116(Pt 9):1827–1835
- Hagmann S, Moradi B, Frank S, Dreher T, Kammerer PW, Richter W, Gotterbarm T (2013a) Different culture media affect growth characteristics, surface marker distribution and chondrogenic differentiation of human bone marrow-derived mesenchymal stromal cells. BMC Musculoskelet Disord 14:223. doi:10.1186/1471-2474-14-223
- Hagmann S, Moradi B, Frank S, Dreher T, Kammerer PW, Richter W, Gotterbarm T (2013b) FGF-2 addition during expansion of human bone marrow-derived stromal cells alters MSC surface marker distribution and chondrogenic differentiation potential. Cell Prolif 46(4):396– 407. doi:10.1111/cpr.12046
- Hagmann S, Frank S, Gotterbarm T, Dreher T, Eckstein V, Moradi B (2014) Fluorescence activated enrichment of CD146+ cells during expansion of human bone-marrow derived mesenchymal stromal cells augments proliferation and GAG/DNA content in chondrogenic media. BMC Musculoskelet Disord 15:322. doi:10.1186/1471-2474-15-322
- Han SA, Lee S, Seong SC, Lee MC (2014) Effects of CD14 macrophages and proinflammatory cytokines on chondrogenesis in osteoarthritic synovium-derived stem cells. Tissue Eng Part A 20(19-20):2680–2691. doi:10.1089/ten.TEA.2013.0656
- Harichandan A, Buhring HJ (2011) Prospective isolation of human MSC. Best Pract Res Clin Haematol 24(1):25–36
- Harichandan A, Sivasubramaniyan K, Buhring HJ (2013) Prospective isolation and characterization of human bone marrow-derived MSCs. Adv Biochem Eng Biotechnol 129:1–17
- Harvanova D, Tothova T, Sarissky M, Amrichova J, Rosocha J (2011) Isolation and characterization of synovial mesenchymal stem cells. Folia Biol (Praha) 57(3):119–124
- Hayes AJ, MacPherson S, Morrison H, Dowthwaite G, Archer CW (2001) The development of articular cartilage: evidence for an appositional growth mechanism. Anat Embryol 203(6):469–479
- Hermida-Gomez T, Fuentes-Boquete I, Gimeno-Longas MJ, Muinos-Lopez E, Diaz-Prado S, de Toro FJ, Blanco FJ (2011) Bone marrow cells immunomagnetically selected for CD271+ antigen promote in vitro the repair of articular cartilage defects. Tissue Eng Part A 17(7-8):1169– 1179. doi:10.1089/ten.TEA.2010.0346
- Holder N (1977) An experimental investigation into the early development of the chick elbow joint. J Embryol Exp Morphol 39:115–127
- Hunziker EB, Rosenberg LC (1996) Repair of partial-thickness defects in articular cartilage: cell recruitment from the synovial membrane. J Bone Joint Surg Am 78(5):721–733
- Hunziker EB, Kapfinger E, Geiss J (2007) The structural architecture of adult mammalian articular cartilage evolves by a synchronized process of tissue resorption and neoformation during postnatal development. Osteoarthritis Cartilage 15(4):403–413
- Hyde G, Boot-Handford RP, Wallis GA (2008) Col2a1 lineage tracing reveals that the meniscus of the knee joint has a complex cellular origin. J Anat 213(5):531–538. doi:10.1111/ j.1469-7580.2008.00966.x
- Im GI (2016) Endogenous cartilage repair by recruitment of stem cells. Tissue Eng Part B Rev 22:160–171
- Jiang Y, Tuan RS (2015) Origin and function of cartilage stem/progenitor cells in osteoarthritis. Nat Rev Rheumatol 11(4):206–212
- Jiang Y, Hu C, Yu S, Yan J, Peng H, Ouyang HW, Tuan RS (2015) Cartilage stem/progenitor cells are activated in osteoarthritis via interleukin-1beta/nerve growth factor signaling. Arthritis Res Ther 17:327

- Jo CH, Ahn HJ, Kim HJ, Seong SC, Lee MC (2007) Surface characterization and chondrogenic differentiation of mesenchymal stromal cells derived from synovium. Cytotherapy 9(4):316– 327. doi:10.1080/14653240701291620
- Johnstone B, Hering TM, Caplan AI, Goldberg VM, Yoo JU (1998) In vitro chondrogenesis of bone marrow-derived mesenchymal progenitor cells. Exp Cell Res 238(1):265–272. doi:10.1006/excr.1997.3858
- Jones EA, English A, Henshaw K, Kinsey SE, Markham AF, Emery P, McGonagle D (2004) Enumeration and phenotypic characterization of synovial fluid multipotential mesenchymal progenitor cells in inflammatory and degenerative arthritis. Arthritis Rheum 50(3):817–827. doi:10.1002/art.20203
- Jones EA, English A, Kinsey SE, Straszynski L, Emery P, Ponchel F, McGonagle D (2006) Optimization of a flow cytometry-based protocol for detection and phenotypic characterization of multipotent mesenchymal stromal cells from human bone marrow. Cytometry B Clin Cytom 70(6):391–399. doi:10.1002/cyto.b.20118
- Jones EA, Crawford A, English A, Henshaw K, Mundy J, Corscadden D, Chapman T, Emery P, Hatton P, McGonagle D (2008) Synovial fluid mesenchymal stem cells in health and early osteoarthritis: detection and functional evaluation at the single-cell level. Arthritis Rheum 58(6):1731–1740
- Jones E, Churchman SM, English A, Buch MH, Horner EA, Burgoyne CH, Reece R, Kinsey S, Emery P, McGonagle D, Ponchel F (2010) Mesenchymal stem cells in rheumatoid synovium: enumeration and functional assessment in relation to synovial inflammation level. Ann Rheum Dis 69(2):450–457. doi:10.1136/ard.2008.106435
- Jurgens WJ, van Dijk A, Doulabi BZ, Niessen FB, Ritt MJ, van Milligen FJ, Helder MN (2009) Freshly isolated stromal cells from the infrapatellar fat pad are suitable for a one-step surgical procedure to regenerate cartilage tissue. Cytotherapy 11(8):1052–1064
- Kaltz N, Ringe J, Holzwarth C, Charbord P, Niemeyer M, Jacobs VR, Peschel C, Haupl T, Oostendorp RA (2010) Novel markers of mesenchymal stem cells defined by genome-wide gene expression analysis of stromal cells from different sources. Exp Cell Res 316(16):2609– 2617. doi:10.1016/j.yexcr.2010.06.002
- Karystinou A et al. (2009) Distinct mesenchymal progenitor cell subsets in the adult human synovium. Rheumatology (Oxford) 48(9):1057–64. doi:10.1093/rheumatology/kep192
- Khan WS, Adesida AB, Hardingham TE (2007) Hypoxic conditions increase hypoxia-inducible transcription factor 2alpha and enhance chondrogenesis in stem cells from the infrapatellar fat pad of osteoarthritis patients. Arthritis Res Ther 9(3):R55. doi:10.1186/ar2211
- Khan WS, Tew SR, Adesida AB, Hardingham TE (2008) Human infrapatellar fat pad-derived stem cells express the pericyte marker 3G5 and show enhanced chondrogenesis after expansion in fibroblast growth factor-2. Arthritis Res Ther 10(4):R74
- Khan IM, Bishop JC, Gilbert S, Archer CW (2009) Clonal chondroprogenitors maintain telomerase activity and Sox9 expression during extended monolayer culture and retain chondrogenic potential. Osteoarthritis Cartilage 17(4):518–528. doi:10.1016/j.joca.2008.08.002
- Khan WS, Adesida AB, Tew SR, Longo UG, Hardingham TE (2012) Fat pad-derived mesenchymal stem cells as a potential source for cell-based adipose tissue repair strategies. Cell Prolif 45(2):111–120
- Koelling S, Kruegel J, Irmer M, Path JR, Sadowski B, Miro X, Miosge N (2009) Migratory chondrogenic progenitor cells from repair tissue during the later stages of human osteoarthritis. Cell Stem Cell 4(4):324–335
- Koyama E, Ochiai T, Rountree RB, Kingsley DM, Enomoto-Iwamoto M, Iwamoto M, Pacifici M (2007) Synovial joint formation during mouse limb skeletogenesis: roles of Indian hedgehog signaling. Ann N Y Acad Sci 1116:100–112. doi:10.1196/annals.1402.063
- Kramer J, Bohrnsen F, Lindner U, Behrens P, Schlenke P, Rohwedel J (2006) In vivo matrixguided human mesenchymal stem cells. Cell Mol Life Sci 63(5):616–626
- Krawetz RJ, Wu YE, Martin L, Rattner JB, Matyas JR, Hart DA (2012) Synovial fluid progenitors expressing CD90+ from normal but not osteoarthritic joints undergo chondrogenic differentiation without micro-mass culture. PLoS One 7(8):e43616. doi:10.1371/journal.pone.0043616

- Kurose R, Ichinohe S, Tajima G, Horiuchi S, Kurose A, Sawai T, Shimamura T (2010) Characterization of human synovial fluid cells of 26 patients with osteoarthritis knee for cartilage repair therapy. Int J Rheum Dis 13(1):68–74. doi:10.1111/j.1756-185X.2009.01456.x
- Kurth TB, Dell'accio F, Crouch V, Augello A, Sharpe PT, De Bari C (2011) Functional mesenchymal stem cell niches in adult mouse knee joint synovium in vivo. Arthritis Rheum 63(5):1289–1300
- Lee CH, Cook JL, Mendelson A, Moioli EK, Yao H, Mao JJ (2010) Regeneration of the articular surface of the rabbit synovial joint by cell homing: a proof of concept study. Lancet 376(9739):440–448
- Lee DH, Sonn CH, Han SB, Oh Y, Lee KM, Lee SH (2012) Synovial fluid CD34(-) CD44(+) CD90(+) mesenchymal stem cell levels are associated with the severity of primary knee osteoarthritis. Osteoarthritis Cartilage 20(2):106–109
- Li T, Longobardi L, Myers TJ, Temple JD, Chandler RL, Ozkan H, Contaldo C, Spagnoli A (2013) Joint TGF-beta type II receptor-expressing cells: ontogeny and characterization as joint progenitors. Stem Cells Dev 22(9):1342–1359. doi:10.1089/scd.2012.0207
- Longobardi L, Li T, Myers TJ, O'Rear L, Ozkan H, Li Y, Contaldo C, Spagnoli A (2012) TGF-beta type II receptor/MCP-5 axis: at the crossroad between joint and growth plate development. Dev Cell 23(1):71–81. doi:10.1016/j.devcel.2012.05.004
- Lopa S, Colombini A, Stanco D, de Girolamo L, Sansone V, Moretti M (2014) Donor-matched mesenchymal stem cells from knee infrapatellar and subcutaneous adipose tissue of osteoarthritic donors display differential chondrogenic and osteogenic commitment. Eur Cell Mater 27:298–311
- López-Ruiz E et al. (2013) Chondrocytes extract from patients with osteoarthritis induces chondrogenesis in infrapatellar fat pad-derived stem cells. Osteoarthritis Cartilage 21(1):246–58. doi:10.1016/j.joca.2012.10.007
- Lv FJ, Tuan RS, Cheung KM, Leung VY (2014) Concise review: the surface markers and identity of human mesenchymal stem cells. Stem Cells 32(6):1408–1419
- Majumdar MK, Banks V, Peluso DP, Morris EA (2000) Isolation, characterization, and chondrogenic potential of human bone marrow-derived multipotential stromal cells. J Cell Physiol 185 (1):98-106. doi:10.1002/1097-4652(200010)185:1<98::AID-JCP9>3.0.CO;2-1. pii: 1002/1097-4652(200010)185:1<98::AID-JCP9>3.0.CO;2-1
- Martin I, Muraglia A, Campanile G, Cancedda R, Quarto R (1997) Fibroblast growth factor-2 supports ex vivo expansion and maintenance of osteogenic precursors from human bone marrow. Endocrinology 138(10):4456–4462. doi:10.1210/endo.138.10.5425
- Martinez C, Hofmann TJ, Marino R, Dominici M, Horwitz EM (2007) Human bone marrow mesenchymal stromal cells express the neural ganglioside GD2: a novel surface marker for the identification of MSCs. Blood 109(10):4245–4248
- Mastrogiacomo M, Cancedda R, Quarto R (2001) Effect of different growth factors on the chondrogenic potential of human bone marrow stromal cells. Osteoarthritis Cartilage 9(Suppl A):S36–S40
- Matsukura Y, Muneta T, Tsuji K, Koga H, Sekiya I (2014) Mesenchymal stem cells in synovial fluid increase after meniscus injury. Clin Orthop Relat Res 472(5):1357–1364. doi:10.1007/s11999-013-3418-4
- Mifune Y, Matsumoto T, Murasawa S, Kawamoto A, Kuroda R, Shoji T, Kuroda T, Fukui T, Kawakami Y, Kurosaka M, Asahara T (2013) Therapeutic superiority for cartilage repair by CD271-positive marrow stromal cell transplantation. Cell Transplant 22(7):1201–1211. doi:10 .3727/096368912X657378
- Mochizuki T, Muneta T, Sakaguchi Y, Nimura A, Yokoyama A, Koga H, Sekiya I (2006) Higher chondrogenic potential of fibrous synovium- and adipose synovium-derived cells compared with subcutaneous fat-derived cells: distinguishing properties of mesenchymal stem cells in humans. Arthritis Rheum 54(3):843–853. doi:10.1002/art.21651
- Morito T, Muneta T, Hara K, Ju YJ, Mochizuki T, Makino H, Umezawa A, Sekiya I (2008) Synovial fluid-derived mesenchymal stem cells increase after intra-articular ligament injury in humans. Rheumatology (Oxford) 47(8):1137–1143

- Muller B, Kohn D (1999) [Indication for and performance of articular cartilage drilling using the Pridie method] Indikation und Durchfuhrung der Knorpel-Knochen-Anbohrung nach Pridie. Orthopade 28(1):4–10
- Nagase T, Muneta T, Ju YJ, Hara K, Morito T, Koga H, Nimura A, Mochizuki T, Sekiya I (2008) Analysis of the chondrogenic potential of human synovial stem cells according to harvest site and culture parameters in knees with medial compartment osteoarthritis. Arthritis Rheum 58(5):1389–1398. doi:10.1002/art.23418
- Nalin AM, Greenlee TK Jr, Sandell LJ (1995) Collagen gene expression during development of avian synovial joints: transient expression of types II and XI collagen genes in the joint capsule. Dev Dyn 203(3):352–362. doi:10.1002/aja.1002030307
- Narcisi R, Cleary MA, Brama PA, Hoogduijn MJ, Tuysuz N, ten Berge D, van Osch GJ (2015) Long-term expansion, enhanced chondrogenic potential, and suppression of endochondral ossification of adult human MSCs via WNT signaling modulation. Stem Cell Rep 4(3):459– 472. doi:10.1016/j.stemcr.2015.01.017
- Nishimura K, Solchaga LA, Caplan AI, Yoo JU, Goldberg VM, Johnstone B (1999) Chondroprogenitor cells of synovial tissue. Arthritis Rheum 42(12):2631–2637
- Ogata Y, Mabuchi Y, Yoshida M, Suto EG, Suzuki N, Muneta T, Sekiya I, Akazawa C (2015) Purified human synovium mesenchymal stem cells as a good resource for cartilage regeneration. PLoS One 10(6):e0129096
- Ozbey O et al. (2014) Characterization of colony-forming cells in adult human articular cartilage. Acta Histochem116(5):763–70. doi:10.1016/j.acthis.2014.01.004
- Pacifici M, Koyama E, Shibukawa Y, Wu C, Tamamura Y, Enomoto-Iwamoto M, Iwamoto M (2006) Cellular and molecular mechanisms of synovial joint and articular cartilage formation. Ann N Y Acad Sci 1068:74–86. doi:10.1196/annals.1346.010
- Pittenger MF, Mackay AM, Beck SC, Jaiswal RK, Douglas R, Mosca JD, Moorman MA, Simonetti DW, Craig S, Marshak DR (1999) Multilineage potential of adult human mesenchymal stem cells. Science 284(5411):143–147
- Pretzel D, Linss S, Rochler S, Endres M, Kaps C, Alsalameh S, Kinne RW (2011) Relative percentage and zonal distribution of mesenchymal progenitor cells in human osteoarthritic and normal cartilage. Arthritis Res Ther 13(2):R64
- Quirici N, Soligo D, Bossolasco P, Servida F, Lumini C, Deliliers GL (2002) Isolation of bone marrow mesenchymal stem cells by anti-nerve growth factor receptor antibodies. Exp Hematol 30(7):783–791
- Richter W (2009) Mesenchymal stem cells and cartilage in situ regeneration. J Intern Med 266(4):390-405
- Sacchetti B, Funari A, Michienzi S, Di Cesare S, Piersanti S, Saggio I, Tagliafico E, Ferrari S, Robey PG, Riminucci M, Bianco P (2007) Self-renewing osteoprogenitors in bone marrow sinusoids can organize a hematopoietic microenvironment. Cell 131(2):324–336. doi:10.1016/j. cell.2007.08.025
- Sakaguchi Y, Sekiya I, Yagishita K, Muneta T (2005) Comparison of human stem cells derived from various mesenchymal tissues: superiority of synovium as a cell source. Arthritis Rheum 52(8):2521–2529
- Sekiya I, Ojima M, Suzuki S, Yamaga M, Horie M, Koga H, Tsuji K, Miyaguchi K, Ogishima S, Tanaka H, Muneta T (2012) Human mesenchymal stem cells in synovial fluid increase in the knee with degenerated cartilage and osteoarthritis. J Orthop Res 30(6):943–949
- Shirasawa S, Sekiya I, Sakaguchi Y, Yagishita K, Ichinose S, Muneta T (2006) In vitro chondrogenesis of human synovium-derived mesenchymal stem cells: optimal condition and comparison with bone marrow-derived cells. J Cell Biochem 97(1):84–97. doi:10.1002/jcb.20546
- Simmons PJ, Torok-Storb B (1991) Identification of stromal cell precursors in human bone marrow by a novel monoclonal antibody, STRO-1. Blood 78(1):55–62
- Sivasubramaniyan K, Lehnen D, Ghazanfari R, Sobiesiak M, Harichandan A, Mortha E, Petkova N, Grimm S, Cerabona F, de Zwart P, Abele H, Aicher WK, Faul C, Kanz L, Buhring HJ (2012) Phenotypic and functional heterogeneity of human bone marrow- and amnion-derived MSC subsets. Ann N Y Acad Sci 1266:94–106

- Sivasubramaniyan K, Harichandan A, Schumann S, Sobiesiak M, Lengerke C, Maurer A, Kalbacher H, Buhring HJ (2013) Prospective isolation of mesenchymal stem cells from human bone marrow using novel antibodies directed against Sushi domain containing 2. Stem Cells Dev 22(13):1944–1954
- Soeda T, Deng JM, de Crombrugghe B, Behringer RR, Nakamura T, Akiyama H (2010) Sox9expressing precursors are the cellular origin of the cruciate ligament of the knee joint and the limb tendons. Genesis 48(11):635–644. doi:10.1002/dvg.20667
- de Sousa EB, Casado PL, Moura Neto V, Duarte ME, Aguiar DP (2014) Synovial fluid and synovial membrane mesenchymal stem cells: latest discoveries and therapeutic perspectives. Stem Cell Res Ther 5(5):112
- Spagnoli A, O'Rear L, Chandler RL, Granero-Molto F, Mortlock DP, Gorska AE, Weis JA, Longobardi L, Chytil A, Shimer K, Moses HL (2007) TGF-beta signaling is essential for joint morphogenesis. J Cell Biol 177(6):1105–1117
- Steadman JR, Rodkey WG, Rodrigo JJ (2001) Microfracture: surgical technique and rehabilitation to treat chondral defects. Clin Orthop Relat Res (391 Suppl):S362–S369
- Storm EE, Kingsley DM (1996) Joint patterning defects caused by single and double mutations in members of the bone morphogenetic protein (BMP) family. Development 122(12): 3969–3979
- Swart GW (2002) Activated leukocyte cell adhesion molecule (CD166/ALCAM): developmental and mechanistic aspects of cell clustering and cell migration. Eur J Cell Biol 81(6): 313–321
- Tong W, Geng Y, Huang Y, Shi Y, Xiang S, Zhang N, Qin L, Shi Q, Chen Q, Dai K, Zhang X (2015) In vivo identification and induction of articular cartilage stem cells by inhibiting NF-kappaB signaling in osteoarthritis. Stem Cells 33(10):3125–3137
- Tormin A, Li O, Brune JC, Walsh S, Schutz B, Ehinger M, Ditzel N, Kassem M, Scheding S (2011) CD146 expression on primary nonhematopoietic bone marrow stem cells is correlated with in situ localization. Blood 117(19):5067–5077
- Tsutsumi S, Shimazu A, Miyazaki K, Pan H, Koike C, Yoshida E, Takagishi K, Kato Y (2001) Retention of multilineage differentiation potential of mesenchymal cells during proliferation in response to FGF. Biochem Biophys Res Commun 288 (2):413-419. doi:10.1006/ bbrc.2001.5777. pii: S0006-291X(01)95777-8
- Ustunel I, Ozenci AM, Sahin Z, Ozbey O, Acar N, Tanriover G, Celik-Ozenci C, Demir R (2008) The immunohistochemical localization of notch receptors and ligands in human articular cartilage, chondroprogenitor culture and ultrastructural characteristics of these progenitor cells. Acta Histochem 110(5):397–407. doi:10.1016/j.acthis.2007.12.005
- Van Landuyt KB, Jones EA, McGonagle D, Luyten FP, Lories RJ (2010) Flow cytometric characterization of freshly isolated and culture expanded human synovial cell populations in patients with chronic arthritis. Arthritis Res Ther 12(1):R15
- Wickham MQ, Erickson GR, Gimble JM, Vail TP, Guilak F (2003) Multipotent stromal cells derived from the infrapatellar fat pad of the knee. Clin Orthop Relat Res 412:196–212
- Williams R, Khan IM, Richardson K, Nelson L, McCarthy HE, Analbelsi T, Singhrao SK, Dowthwaite GP, Jones RE, Baird DM, Lewis H, Roberts S, Shaw HM, Dudhia J, Fairclough J, Briggs T, Archer CW (2010) Identification and clonal characterisation of a progenitor cell subpopulation in normal human articular cartilage. PLoS One 5(10):e13246. doi:10.1371/journal. pone.0013246
- Zhang Q, Cigan AD, Marrero L, Lopreore C, Liu S, Ge D, Savoie FH, You Z (2011) Expression of doublecortin reveals articular chondrocyte lineage in mouse embryonic limbs. Genesis 49(2):75–82. doi:10.1002/dvg.20702
- Zuk PA, Zhu M, Ashjian P, De Ugarte DA, Huang JI, Mizuno H, Alfonso ZC, Fraser JK, Benhaim P, Hedrick MH (2002) Human adipose tissue is a source of multipotent stem cells. Mol Biol Cell 13(12):4279–4295. doi:10.1091/mbc.E02-02-0105

Chondrogenic Progenitor Cells and Cartilage Repair

Diana Oellerich and Nicolai Miosge

Abstract

Osteoarthritis and rheumatoid arthritis are the two most common diseases of joints causing cartilage and bone destruction leading to the loss of joint function. Causal therapies are still challenging, and to date various cell biological methods applying chondrocytes or mesenchymal stem cells only generate fibrocartilaginous repair tissue with less good mechano-biological properties.

In osteoarthritis and rheumatoid arthritis, a specific cell type with stem cell characteristics including migratory activity, clonogenicity, and multipotency could be characterized, which we named "chondrogenic progenitor cells." These cells, involved in regeneration processes, are largely unsuccessful mainly because they produce collagen type I. Manipulation of these progenitor cells, which are already present in diseased cartilage tissue, could be a promising approach for cartilage repair. Several chondrogenic pathways and interacting partners have been already identified. The transcription factors Runx2 and Sox9 play an important role by influencing the collagen II production. Interleukins and TGF- β might also play an important role in the regulation of Runx2. Furthermore, it could be shown that various other factors like mechanical stimulation or components of the pericellular matrix prompt chondrogenic progenitor cells to trigger chondrogenesis. The differentiation potential of chondrogenic progenitor cells seems to be affected by calcium homeostasis, including calcium regulatory mechanisms. However, several challenges remain regarding the elucidation of the regulatory pathways that determine chondrogenic progenitor cells to become more chondrogenic.

D. Oellerich • N. Miosge (\boxtimes)

Georg August University, Tissue Regeneration Work Group, Department of Prosthodontics, Robert-Koch-Str. 40, 37075 Göttingen, Germany e-mail: diana.oellerich@gmx.de; nmiosge@gwdg.de

[©] Springer International Publishing AG 2017

S. Grässel, A. Aszodi (eds.), Cartilage, DOI 10.1007/978-3-319-53316-2_3

3.1 Osteoarthritis and Rheumatoid Arthritis

Osteoarthritis (Fig. 3.1) is a whole-organ disease, also affecting the synovium, subchondral bone, and meniscus (Goldring and Otero 2011; Englund et al. 2012; Goldring and Goldring 2010). As a degenerative joint disease, it shows a progressive loss of the articular cartilage (Loeser et al. 2012). The synovium is responsible for producing inflammation mediators, which can also be found in osteoarthritis (Bougault et al. 2012). Modified cell-matrix interactions result in destroyed tissue integrity and primarily affect the hyaline cartilage. However, this also applies to other tissues, especially the subchondral bone. An eburnation of the subchondral bone can arise (Goldring and Otero 2011; Goldring and Goldring 2010). An abnormal remodeling of the subchondral bone often leads to a thicker, but mechanically less stable, tissue (Lories and Luyten 2011). Osteoarthritis often originates from



Fig. 3.1 (a) Immunohistochemistry for biglycan in healthy human articular cartilage of the knee joint. (b) Histology of late-stage OA, please note the numerous tidemark duplications and the deep surface fissures. (c) Fibrocartilaginous repair tissue, where the chondrogenic progenitor cells are found. (d) Breaks in the tidemark with mesenchymal tissue entering the cartilage tissue

meniscal lesions (Englund et al. 2012). Even when meniscal substitutes, such as allografts or bioengineered substitutes, are used, there is no protective effect against the development of osteoarthritis (Hommen et al. 2007). No markers have been discovered to diagnose early stages of osteoarthritis. At present, no therapy to treat the causes of osteoarthritis is available that could promise a complete cure (Lohmander and Roos 2007; Lohmander et al. 2014; Musumeci et al. 2014). In the late stages of refractory osteoarthritis, knee replacement is the gold standard of treatment (Johnson and Hunter 2014).

Various cellular mechanisms cause rheumatoid arthritis and result in cartilage and bone destruction. Inflammation plays an important role in the disease process. The inflamed synovium produces a tumor-like pannus tissue, which destroys the cartilage tissue. Proinflammatory cytokines, such as IL-1 β , IL-6, TNF- α , and matrix metalloproteinases, are produced by fibroblast-like synoviocytes and synovial macrophages (Karouzakis et al. 2006). The RANKL-dependent induction of osteoclasts is primarily responsible for bone destruction (Kim et al. 2014). In the synovium in rheumatoid arthritis, CD4+ T cells accumulate (Franz et al. 1998; Blaschke et al. 2003; Toh and Miossec 2007).

3.2 Regeneration Attempts in Cartilage Tissue

Chondrocytes are the only cell source found in healthy articular cartilage (Muir 1995; Kock et al. 2012). These produce collagen type II, which together with aggrecan is mainly responsible for the high mechanical resilience of articular cartilage tissue. Cartilage has a very low potential for intrinsic self-repair and regeneration in its mature tissue, because chondrocytes are believed to have no capacity for migration, proliferation, and repair (Tew et al. 2001; Johnstone et al. 2013; Redman et al. 2005).

Most surgical therapies aim to stimulate cells from the bone marrow by microfracture, abrasion arthroplasty, and Pridie drilling (Buckwalter and Mankin 1998; Hochberg et al. 2012; Lohmander and Roos 2007; Minas 1999; Steadman et al. 2002; Steinwachs et al. 2008; Muller and Kohn 1999). However, these treatments attempt to support fibrocartilaginous repair tissue and cannot induce hyaline cartilage (Ronn et al. 2011; Becher et al. 2010). Other treatment options such as osteochondral autologous transplantation or autologous chondrocyte implantation also try to recover the articular surfaces (Muller et al. 2010; Vasiliadis et al. 2010).

3.2.1 Mesenchymal Stem Cells and Osteoarthritis

Tissue regeneration should focus on generating a repair tissue, which exerts the same mechano-biological properties and assimilates with the native tissue (Redman et al. 2005). Currently, one focus uses stem cell-based therapies to induce regeneration. The use of mesenchymal stem cells targets restores hyaline articular cartilage. Mesenchymal stem cells can be found in differentiated tissues and fulfil
various tasks in the adult human body. They are characterized by their multi-lineage potential, which was demonstrated first by Pittenger et al. (1999), who isolated mesenchymal stem cells from bone marrow. Currently, mesenchymal stem cells can be obtained from diverse adult tissues (Kuhn and Tuan 2010). They have the capability to differentiate into various mesenchymal phenotypes including muscle, ligament, tendon, adipose, stroma, bone, and cartilage (Cai et al. 2004; Caplan 2007). Furthermore, they possess immunomodulating properties, antiinflammation effects, and self-renewal capacities (Bonfield et al. 2010; Chamberlain et al. 2007; Chen and Tuan 2008). Stem cells are located in a yet-to-be-defined niche and remain quiescent except for rare cell divisions (Fuchs et al. 2004).

One of the first attempts to achieve cartilage repair in osteoarthritis was performed by Wakitani et al. in 1994. Large, full-thickness defects of the articular cartilage in the knees of rabbits were repaired by osteochondral progenitor cells (Wakitani et al. 1994). Later, a clinical trial was designed to investigate the influence of mesenchymal stem cells on damaged human cartilage. Human autologous culture expanded bone marrow mesenchymal cells were transplanted into the osteoarthritic knee joints of patients. A cartilage-like tissue formed after 42 weeks. However, symptoms were not significantly improved (Wakitani et al. 2002).

Further essential findings concerning stem cell therapy of osteoarthritis were obtained by Murphy et al. (2003). They explored the role of implanted adult mesenchymal stem cells in tissue repair and regeneration in an injured joint in a goat model. Regeneration of the meniscal tissue was stimulated, and its progressive destruction was prevented. However, none of the stem cells were located in the diseased cartilage tissue.

Another study investigated the benefit of mesenchymal stem cell treatment in patients suffering from moderate to late-stage osteoarthritis. No complete regeneration of cartilage and no efficient long-term success were obtained (Davatchi et al. 2011; Cucchiarini et al. 2014).

3.2.2 Mesenchymal Stem Cells and Rheumatoid Arthritis

Rheumatoid arthritis involves chronic inflammation of the synovium, which induces cartilage and bone erosion (De Bari 2015). Mesenchymal stem cells can also be found in the synovium (De Bari et al. 2001). In rheumatoid arthritis, the immune modulatory properties of mesenchymal stem cells seem to be very important (El-Jawhari et al. 2014). The effects of mesenchymal stem cells seem to be repressed by the inflammatory milieu. Rheumatoid arthritis patients show a lower prevalence of mesenchymal stem cells than is seen in osteoarthritis patients (Jones et al. 2004). Furthermore, there is a negative correlation between the chondrogenic and clonogenic capacities of synovial mesenchymal stem cells and the magnitude of synovitis in rheumatoid arthritis (Jones et al. 2010). In addition, there is an interdependency between infiltrating inflammatory/immune cells and resident fibroblast-like synoviocytes. A proliferation of fibroblast-like synoviocytes, the major pathogenic component in rheumatoid arthritis, is observed in rheumatoid arthritis (Li and

Makarov 2006). These cells support the development of harmful pannus, which results in damaged articular cartilage and bone (Naylor et al. 2013). A recent study demonstrated that interactions between fibroblast-like synoviocytes and mesenchymal stem cells are possible. The placental growth factor, which occurs at higher levels in joints suffering from rheumatoid arthritis, could attract bone marrow mesenchymal stem cells to the synovium, where interactions with the resident fibroblast-like synoviocytes may lead to angiogenesis and chronic synovitis by further enhancing the secretion of placental growth factor (Park et al. 2014). However, the relationship between fibroblast-like synoviocytes and mesenchymal stem cells remains unclear. It might be possible that they represent different functional stages of the same lineage or that they represent the same cell type with functional specialization and diversification according to their positional information and environmental cues (De Bari et al. 2001).

Recent clinical studies have explored the benefit of mesenchymal stem cell treatment in rheumatoid arthritis patients. The intravenous injection of umbilical cord mesenchymal stem cells in addition to disease-modifying antirheumatic drugs induced a significant clinical improvement in patients suffering from active rheumatoid arthritis and in whom conventional treatment was ineffective (Wang et al. 2013). However, larger multicenter clinical studies are needed to provide safe treatment recommendations.

3.3 Chondrogenic Progenitor Cells

Restoring fully functional hyaline cartilage has not been achieved to date via chondrocytes or mesenchymal stem cells. Various experiments and methods have only generated fibrocartilaginous repair tissue instead of stable hyaline cartilage (Cucchiarini et al. 2014). Fibrocartilaginous repair tissue shows morphologically distinct cell types (Kouri et al. 1996). In the late stages of osteoarthritis, single chondrocytes and cells, which are organized in aggregates, are found. Furthermore, comprised chondrocytes undergoing a degenerative process exist in all zones of the cartilage. The most common cells are elongated secretory type 2 cells, which display a secretory phenotype (Kouri et al. 1996). These cells have been named fibroblast-like chondrocytes (Tesche and Miosge 2005). Koelling et al. (2009) were able to demonstrate that these cells show typical stem cell characteristics, including clonogenicity, multipotency, and migratory activity, and named them *chondrogenic* progenitor cells. They are also referred to as osteochondroprogenitor cells (Khan et al. 2009). In osteoarthritis and rheumatoid arthritis, these chondrogenic progenitor cells (Table 3.1) are involved in regeneration efforts that are largely unsuccessful in diseased cartilage tissue (Schminke and Miosge 2014).

Recently the presence of migratory progenitor cells in diseased tissues has been explored because they may play an important role in tissue regeneration and could be a promising target for cell-based therapy. Migratory progenitor cells show stem cell characteristics and possess great chondrogenic potential. The migratory potential is an important feature. The mechanism of cell migration can be found in numerous

| Cell types | Mesenchymal stem cells | Chondrogenic progenitor cells |
|---|---|---|
| Origin | Adult tissue | Osteoarthritic cartilage |
| Self-renewal | Slightly limited self-renewal | Limited self-renewal |
| Differentiation potential/ preclinical aspects | Multipotent; mainly differentiation into the cell types of the mesodermal lineage | Multipotent; already determined to the osteochondrogenic lineage |
| Stem cell marker positivity (CD, Stro-1) | Stro-1, CD13, CD29, CD44, CD49a, CD73 , CD90 , CD105 , CD114, CD166 (while no expression of CD14, CD19, CD34, and CD45) | Stro-1, CD13, CD29, CD44, CD73, CD90 (while no expression of CD18, CD31, CD34, CD117, CD271) |
| Immunity/preclinical aspects | Less immunogenic; difficult to maintain undifferentiated in cell culture | Unknown; easy to isolate and differentiate into chondrocytes |

Table 3.1 Mesenchymal stem cells and chondrogenic progenitor cells under investigation for cartilage repair (modified from Muhammad et al. 2013)

biological processes (Theveneau and Mayor 2013). Migration is a relevant characteristic of epithelial cells, i.e., during wound healing of the skin (Blanpain and Fuchs 2014). Migration is also indispensable to mesenchymal stem cells. Diverse repair processes in one's lifetime are dependent on the migration of these cells (Sohni and Verfaillie 2013). Hematopoiesis and bone regeneration require cell migration (Sahin and Buitenhuis 2012; Pignolo and Kassem 2011). Additionally, progenitor cells, which are involved in basic biological processes of the stem cell niche, migrate (Augello et al. 2010). Progenitor cell populations that were generated from patient tissue from late stages of osteoarthritis possessed great migratory potential, at least in vitro and ex vivo (Schminke and Miosge 2014; Muhammad et al. 2013).

Chondrogenic progenitor cells show stem cell marker positivity for Stro-1 and CD29 and also for CD13, CD44, CD73, and CD90. They are negative for CD31, CD34, CD117, and CD271. Cells isolated from the superficial zone of healthy cartilage tissue in vivo also show stem cell marker positivity and could be related to chondrogenic progenitor cells. So far, this observation has not been confirmed (Dowthwaite et al. 2004). Chondrogenic progenitor cells in the late stages of osteoarthritis exhibit multi-differentiation potential to become adipocytes, cells of the osteoblastic lineage and chondrocytes. They can be cloned and expanded for up to 60 population doublings. Chondrogenic progenitor cells can exist as cells of the chondrogenic lineage if they are simply placed in 3D alginate culture (Koelling and Miosge 2009).

3.3.1 Chondrogenic Progenitor Cells from Osteoarthritic Patients (CPCs)

Chondrogenic progenitor cells are a subpopulation of cells that are localized in the repair tissue of advanced stages of osteoarthritis (Koelling et al. 2009). The

compositions of collagens change in the late stages of osteoarthritis. Collagen types I and III are found in the fibrocartilaginous cartilage (Sandell and Aigner 2001; Poole 1999). Furthermore, a reduction of collagen type II is observed by quantitative immunohistochemistry (Miosge et al. 1998). In contrast to the microarray experiments, mRNAs of cartilage-specific collagens are upregulated, and increased anabolism is observed (Aigner et al. 2006). The change of the matrix composition may have a distinct influence on the deficient functioning of the repair tissue. Further investigations showed an increased expression level of proteoglycans such as biglycan, decorin, and perlecan (Tesche and Miosge 2005; Bock et al. 2001). This might reflect compensation for the loss of matrix molecules and the stabilization of the extracellular matrix (Tesche and Miosge 2005).

The transcription factors *runt-related transcription factor 2* (Runx2) and *sexdetermining region Y-box 9* (Sox9) play an important role in the regulation mechanisms of chondrogenic progenitor cells (Koelling et al. 2009; Koelling and Miosge 2010). Sox9 is involved in the development of chondrocytes, operates the synthesis of cartilage-specific matrix components, and inhibits the beginning of chondral ossification (de Crombrugghe et al. 2001). Runx9 coordinates the development of osteoblasts and is essential for bone formation (Stein et al. 2004).

An ex vivo experiment using siRNA in three-dimensional culture downregulation of the osteogenic transcription factor Runx2 resulted in a simultaneous upregulation of the chondrogenic transcription factor Sox9. Hence, COL2A1 mRNA was detected (Koelling and Miosge 2010; Koelling et al. 2009; Muhammad et al. 2014). A Sox9 knockdown results in reduced aggrecan and Runx2 expression (unpublished observation). Mass spectrometry analysis was used to identify proteins, which are involved in the signal transduction and transcription of Sox9 and Runx2. Overexpression of DDX5, HSPA8, RAB5C, and YWHAE resulted in enhanced gene expression of Sox9. HSPA8 also enhanced the gene expression of Runx2, which was downregulated by YWHAE, and the chondrogenic potential of the chondrogenic progenitor cells was increased. A knockdown of LEMD2 and TMPO leads to an upregulation of Sox9. Further indicators of increased chondrogenic potential were the enhanced expression of the extracellular component ACAN and the decreased expression of COL1A1.

The pericellular matrix with laminins and nidogen-2 may also play an important role in the regulation mechanisms of chondrogenic progenitor cells. It has been shown that chondrogenic progenitor cells produce high levels of laminin- α 1, laminin- α 5, and nidogen-2 in their pericellular matrix. Laminin- α 1 regulates collagen expression by enhancing collagen type II and decreasing collagen type I expression. Nidogen-2 upregulates Sox9 expression. A knockdown of nidogen-2 results in reduced Sox9 expression and enhanced Runx2 expression. Laminins and nidogen-2 guide chondrogenic progenitor cells toward chondrogenesis (Schminke et al. 2016a). This fact highlights the importance of the extracellular matrix components on chondrogenic progenitor cells and in stem cell biology (Fuchs et al. 2004; Fuchs 2008). The population of chondrogenic progenitor cells is not present in healthy cartilage.

Mechanical stimulation has an influence on chondrocytes. The primary cilium, a mechanosensor, also seems to be involved in mechano-transduction in chondrocytes.

The mechanical load enhances ciliogenesis in the growth plate. The expression and localization of key members of the Ihh-PTHrP loop is altered, resulting in decreased proliferation and a switch from proliferation to differentiation. Abnormal chondrocyte morphology and organization is also observed (Rais et al. 2015; Muhammad et al. 2012).

Furthermore, calcium signaling is important for chondrogenesis. A recent study suggested that calcium homeostasis, including calcium regulatory mechanisms, has an influence on the differentiation potential of chondrogenic progenitor cells. An autocrine/paracrine purinergic mechanism plays an important role in driving calcium oscillations in these cells (Matta et al. 2015). Furthermore, the external influence of the sympathetic nervous system on chondrogenic progenitor cell-dependent chondrogenesis is important. A norepinephrine-dependent inhibition of chondrogenesis and acceleration of hypertrophic differentiation was recently discovered (Jenei-Lanzl et al. 2014).

3.3.2 Chondrogenic Progenitor Cells from Rheumatoid Arthritis Patients

Chondrogenic progenitor cells can also be isolated from diseased cartilage tissue of patients who suffer from rheumatoid arthritis. Interleukins, which are commonly found in the inflamed rheumatoid tissue of affected joints, exert a negative influence on these cells. This results in a less chondrogenic phenotype. High levels of matrix metalloproteinases and proinflammatory cytokines, such as tumor necrosis factor α (TNF- α), which are influenced by IL-17, are produced by these chondrogenic progenitor cells. IL-17 A/F leads to the upregulation of Runx2 protein and enhanced IL-6 protein and MMP3 mRNA levels. Blocking antibodies against IL-17 improved the repair potential of the progenitor cells. When chondrogenic progenitor cells are treated with the antihuman IL-17 antibody secukinumab or the anti-TNF- α -antibody adalimumab, a reduction of the proinflammatory IL-6 protein levels and a positive influence on the secretion of anti-inflammatory IL-10 protein are observed. Runx2 protein is also reduced by the same antibodies, which promote chondrogenesis. The chondrogenic capacity of the chondrogenic progenitor cells can be improved again by anti-inflammatory agents. Again, progenitor cells are distinguished by their high migration potential. They are able to repopulate diseased cartilage tissue ex vivo. Inflammatory mediators have a remarkable influence on these progenitor cells and their ability to migrate (Schminke et al. 2016b).

3.3.3 Meniscus Progenitor Cells (MPCs)

In the inner, avascular part of diseased human menisci, meniscus progenitor cells can be found. They are normally distinguished by the production of collagen type I, and they display a fibrocartilaginous nature and high migration potential (Muhammad et al. 2014). During the investigation of diseased human menisci from

patients who were suffering from advanced stages of osteoarthritis, strongly affected menisci exhibit a downregulation of TGF- β and Smad2, resulting in the upregulation of Runx2. These facts support the assumption that meniscus progenitor cells also underlie a fine-tuned interaction between Runx2 and Sox9. Chondrogenic differentiation can be initiated by a knockdown of Runx2, which enhances p-Smad2. On the other hand, BMP2 stimulation of meniscus progenitor cells results in lower Smad2 levels and supports a change of the cells toward the osteogenic lineage (Muhammad et al. 2014). Additionally, a study in mice ascertained that TGF- β signaling directs knee morphogenesis and is important for meniscus development (Pazin et al. 2012).

Conclusion

Regenerative therapies aim to substitute diseased tissue with native-like functional tissue. Until now, nearly all attempts have only managed to achieve the generation of a fibrocartilaginous repair tissue instead of fully functional, collagen type II-enriched hyaline cartilage. To date, treatment approaches using mesenchymal stem cells have not obtained satisfying long-term results. A new strategy is not to transplant stem cells into diseased cartilage tissue but to manipulate resident cells with stem cell characteristics, which are already present in situ and are active in their physiological response to the cell biological stimuli of the diseased tissue (Muhammad et al. 2013). Chondrogenic progenitor cells could be a promising target for cartilage repair (Fig. 3.2).

Published successes provide insight into the chondrogenic pathways and interacting partners of chondrogenic progenitor cells. Two master regulators, Runx2



Fig. 3.2 The concept of chondrogenic progenitor cells in situ. Cells derived from the stem cell in its niche are already pre-determined as osteochondroprogenitor cells. In OA cartilage, under the control of runx2, they become collagen type I producing cells. If these cells would be targeted in situ to switch to collagen type II production by enhancing Sox9 expression, they would help to regenerate a more hyaline-like cartilage tissu

and Sox9, have already been identified. A knockdown of Runx2 leads to the upregulation of Sox9, aggrecan, and collagen type II (Koelling et al. 2009). The pericellular matrix with laminins and nidogen-2 is also involved in regulation mechanisms. Laminin-al and nidogen-2 guide chondrogenic progenitor cells toward chondrogenesis (Schminke et al. 2016a). Furthermore, mechanical stimulation has an influence on the chondrogenic differentiation potential of chondrogenic progenitor cells. An important role is played here by the primary cilium, which is necessary for mechano-transduction in chondrocytes (Rais et al. 2015; Muhammad et al. 2012). Calcium homeostasis may have autocrine/paracrine purinergic mechanisms that affect the calcium oscillations in these cells (Matta et al. 2015). In addition, the sympathetic nervous system was induced to have a norepinephrine-dependent inhibition of chondrogenesis (Jenei-Lanzl et al. 2014). Interleukins, especially IL-17, influence chondrogenic progenitor cells through the upregulation of Runx2 and drive them toward a less chondrogenic phenotype. Repair potential can be improved again by blocking antibodies against IL-17 in rheumatoid arthritis (Schminke et al. 2016b). The downregulation of TGF- β is also associated with an upregulation of Runx2 in meniscus progenitor cells (Muhammad et al. 2014).

Against the background outlined here, further investigations should focus on manipulating chondrogenic progenitor cells in situ with the help of small modifying molecules to improve their chondrogenic potential. It remains to be demonstrated whether manipulated chondrogenic progenitor cells produce an extracellular matrix, which provides repair tissue with better mechanical stress resistance than fibrocartilaginous tissue. Moreover, there is still a lack of knowledge about the behavior of manipulated chondrogenic progenitor cells in vivo. It has to be shown that they survive and maintain their favorable characteristics in the hostile microenvironment of the diseased organ (Koelling and Miosge 2009). In addition, the influence of age, gender, and body weight should not be disregarded (Gharibi et al. 2014; Murphy et al. 2002). Overall, further knowledge and understanding of the mechanisms that contribute to the regulation of stemness, multipotency, and differentiation have to be accomplished to allow a "restitutio ad integrum" in diseased cartilage tissue.

References

Aigner T, Fundel K, Saas J, Gebhard PM, Haag J, Weiss T, Zien A, Obermayr F, Zimmer R, Bartnik E (2006) Large-scale gene expression profiling reveals major pathogenetic pathways of cartilage degeneration in osteoarthritis. Arthritis Rheum 54(11):3533–3544. doi:10.1002/ art.22174

Augello A, Kurth TB, De Bari C (2010) Mesenchymal stem cells: a perspective from in vitro cultures to in vivo migration and niches. Eur Cell Mater 20:121–133

- Becher C, Driessen A, Hess T, Longo UG, Maffulli N, Thermann H (2010) Microfracture for chondral defects of the talus: maintenance of early results at midterm follow-up. Knee Surg Sports Traumatol Arthrosc 18(5):656–663. doi:10.1007/s00167-009-1036-1
- Blanpain C, Fuchs E (2014) Stem cell plasticity. Plasticity of epithelial stem cells in tissue regeneration. Science 344(6189):1242281. doi:10.1126/science.1242281

- Blaschke S, Middel P, Dorner BG, Blaschke V, Hummel KM, Kroczek RA, Reich K, Benoehr P, Koziolek M, Muller GA (2003) Expression of activation-induced, T cell-derived, and chemokine-related cytokine/lymphotactin and its functional role in rheumatoid arthritis. Arthritis Rheum 48(7):1858–1872. doi:10.1002/art.11171
- Bock HC, Michaeli P, Bode C, Schultz W, Kresse H, Herken R, Miosge N (2001) The small proteoglycans decorin and biglycan in human articular cartilage of late-stage osteoarthritis. Osteoarthritis Cartilage 9(7):654–663. doi:10.1053/joca.2001.0420
- Bonfield TL, Nolan Koloze MT, Lennon DP, Caplan AI (2010) Defining human mesenchymal stem cell efficacy in vivo. J Inflamm 7:51. doi:10.1186/1476-9255-7-51
- Bougault C, Gosset M, Houard X, Salvat C, Godmann L, Pap T, Jacques C, Berenbaum F (2012) Stress-induced cartilage degradation does not depend on the NLRP3 inflammasome in human osteoarthritis and mouse models. Arthritis Rheum 64(12):3972–3981. doi:10.1002/art.34678
- Buckwalter JA, Mankin HJ (1998) Articular cartilage repair and transplantation. Arthritis Rheum 41(8):1331–1342. doi:10.1002/1529-0131(199808)41:8<1331::AID-ART2>3.0.CO;2-J
- Cai J, Weiss ML, Rao MS (2004) In search of "stemness". Exp Hematol 32(7):585–598. doi:10.1016/j.exphem.2004.03.013
- Caplan AI (2007) Adult mesenchymal stem cells for tissue engineering versus regenerative medicine. J Cell Physiol 213(2):341–347. doi:10.1002/jcp.21200
- Chamberlain G, Fox J, Ashton B, Middleton J (2007) Concise review: mesenchymal stem cells: their phenotype, differentiation capacity, immunological features, and potential for homing. Stem Cells 25(11):2739–2749. doi:10.1634/stemcells.2007-0197
- Chen FH, Tuan RS (2008) Mesenchymal stem cells in arthritic diseases. Arthritis Res Ther 10(5):223. doi:10.1186/ar2514
- de Crombrugghe B, Lefebvre V, Nakashima K (2001) Regulatory mechanisms in the pathways of cartilage and bone formation. Curr Opin Cell Biol 13(6):721–727
- Cucchiarini M, Madry H, Guilak F, Saris DB, Stoddart MJ, Koon Wong M, Roughley P (2014) A vision on the future of articular cartilage repair. Eur Cell Mater 27:12–16
- Davatchi F, Abdollahi BS, Mohyeddin M, Shahram F, Nikbin B (2011) Mesenchymal stem cell therapy for knee osteoarthritis. Preliminary report of four patients. Int J Rheum Dis 14(2):211– 215. doi:10.1111/j.1756-185X.2011.01599.x
- De Bari C (2015) Are mesenchymal stem cells in rheumatoid arthritis the good or bad guys? Arthritis Res Ther 17:113. doi:10.1186/s13075-015-0634-1
- De Bari C, Dell'Accio F, Tylzanowski P, Luyten FP (2001) Multipotent mesenchymal stem cells from adult human synovial membrane. Arthritis Rheum 44(8):1928–1942. doi:10.1002/ 1529-0131(200108)44:8<1928::AID-ART331>3.0.CO;2-P
- Dowthwaite GP, Bishop JC, Redman SN, Khan IM, Rooney P, Evans DJ, Haughton L, Bayram Z, Boyer S, Thomson B, Wolfe MS, Archer CW (2004) The surface of articular cartilage contains a progenitor cell population. J Cell Sci 117(Pt 6):889–897. doi:10.1242/jcs.00912
- El-Jawhari JJ, El-Sherbiny YM, Jones EA, McGonagle D (2014) Mesenchymal stem cells, autoimmunity and rheumatoid arthritis. QJM 107(7):505–514. doi:10.1093/qjmed/hcu033
- Englund M, Roemer FW, Hayashi D, Crema MD, Guermazi A (2012) Meniscus pathology, osteoarthritis and the treatment controversy. Nat Rev Rheumatol 8(7):412–419. doi:10.1038/ nrrheum.2012.69
- Franz JK, Kolb SA, Hummel KM, Lahrtz F, Neidhart M, Aicher WK, Pap T, Gay RE, Fontana A, Gay S (1998) Interleukin-16, produced by synovial fibroblasts, mediates chemoattraction for CD4+ T lymphocytes in rheumatoid arthritis. Eur J Immunol 28(9):2661–2671. doi:10.1002/ (SICI)1521-4141(199809)28:09<2661::AID-IMMU2661>3.0.CO;2-N
- Fuchs E (2008) Skin stem cells: rising to the surface. J Cell Biol 180(2):273–284. doi:10.1083/ jcb.200708185
- Fuchs E, Tumbar T, Guasch G (2004) Socializing with the neighbors: stem cells and their niche. Cell 116(6):769–778
- Gharibi B, Farzadi S, Ghuman M, Hughes FJ (2014) Inhibition of Akt/mTOR attenuates agerelated changes in mesenchymal stem cells. Stem Cells 32(8):2256–2266. doi:10.1002/ stem.1709

- Goldring MB, Goldring SR (2010) Articular cartilage and subchondral bone in the pathogenesis of osteoarthritis. Ann N Y Acad Sci 1192:230–237. doi:10.1111/j.1749-6632.2009.05240.x
- Goldring MB, Otero M (2011) Inflammation in osteoarthritis. Curr Opin Rheumatol 23(5):471– 478. doi:10.1097/BOR.0b013e328349c2b1
- Hochberg MC, Altman RD, April KT, Benkhalti M, Guyatt G, McGowan J, Towheed T, Welch V, Wells G, Tugwell P, American College of Rheumatology (2012) American College of Rheumatology 2012 recommendations for the use of nonpharmacologic and pharmacologic therapies in osteoarthritis of the hand, hip, and knee. Arthritis Care Res 64(4):465–474
- Hommen JP, Applegate GR, Del Pizzo W (2007) Meniscus allograft transplantation: ten-year results of cryopreserved allografts. Arthroscopy 23(4):388–393. doi:10.1016/j.arthro.2006.11.032
- Jenei-Lanzl Z, Grassel S, Pongratz G, Kees F, Miosge N, Angele P, Straub RH (2014) Norepinephrine inhibition of mesenchymal stem cell and chondrogenic progenitor cell chondrogenesis and acceleration of chondrogenic hypertrophy. Arthritis Rheumatol 66(9):2472– 2481. doi:10.1002/art.38695
- Johnson VL, Hunter DJ (2014) The epidemiology of osteoarthritis. Best Pract Res Clin Rheumatol 28(1):5–15. doi:10.1016/j.berh.2014.01.004
- Johnstone B, Alini M, Cucchiarini M, Dodge GR, Eglin D, Guilak F, Madry H, Mata A, Mauck RL, Semino CE, Stoddart MJ (2013) Tissue engineering for articular cartilage repair--the state of the art. Eur Cell Mater 25:248–267
- Jones EA, English A, Henshaw K, Kinsey SE, Markham AF, Emery P, McGonagle D (2004) Enumeration and phenotypic characterization of synovial fluid multipotential mesenchymal progenitor cells in inflammatory and degenerative arthritis. Arthritis Rheum 50(3):817–827. doi:10.1002/art.20203
- Jones E, Churchman SM, English A, Buch MH, Horner EA, Burgoyne CH, Reece R, Kinsey S, Emery P, McGonagle D, Ponchel F (2010) Mesenchymal stem cells in rheumatoid synovium: enumeration and functional assessment in relation to synovial inflammation level. Ann Rheum Dis 69(2):450–457. doi:10.1136/ard.2008.106435
- Karouzakis E, Neidhart M, Gay RE, Gay S (2006) Molecular and cellular basis of rheumatoid joint destruction. Immunol Lett 106(1):8–13. doi:10.1016/j.imlet.2006.04.011
- Khan IM, Williams R, Archer CW (2009) One flew over the progenitor's nest: migratory cells find a home in osteoarthritic cartilage. Cell Stem Cell 4(4):282–284. doi:10.1016/j.stem.2009.03.007
- Kim HR, Kim KW, Kim BM, Jung HG, Cho ML, Lee SH (2014) Reciprocal activation of CD4+ T cells and synovial fibroblasts by stromal cell-derived factor 1 promotes RANKL expression and osteoclastogenesis in rheumatoid arthritis. Arthritis Rheumatol 66(3):538–548. doi:10.1002/art.38286
- Kock L, van Donkelaar CC, Ito K (2012) Tissue engineering of functional articular cartilage: the current status. Cell Tissue Res 347(3):613–627. doi:10.1007/s00441-011-1243-1
- Koelling S, Miosge N (2009) Stem cell therapy for cartilage regeneration in osteoarthritis. Expert Opin Biol Ther 9(11):1399–1405. doi:10.1517/14712590903246370
- Koelling S, Miosge N (2010) Sex differences of chondrogenic progenitor cells in late stages of osteoarthritis. Arthritis Rheum 62(4):1077–1087. doi:10.1002/art.27311
- Koelling S, Kruegel J, Irmer M, Path JR, Sadowski B, Miro X, Miosge N (2009) Migratory chondrogenic progenitor cells from repair tissue during the later stages of human osteoarthritis. Cell Stem Cell 4(4):324–335. doi:10.1016/j.stem.2009.01.015
- Kouri JB, Jimenez SA, Quintero M, Chico A (1996) Ultrastructural study of chondrocytes from fibrillated and non-fibrillated human osteoarthritic cartilage. Osteoarthritis Cartilage 4(2):111–125
- Kuhn NZ, Tuan RS (2010) Regulation of stemness and stem cell niche of mesenchymal stem cells: implications in tumorigenesis and metastasis. J Cell Physiol 222(2):268–277. doi:10.1002/jcp.21940
- Li X, Makarov SS (2006) An essential role of NF-kappaB in the "tumor-like" phenotype of arthritic synoviocytes. Proc Natl Acad Sci U S A 103(46):17432–17437. doi:10.1073/ pnas.0607939103
- Loeser RF, Goldring SR, Scanzello CR, Goldring MB (2012) Osteoarthritis: a disease of the joint as an organ. Arthritis Rheum 64(6):1697–1707. doi:10.1002/art.34453
- Lohmander LS, Roos EM (2007) Clinical update: treating osteoarthritis. Lancet 370(9605):2082–2084. doi:10.1016/S0140-6736(07)61879-0

- Lohmander LS, Hellot S, Dreher D, Krantz EF, Kruger DS, Guermazi A, Eckstein F (2014) Intraarticular sprifermin (recombinant human fibroblast growth factor 18) in knee osteoarthritis: a randomized, double-blind, placebo-controlled trial. Arthritis Rheumatol 66(7):1820– 1831. doi:10.1002/art.38614
- Lories RJ, Luyten FP (2011) The bone-cartilage unit in osteoarthritis. Nat Rev Rheumatol 7(1):43–49. doi:10.1038/nrrheum.2010.197
- Matta C, Fodor J, Miosge N, Takacs R, Juhasz T, Rybaltovszki H, Toth A, Csernoch L, Zakany R (2015) Purinergic signalling is required for calcium oscillations in migratory chondrogenic progenitor cells. Pflugers Arch 467(2):429–442. doi:10.1007/s00424-014-1529-8
- Minas T (1999) The role of cartilage repair techniques, including chondrocyte transplantation, in focal chondral knee damage. Instr Course Lect 48:629–643
- Miosge N, Waletzko K, Bode C, Quondamatteo F, Schultz W, Herken R (1998) Light and electron microscopic in-situ hybridization of collagen type I and type II mRNA in the fibrocartilaginous tissue of late-stage osteoarthritis. Osteoarthritis Cartilage 6(4):278–285. doi:10.1053/ joca.1998.0121
- Muhammad H, Rais Y, Miosge N, Ornan EM (2012) The primary cilium as a dual sensor of mechanochemical signals in chondrocytes. Cell Mol Life Sci 69(13):2101–2107. doi:10.1007/ s00018-011-0911-3
- Muhammad H, Schminke B, Miosge N (2013) Current concepts in stem cell therapy for articular cartilage repair. Expert Opin Biol Ther 13(4):541–548. doi:10.1517/14712598.2013.758707
- Muhammad H, Schminke B, Bode C, Roth M, Albert J, von der Heyde S, Rosen V, Miosge N (2014) Human migratory meniscus progenitor cells are controlled via the TGF-beta pathway. Stem Cell Rep 3(5):789–803. doi:10.1016/j.stemcr.2014.08.010
- Muir H (1995) The chondrocyte, architect of cartilage. Biomechanics, structure, function and molecular biology of cartilage matrix macromolecules. Bioessays 17(12):1039–1048. doi: 10.1002/bies.950171208
- Muller B, Kohn D (1999) Indication for and performance of articular cartilage drilling using the Pridie method. Orthopade 28(1):4–10
- Muller S, Breederveld RS, Tuinebreijer WE (2010) Results of osteochondral autologous transplantation in the knee. Open Orthop J 4:111–114. doi:10.2174/1874325001004020111
- Murphy JM, Dixon K, Beck S, Fabian D, Feldman A, Barry F (2002) Reduced chondrogenic and adipogenic activity of mesenchymal stem cells from patients with advanced osteoarthritis. Arthritis Rheum 46(3):704–713. doi:10.1002/art.10118
- Murphy JM, Fink DJ, Hunziker EB, Barry FP (2003) Stem cell therapy in a caprine model of osteoarthritis. Arthritis Rheum 48(12):3464–3474. doi:10.1002/art.11365
- Musumeci G, Castrogiovanni P, Leonardi R, Trovato FM, Szychlinska MA, Di Giunta A, Loreto C, Castorina S (2014) New perspectives for articular cartilage repair treatment through tissue engineering: a contemporary review. World J Orthod 5(2):80–88. doi:10.5312/ wjo.v5.i2.80
- Naylor AJ, Filer A, Buckley CD (2013) The role of stromal cells in the persistence of chronic inflammation. Clin Exp Immunol 171(1):30–35. doi:10.1111/j.1365-2249.2012.04634.x
- Park SJ, Kim KJ, Kim WU, Cho CS (2014) Interaction of mesenchymal stem cells with fibroblastlike synoviocytes via cadherin-11 promotes angiogenesis by enhanced secretion of placental growth factor. J Immunol 192(7):3003–3010. doi:10.4049/jimmunol.1302177
- Pazin DE, Gamer LW, Cox KA, Rosen V (2012) Molecular profiling of synovial joints: use of microarray analysis to identify factors that direct the development of the knee and elbow. Dev Dyn 241(11):1816–1826. doi:10.1002/dvdy.23861
- Pignolo RJ, Kassem M (2011) Circulating osteogenic cells: implications for injury, repair, and regeneration. J Bone Miner Res 26(8):1685–1693. doi:10.1002/jbmr.370
- Pittenger MF, Mackay AM, Beck SC, Jaiswal RK, Douglas R, Mosca JD, Moorman MA, Simonetti DW, Craig S, Marshak DR (1999) Multilineage potential of adult human mesenchymal stem cells. Science 284(5411):143–147
- Poole AR (1999) An introduction to the pathophysiology of osteoarthritis. Front Biosci 4:D662–D670

- Rais Y, Reich A, Simsa-Maziel S, Moshe M, Idelevich A, Kfir T, Miosge N, Monsonego-Ornan E (2015) The growth plate's response to load is partially mediated by mechano-sensing via the chondrocytic primary cilium. Cell Mol Life Sci 72(3):597–615. doi:10.1007/s00018-014-1690-4
- Redman SN, Oldfield SF, Archer CW (2005) Current strategies for articular cartilage repair. Eur Cell Mater 9:23–32. discussion 23-32
- Ronn K, Reischl N, Gautier E, Jacobi M (2011) Current surgical treatment of knee osteoarthritis. Arthritis 2011:454873. doi:10.1155/2011/454873
- Sahin AO, Buitenhuis M (2012) Molecular mechanisms underlying adhesion and migration of hematopoietic stem cells. Cell Adh Migr 6(1):39–48. doi:10.4161/cam.18975
- Sandell LJ, Aigner T (2001) Articular cartilage and changes in arthritis. An introduction: cell biology of osteoarthritis. Arthritis Res 3(2):107–113
- Schminke B, Miosge N (2014) Cartilage repair in vivo: the role of migratory progenitor cells. Curr Rheumatol Rep 16(11):461. doi:10.1007/s11926-014-0461-4
- Schminke B, Frese J, Bode C, Goldring MB, Miosge N (2016a) Laminins and nidogens in the pericellular matrix of chondrocytes: their role in osteoarthritis and chondrogenic differentiation. Am J Pathol 186(2):410–418. doi:10.1016/j.ajpath.2015.10.014
- Schminke B, Trautmann S, Mai B, Miosge N, Blaschke S (2016b) Interleukin 17 inhibits progenitor cells in rheumatoid arthritis cartilage. Eur J Immunol 46(2):440–445. doi:10.1002/ eji.201545910
- Sohni A, Verfaillie CM (2013) Mesenchymal stem cells migration homing and tracking. Stem Cells Int 2013:130763. doi:10.1155/2013/130763
- Steadman JR, Rodkey WG, Briggs KK (2002) Microfracture to treat full-thickness chondral defects: surgical technique, rehabilitation, and outcomes. J Knee Surg 15(3):170–176
- Stein GS, Lian JB, van Wijnen AJ, Stein JL, Montecino M, Javed A, Zaidi SK, Young DW, Choi JY, Pockwinse SM (2004) Runx2 control of organization, assembly and activity of the regulatory machinery for skeletal gene expression. Oncogene 23(24):4315–4329. doi:10.1038/ sj.onc.1207676
- Steinwachs MR, Guggi T, Kreuz PC (2008) Marrow stimulation techniques. Injury 39(Suppl 1):S26–S31. doi:10.1016/j.injury.2008.01.042
- Tesche F, Miosge N (2005) New aspects of the pathogenesis of osteoarthritis: the role of fibroblastlike chondrocytes in late stages of the disease. Histol Histopathol 20(1):329–337
- Tew S, Redman S, Kwan A, Walker E, Khan I, Dowthwaite G, Thomson B, Archer CW (2001) Differences in repair responses between immature and mature cartilage. Clin Orthop Relat Res (391 Suppl):S142–S152
- Theveneau E, Mayor R (2013) Collective cell migration of epithelial and mesenchymal cells. Cell Mol Life Sci 70(19):3481–3492. doi:10.1007/s00018-012-1251-7
- Toh ML, Miossec P (2007) The role of T cells in rheumatoid arthritis: new subsets and new targets. Curr Opin Rheumatol 19(3):284–288. doi:10.1097/BOR.0b013e32805e87e0
- Vasiliadis HS, Danielson B, Ljungberg M, McKeon B, Lindahl A, Peterson L (2010) Autologous chondrocyte implantation in cartilage lesions of the knee: long-term evaluation with magnetic resonance imaging and delayed gadolinium-enhanced magnetic resonance imaging technique. Am J Sports Med 38(5):943–949. doi:10.1177/0363546509358266
- Wakitani S, Goto T, Pineda SJ, Young RG, Mansour JM, Caplan AI, Goldberg VM (1994) Mesenchymal cell-based repair of large, full-thickness defects of articular cartilage. J Bone Joint Surg Am 76(4):579–592
- Wakitani S, Imoto K, Yamamoto T, Saito M, Murata N, Yoneda M (2002) Human autologous culture expanded bone marrow mesenchymal cell transplantation for repair of cartilage defects in osteoarthritic knees. Osteoarthritis Cartilage 10(3):199–206. doi:10.1053/joca.2001.0504
- Wang L, Wang L, Cong X, Liu G, Zhou J, Bai B, Li Y, Bai W, Li M, Ji H, Zhu D, Wu M, Liu Y (2013) Human umbilical cord mesenchymal stem cell therapy for patients with active rheumatoid arthritis: safety and efficacy. Stem Cells Dev 22(24):3192–3202. doi:10.1089/ scd.2013.0023

Induced Pluripotent Stem Cells and Cartilage Regeneration

Solvig Diederichs and Wiltrud Richter

Abstract

Induced pluripotency has attracted enormous scientific and public attention and has quickly entered all fields of research not only those that had so far lacked an easily accessible cell source for cell therapies and in vitro models. The attractiveness of induced pluripotent stem (iPS) cells for cartilage regeneration roots from their immense expandability and their intrinsic ability to give rise to any adult tissue including stable hyaline cartilage. Thus, iPS cells offer to overcome the extremely limited supply of human articular chondrocytes and the restricted differentiation capacity of mesenchymal stem cells from bone marrow or adipose tissue that differentiate along the endochondral pathway and form mineralized bone upon ectopic implantation. Beyond being a potential alternative cell source for articular chondrocyte implantation, iPS cells are particularly promising for in vitro modeling of genetic diseases and for drug testing. Reprogramming patient-specific cells with a genetic predisposition and engineering disease-specific genetic variations into healthy control iPS cells promises to recapitulate "diseases in a dish" more realistically than immortalized human cell lines and will be an invaluable complementation for animal models. Whether iPS cells will satisfy these tremendous expectations will depend on our ability to upscale iPS cell culture, to derive sufficient amounts of relevant cell types like chondrocytes from iPS cells with acceptable efforts, and to find clinically safe reprogramming techniques for iPS cell-based therapies.

S. Diederichs (🖂) • W. Richter

Research Centre for Experimental Orthopaedics, Orthopaedic University Hospital Heidelberg, Heidelberg, Germany

e-mail: solvig.diederichs@med.uni-heidelberg.de; wiltrud.richter@med.uni-heidelberg.de

[©] Springer International Publishing AG 2017

S. Grässel, A. Aszodi (eds.), Cartilage, DOI 10.1007/978-3-319-53316-2_4

4.1 Introduction

The era of induced pluripotent stem cells (iPS cells) started with the breakthrough report by Takahashi and Yamanaka who reprogrammed somatic cells back to pluripotent stem cells by retrovirally overexpressing a defined set of transcription factors (Takahashi and Yamanaka 2006). The search for such factors present in the egg cytoplasm had been initiated by John Gurdon who managed to obtain normally developed *Xenopus* frogs by transferring somatic intestinal cell nuclei into an egg, thus proving that some components within the egg cytoplasm were capable to reprogram a somatic nucleus to pluripotency. Yamanaka's cumbersome exclusion experiments surprisingly uncovered that only four factors-Oct3/4, Sox2, Klf4, and c-Myc-were sufficient to initiate transcription of the complete pluripotency network within somatic cells, to erase the epigenetic marks accrued during differentiation and to, thus, reprogram committed cells back to pluripotency. For proving that cellular differentiation is not an irreversible process and that mature cells can be reprogrammed to become pluripotent, John Gurdon and Shinya Yamanaka were awarded the 2012 Nobel Prize for Physiology or Medicine. Extensive research is currently going on in the new field of induced pluripotency addressing various aspects of iPS technology, such as cell sourcing, reprogramming efficiency, and the development of sophisticated reprogramming strategies. Furthermore, elucidation of the pluripotency transcription network that enables stem cells to infinitely proliferate and maintain the undifferentiated status is ongoing.

The simplicity and reproducibility of cellular reprogramming by ectopic expression of few defined transcription factors has stimulated iPS cell research to spread beyond basic stem cell science into applicational research. Soon, iPS cells became a universal tool holding prospect for cell therapy, for diagnosis and drug development, and even for the rescue of endangered species (Fig. 4.1). Thus, iPS cells also entered the field of regenerative medicine where iPS cells represent the only practical cell source, e.g., in neurobiology or cardiology. Cartilage regeneration may also benefit from iPS cells since they promise to overcome limitations of the current cell sources which are articular chondrocytes and mesenchymal stromal cells. Chondrocyte availability suffers from the invasiveness of tissue harvest requiring creation of a new defect in a tissue with low intrinsic regeneration capacity. Thus, only small amounts of cartilage tissue can be harvested for cell isolation, and chondrocytes need to be expanded in vitro in order to provide sufficient cell quantities. Upon in vitro culture, however, chondrocytes dedifferentiate and lose their capacity to form cartilage, thus severely limiting the amount of chondrocytes available for cell therapy and research. The high proliferative capacity of stem cells offers to overcome the limited availability of primary chondrocytes.

Multipotent mesenchymal stromal cells (MSCs) that can be isolated from bone marrow (Bianco and Gehron 2000; Caplan 1991), adipose tissue (Zuk et al. 2001), or umbilical cord blood (Erices et al. 2000) are currently the best described stem cell source for cartilage regeneration and closest to clinical translation. The harvest procedures for bone marrow and adipose tissue are less invasive than an articular cartilage biopsy, and MSCs can be expanded for some passages before they lose their



Fig. 4.1 Promise of iPS cells for cartilage regeneration. Body cells like fibroblasts, lymphocytes, or cord blood cells can be reprogrammed into iPS cells that can be cryopreserved. Upon need, iPS cells can be revitalized and differentiated into chondrocytes for cell-based therapies. Importantly, iPS cells are particularly promising for basic research to elucidate cartilage development and pathomechanisms of genetic cartilage diseases in order to develop novel treatment strategies

chondrogenic capacity (Bruder et al. 1997; Narcisi et al. 2015). Although bone marrow-derived MSCs are considered to contain skeletal stem cells (Bianco 2014) which are the developmental progenitors of articular chondrocytes, the cartilage they form in vitro is not hyaline but comprises also fibrous components (collagen type I), becomes hypertrophic, and expresses markers of calcifying cartilage (Dickhut et al. 2010; Pelttari et al. 2006). Upon ectopic implantation, such cartilage becomes mineralized and forms endochondral bone (Pelttari et al. 2006; Janicki et al. 2010). Thus, MSCs seem to be primed for endochondral development and incapable to form stable hyaline cartilage in vitro. The intrinsic capability of iPS cells to form any adult tissue including hyaline articular cartilage and their unlimited proliferative capacity thus make iPS cells a highly attractive cell source that could potentially provide access to ample amounts of in vitro-generated cartilage. Furthermore, iPS cells offer the new possibility of in vitro disease modeling, for which both primary chondrocytes and MSCs are inapplicable, because upon genetic manipulation and subcloning they easily lose their capacity to form cartilage.

Thus, iPS cells are attractive for cartilage regeneration because their abundant availability may solve the limited supply of articular chondrocytes, their intrinsic capacity to form stable hyaline cartilage may overcome the endochondral priming of MSCs, and their stable clonal expansion capacity makes in vitro models accessible for elucidating pathomechanisms and for drug development and highthroughput screenings. In the following sections, the status and recent major achievements to utilize iPS cells for cell-based therapies, to derive therapeutically relevant cells from iPS cells, and to establish iPS cell-based in vitro disease models will be presented, and future perspectives will be discussed.

4.2 Making iPS Cells Available for Cell-Based Therapies

Optimal cell sourcing is a central question for all cell-based therapies. In line with the current trend of personalized medicine, autologous strategies are preferred over allogeneic application. But even if virtually any cell is reprogrammable, it practicality requires a cell source that is easily accessible by a noninvasive procedure. Skin fibroblasts that were originally used by Yamanaka have probably been reprogrammed most frequently, but a skin biopsy can cause scar formation and is afflicted with the risk of infection. Thus, skin fibroblasts may not necessarily be the best tissue source for reprogramming. Peripheral blood on the other hand is more easily accessible, and blood withdrawal is less invasive than a skin biopsy and very well accepted by most patients. T lymphocytes have been reprogrammed with good yields (Okita et al. 2013; Seki et al. 2011), and the rearrangements of the T-cell receptor DNA which are specific for each individual T lymphocyte make the clonality of the resulting iPS cells easy to monitor under GMP culture conditions. Thus, in those scenarios where personalized iPS cell-based therapies are supposed to be generated, peripheral blood holds prospect to become a popular and easily accessible source.

However, all cells accrue genetic mutations during their lifetime, and these mutations persist upon reprogramming (Gore et al. 2011) and may increase the risk of tumorigenicity. Thus, embryonic cells would be preferable over adult cells making cord blood cells highly attractive for reprogramming because of their developmental youth and low genetic modification rate (Giorgetti et al. 2009; Haase et al. 2009; Takenaka et al. 2010). Of course, autologous cord blood is currently not available for every patient, but cryopreservation of cord blood has become increasingly popular, and cord blood banks are growing in many countries. In the future, iPS cells could be routinely generated and cryopreserved from cord blood to provide a cell pool for biologic treatment strategies in any organ system including cartilage. Banking iPS cells instead of cord blood will also facilitate a more prompt availability. Indeed, generating a validated and well-characterized iPS cell line is a timeconsuming process that can take months and should, thus, preferentially be already accomplished when a diagnosis calls for cell-based therapy. Of course, the generation of therapeutically active cells or cartilage tissue from iPS cells to treat injured or degraded cartilage will also take a few weeks, but this appears an absolutely acceptable period to wait for implantation of a cartilage regenerate.

Generating iPS cell banks to cover the complete population, however, might seem enormously challenging, and thus, despite a general preference for autologous

strategies, feasibility and practicality aspects support allogeneic approaches. Especially for regenerating cartilage, which is avascular and alymphatic and has been suggested to be immune privileged to some degree (Elves and Zervas 1974; Hunt et al. 2014; Levy et al. 2013), allogeneic treatment might be well tolerated. Japanese stem cell researchers are already establishing an iPS cell bank containing HLA-homozygous donor cells under the assumption that matching at least the three major HLA markers will result in less immune rejection (Takahashi and Yamanaka 2013; Taylor et al. 2012; Turner et al. 2013; Yamanaka 2012). Of course, matching major HLA types such as HLA-A, -B, and -DR might not completely rule out an immune reaction when other HLA loci or minor histocompatibility antigens mismatch (Tsumaki et al. 2015; Turner et al. 2013). Future studies will reveal whether the potential immune privileges of cartilage may allow imprecise donor matching, thus increasing the availability of suitable donor cells and overcoming potential graft shortage. Of course, allogeneic iPS cell banks would ideally only contain cells from healthy donors. This criterion may prove challenging for cord blood cells, since donors might seem healthy at birth but can still develop degenerative or other diseases with age. Our continuously advancing ability to assess the cellular genetic and epigenetic profile together with our progressing insight into disease-causing cell transformations will help setting up sophisticated quality controls to ensure graft safety.

In summary, their accessibility and genetic integrity make cord blood cells an ideal source for generating iPS cells. When an autologous approach is preferred but autologous cord blood is not available, peripheral blood seems the most obvious alternative choice for conversion into personalized iPS cells. To accelerate the availability of cell-based therapies upon need, iPS cell banks are highly promising. Practicality and feasibility have encouraged allogeneic instead of autologous approaches. An allogeneic iPS cell bank is currently being established in Japan and is meant to largely cover the population with 100 HLA-homozygous cell lines. Thus, having allogeneic iPS cells available for cartilage regeneration seems only a small step away. In the future, iPS cells could be generated and banked routinely and could then upon need provide an autologous cell pool for biologic treatment strategies in any organ system including cartilage that might be injured or diseased throughout life.

4.3 Therapeutic Cells from iPS Cells

The euphoria of having a universal cell source available intuitively suggests utilizing these cells therapeutically for cell and tissue transplantation strategies. Their unlimited proliferative capacity is one main quality defining iPS cells. However, the disadvantage of this highly praised property is that upon implantation, iPS cells could potentially proliferate without inhibition and form tumors. Indeed, the formation of teratomas—benign tumors containing tissues originating from all three germ lines—is the most stringent quality test for newly reprogrammed human iPS cell lines to verify their pluripotency. Thus, iPS cells cannot be implanted directly but must be differentiated into offspring with a restricted proliferative capacity, and contamination of these progeny with pluripotent stem cells must be excluded.

The most obvious strategy to utilize iPS cells for cell implantation therapy in cartilage is to differentiate them into cartilage-forming chondrocytes. It took surprisingly long until the first study reported human iPS cell chondrogenesis (Medvedev et al. 2011) 4 years after human iPS cells were generated for the first time (Park et al. 2008; Takahashi et al. 2007; Yu et al. 2007). In this study by Medvedev et al., iPS cells were initially differentiated as embryoid bodies-small cell aggregates in which tissues from all three germ lines are formed spontaneously like in an embryo but far less coordinated (Medvedev et al. 2011). Cells dissociated from these embryoid bodies were subsequently subjected to chondrogenic stimuli in 3D culture where they were supposed to self-aggregate and deposit cartilaginous matrix. Hallmarks of chondrogenesis were indeed detected, but the amount of proteoglycans and collagen type II remained disappointingly low. Several alternative methods to induce in vitro chondrogenesis of human iPS cells have been reported since then. Most of them omitted embryoid body culture arguing that this method is uncontrollable and overly laborious. Co-culture with chondrocytes was a plausible attempt based on the assumption that direct cell-cell contact and/or the soluble factors produced by chondrocytes might be able to instruct iPS cells to differentiate efficiently into the chondrogenic lineage (Qu et al. 2013; Wei et al. 2012). However, the cartilage formation efficiency of such chondrogenic iPS cell offspring still remained low. Moreover, the instructive influence of chondrocytes remained debatable, because the switch of culture conditions even without the influence of chondrocytes is known to induce iPS cell differentiation and the studies did not include a chondrocyte-free control group. Compared to the strong induction of spontaneous iPS cell differentiation by discontinuing high-dose bFGF treatment and increasing medium exchange intervals (Diederichs and Tuan 2014; Frobel et al. 2014; Guzzo et al. 2013; Lian et al. 2010), the actual effect of co-culture with chondrocytes appears minor.

In vitro recapitulation of embryonic cartilage development is currently considered to be the most promising and successful method for iPS cell differentiation (Umeda et al. 2012; Wu et al. 2013; Yamashita et al. 2015). This strategy requires sequential induction of multiple developmental phases including a mesendodermal and a mesodermal stage before induction into the chondrogenic lineage. Two recent publications suggest that such an approach may be capable to induce formation of stable hyaline cartilage from human iPS cells with a quality that might well match that of primary chondrocytes (Umeda et al. 2012; Yamashita et al. 2015). Wnt agonists and activin A are typically used to enhance formation of a mesodermal subpopulation, but the optimal growth factor cocktail has yet to be found. Thus, both studies relied on an enrichment strategy to exclude mis-differentiated cells that would otherwise impede efficient chondrogenesis. The Nakayama group enriched cells expressing platelet-derived growth factor receptor- α (PDGFR α) but not vascular endothelial growth factor receptor-2 (FLK1 or KDR) (Umeda et al. 2012). These surface receptors had been suggested in previous studies with mouse embryonic stem cells to define the mesodermal lineage (Tanaka et al. 2009). The Tsumaki

group chose an approach that did not rely on an uncertain correlation between surface markers and chondrogenic capacity. Instead, they argued that embryonic cartilage formation is initiated by mesenchymal condensation and consequently isolated such condensations early after chondrogenic induction of mesodermally predifferentiated human iPS cells (Yamashita et al. 2015). The chondrogenically induced human iPS cells deposited ample cartilaginous matrix rich in collagen type II and proteoglycans in both studies. Fibrous collagen type I and hypertrophic collagen type X that usually both accompany collagen type II deposition during chondrogenesis of human MSCs seemed to remain largely absent. In vivo, the Tsumaki group reported that their human iPS cell-derived cartilage nodules remained stable for up to 12 months at ectopic sites in immune-deficient mice. Moreover, neocartilage survived and integrated into focal cartilage defects in both immune-deficient rats and in immunosuppressed mini-pigs without any signs of tumor formation.

Thus, first highly promising reports suggest that hyaline cartilage can be generated from human iPS cells via a mesodermal pre-differentiation followed by chondrogenic induction. Qualitatively, the iPS cell-derived chondrocytes seemed to match the chondrogenic capacity of primary chondrocytes and to outperform MSCs that can only be induced to form fibrous and hypertrophic cartilage in vitro. However, the work has now to be repeated by other groups with independent iPS cells from different sources. Moreover, optimal mesodermal stimuli have yet to be found in order to obtain homogenous differentiation, and enrichment strategies which make the protocols impracticable for routine high-throughput application have to be avoided.

Instead of completing chondrogenesis and generating cartilage-producing chondrocytes, the differentiation process of iPS cells could also be arrested at a specific (meta-)stable intermediate that comprises the capacity to mature after implantation and give rise to therapeutically active cells under orthotopic instruction. IPS cell-derived mesenchymal progenitor cells (iMPCs) could potentially represent such a therapeutically interesting intermediate. iMPCs are immature cells in a transitional state of development (between stem cells and terminally differentiated cells), and their restricted capacity to self-renew and differentiate into only musculoskeletal tissues should reduce tumorigenicity. Thus, iMPCs may be highly attractive with regard to safety aspects and therapeutic applicability. Various methods have been reported to generate iMPCs including embryoid body outgrowth culture (Diederichs and Tuan 2014; Frobel et al. 2014; Koyama et al. 2013; Li et al. 2010), spontaneous differentiation of iPS cells (Diederichs and Tuan 2014; Frobel et al. 2014; Guzzo et al. 2013; Hynes et al. 2014; Kang et al. 2015; Lian et al. 2010), and co-culture with MSCs (Diederichs and Tuan 2014) (Fig. 4.2). iMPCs resemble MSCs by expressing a similar surface marker profile (CD44⁺, CD73⁺, CD90⁺, CD105⁺, CD34⁻, CD45⁻) and a general capacity for in vitro differentiation into the three mesenchymal lineages (cartilage, bone, fat; Fig. 4.3). Thus, iMPCs and equivalent derivatives from embryonic stem cells have frequently been termed "MSCs" (Chen et al. 2010; Hynes et al. 2014; Kang et al. 2015; Kimbrel et al. 2014). Direct comparison with MSCs from the same donor, however, showed that despite evident similarities, iMPCs are not completely



Fig. 4.3 iMPCs are MSC-like cells. (a) Morphology of iPS cell colonies, iMPCs and MSCs, as indicated. (b) iMPCs express the surface markers CD73, CD105, and CD90 that are typical for MSCs. (c) iMPCs are capable to differentiate in vitro into the three mesenchymal lineages and deposit collagen type II after chondrogenic stimulation (*top*), mineralize their matrix under osteogenic stimulation (Alizarin red staining, *middle*), and form intracellular lipid vesicles upon adipogenic stimulation (Oil Red O staining, *bottom*)

equivalent to MSCs (Diederichs and Tuan 2014; Frobel et al. 2014). The most striking differences were their DNA methylation profile, which was still more closely related to iPS cells than to MSCs (Frobel et al. 2014), and their inferior inducibility to differentiate into the three mesenchymal lineages (Diederichs and Tuan 2014; Frobel et al. 2014). Stimuli that easily induced MSC differentiation were less efficient for iMPCs (Diederichs and Tuan 2014), and this was also true for chondrogenesis which was less robust with iMPCs than with MSCs when induced with the same stimuli. Although additional treatment with BMPs can enhance iMPC chondrogenesis, deposition of cartilaginous matrix was still unsatisfactory (Guzzo et al. 2013; Saitta et al. 2014), and the ideal chondrogenic stimuli for iMPCs remain to be discovered. Whether the orthotopic articular environment in vivo might be able to more efficiently instruct iMPCs to differentiate into chondrocytes and form stable hyaline cartilage has yet to be investigated. In addition to finding adequate chondrogenic stimuli, iMPC generation strategies could also be adapted to yield more potent iMPC populations that can be more easily induced to differentiate. This overlaps with the developmental strategies mentioned above where growth factor cocktails are investigated with the aim to enhance formation of mesodermal/mesenchymal subpopulations without yielding an ideal combination yet.

In the recent years, cell implantation research has faced a paradigm shift with regard to the supposed activity and function of implanted cells. The original idea that implanted cells actively generate new tissue and their differentiation into tissue-forming cells is the major requirement for stem cells has been challenged by observations of beneficial effects of cell therapies even without long-term cell engraftment (Boukhechba et al. 2011; Giannoni et al. 2010; Niemietz et al. 2014; Seebach et al. 2014; Zimmermann et al. 2011). The paracrine activity of implanted cells secreting cytokines including growth and differentiation factors as well as immunomodulatory substances becomes increasingly recognized (Caplan and Dennis 2006; Liang et al. 2014; Ranganath et al. 2012). Thus, the major contribution of transplanted cells might be to control the inflammatory response, to attract endogenous repair cells to the defect site, and to support the intrinsic regenerative capacity. This general reformation of cell therapy will also significantly influence iPS cell research, and, consequently, the paracrine activity of iPS cell derivatives is currently being investigated (Lian et al. 2010). The presently most interesting questions are whether iPS cell-derived chondrocytes or iMPCs stably engraft into a cartilage defect, whether their paracrine activity might outweigh their capacity to form cartilage, and which iPS cell descendent is most potent to regenerate cartilage.

In summary, highly promising results have been achieved for differentiating iPS cells into chondrocytes that form stable hyaline cartilage of high quality without undesired fibrous or hypertrophic features. The next step will be to simplify the current differentiation strategies and to improve their efficiency in order to yield therapeutically active cells with feasible efforts and costs. The key will thereby be to either find factors stringent enough to force the complete iPS cell population into homogeneous differentiation or else to develop culture conditions that automatically enrich for specific subpopulations—for example, by size exclusion of cell condensates—without manual intervention or sorting.

Instead of completely differentiating iPS cells into chondrocytes, mesenchymal intermediates like iMPCs may be simpler to generate in shorter time. Whether iMPCs may become an attractive therapeutic cell source will depend on our ability to find stimuli that reproducibly induce iMPCs with a robust chondrogenic activity and/or a high paracrine activity. In comparison to bone marrow-derived MSCs, that seem to be primed for endochondral development, iPS cells and their offspring should overcome the risk of ectopic bone formation in the joint. Moreover, given the biologic variability of MSC activity, which is strongly dependent on donor age (Dexheimer et al. 2011; Majors et al. 1997; Muschler et al. 2001), general health status (Murphy et al. 2002), and harvest procedure (Muschler et al. 2001), iPS cells may in the future comprise a more robust and less vulnerable cell source. Thus, iPS cells are attractive for cell-based cartilage regeneration and promise to overcome limited supply and activity of currently available cell sources.

4.4 Disease in a Dish: In Vitro iPS Cell Models

Generating in vitro disease models is often considered to be the most promising future iPS cell application. For elucidating pathomechanisms and for drug development, stable cell lines can be utilized for a multitude of experiments and screens. The cumbersome process of iPS cell generation that might be uneconomic for treating just one specific patient can here become profitable. Animal models have enormously contributed to elucidate pathomechanisms and will certainly remain indispensable for preclinical drug testing in the near future. In vitro models are, however, highly valuable complements because they allow screens of large substance libraries. Moreover, they represent the only possibility for preclinical tests and investigations in the human system which can considerably deviate from the regulatory mechanisms observed in animal models that often recapitulate human diseases incompletely. Because of the differences in size and movement between mice and human, the cartilage composition deviates. Mouse cartilage is necessarily much thinner (0.03 mm on the femur) than human cartilage (2.2 mm on the femur), and chondrocytes represent 15-40% of the mouse femoral cartilage volume but only 2% in humans (Aigner et al. 2010). Thus, human chondrocytes produce far more cartilaginous matrix than mouse chondrocytes. Like for cell-based therapies, primary tissue-specific cells like chondrocytes would be ideal for disease models but are not available in sufficient amounts. The obtainable number and the expandability of disease-specific chondrocytes that could potentially be isolated from specific patients are severely limited and allow only very few experiments. MSCs, that are available in larger quantities and can be expanded to some degree without losing quality, cannot be differentiated into stable hyaline cartilage in vitro. Immortalized cell lines are available in ample quantities allowing large screens, but their phenotype and behavior are often abnormal, thus limiting the transferability of observations from immortalized cell lines to equivalent tissue cells.

Human iPS cells offer the attractive possibility to derive disease-specific cell lines with unlimited expandability. They can thus substantially contribute to the investigation of pathomechanisms and to the development of novel drugs not only for common diseases but also for rare disorders that have been hard to investigate. This includes of course osteoarthritis, the most frequent degenerative joint disease, but also the numerous hereditary osteochondral dysplasias which result from genetic disorders causing defective cartilage and bone differentiation, formation, and growth (Ikegawa 2006). Malformations can be manifest already at birth as in achondroplasia and Kniest dysplasia that cause dwarfism, as well as in the lethal forms of achondrogenesis, thanatophoric dysplasia, and campomelic dysplasia. Other dysplasias like multiple epiphyseal dysplasia cause early onset of osteoarthritis or cause anarchical cartilage and fibrous tissue-like enchondromatosis or multiple osteochondromatosis where multiple cartilage cysts (enchondroma) or bone tumors (exostoses) are formed. For many dysplasias, the disease-causing mutations are already known (Ikegawa 2006). The exact mechanisms, on the other hand, how, for example, defective collagens cause the multiple symptoms of collagenopathies, or how a gain of function mutation of an FGF receptor can result in dwarfism or lethal respiratory insufficiency, and how such conditions can be treated, are currently largely unknown. IPS technology now offers the possibility to generate and study cell lines carrying the disease-causing mutations. In addition to reprogramming patient-specific cells, novel gene editing methods allow introducing genetic defects into well-characterized iPS cell lines. This will considerably increase the comparability between independent studies and the quality of the results. Since iPS cells are not committed to a certain developmental lineage, the effects of a certain disease or drug can be studied in various cells or tissues with an identical genotype that are derived from the same iPS cell clone.

The main challenge when attempting to establish an iPS cell-based in vitro model is to establish a phenotype that correctly recapitulates the disease in vitro. This is one major limitation for studies attempting to establish an iPS cell-based model for osteoarthritis which is not necessarily caused by a genetic defect but rather by multiple concurrent factors including age, mechanical overload, or trauma and genetic predisposition. Thus, even iPS cells reprogrammed from osteoarthritic chondrocytes have so far not been shown to recapitulate any relevant disease mechanism in the dish (Kim et al. 2011; Lee et al. 2014; Wei et al. 2012). Obviously, genetic diseases where the causative mutation is known like skeletal dysplasias seem far better suitable for iPS cell models. However, even with known genetic mutations, obtaining a phenotype can prove challenging, because the genetic mutation alone might not be sufficient to induce a functional defect under basal culture conditions without additional stressors (Kim et al. 2013). Diseases that manifest after childhood or at an advanced age might be hard to recapitulate in short-term culture. To imitate aging, cellular stressors could be applied or trophic factors

withdrawn. Such strategies may help establishing an iPS cell model even for osteoarthritis in the near future.

With respect to diseases caused mainly by genetic defects, first very promising results have been reported. In a remarkable study, the Tsumaki group has reprogrammed fibroblasts from patients with thanatophoric dysplasia and achondroplasia caused by FGFR3 gain of function mutations (Yamashita et al. 2014). During in vitro differentiation, the disease-specific iPS cells were less able to form cartilaginous tissue, exhibited decreased proliferation, and increased apoptosis. The phenotype was rescued by knocking down the causative FGFR3 activity, thus proving that the FGFR3 mutation in these cells caused the phenotype. Once they had verified that the diseased iPS cells recapitulated the main abnormalities typical for patients and models of FGFR3-related diseases, the authors went on to screen for substances capable to rescue the diseased phenotype and found statins to improve cell survival and increase proliferation and cartilage formation. In a chondrodysplasia mouse model, statins were subsequently found to rescue reduced bone growth. Statins are clinically approved to reduce cholesterol levels but have so far not been considered for the treatment of chondrodysplasia. Thus, this iPS cell model has enabled the discovery of a new treatment strategy, has given first insights into the mechanisms of statin effects on chondrodysplasia cells, and has a high potential for further mechanistic studies of FGFR3-related chondrodysplasia.

Another interesting study focused on neonatal-onset multisystem inflammatory disease (NOMID) which is an inherited auto-inflammatory disease caused by NLRP3 mutations (Yokoyama et al. 2015). Utilizing their NOMID iPS cells as an in vitro disease model, the authors assessed the question why the epiphyseal overgrowth in NOMID is resistant to anti-IL-1ß therapy while systemic inflammation can effectively be controlled with anti-IL-1ß treatment. The data suggested that in chondrocytes with disease-causing mutations in NLRP3, the chondrogenic master transcription factor SOX9 was abnormally strongly expressed, induced by the cAMP/PKA/CREB signaling pathway, and this may cause overproduction of extracellular matrix independently of the NLRP3 inflammasome. This study excellently demonstrated the strength of iPS model to elucidate mechanisms and to thus promote the development of new therapies. Another very interesting aspect of this study was the derivation of isogenic healthy control cells enabled by the genetic mosaicism of the disease. In such cases like NOMID, the primary patient cell population comprises both wild-type and mutated cells, and, thus, iPS cell clones can be derived that differ only in the altered chromosome or mutated gene and the integration sites of the introduced pluripotency factors. Indeed, isogenic cells seem the ideal controls, given the increasingly realized fact that every person carries diseaserelevant single-nucleotide polymorphisms (Abecasis et al. 2010) meaning that essentially no iPS cell line could be defined as "healthy." Such isogenic controls can also easily be obtained when the disease-causing mutation is introduced into a welldescribed control iPS cell line or when the mutation of a patient-derived cell line is corrected via gene editing techniques.

A third insightful study established an iPS cell model for type II collagenopathy (Okada et al. 2015), which comprises a number of diseases caused by heterozygous mutations in the collagen type II gene. While previous investigations had been limited by difficulties to obtain live chondrocytes from patients, the authors generated iPS cells from dermal fibroblasts. In addition, they also directly induced the chondrocyte phenotype in diseased fibroblasts by ectopically expressing two pluripotency factors KLF4 and C-MYC in combination with the chondrogenic master transcription factor SOX9. This technique of direct conversion of fibroblasts into chondrocytes is another potential cell source for cartilage regeneration. Bypassing the stage of pluripotency, direct conversion certainly promises to be a quicker way of generating chondrocytes from an unrelated cell source than reprogramming and subsequent differentiation into chondrocytes. However, only at the iPS cell stage, cells are indefinitely expandable, and thus limited cell numbers become an issue yet again. The authors used direct lineage conversion into so-called iChon cells complementary to the iPS technology. They showed that both diseased iChon cells and chondrocytes differentiated from diseased iPS cells recapitulated the expected pathological features. Collagen type II secretion was severely limited and accompanied by increased expression of stress markers of the endoplasmic reticulum as well as increased apoptosis. Cartilage in teratomas generated from diseased iPS cells in vivo resembled the pathological cartilage tissue in patients with type II collagenopathy. Intracellular retention of collagen type II was accompanied by an apparently decreased density of the extracellular matrix. In general, the amount of cartilage in disease-specific teratomas was decreased compared to teratomas obtained from control iPS cells. Beyond successful recapitulation of the disease phenotype, the data gave new insights into the pathomechanisms of type II collagenopathy and suggested that cellular stress and apoptosis are the central issues of this disorder rather than chondrogenic maturation which seemed to be less affected. Moreover, chemically increasing protein folding seemed to reduce cellular stress and apoptosis, while challenging the cells by forcing chondrogenic differentiation and collagen type II synthesis further increased cell stress and apoptosis.

This study demonstrated that the iPS cell technology can be successfully complemented by direct lineage conversion that can quickly provide a certain amount of cells for a number of assays. Generating iPS cells and subsequent differentiation into the cells of interest can take several months, but the infinite expansion capacity of iPS cells enables high-throughput drug screening. Moreover, disease-specific teratomas even allow in vivo studies in human tissues to test specific drugs and novel therapies.

In summary, in vitro iPS cell models are particularly valuable for cartilage diseases like skeletal dysplasias that are directly caused by genetic variations. Pathomechanisms that are not correctly recapitulated in animal models can be investigated in vitro, and high-throughput screenings can be performed to find new therapeutic substances. Novel gene editing methods enable generating adequate controls, and first studies have already yielded promising results by finding additional effective therapeutics (Yamashita et al. 2014) or by enlightening the mechanism behind a phenotype resistant to current therapy (Okada et al. 2015; Yokoyama et al. 2015).

4.5 Challenges and Gaps of Knowledge

The new field of iPS research has evolved tremendously in the past years. The possibilities of a universal cell source with infinite expandability that can be generated by simply overexpressing a defined set of transcription factors currently seem tremendous. Certainly, to harness the full potential of iPS cells, we need a more detailed understanding of the highly artificial and massively invasive cellular reprogramming process. Stringent quality controls and elucidation of the pluripotency network will help minimizing iPS cell variability that can arise from diverse origins. These include the variable presence of preexisting mutations in the originating cells (Gore et al. 2011), incomplete reprogramming including retained epigenetic memory and aberrant reprogramming of DNA methylation (Lister et al. 2011), newly arising mutations and changes of gene copy number during reprogramming and culture (Gore et al. 2011; Laurent et al. 2011), as well as aberrations in imprinting (Trounson et al. 2012). Of course, such variants can considerably affect the capacity of iPS cells to differentiate into cells desired for cell-based therapy or disease models. Our continuously advancing ability to assess whole-genome DNA methylation, genome-wide expression, and the complete cellular metabolome will help setting up sophisticated quality controls and thus to standardize cellular reprogramming (Muller et al. 2011).

With regard to clinical safety, a plethora of new techniques has been developed to overcome the potential risks of iPS cells. To avoid retroviral transgene delivery that results in random integrations into the genome, thus risking insertional mutagenesis and proto-oncogene activation, non-integrating vectors can be used. Adenoviral vectors (Stadtfeld et al. 2008), nonviral and episomal vectors (Yu et al. 2009), expression plasmids (Okita et al. 2008), and piggyBac transposition (Woltjen et al. 2009) have already been successfully applied for cellular reprogramming. Also, the Crerecombinase system (Soldner et al. 2009) can be used to excise the transgenes after iPS cell generation when the endogenous pluripotency network is activated. Small molecule-based chemical reprogramming (Hou et al. 2013; Shu et al. 2013), proteinbased strategies (Zhou et al. 2009), and introduction of microRNA clusters (Anokye-Danso et al. 2011; Lin et al. 2008) promise to eliminate many drawbacks by completely avoiding genetic manipulation and the ectopic expression of oncogenes.

Significant progress has also been made with regard to the inherent risk of tumorigenicity which requires all iPS cells to be removed from cell-based therapeutics before implantation. Sophisticated purification strategies using lineage-specific reporters (Tabar and Studer 2014) enable identification and enrichment of specific cells, i.e., cells committed to a certain lineage or residual pluripotent cells (Tang et al. 2011). Also, so-called suicide genes have been proposed for eliminating all transplanted cells in case of any adverse effects (Di Stasi et al. 2011).

Furthermore, xeno-free and GMP-compliant culture conditions have been developed in order to generate clinical-grade iPS cells (Nakagawa et al. 2014). Suspension culture and large-scale culture conditions are being tested for iPS cells and their offspring in order to overcome the current elaborate culture in small colonies and to provide large cell quantities (Haraguchi et al. 2013; McLaren et al. 2013; Olmer et al. 2012). In summary, the major current challenges are the considerable variability of independently derived iPS cells and the generation of sufficient amounts of relevant iPS cell progeny with acceptable efforts. For clinical application, safety issues are of major concern, and insertional mutagenesis, potential tumorigenesis, and ectopic tissue growth need to be controlled. Sophisticated technologies and continuously increasing insights into stem cell biology have already brought us close to overcoming these hurdles in the near future.

4.6 Summary

IPS cells are on the verge of becoming a valuable tool for cartilage regeneration. First, iPS cell-based in vitro models of genetic cartilage diseases have resulted in novel insights into pathomechanisms induced, for example, by a constitutively active FGF receptor or misfolded collagen. A new treatment strategy for FGFRrelated skeletal dysplasia with statins has been proposed and can now be tested. The promising results of these first cartilage-related iPS cell-based disease models will certainly spur the generation of models for other cartilage diseases including dysplasias caused by mutations in other collagens (types IX, XI), SOX9, filamin B, COMP, matrilin, PTHR1, or the sulfate transporter SLC26A2. High-throughput screens that had so far relied on immortalized cell lines with atypical phenotypes can now be performed with human disease-specific cells. Also in vivo tests with human tissues formed in iPS cell-derived teratomas are possible. Thus, the true strength of iPS cells for cartilage regeneration currently is to serve as the most realistic human in vitro model for disease mechanisms as well as for normal human development. Thus, iPS cell research will increase our understanding of the processes and regulatory mechanisms during chondrogenic differentiation, cartilage formation, and tissue homeostasis and will teach us to selectively manipulate these processes.

In addition, iPS cells are also highly attractive for cell-based therapies and offer to overcome the invasive tissue harvest procedure and limited supply for autologous chondrocyte implantation. Whether mesenchymal intermediates like iMPCs may replace iPS cell-derived chondrocytes as the most attractive therapeutic cells will depend on our ability to improve iMPC generation strategies to give rise to homogeneous, highly active cells. Stringent standardization of reprogramming and iMPC derivation could be envisioned to yield "super MSCs" with high regenerative capacity that is not afflicted with endochondral priming and donor variability. Of course, the regulatory hurdles for novel cell therapies are high and have so far prevented the translation of MSC-based therapies into clinical application. In contrast to MSCs, however, iPS cells are the only potential cell source for life-threatening diseases like cardiac insufficiency, kidney failure, Parkinson's disease, or stroke damage, and such fatal maladies might spur the approval for clinical application. Our recent advances to generate clinical-grade iPS cells are very promising, and the establishment of iPS cell banks as well as first in man clinical phase I trials for macular degeneration indicates that iPS cell therapy might one day become reality.

References

- Abecasis GR, Altshuler D, Auton A, Brooks LD, Durbin RM, Gibbs RA, Hurles ME, McVean GA (2010) A map of human genome variation from population-scale sequencing. Nature 467(7319):1061–1073. doi:10.1038/nature09534
- Aigner T, Cook JL, Gerwin N, Glasson SS, Laverty S, Little CB, McIlwraith W, Kraus VB (2010) Histopathology atlas of animal model systems – overview of guiding principles. Osteoarthritis Cartilage 18(Suppl 3):S2–S6. doi:http://dx.doi.org/10.1016/j.joca.2010.07.013
- Anokye-Danso F, Trivedi CM, Juhr D, Gupta M, Cui Z, Tian Y, Zhang Y, Yang W, Gruber PJ, Epstein JA, Morrisey EE (2011) Highly efficient miRNA-mediated reprogramming of mouse and human somatic cells to pluripotency. Cell Stem Cell 8(4):376–388. doi:10.1016/j. stem.2011.03.001
- Bianco P (2014) "Mesenchymal" stem cells. Annu Rev Cell Dev Biol 30:677–704. doi:10.1146/ annurev-cellbio-100913-013132
- Bianco P, Gehron RP (2000) Marrow stromal stem cells. J Clin Invest 105(12):1663-1668
- Boukhechba F, Balaguer T, Bouvet-Gerbettaz S, Michiels JF, Bouler JM, Carle GF, Scimeca JC, Rochet N (2011) Fate of bone marrow stromal cells in a syngenic model of bone formation. Tissue Eng Part A 17(17-18):2267–2278. doi:10.1089/ten.TEA.2010.0461
- Bruder SP, Jaiswal N, Haynesworth SE (1997) Growth kinetics, self-renewal, and the osteogenic potential of purified human mesenchymal stem cells during extensive subcultivation and following cryopreservation. J Cell Biochem 64(2):278–294
- Caplan AI (1991) Mesenchymal stem cells. J Orthop Res 9:641-650
- Caplan AI, Dennis JE (2006) Mesenchymal stem cells as trophic mediators. J Cell Biochem 98(5):1076–1084. doi:10.1002/jcb.20886
- Chen TS, Lai RC, Lee MM, Choo AB, Lee CN, Lim SK (2010) Mesenchymal stem cell secretes microparticles enriched in pre-microRNAs. Nucleic Acids Res 38(1):215–224. doi:10.1093/ nar/gkp857
- Dexheimer V, Mueller S, Braatz F, Richter W (2011) Reduced reactivation from dormancy but maintained lineage choice of human mesenchymal stem cells with donor age. PLoS One 6(8):e22980. doi:10.1371/journal.pone.0022980
- Di Stasi A, Tey SK, Dotti G, Fujita Y, Kennedy-Nasser A, Martinez C, Straathof K, Liu E, Durett AG, Grilley B, Liu H, Cruz CR, Savoldo B, Gee AP, Schindler J, Krance RA, Heslop HE, Spencer DM, Rooney CM, Brenner MK (2011) Inducible apoptosis as a safety switch for adoptive cell therapy. N Engl J Med 365(18):1673–1683. doi:10.1056/NEJMoa1106152
- Dickhut A, Dexheimer V, Martin K, Lauinger R, Heisel C, Richter W (2010) Chondrogenesis of human mesenchymal stem cells by local transforming growth factor-beta delivery in a biphasic resorbable carrier. Tissue Eng Part A 16(2):453–464. doi:10.1089/ten.TEA.2009.0168
- Diederichs S, Tuan RS (2014) Functional comparison of human-induced pluripotent stem cellderived mesenchymal cells and bone marrow-derived mesenchymal stromal cells from the same donor. Stem Cells Dev 23(14):1594–1610. doi:10.1089/scd.2013.0477
- Elves MW, Zervas J (1974) An investigation into the immunogenicity of various components of osteoarticular grafts. Br J Exp Pathol 55(4):344–351
- Erices A, Conget P, Minguell JJ (2000) Mesenchymal progenitor cells in human umbilical cord blood. Br J Haematol 109(1):235–242
- Frobel J, Hemeda H, Lenz M, Abagnale G, Joussen S, Denecke B, Saric T, Zenke M, Wagner W (2014) Epigenetic rejuvenation of mesenchymal stromal cells derived from induced pluripotent stem cells. Stem Cell Rep 3(3):414–422. doi:10.1016/j.stemcr.2014.07.003
- Giannoni P, Scaglione S, Daga A, Ilengo C, Cilli M, Quarto R (2010) Short-time survival and engraftment of bone marrow stromal cells in an ectopic model of bone regeneration. Tissue Eng Part A 16(2):489–499. doi:10.1089/ten.TEA.2009.0041
- Giorgetti A, Montserrat N, Aasen T, Gonzalez F, Rodriguez-Piza I, Vassena R, Raya A, Boue S, Barrero MJ, Corbella BA, Torrabadella M, Veiga A, Izpisua Belmonte JC (2009) Generation of induced pluripotent stem cells from human cord blood using OCT4 and SOX2. Cell Stem Cell 5(4):353–357. doi:10.1016/j.stem.2009.09.008

- Gore A, Li Z, Fung HL, Young JE, Agarwal S, Antosiewicz-Bourget J, Canto I, Giorgetti A, Israel MA, Kiskinis E, Lee JH, Loh YH, Manos PD, Montserrat N, Panopoulos AD, Ruiz S, Wilbert ML, Yu J, Kirkness EF, Izpisua Belmonte JC, Rossi DJ, Thomson JA, Eggan K, Daley GQ, Goldstein LS, Zhang K (2011) Somatic coding mutations in human induced pluripotent stem cells. Nature 471(7336):63–67. doi:10.1038/nature09805
- Guzzo RM, Gibson J, Xu RH, Lee FY, Drissi H (2013) Efficient differentiation of human iPSCderived mesenchymal stem cells to chondroprogenitor cells. J Cell Biochem 114(2):480–490. doi:10.1002/jcb.24388
- Haase A, Olmer R, Schwanke K, Wunderlich S, Merkert S, Hess C, Zweigerdt R, Gruh I, Meyer J, Wagner S, Maier LS, Han DW, Glage S, Miller K, Fischer P, Scholer HR, Martin U (2009) Generation of induced pluripotent stem cells from human cord blood. Cell Stem Cell 5(4):434– 441. doi:10.1016/j.stem.2009.08.021
- Haraguchi Y, Matsuura K, Shimizu T, Yamato M, Okano T (2013) Simple suspension culture system of human iPS cells maintaining their pluripotency for cardiac cell sheet engineering. J Tissue Eng Regen Med. doi:10.1002/term.1761
- Hou P, Li Y, Zhang X, Liu C, Guan J, Li H, Zhao T, Ye J, Yang W, Liu K, Ge J, Xu J, Zhang Q, Zhao Y, Deng H (2013) Pluripotent stem cells induced from mouse somatic cells by small-molecule compounds. Science 341(6146):651–654. doi:10.1126/science.1239278
- Hunt HE, Sadr K, Deyoung AJ, Gortz S, Bugbee WD (2014) The role of immunologic response in fresh osteochondral allografting of the knee. Am J Sports Med 42(4):886–891. doi:10.1177/0363546513518733
- Hynes K, Menicanin D, Mrozik K, Gronthos S, Bartold PM (2014) Generation of functional mesenchymal stem cells from different induced pluripotent stem cell lines. Stem Cells Dev 23(10):1084–1096. doi:10.1089/scd.2013.0111
- Ikegawa S (2006) Genetic analysis of skeletal dysplasia: recent advances and perspectives in the post-genome-sequence era. J Hum Genet 51(7):581–586. doi:10.1007/s10038-006-0401-x
- Janicki P, Kasten P, Kleinschmidt K, Luginbuehl R, Richter W (2010) Chondrogenic pre-induction of human mesenchymal stem cells on beta-TCP: enhanced bone quality by endochondral heterotopic bone formation. Acta Biomater 6(8):3292–3301. doi:10.1016/j.actbio.2010.01.037
- Kang R, Zhou Y, Tan S, Zhou G, Aagaard L, Xie L, Bunger C, Bolund L, Luo Y (2015) Mesenchymal stem cells derived from human induced pluripotent stem cells retain adequate osteogenicity and chondrogenicity but less adipogenicity. Stem Cell Res Ther 6(1):144. doi:10.1186/ s13287-015-0137-7
- Kim MJ, Son MJ, Son MY, Seol B, Kim J, Park J, Kim JH, Kim YH, Park SA, Lee CH, Lee KS, Han YM, Chang JS, Cho YS (2011) Generation of human induced pluripotent stem cells from osteoarthritis patient-derived synovial cells. Arthritis Rheum 63(10):3010–3021. doi:10.1002/ art.30488
- Kim C, Wong J, Wen J, Wang S, Wang C, Spiering S, Kan NG, Forcales S, Puri PL, Leone TC, Marine JE, Calkins H, Kelly DP, Judge DP, Chen HS (2013) Studying arrhythmogenic right ventricular dysplasia with patient-specific iPSCs. Nature 494(7435):105–110. doi:10.1038/nature11799
- Kimbrel EA, Kouris NA, Yavanian GJ, Chu J, Qin Y, Chan A, Singh RP, McCurdy D, Gordon L, Levinson RD, Lanza R (2014) Mesenchymal stem cell population derived from human pluripotent stem cells displays potent immunomodulatory and therapeutic properties. Stem Cells Dev 23(14):1611–1624. doi:10.1089/scd.2013.0554
- Koyama N, Miura M, Nakao K, Kondo E, Fujii T, Taura D, Kanamoto N, Sone M, Yasoda A, Arai H, Bessho K, Nakao K (2013) Human induced pluripotent stem cells differentiated into chondrogenic lineage via generation of mesenchymal progenitor cells. Stem Cells Dev 22(1):102– 113. doi:10.1089/scd.2012.0127
- Laurent LC, Ulitsky I, Slavin I, Tran H, Schork A, Morey R, Lynch C, Harness JV, Lee S, Barrero MJ, Ku S, Martynova M, Semechkin R, Galat V, Gottesfeld J, Izpisua Belmonte JC, Murry C, Keirstead HS, Park HS, Schmidt U, Laslett AL, Muller FJ, Nievergelt CM, Shamir R, Loring JF (2011) Dynamic changes in the copy number of pluripotency and cell proliferation genes in human ESCs and iPSCs during reprogramming and time in culture. Cell Stem Cell 8(1):106–118. doi:10.1016/j.stem.2010.12.003

- Lee J, Kim Y, Yi H, Diecke S, Kim J, Jung H, Rim YA, Jung SM, Kim M, Kim YG, Park SH, Kim HY, Ju JH (2014) Generation of disease-specific induced pluripotent stem cells from patients with rheumatoid arthritis and osteoarthritis. Arthritis Res Ther 16(1):R41. doi:10.1186/ar4470
- Levy YD, Gortz S, Pulido PA, McCauley JC, Bugbee WD (2013) Do fresh osteochondral allografts successfully treat femoral condyle lesions? Clin Orthop Relat Res 471(1):231–237. doi:10.1007/s11999-012-2556-4
- Li F, Bronson S, Niyibizi C (2010) Derivation of murine induced pluripotent stem cells (iPS) and assessment of their differentiation toward osteogenic lineage. J Cell Biochem 109(4):643–652. doi:10.1002/jcb.22440
- Lian Q, Zhang Y, Zhang J, Zhang HK, Wu X, Lam FF, Kang S, Xia JC, Lai WH, Au KW, Chow YY, Siu CW, Lee CN, Tse HF (2010) Functional mesenchymal stem cells derived from human induced pluripotent stem cells attenuate limb ischemia in mice. Circulation 121(9):1113–1123
- Liang X, Ding Y, Zhang Y, Tse HF, Lian Q (2014) Paracrine mechanisms of mesenchymal stem cell-based therapy: current status and perspectives. Cell Transplant 23(9):1045–1059. doi:10.3 727/096368913X667709
- Lin SL, Chang DC, Chang-Lin S, Lin CH, Wu DT, Chen DT, Ying SY (2008) Mir-302 reprograms human skin cancer cells into a pluripotent ES-cell-like state. RNA 14(10):2115–2124. doi:10.1261/rna.1162708
- Lister R, Pelizzola M, Kida YS, Hawkins RD, Nery JR, Hon G, Antosiewicz-Bourget J, O'Malley R, Castanon R, Klugman S, Downes M, Yu R, Stewart R, Ren B, Thomson JA, Evans RM, Ecker JR (2011) Hotspots of aberrant epigenomic reprogramming in human induced pluripotent stem cells. Nature 471(7336):68–73. doi:10.1038/nature09798
- Majors AK, Boehm CA, Nitto H, Midura RJ, Muschler GF (1997) Characterization of human bone marrow stromal cells with respect to osteoblastic differentiation. J Orthop Res 15(4):546–557
- McLaren D, Gorba T, Marguerie de Rotrou A, Pillai G, Chappell C, Stacey A, Lingard S, Falk A, Smith A, Koch P, Brustle O, Vickers R, Tinsley J, Flanders D, Bello P, Craig S (2013) Automated large-scale culture and medium-throughput chemical screen for modulators of proliferation and viability of human induced pluripotent stem cell-derived neuroepithelial-like stem cells. J Biomol Screen 18(3):258–268. doi:10.1177/1087057112461446
- Medvedev SP, Grigor'eva EV, Shevchenko AI, Malakhova AA, Dementyeva EV, Shilov AA, Pokushalov EA, Zaidman AM, Aleksandrova MA, Plotnikov EY, Sukhikh GT, Zakian SM (2011) Human induced pluripotent stem cells derived from fetal neural stem cells successfully undergo directed differentiation into cartilage. Stem Cells Dev 20(6):1099–1112. doi:10.1089/ scd.2010.0249
- Muller FJ, Schuldt BM, Williams R, Mason D, Altun G, Papapetrou EP, Danner S, Goldmann JE, Herbst A, Schmidt NO, Aldenhoff JB, Laurent LC, Loring JF (2011) A bioinformatic assay for pluripotency in human cells. Nat Methods 8(4):315–317. doi:10.1038/nmeth.1580
- Murphy JM, Dixon K, Beck S, Fabian D, Feldman A, Barry F (2002) Reduced chondrogenic and adipogenic activity of mesenchymal stem cells from patients with advanced osteoarthritis. Arthritis Rheum 46(3):704–713. doi:10.1002/art.10118
- Muschler GF, Nitto H, Boehm CA, Easley KA (2001) Age- and gender-related changes in the cellularity of human bone marrow and the prevalence of osteoblastic progenitors. J Orthop Res 19(1):117–125
- Nakagawa M, Taniguchi Y, Senda S, Takizawa N, Ichisaka T, Asano K, Morizane A, Doi D, Takahashi J, Nishizawa M, Yoshida Y, Toyoda T, Osafune K, Sekiguchi K, Yamanaka S (2014) A novel efficient feeder-free culture system for the derivation of human induced pluripotent stem cells. Sci Rep 4:3594. doi:10.1038/srep03594
- Narcisi R, Cleary MA, Brama PA, Hoogduijn MJ, Tuysuz N, Ten Berge D, van Osch GJ (2015) Long-term expansion, enhanced chondrogenic potential, and suppression of endochondral ossification of adult human MSCs via WNT signaling modulation. Stem Cell Rep 4(3):459– 472. doi:10.1016/j.stemcr.2015.01.017
- Niemietz T, Zass G, Hagmann S, Diederichs S, Gotterbarm T, Richter W (2014) Xenogeneic transplantation of articular chondrocytes into full-thickness articular cartilage defects in minipigs:

fate of cells and the role of macrophages. Cell Tissue Res 358(3):749-761. doi:10.1007/s00441-014-1982-x

- Okada M, Ikegawa S, Morioka M, Yamashita A, Saito A, Sawai H, Murotsuki J, Ohashi H, Okamoto T, Nishimura G, Imaizumi K, Tsumaki N (2015) Modeling type II collagenopathy skeletal dysplasia by directed conversion and induced pluripotent stem cells. Hum Mol Genet 24(2):299–313. doi:10.1093/hmg/ddu444
- Okita K, Nakagawa M, Hyenjong H, Ichisaka T, Yamanaka S (2008) Generation of mouse induced pluripotent stem cells without viral vectors. Science 322(5903):949–953
- Okita K, Yamakawa T, Matsumura Y, Sato Y, Amano N, Watanabe A, Goshima N, Yamanaka S (2013) An efficient nonviral method to generate integration-free human-induced pluripotent stem cells from cord blood and peripheral blood cells. Stem Cells 31(3):458–466. doi:10.1002/ stem.1293
- Olmer R, Lange A, Selzer S, Kasper C, Haverich A, Martin U, Zweigerdt R (2012) Suspension culture of human pluripotent stem cells in controlled, stirred bioreactors. Tissue Eng Part C Methods 18(10):772–784. doi:10.1089/ten.TEC.2011.0717
- Park IH, Zhao R, West JA, Yabuuchi A, Huo H, Ince TA, Lerou PH, Lensch MW, Daley GQ (2008) Reprogramming of human somatic cells to pluripotency with defined factors. Nature 451(7175):141–146
- Pelttari K, Winter A, Steck E, Goetzke K, Hennig T, Ochs BG, Aigner T, Richter W (2006) Premature induction of hypertrophy during in vitro chondrogenesis of human mesenchymal stem cells correlates with calcification and vascular invasion after ectopic transplantation in SCID mice. Arthritis Rheum 54(10):3254–3266. doi:10.1002/art.22136
- Qu C, Puttonen KA, Lindeberg H, Ruponen M, Hovatta O, Koistinaho J, Lammi MJ (2013) Chondrogenic differentiation of human pluripotent stem cells in chondrocyte co-culture. Int J Biochem Cell Biol 45(8):1802–1812. doi:10.1016/j.biocel.2013.05.029
- Ranganath SH, Levy O, Inamdar MS, Karp JM (2012) Harnessing the mesenchymal stem cell secretome for the treatment of cardiovascular disease. Cell Stem Cell 10(3):244–258. doi:10.1016/j.stem.2012.02.005
- Saitta B, Passarini J, Sareen D, Ornelas L, Sahabian A, Argade S, Krakow D, Cohn DH, Svendsen CN, Rimoin DL (2014) Patient-derived skeletal dysplasia induced pluripotent stem cells display abnormal chondrogenic marker expression and regulation by BMP2 and TGFbeta1. Stem Cells Dev 23(13):1464–1478. doi:10.1089/scd.2014.0014
- Seebach E, Freischmidt H, Holschbach J, Fellenberg J, Richter W (2014) Mesenchymal stroma cells trigger early attraction of M1 macrophages and endothelial cells into fibrin hydrogels, stimulating long bone healing without long-term engraftment. Acta Biomater 10(11):4730– 4741. doi:10.1016/j.actbio.2014.07.017
- Seki T, Yuasa S, Fukuda K (2011) Derivation of induced pluripotent stem cells from human peripheral circulating T cells. Curr Protoc Stem Cell Biol Chapter 4:Unit4A.3. doi:10.1002/9780470151808.sc04a03s18
- Shu J, Wu C, Wu Y, Li Z, Shao S, Zhao W, Tang X, Yang H, Shen L, Zuo X, Yang W, Shi Y, Chi X, Zhang H, Gao G, Shu Y, Yuan K, He W, Tang C, Zhao Y, Deng H (2013) Induction of pluripotency in mouse somatic cells with lineage specifiers. Cell 153(5):963–975. doi:10.1016/j. cell.2013.05.001
- Soldner F, Hockemeyer D, Beard C, Gao Q, Bell GW, Cook EG, Hargus G, Blak A, Cooper O, Mitalipova M, Isacson O, Jaenisch R (2009) Parkinson's disease patient-derived induced pluripotent stem cells free of viral reprogramming factors. Cell 136(5):964–977
- Stadtfeld M, Nagaya M, Utikal J, Weir G, Hochedlinger K (2008) Induced pluripotent stem cells generated without viral integration. Science 322(5903):945–949
- Tabar V, Studer L (2014) Pluripotent stem cells in regenerative medicine: challenges and recent progress. Nat Rev Genet 15(2):82–92. doi:10.1038/nrg3563
- Takahashi K, Yamanaka S (2006) Induction of pluripotent stem cells from mouse embryonic and adult fibroblast cultures by defined factors. Cell 126(4):663–676. doi:10.1016/j.cell.2006.07.024
- Takahashi K, Yamanaka S (2013) Induced pluripotent stem cells in medicine and biology. Development 140(12):2457–2461. doi:10.1242/dev.092551

- Takahashi K, Tanabe K, Ohnuki M, Narita M, Ichisaka T, Tomoda K, Yamanaka S (2007) Induction of pluripotent stem cells from adult human fibroblasts by defined factors. Cell 131(5):861–872
- Takenaka C, Nishishita N, Takada N, Jakt LM, Kawamata S (2010) Effective generation of iPS cells from CD34+ cord blood cells by inhibition of p53. Exp Hematol 38(2):154–162. doi:10.1016/j.exphem.2009.11.003
- Tanaka M, Jokubaitis V, Wood C, Wang Y, Brouard N, Pera M, Hearn M, Simmons P, Nakayama N (2009) BMP inhibition stimulates WNT-dependent generation of chondrogenic mesoderm from embryonic stem cells. Stem Cell Res 3(2-3):126–141. doi:10.1016/j.scr.2009.07.001
- Tang C, Lee AS, Volkmer JP, Sahoo D, Nag D, Mosley AR, Inlay MA, Ardehali R, Chavez SL, Pera RR, Behr B, Wu JC, Weissman IL, Drukker M (2011) An antibody against SSEA-5 glycan on human pluripotent stem cells enables removal of teratoma-forming cells. Nat Biotechnol 29(9):829–834. doi:10.1038/nbt.1947
- Taylor CJ, Peacock S, Chaudhry AN, Bradley JA, Bolton EM (2012) Generating an iPSC bank for HLA-matched tissue transplantation based on known donor and recipient HLA types. Cell Stem Cell 11(2):147–152. doi:10.1016/j.stem.2012.07.014
- Trounson A, Shepard KA, DeWitt ND (2012) Human disease modeling with induced pluripotent stem cells. Curr Opin Genet Dev 22(5):509–516. doi:10.1016/j.gde.2012.07.004
- Tsumaki N, Okada M, Yamashita A (2015) iPS cell technologies and cartilage regeneration. Bone 70:48–54. doi:10.1016/j.bone.2014.07.011
- Turner M, Leslie S, Martin Nicholas G, Peschanski M, Rao M, Taylor Craig J, Trounson A, Turner D, Yamanaka S, Wilmut I (2013) Toward the development of a global induced pluripotent stem cell library. Cell Stem Cell 13(4):382–384. doi:http://dx.doi.org/10.1016/j.stem.2013.08.003
- Umeda K, Zhao J, Simmons P, Stanley E, Elefanty A, Nakayama N (2012) Human chondrogenic paraxial mesoderm, directed specification and prospective isolation from pluripotent stem cells. Sci Rep 2:455. doi:10.1038/srep00455
- Wei Y, Zeng W, Wan R, Wang J, Zhou Q, Qiu S, Singh SR (2012) Chondrogenic differentiation of induced pluripotent stem cells from osteoarthritic chondrocytes in alginate matrix. Eur Cell Mater 23:1–12
- Woltjen K, Michael IP, Mohseni P, Desai R, Mileikovsky M, Hamalainen R, Cowling R, Wang W, Liu P, Gertsenstein M, Kaji K, Sung HK, Nagy A (2009) piggyBac transposition reprograms fibroblasts to induced pluripotent stem cells. Nature 458(7239):766–770
- Wu L, Bluguermann C, Kyupelyan L, Latour B, Gonzalez S, Shah S, Galic Z, Ge S, Zhu Y, Petrigliano FA, Nsair A, Miriuka SG, Li X, Lyons KM, Crooks GM, McAllister DR, Van Handel B, Adams JS, Evseenko D (2013) Human developmental chondrogenesis as a basis for engineering chondrocytes from pluripotent stem cells. Stem Cell Rep 1(6):575–589. doi:10.1016/j.stemcr.2013.10.012
- Yamanaka S (2012) Induced pluripotent stem cells: past, present, and future. Cell Stem Cell 10(6):678–684. doi:10.1016/j.stem.2012.05.005
- Yamashita A, Morioka M, Kishi H, Kimura T, Yahara Y, Okada M, Fujita K, Sawai H, Ikegawa S, Tsumaki N (2014) Statin treatment rescues FGFR3 skeletal dysplasia phenotypes. Nature 513(7519):507–511. doi:10.1038/nature13775
- Yamashita A, Morioka M, Yahara Y, Okada M, Kobayashi T, Kuriyama S, Matsuda S, Tsumaki N (2015) Generation of scaffoldless hyaline cartilaginous tissue from human iPSCs. Stem Cell Rep 4(3):404–418. doi:10.1016/j.stemcr.2015.01.016
- Yokoyama K, Ikeya M, Umeda K, Oda H, Nodomi S, Nasu A, Matsumoto Y, Izawa K, Horigome K, Kusaka T, Tanaka T, Saito MK, Yasumi T, Nishikomori R, Ohara O, Nakayama N, Nakahata T, Heike T, Toguchida J (2015) Enhanced chondrogenesis of induced pluripotent stem cells from patients with neonatal-onset multisystem inflammatory disease occurs via the caspase 1-independent cAMP/protein kinase A/CREB pathway. Arthritis Rheumatol 67(1):302–314. doi:10.1002/art.38912
- Yu J, Vodyanik MA, Smuga-Otto K, Antosiewicz-Bourget J, Frane JL, Tian S, Nie J, Jonsdottir GA, Ruotti V, Stewart R, Slukvin II, Thomson JA (2007) Induced pluripotent stem cell lines derived from human somatic cells. Science 318(5858):1917–1920. doi:10.1126/ science.1151526

- Yu J, Hu K, Smuga-Otto K, Tian S, Stewart R, Slukvin II, Thomson JA (2009) Human induced pluripotent stem cells free of vector and transgene sequences. Science 324(5928):797–801
- Zhou H, Wu S, Joo JY, Zhu S, Han DW, Lin T, Trauger S, Bien G, Yao S, Zhu Y, Siuzdak G, Scholer HR, Duan L, Ding S (2009) Generation of induced pluripotent stem cells using recombinant proteins. Cell Stem Cell 4(5):381–384
- Zimmermann CE, Gierloff M, Hedderich J, Acil Y, Wiltfang J, Terheyden H (2011) Survival of transplanted rat bone marrow-derived osteogenic stem cells in vivo. Tissue Eng Part A 17(7-8):1147–1156. doi:10.1089/ten.TEA.2009.0577
- Zuk PA, Zhu M, Mizuno H, Huang J, Futrell JW, Katz AJ, Benhaim P, Lorenz HP, Hedrick MH (2001) Multilineage cells from human adipose tissue: implications for cell-based therapies. Tissue Eng 7(2):211–228. doi:10.1089/107632701300062859

Cell-Based Cartilage Regeneration

5

Johannes Zellner and Peter Angele

Abstract

Since decades various regenerative cell-based treatment options have been developed for cartilage repair. With the introduction of the autologous chondrocyte transplantation, also large-sized chondral defects can be successfully addressed. This chapter gives a short overview about current procedures for cellbased treatment strategies like bone marrow stimulation techniques, osteochondral transplantation, and chondrocyte transplantation. Requirements and outcome parameters for a successful treatment and future directions in cartilage regeneration are discussed. Finally treatment recommendations according to cartilage defect size and depth are given.

5.1 Introduction

Articular cartilage injuries are common. They can result from acute traumatic injuries, posttraumatic or early degenerative changes, osteochondritis dissecans, or avascular necrosis. Numerous reports analyzing high numbers of arthroscopies show cartilage lesions in up to 60% of the patients (Widuchowski et al. 2007). The incidence of chondral injuries indicates the high impact on the society, as it is

J. Zellner

University Medical Center Regensburg, Department of Trauma Surgery, Regensburg, Germany e-mail: johannes.zellner@ukr.de

P. Angele (⊠)
University Medical Center Regensburg, Department of Trauma Surgery, Regensburg, Germany

Sporthopaedicum Regensburg, Regensburg, Germany e-mail: peter.angele@ukr.de

[©] Springer International Publishing AG 2017 S. Grässel, A. Aszodi (eds.), *Cartilage*, DOI 10.1007/978-3-319-53316-2_5

generally agreed that the persistence of cartilage defects is a risk factor for joint dysfunction, which finally may lead to severe osteoarthritis.

Cartilage lesions can remain symptomless over a long period of time. This can cause a delayed diagnosis and late treatment of cartilage injuries, which may have negative consequences for joint recovery. Clearly, this emphasizes the importance of an adequate regenerative treatment of cartilage lesions at the earliest time point in order to prevent onset and development of osteoarthritis (Zellner et al. 2015).

Since decades regenerative treatment options for small- and middle-sized cartilage lesions were developed like, e.g., Pridie drilling or microfracture. With the introduction of the "autologous chondrocyte transplantation (ACT)" technique by Brittberg et al. in 1994, also large-sized cartilage defects can be successfully addressed via a regenerative approach (Brittberg et al. 1994).

Recently many randomized clinical trials investigated the efficiency and the quality of different cartilage repair procedures. Most of them enrolled young and active patients with "ideal" chondral defects that were focal and isolated with clearly defined borders (Lattermann and Luckett 2011). However, in reality, most patients that present with clinically symptomatic chondral lesions do not fulfill these criteria. Consecutively a detailed analysis and assessment of the cartilage defect and all underlying pathologies should be performed. Specific comorbidities have to be taken into account prior to performing regenerative cartilage repair as they may require additional concomitant or staged surgical procedures.

All cell-based cartilage repair strategies like bone marrow stimulating techniques or autologous chondrocyte transplantation require a correction of the comorbidities like malalignment, meniscal deficiency, instability, or pathologies of the subchondral bone. Only if the comorbidities are addressed sufficiently, the chance for appropriate cartilage regeneration is achievable.

Axis deviations can cause overload of an affected joint compartment. For cartilage treatment, malalignment needs be corrected to restore normal load distribution that allows the repair tissue to adjust to physiological loads. Corrected patellofemoral and tibiofemoral alignment improves clinical outcome when realignment operations are performed concurrently with the cartilage repair or as a staged procedure (Behery et al. 2014).

Another factor contributing to successful cartilage regeneration is the meniscal status. As the menisci are critical for shock absorption and load distribution in the knee joint, meniscal deficiency also affects cartilage regeneration (Makris et al. 2011). Meniscal lesions should be treated adequately in combination with cartilage regeneration. Because of the direct correlation between the lost amount of meniscus tissue and the increase of load on the surrounding cartilage, as much meniscus substance as possible should be restored (McDermott and Amis 2006) which can be achieved by suturing or limited partial resection. In case of previous subtotal meniscussed in order to restore a normal joint physiology for cartilage regeneration.

It has been clearly shown that joint instability contributes to a significant increase in cartilage lesions. In the long term, ACL insufficiency is a negative predictor for development of knee osteoarthritis. Therefore the correction of ligamentous instability by ligament reconstruction is a mandatory requirement for regenerative cartilage treatment.

Besides the alignment, meniscal integrity, and knee stability, the status of the subchondral bone is crucial for a successful cartilage repair. It has been shown that the adjacent bone quality affects the regeneration of the cartilage defect (Gomoll et al. 2010). So the state of the subchondral bone needs also to be taken into account for planning a regenerative cartilage repair procedure.

In conclusion, analysis, evaluation, and correction of all comorbidities and underlying pathologies are mandatory requirements for a good clinical outcome after cell-based cartilage regeneration procedures.

5.2 Regenerative Treatment Options for Cartilage Repair

5.2.1 Bone Marrow Stimulation Techniques

Due to their simplicity and low costs, bone marrow stimulation techniques are the most commonly used procedures among regenerative options for cartilage treatment worldwide. Developed by Steadman in the 1980s as an enhancement of tissue response techniques like drilling (Pridie) and abrasion arthroplasty (Johnson), this widely used procedure is generally regarded as safe and effective. Cartilage repair with the microfracture technique involves several systematic steps, including debridement to a stable cartilage margin creating a stable defect containment, careful removal of the calcified cartilage layer with special curettes or shavers, and homogeneous microfracture penetrations within the cartilage defect with specific awls perpendicular to the subchondral bone plate. This procedure results in a complete defect filling by a well-anchored mesenchymal clot (Mithoefer et al. 2009b) (Fig. 5.1).

The aim is to recruit bone marrow cells via creating a communication between cartilage lesions and subchondral bone to get access to potential cartilage precursor cells. Stem cells migrate from the marrow cavity to the fibrin clot of the defect and promote the formation of a fibrocartilaginous tissue (Marcacci et al. 2013). Arthroscopically performed microfracturing is a cost-saving procedure with a low complication rate and mainly successful for small defects. However the development of tissue hypertrophy or formation of soft scar tissue that lacks the mechanical characteristics of hyaline cartilage are disadvantages of this specific treatment (Fortier et al. 2012).

According to the recent literature, good indications for treatment of chondral lesions with microfracture are:

- Small defect size (<3 cm²)
- Full-thickness, traumatic cartilage defects (Outerbridge grades III and IV) with full containment
- Intact articulating joint surface
- Patient age between 18 and 50 years



Fig 5.1 (a) Schematic drawing of cartilage repair using the microfracture technique. After debridement in order to create stable defect edges (*upper left*), the calcified cartilage layer is removed (*upper right*), and microfracture penetrations are placed in the defect with a distance of 3–4 mm (*lower left*), resulting in defect filling with a mesenchymal clot (*lower right*) (adopted from Mithoefer et al. 2009b). (b) Arthroscopic view of a microfracture procedure for treatment of a full-size cartilage defect at the lateral femoral condyle after preparation of stable defect edges (containment)

- No or corrected concomitant morbidity (axis deviation, instability, meniscus injuries, subchondral bone pathology)
- · Full range of motion of the affected joint

An important factor for a successful outcome after microfracture treatment seems to be the postoperative rehabilitation protocol. Partial weight bearing with no more than 20 kg and the use of CPM is recommended for 6 weeks without limitation of the range of motion (Steadman et al. 2001).

In a systematic review, Mithoefer et al. (2009b) observed that microfracturing provides effective functional improvement for at least 2 years. In smaller defects, microfracture shows promising results concerning mobility, reduction of pain, and return to sport (Kon et al. 2011). Positive prognostic factors for a successful cartilage treatment with microfracture are size smaller than 3 cm², BMI less than 30 kg/m², femoral defects, age younger than 40, interval of pain less than 1 year, and no previous knee surgery. However, recent reports show that over time the results of microfracture are getting worse especially in active patients and larger chondral defects 5 years after the procedure. Additionally, the effects of microfracture are patient age related, meaning that older patients do not seem to profit at the same extent from this specific treatment as compared to young patients (Kreuz et al. 2006; Mithoefer et al. 2009a). The repair tissue response is often unpredictable; fibrous, soft, spongiform tissue combined with central degeneration is frequently found, and
patients may have to adjust their activity level to that of their knee function (Nehrer et al. 1999). Another reason for the deterioration of the clinical outcome after microfracture over time might be the development of subchondral sclerosis and cysts or the formation of intralesional osteophytes. Consecutively a complication rate of up to 50% after microfracturing is described in literature after 5 years. Recent results published in the literature recommend using these procedures only for the treatment of acute and small lesions and not in large cartilage defects anymore.

Further developments in the field of bone marrow stimulating techniques try to overcome the shortcomings of the procedures. In combination with the microfracture technique, coverage of the prepared and treated chondral defect site by a biomaterial is becoming more and more popular. This enhanced procedure was first described by Gille et al. (2010). The so-called autologous matrix-induced chondrogenesis (AMIC) reveals promising results in terms of functional outcome. In a prospective study, Gille et al. investigated 27 patients up to 62 months with a mean defect size of 4.2 cm². 87% of the patients showed an increase in functional outcome scores like ICRS, Tegner, Cincinnati et al. compared to the preoperative status. In another study, the same authors detected a significant decrease of pain in the VAS after 1 and 2 years postoperatively (Gille et al. 2013). Kusano et al. (2012) also detected significant improvements in functional scores and pain reduction after 29 months, but MRI findings showed generally incomplete or inhomogeneous tissue filling. Comparing AMIC with the original microfracturing technique, Anders et al. (2013) found no significant differences in the IKDC or Cincinnati score at 1- or 2-year follow-ups. A recent study has shown an improvement in repair tissue quality by enhancing microfracture with a chitosan-based biomaterial (BST-CarGel; Piramal, Laval, Quebec, Canada) (Stanish et al. 2013). Mixed with autologous blood, it stabilizes the clot and enhances marrow-derived repair in the microfractured cartilage lesion. Using this technique, Stanish et al. observed an equivalent clinical benefit compared to microfracturing alone, but a greater defect filling and superior repair tissue quality in MRI evaluation. Further studies and long-term results will show whether enhanced microfracture techniques are really capable to overcome the shortcomings of the original procedure regarding the development of intralesional osteophytes or formation of subchondral cysts. However, there is doubt whether these modifications make microfracture-based techniques more appropriate for treatment of large-sized chondral defects.

Recently the technique of microfracturing has been modified to a microdrilling method. The idea of drilling holes through the damaged cartilage area into the subchondral bone marrow space to stimulate repair tissue was first described by Pridie. Thermal necrosis was a potential disadvantage that could affect the outcome. The improved modern microdrilling version with arthroscopically applicable narrowcaliber drills up to 4 mm in depth is more reproducible and less traumatic. In an animal model, Chen et al. compared this "micro-Pridie" drilling method histologically with standard microfracturing. While microfracture caused compacted bone formation around the created holes that sealed them off from viable bone marrow, drilling cleanly removed the bone from the holes and provided access channels to marrow stroma. Heat necrosis was not seen in the drilling group (Chen et al. 2009). Furthermore Eldracher et al. demonstrated improved osteochondral repair by application of smaller drill holes that reflect the physiological trabecular distance in a translational sheep model. They conclude to use small-diameter bone-cutting devices for subchondral drilling (Eldracher et al. 2014). However, no prospective clinical trial has shown significant improvement of the microdrilling method over the original microfracture technique yet.

5.2.2 Autologous Osteochondral Transplantation

Focal (osteo-)chondral defects may also be addressed with osteochondral autograft transplantation (OAT). It is the only method to transfer native hyaline articular cartilage into the defect area. Harvesting and subsequent implantation of autologous osteochondral plugs is performed in a one-step procedure. The plugs are frequently taken via a small incision from a non-weight-bearing area such as the medial or lateral margin of the trochlea or the intercondylar notch. This procedure guarantees a tissue transfer of viable osteochondral units that aims to integrate via bone-tobone healing, since the mature cartilage tissue has limited healing potential and rarely fully heals and integrates with surrounding cartilage. The fast bone-to-bone integration allows a rehab program with a rapid increase in weight bearing. In the early 1990s, Hangody conceived and perfected the mosaicplasty technique, which uses multiple small-diameter osteochondral plugs that can be implanted also through an arthroscopic approach, and good results have been reported at long-term followup, particularly for defects up to 4 cm^2 (Hangody et al. 2010). Especially deep focal chondral defects which affect the subchondral plate or small cartilage lesions with pathologies of the subchondral bone like cysts may be responsive to a treatment with OAT. In controlled randomized prospective studies, Gudas et al. (2012) showed significantly better clinical results after 12, 24, and 36 months comparing OAT versus microfracture. Compared to other regenerative treatment options, OAT requires the shortest postoperative time of partial weight bearing during the rehabilitation period. Consecutively, time to return to sport is diminished. However, with increasing defect size, complication rate rises due to integration problems and donor site morbidity. Therefore, treatment of chondral defects larger than 3-4 cm² with OAT is no longer recommended in literature.

5.2.3 Autologous Chondrocyte Transplantation

The treatment of choice for large full-thickness articular cartilage defects is the matrix-guided autologous chondrocyte transplantation (MACT).

Brittberg et al. (1994) first introduced the technique of the autologous chondrocyte transplantation (ACT) in 1994. Particularly for treatment of cartilage defects larger than 3 cm², the ACT method revealed superior long-term success (Bentley et al. 2012). The conventional technique is accompanied with periosteum harvest and fixation over the cartilage defects via large skin incisions. Autologous chondrocytes were injected underneath the periosteal flap. Hypertrophy of the periosteum with high rate of revision arthroscopies and the risk of transplant failure of up to 20% were major drawbacks of the conventional autologous chondrocyte transplantation technique.

MACT was developed to address these problems. In a first arthroscopy, small osteochondral plugs are taken from the non-weight-bearing cartilage adjacent to the lateral femoral notch. Then the chondrocytes are isolated, cultured, and seeded on biodegradable scaffolds. Approximately 3 weeks after the first arthroscopy, the cell-seeded scaffolds are implanted into cartilage defects. Therefore, the lesion is pre-pared by removal of the calcified cartilage layer and creation of containment with stable rims of the defect. The cell-matrix construct is then fixed in the defect with sutures or biodegradable devices like plugs or anchors (Fig. 5.2).

Another technique uses self-adhering chondrospheres to fill the defect. These further developments of the ACT technique enable to minimize the incision and to perform the procedure in a "mini-open" way or arthroscopically. Consecutively the rehabilitation time was reduced and the complication rate diminished.

With the new MACT technique, also some other disadvantages of the ACT were eliminated (Harris et al. 2011). The rate of hypertrophy of the transplant that may be caused by the biologically active periosteal flap was reduced by the matrix-guided technique.

An advantage of the second and third generation of the autologous chondrocyte transplantation is the scaffold-based technique that also simplified the surgical procedure. The biomaterial represents a temporary 3D structure of biodegradable



Fig. 5.2 *Upper row:* arthroscopic evaluation of a full-size cartilage defect at the lateral femoral condyle and harvesting of osteochondral plugs from the medial edge of the lateral notch border for further culture of autologous chondrocytes. *Lower row:* 3 weeks after cell harvest, the defect at the lateral femoral condyle is prepared via mini-arthrotomy. After cutting the autologous chondrocyte transplant to the correct size, the cell-seeded implant is placed into the defect and fixed with sutures

polymers, which favors the growth of the specific cartilaginous cell type. An ideal scaffold should mimic the biological and structural properties of native cartilage in order to enable cell infiltration, attachment, proliferation, and differentiation. The matrix should be biocompatible and biodegradable in order to support initial tissue formation and then to be gradually replaced by the regenerated tissue. The different three-dimensional biomaterials support the redifferentiation process, cell protection in the initial phase, and a homogenous cell distribution in the defect.

Compared to other reconstructive therapy options for cartilage defects like microfracturing, MACT restores the cartilage defect up to date with the best quality of the regenerated tissue (Vavken and Samartzis 2010).

Especially for full-thickness cartilage defects larger than 4 cm^2 , MACT is the recommended therapy in literature (Niemeyer et al. 2016). Other cartilage therapy procedures failed to improve the clinical outcome of cartilage defects of that size.

In a controlled randomized prospective study for large-sized chondral defects (4–10 cm²), the outcome after MACT was significantly better after 2 years compared to microfracture (Basad et al. 2010). Similar long-term results were seen for active patients comparing MACT with microfracture. In another randomized prospective study, Crawford et al. saw significantly more therapy responder in the MACT group compared to the microfracture group after 6, 12, or 24 months. These results correlated to the clinical and functional outcome of the patients in the KOOS and IKDC scores (Crawford et al. 2012).

The reason for the superior results after MACT compared to microfracture might be the better defect filling, the histological results, and the lack of osteophytes in the defect site or the regenerated tissue, which can be predominantly detected 4 or 5 years after microfracture (Zellner et al. 2015).

However, if microfracture fails as primary procedure for treatment of a chondral defect, the risk of treatment failure after the secondary performed MACT increases significantly. For that reason some authors do not recommend microfracture as a first-line treatment especially for larger defects. On the other hand, there are reports in literature, which reveal good results of MACT even as a second-line therapy procedure. Additionally the age-related effects of a cartilage therapy seem to be less significant with MACT in comparison to microfracture.

In a controlled randomized prospective study, Bentley et al. (2012) showed significantly better outcome results after ACI compared to OAT. The best clinical results of MACT were observed for traumatic chondral lesions and for osteochondrosis dissecans. On the other hand, degenerative cartilage defects and chronic lesions are still difficult to treat, especially when patients with a long history of pain show a significantly worse outcome after MACT (Angele et al. 2015).

In a published study, Vanlauwe et al. compared ACI with microfracture and showed a significant improvement of patients' outcome after MACT when the symptoms of the cartilage lesion did not last more than 3 years. On the other hand, in patients with clinical symptoms more than 3 years, ACI failed to improve the functional outcome significantly compared to microfracture (Vanlauwe et al. 2011).

The earlier a biological cartilage repair is performed, the better are the clinical results. Consecutively primary cartilage defects should be treated as soon as possible to improve the long-term outcome.

Another problem for biological cartilage repair is the localization of the defect. Results of all treatment options behind the patella are worse than in other parts of the knee joint (Niemeyer et al. 2011). Also in the original description of the ACI technique, Brittberg et al. observed significant more treatment failures for defects in the patellofemoral compartment. Probably the special biomechanical situation in the retropatellar area is the reason for the higher rate of cartilage treatment failure. As this is not a problem of a specific cartilage repair procedure, the necessity arises to address all pathologies for a successful cartilage treatment behind the patella. Comparable to malalignment of the leg axis, knee instability, or meniscal tears in the femorotibial compartment, all pathologies like maltracking of the patella or dysplasia in the retropatellar area should be corrected.

As mentioned above the status of the subchondral bone is crucial for successful cartilage regeneration. For deep osteochondral defects like in osteochondritis dissecans, the MACT can be combined with bone augmentation like cancellous bone grafting or autologous bone transplantation, e.g., from the iliac crest. After reconstruction of the osseous part, the defect is covered by MACT.

Macroscopic and histological findings play an important role after MACT. For the evaluation of the quality of the regenerated tissue, not only histological findings but also the amount of defect filling, the surface quality, and the integration into the surrounding native cartilage are important (Nehrer et al. 1999).

It has been shown that complete defect filling with differentiated tissue correlates with good clinical results. On the other hand, incomplete defect filling with undifferentiated scar tissue reveals unsatisfying scoring results with ongoing pain and worse joint function of the patients (Henderson et al. 2007). This effect can be particularly seen in larger chondral defects. In a pilot study, we reported that the transplant quality is adequate at the time of surgery of MACT. We retrospectively reviewed 125 patients with large localized cartilage defects (mean defect size 5 cm²) of the knee who were treated with MACT. Portions of the cell-matrix constructs that were not implanted in the cartilage defects were further cultured and tested for their potential to form articular cartilage. In vitro assessment of the cell-matrix implants showed chondrogenic differentiation with positive staining for glycosaminoglycans and collagen II in all cultures. Enzyme-linked immunosorbent assay confirmed an increase of collagen II production. Clinically, we observed an improvement in median IKDC score from 41 to 67 points at the last follow-up indicating that cartilage extracellular matrix deposition shows adequate implant quality for MACT at the time of implantation and justifies the use for treatment of large cartilage defects (Zellner et al. 2013).

Besides regulatory restrictions and high costs, a disadvantage of today's autologous chondrocyte transplantation is the necessity of two steps for the surgical procedure. After cell harvest a certain time of cultivation and expansion of the chondrocytes is mandatory prior to the application. Consecutively, the patient needs two operations plus the phenotype and quality of the transplanted chondrocytes might be affected. Future directions are aiming for the development of one-step procedures.

Appropriate cell types might help to affect the complexity of ACI and simplify surgical procedures. Alternative cell sources are allogenic cells or mesenchymal stem cells. Allogenic chondrocytes can help to reduce donor site morbidity. In combination with a biocompatible and chondroinductive matrix, allogenic chondrocytes harvested from neonatal donors or from donor's knee joints within 24 h of death may be used in a single-stage procedure (Farr et al. 2014). Preliminary results demonstrated a safe and effective treatment for cartilage defects with a mean lesion size of 2.7 cm². Clinical outcomes showed significant improvement over baseline and favorable histological repair tissue 2 years postoperatively. Dhollander et al. reported of midterm results after implantation of alginate beads containing human mature allogenic chondrocytes in cartilage lesions of the knee. Twenty-one patients were followed for an average period of 6.3 years, and a significant improvement in WOMAC and VAS scores was observed. However, four failures occurred and MRI evaluation with the MOCART score only revealed moderate values (Dhollander et al. 2012).

Autologous adult mesenchymal stem cells (MSCs) are a potential cell source for a single-step cell-based treatment of large cartilage defects. MSCs have a better proliferation rate than chondrocytes and a high potential for differentiation into several lineages including chondrogenesis. Autologous MSCs can derive from many sources. Particularly, bone marrow-derived MSCs (BMSCs) combine many advantages as they are easy to isolate and to store. Extensive preclinical and clinical work has shown that BMSCs can differentiate into cartilage among other tissues. Other potential sources for MSCs are adipose tissue, muscle, synovium, periosteum, and umbilical cord. Nejadnik et al. (2010) analyzed the clinical outcome of patients treated with autologous BMSCs compared to patients treated with first-generation ACI for large cartilage defects in the knee. After 2 years a similar functional outcome regarding IKDC, Lysholm, or Tegner scores was found. The authors concluded that using BMSCs for articular cartilage repair is as effective as chondrocytes. In addition, it required one less knee surgery, reduced costs, and minimized donor site morbidity. However, in some countries, regulatory burdens might be a problem for implementing the use of autologous mesenchymal stem cells into daily clinical practice.

5.2.4 Current Treatment Recommendations for Chondral Injuries

In their current review, Niemeyer et al. (2016) provide a concise overview on important scientific background issues and the results of clinical studies discussing advantages and disadvantages of ACI and other cartilage treatment options. They describe the biology and function of healthy articular cartilage, the present state of knowledge concerning potential consequences of primary cartilage lesions, and the suitable indication for ACI. Based on current evidence, an indication for ACI is given for symptomatic cartilage defects starting from defect sizes of more than 3–4 cm² in the case of young and active sports patients at 2.5 cm². Smaller lesions are supposed to be treated by bone marrow stimulating techniques like microfracturing. However, the status of the subchondral bone will influence the decision-making process for cartilage therapy. Smaller defects with pathologies of the whole osteochondral unit are best treated with OAT. For large and deep osteochondral lesions, a combination of MACT and bone augmentation techniques is the favorable treatment option.

Conclusions

With increasing knowledge cell-based cartilage regeneration becomes a more and more routinely used technique with well-predictable outcome and results. As research activities are increasing in the field of regenerative joint therapy, recent developments help to overcome remaining limitations step by step. Simplification of regulatory burdens is needed to transfer rising knowledge and developments into daily clinical practice.

References

- Anders S, Volz M, Frick H, Gellissen J (2013) A randomized, controlled trial comparing autologous matrix-induced chondrogenesis (AMIC(R)) to microfracture: analysis of 1- and 2-year follow-up data of 2 centers. Open Orthop J 7:133–143. doi:10.2174/18743250013070 10133
- Angele P, Fritz J, Albrecht D, Koh J, Zellner J (2015) Defect type, localization and marker gene expression determines early adverse events of matrix-associated autologous chondrocyte implantation. Injury 46(Suppl 4):S2–S9. doi:10.1016/S0020-1383(15)30012-7
- Basad E, Ishaque B, Bachmann G, Sturz H, Steinmeyer J (2010) Matrix-induced autologous chondrocyte implantation versus microfracture in the treatment of cartilage defects of the knee: a 2-year randomised study. Knee Surg Sports Traumatol Arthrosc 18(4):519–527. doi:10.1007/ s00167-009-1028-1
- Behery O, Siston RA, Harris JD, Flanigan DC (2014) Treatment of cartilage defects of the knee: expanding on the existing algorithm. Clin J Sport Med 24(1):21–30. doi:10.1097/ JSM.000000000000004
- Bentley G, Biant LC, Vijayan S, Macmull S, Skinner JA, Carrington RW (2012) Minimum tenyear results of a prospective randomised study of autologous chondrocyte implantation versus mosaicplasty for symptomatic articular cartilage lesions of the knee. J Bone Joint Surg 94(4):504–509. doi:10.1302/0301-620X.94B4.27495
- Brittberg M, Lindahl A, Nilsson A, Ohlsson C, Isaksson O, Peterson L (1994) Treatment of deep cartilage defects in the knee with autologous chondrocyte transplantation. N Engl J Med 331(14):889–895. doi:10.1056/NEJM199410063311401
- Chen H, Sun J, Hoemann CD, Lascau-Coman V, Ouyang W, McKee MD, Shive MS, Buschmann MD (2009) Drilling and microfracture lead to different bone structure and necrosis during bone-marrow stimulation for cartilage repair. J Orthop Res 27(11):1432–1438. doi:10.1002/ jor.20905
- Crawford DC, DeBerardino TM, Williams RJ 3rd (2012) NeoCart, an autologous cartilage tissue implant, compared with microfracture for treatment of distal femoral cartilage lesions: an FDA phase-II prospective, randomized clinical trial after two years. J Bone Joint Surg Am 94(11):979–989. doi:10.2106/JBJS.K.00533

- Dhollander AA, Verdonk PC, Lambrecht S, Verdonk R, Elewaut D, Verbruggen G, Almqvist KF (2012) Midterm results of the treatment of cartilage defects in the knee using alginate beads containing human mature allogenic chondrocytes. Am J Sports Med 40(1):75–82. doi:10.1177/0363546511423013
- Eldracher M, Orth P, Cucchiarini M, Pape D, Madry H (2014) Small subchondral drill holes improve marrow stimulation of articular cartilage defects. Am J Sports Med 42(11):2741–2750. doi:10.1177/0363546514547029
- Farr J, Tabet SK, Margerrison E, Cole BJ (2014) Clinical, radiographic, and histological outcomes after cartilage repair with particulated juvenile articular cartilage: a 2-year prospective study. Am J Sports Med 42(6):1417–1425. doi:10.1177/0363546514528671
- Fortier LA, Cole BJ, McIlwraith CW (2012) Science and animal models of marrow stimulation for cartilage repair. J Knee Surg 25(1):3–8
- Gille J, Schuseil E, Wimmer J, Gellissen J, Schulz AP, Behrens P (2010) Mid-term results of autologous matrix-induced chondrogenesis for treatment of focal cartilage defects in the knee. Knee Surg Sports Traumatol Arthrosc 18(11):1456–1464. doi:10.1007/s00167-010-1042-3
- Gille J, Behrens P, Volpi P, de Girolamo L, Reiss E, Zoch W, Anders S (2013) Outcome of autologous matrix induced chondrogenesis (AMIC) in cartilage knee surgery: data of the AMIC Registry. Arch Orthop Trauma Surg 133(1):87–93. doi:10.1007/s00402-012-1621-5
- Gomoll AH, Madry H, Knutsen G, van Dijk N, Seil R, Brittberg M, Kon E (2010) The subchondral bone in articular cartilage repair: current problems in the surgical management. Knee Surg Sports Traumatol Arthrosc 18(4):434–447. doi:10.1007/s00167-010-1072-x
- Gudas R, Gudaite A, Pocius A, Gudiene A, Cekanauskas E, Monastyreckiene E, Basevicius A (2012) Ten-year follow-up of a prospective, randomized clinical study of mosaic osteochondral autologous transplantation versus microfracture for the treatment of osteochondral defects in the knee joint of athletes. Am J Sports Med 40(11):2499–2508. doi:10.1177/0363546512458763
- Hangody L, Dobos J, Balo E, Panics G, Hangody LR, Berkes I (2010) Clinical experiences with autologous osteochondral mosaicplasty in an athletic population: a 17-year prospective multicenter study. Am J Sports Med 38(6):1125–1133. doi:10.1177/0363546509360405
- Harris JD, Siston RA, Brophy RH, Lattermann C, Carey JL, Flanigan DC (2011) Failures, reoperations, and complications after autologous chondrocyte implantation--a systematic review. Osteoarthritis Cartilage 19(7):779–791
- Henderson I, Lavigne P, Valenzuela H, Oakes B (2007) Autologous chondrocyte implantation: superior biologic properties of hyaline cartilage repairs. Clin Orthop Relat Res 455:253–261. doi:10.1097/01.blo.0000238829.42563.56
- Kon E, Filardo G, Berruto M, Benazzo F, Zanon G, Della Villa S, Marcacci M (2011) Articular cartilage treatment in high-level male soccer players: a prospective comparative study of arthroscopic second-generation autologous chondrocyte implantation versus microfracture. Am J Sports Med 39(12):2549–2557. doi:10.1177/0363546511420688
- Kreuz PC, Erggelet C, Steinwachs MR, Krause SJ, Lahm A, Niemeyer P, Ghanem N, Uhl M, Sudkamp N (2006) Is microfracture of chondral defects in the knee associated with different results in patients aged 40 years or younger? Arthroscopy 22(11):1180–1186. doi:10.1016/j. arthro.2006.06.020
- Kusano T, Jakob RP, Gautier E, Magnussen RA, Hoogewoud H, Jacobi M (2012) Treatment of isolated chondral and osteochondral defects in the knee by autologous matrix-induced chondrogenesis (AMIC). Knee Surg Sports Traumatol Arthrosc 20(10):2109–2115. doi:10.1007/ s00167-011-1840-2

Lattermann C, Luckett MR (2011) Staging and comorbidities. J Knee Surg 24(4):217-224

- Makris EA, Hadidi P, Athanasiou KA (2011) The knee meniscus: structure-function, pathophysiology, current repair techniques, and prospects for regeneration. Biomaterials 32(30):7411– 7431. doi:10.1016/j.biomaterials.2011.06.037
- Marcacci M, Filardo G, Kon E (2013) Treatment of cartilage lesions: what works and why? Injury 44(Suppl 1):S11–S15. doi:10.1016/S0020-1383(13)70004-4
- McDermott ID, Amis AA (2006) The consequences of meniscectomy. J Bone Joint Surg 88(12):1549–1556. doi:10.1302/0301-620X.88B12.18140

- Mithoefer K, Hambly K, Della Villa S, Silvers H, Mandelbaum BR (2009a) Return to sports participation after articular cartilage repair in the knee: scientific evidence. Am J Sports Med 37(Suppl 1):167S–176S. doi:10.1177/0363546509351650
- Mithoefer K, McAdams T, Williams RJ, Kreuz PC, Mandelbaum BR (2009b) Clinical efficacy of the microfracture technique for articular cartilage repair in the knee: an evidence-based systematic analysis. Am J Sports Med 37(10):2053–2063. doi:10.1177/0363546508328414
- Nehrer S, Spector M, Minas T (1999) Histologic analysis of tissue after failed cartilage repair procedures. Clin Orthop Relat Res 365:149–162
- Nejadnik H, Hui JH, Feng Choong EP, Tai BC, Lee EH (2010) Autologous bone marrow-derived mesenchymal stem cells versus autologous chondrocyte implantation: an observational cohort study. Am J Sports Med 38(6):1110–1116. doi:10.1177/0363546509359067
- Niemeyer P, Koestler W, Sudkamp NP (2011) Problems and complications of surgical techniques for treatment of full-thickness cartilage defects. Z Orthop Unfall 149(1):45–51. doi:10.105 5/s-0030-1250104
- Niemeyer P, Albrecht D, Andereya S, Angele P, Ateschrang A, Aurich M, Baumann M, Bosch U, Erggelet C, Fickert S, Gebhard H, Gelse K, Gunther D, Hoburg A, Kasten P, Kolombe T, Madry H, Marlovits S, Meenen NM, Muller PE, Noth U, Petersen JP, Pietschmann M, Richter W, Rolauffs B, Rhunau K, Schewe B, Steinert A, Steinwachs MR, Welsch GH, Zinser W, Fritz J (2016) Autologous chondrocyte implantation (ACI) for cartilage defects of the knee: a guide-line by the working group "Clinical Tissue Regeneration" of the German Society of Orthopaedics and Trauma (DGOU). Knee 23(3):426–435. doi:10.1016/j.knee.2016.02.001
- Stanish WD, McCormack R, Forriol F, Mohtadi N, Pelet S, Desnoyers J, Restrepo A, Shive MS (2013) Novel scaffold-based BST-CarGel treatment results in superior cartilage repair compared with microfracture in a randomized controlled trial. J Bone Joint Surg Am 95(18):1640– 1650. doi:10.2106/JBJS.L.01345
- Steadman JR, Rodkey WG, Rodrigo JJ (2001) Microfracture: surgical technique and rehabilitation to treat chondral defects. Clin Orthop Relat Res (391 Suppl):S362–S369
- Vanlauwe J, Saris DB, Victor J, Almqvist KF, Bellemans J, Luyten FP, TIG/ACT/01/2000&EXT Study Group (2011) Five-year outcome of characterized chondrocyte implantation versus microfracture for symptomatic cartilage defects of the knee: early treatment matters. Am J Sports Med 39(12):2566–2574. doi:10.1177/0363546511422220
- Vavken P, Samartzis D (2010) Effectiveness of autologous chondrocyte implantation in cartilage repair of the knee: a systematic review of controlled trials. Osteoarthritis Cartilage 18(6):857– 863. doi:10.1016/j.joca.2010.03.005
- Widuchowski W, Widuchowski J, Trzaska T (2007) Articular cartilage defects: study of 25,124 knee arthroscopies. Knee 14(3):177–182. doi:10.1016/j.knee.2007.02.001
- Zellner J, Angele P, Zeman F, Kujat R, Nerlich M (2013) Is the transplant quality at the time of surgery adequate for matrix-guided autologous cartilage transplantation? A pilot study. Clin Orthop Relat Res 471(9):2852–2861. doi:10.1007/s11999-013-2958-y
- Zellner J, Krutsch W, Pfeifer CG, Koch M, Nerlich M, Angele P (2015) Autologous chondrocyte implantation for cartilage repair: current perspectives. Orthop Res Rev 7:149–158

Growth Factors in Cartilage Repair

6

Maximilian A. Meyer, Atsushi Urita, Brian J. Cole, and Susanna Chubinskaya

Abstract

The articular cartilage does not heal completely after injury, predisposing patients to accelerated progression of degenerative joint disease. While surgical intervention can address chondral defects and yield positive functional outcomes, substantial research has gone into the use of growth factors to augment cartilage repair and preclude or postpone the need for operative management. This chapter describes the growth factors with the most promising in vitro and in vivo data in cartilage repair, namely, bone morphogenetic protein-7, transforming growth factor- β , fibroblast growth factor-18, connective tissue growth factor, insulin-like growth factor-1, and recent advancements with autologous solutions of growth factors, such as platelet-rich plasma. Each section provides a background on mechanism of action, summarizes pivotal basic science research, and describes the results of clinical application in animal and human models of chondral disease.

6.1 Introduction

The optimization of cartilage repair both in the setting of acute, post-traumatic chondral injury and in halting the progression of chronic degenerative disease remains a challenge to clinicians and researchers alike. Cartilage tissue is

M.A. Meyer (🖂) • A. Urita • B.J. Cole • S. Chubinskaya

Midwest Orthopaedics at Rush, Rush University Medical Center, Chicago, IL 60612, USA e-mail: maxameyer@gmail.com; atsushi.urita@rushortho.com; brian.cole@rushortho.com; susanna_chubinskaya@rush.edu

[©] Springer International Publishing AG 2017

S. Grässel, A. Aszodi (eds.), Cartilage, DOI 10.1007/978-3-319-53316-2_6

avascular, aneural, and alymphatic and receives all nutrients from the synovium via diffusion. The articular cartilage therefore does not heal completely after traumatic injury. Instead, spontaneous repair of cartilage produces tissue that fails to integrate with surrounding native cartilage and is inferior in both structure and function. The result is a nonuniform articular joint lining with greater susceptibility to inflammation, further injury, and progression of osteoarthritis (OA). Epidemiological consequences are profound, as knee OA affects between 19-28% of Americans over age 45 (Felson et al. 1987; Jordan et al. 2007). Of this number, 13–18% of patients had an identifiable acute trauma to the joint (Kern 1988), and data shows that post-traumatic arthritis can develop within 10 years of the initial insult (Roos et al. 1995). While arthroplasty addresses endstage disease, there remains a window for intervention with biologic agents to prevent progression of OA after initial chondral injury. Ideally, such intervention must recreate a cartilage that is thin and compression resistant, has a low coefficient of friction, distributes load, and lasts for many years. To date, the use of recombinant growth factors has vielded appreciable cartilage repair both in vitro and in vivo and holds significant clinical promise for patients suffering from chondral injury.

Growth factors are circulating, biologically active polypeptides that stimulate and promote chondrocyte growth and differentiation (Goldring et al. 2006). They drive chondrocyte synthesis of extracellular matrix components, such as proteoglycans, aggrecan, and type II collagen. Aside from anabolic effects, many growth factors also have anti-catabolic effects on cartilage tissue by decreasing expression of local cytokines such as interleukin-1 (IL-1) and matrix metalloproteinases (MMPs) (Fortier et al. 2011; Pascual-Garrido and Chubinskaya 2015; Chubinskaya et al. 2011). Growth factors do not function in a vacuum, but rather have complex signaling interactions with other mediators. As described by Giannoudis, an accurate model for cartilage repair also takes into account chondrocyte interactions with the synovial environment and host bone, all of which are affected by mechanical forces and circulating cytokines (Fig. 6.1) (Giannoudis et al. 2008). For example, the presence of inflammatory cytokines (IL-1) and oxidative stress has recently been shown to decrease chondrocyte responsiveness to growth factor augmentation (Elshaier et al. 2009; Loeser et al. 2014). Thus, maximizing the chondroprotective capacity of growth factors necessitates understanding not only the expression of growth factors in healthy and in diseased cartilage but also their complex interactions with the surrounding synovial environment. The goal of this chapter is to familiarize the reader with growth factors showing the greatest clinical promise in cartilage repair and to provide an update on evidence supporting their use in chondral disease.



Fig. 6.1 Evaluation and treatment of articular cartilage defects have shifted toward treating the entire joint as an organ. In order to maintain integrity over time and to heal after minor damage, articular relies on interactions with several different local and systemic factors

6.2 Bone Morphogenetic Proteins

Of all growth factors, bone morphogenetic proteins (BMPs) have the most robust evidence from both in vitro and in vivo studies supporting their use for cartilage regeneration and repair. BMPs are members of the transforming growth factor- β (TGF- β) superfamily that have wide-ranging biological activities, including the regulation of cellular proliferation, apoptosis, differentiation and migration, embry-onic development, and the maintenance of tissue homeostasis during adult life (Goumans and Mummery 2000; Massague and Chen 2000; Itoh et al. 2000). In chondrocytes, BMPs stimulate cartilage synthesis and decrease activity of catabolic cytokines, such as IL-1, IL-6, IL-8, MMP-1, and MMP-13 (Badlani et al. 2009; Elshaier et al. 2009; Chubinskaya et al. 2007b, 2008; Im et al. 2003). The ability of BMPs to induce an anabolic response in cartilage in vitro has been documented using different BMPs in multiple species, including human, bovine, rat, rabbit, and mouse, and a variety of culture conditions.

Though the BMP family has several growth factors, most data to date explores the use of BMP-7, also known as osteogenic protein 1 (OP-1). In vitro augmentation of both human and animal chondrocytes with BMP-7 has yielded increased production of cartilage-specific extracellular proteins, such as collagens type II and VI, aggrecan, decorin, fibronectin, and hyaluronan [HA] via upregulation of enzymes such as hyaluronan synthase (Nishida et al. 2000b; Chubinskaya et al. 2007a). When applied to other cell types in the knee, BMP-7 has been shown to increase ECM synthesis in synovial and bone marrow-derived MSCs, both alone and in combination with TGF- β (Miyamoto et al. 2007; Shen et al. 2010). This profound anabolic response stems from BMP-7 regulatory properties as a modulator of other growth factors, such as insulin-like growth factor-1 and fibroblast growth factor, as well as their receptors, kinases involved in signaling, inhibitory binding proteins, and downstream transcription factors (Chubinskaya et al. 2011). Furthermore, BMP has been shown to restore tissue responsiveness to IGF-1 (Chubinskaya et al. 2007a). BMP-7 also downregulates catabolic mediators (IL-1, IL-6, IL-8, IL-11, and tumor necrosis factor $[TNF]-\alpha$) and inhibits both baseline and cytokine-induced expression of MMP-1 and MMP-13 (Im et al. 2003). Lastly, it modulates expression of receptors for certain matrix components, such as CD44 (Nishida et al. 2000b), and the synthesis of chondrocyte cytoskeleton proteins, such as talin, paxillin, and focal adhesion kinase (Vinall et al. 2002), bolstering the cartilage scaffold and strengthening newly formed tissue. While several growth factors have shown decreased efficacy with aged or diseased chondrocytes, BMP-7 induces an anabolic response across a variety of age groups and different stages of OA (Chubinskaya et al. 2007b). Despite its anabolic capacity, BMP-7 has not been shown to induce chondrocyte hypertrophy or other changes in chondrocytic phenotype, nor have BMP-7-treated animal knees displayed any histological evidence of uncontrolled fibroblast proliferation or radiographically detectable osteophyte formation (Fortier et al. 2011).

Animal studies have provided substantial evidence for the use of BMP-7 in vivo. A study of New Zealand white rabbits with femoral condyle defects showed extensive regeneration of both subchondral bone and a hyaline-like cartilage layer when treated with BMP-7 compared to fibrocartilage tissue fill-in defects left empty or treated with collagen only (Grgic et al. 1997). In a sheep study, continuous presence of BMP-7 led to markedly improved gross and microscopic cartilage healing of focal condylar and trochlear defects, suggesting the growth factor attracted mesenchymal-like cells originating from the synovium into the defect area (Jelic et al. 2001). In addition to applications of recombinant BMPs, rabbit studies evaluating the local application of a BMP gene to periosteal-derived allogenic mesenchymal stem cells via a retroviral vector showed complete or near-complete bone and cartilage regeneration of oACL tears, BMP-7 injections promoted significantly improved tissue healing and prevented progression of OA compared to placebo injection (Hayashi et al. 2008). BMP-7 has also been tested in conjunction with

existing cartilage restoration surgery, improving gross and histological outcomes with mosaicplasty (osteochondral autograft) (Shimmin et al. 2003) and with microfracture (Kuo et al. 2006). Though BMP-7 appears effective in several different treatment models, the timing of BMP-7 application after joint injury affects its efficacy. In a model of post-traumatic osteochondral defects, Hurtig and colleagues showed increased cartilage healing and chondroprotective effects of BMP-7 augmentation both immediately and at 3 weeks after injury, but this effect was diminished at 12 weeks after injury (Fig. 6.2) (Hurtig et al. 2009). This finding suggests that a window of opportunity exists shortly after cartilage injury where growth factor intervention used as a disease-modifying agent may prevent progression of post-traumatic OA. At later time points post-injury, the natural history of disease may become more challenging to overcome with growth factor therapy alone.

Human trials of intra-articular BMP-7 injection have been underway over the past half-decade. In 2010, Hunter and colleagues published results of a phase I randomized controlled trial of various dosing regiments of BMP-7 injections in patients over 40 years old with symptomatic knee OA (Hunter et al. 2010). Despite higher rates of injection at the site of pain, there were no differences in toxicity or adverse events between BMP-7 and placebo. Furthermore, patients receiving BMP-7 injections at midrange dose reported a symptomatic improvement and anti-pain effects, though this was not the primary objective of the study. Unfortunately, a phase II clinical OA study did not appear to be successful; however, the reasons for such outcome have yet to be determined.



Adapted form Hurtig et al, 2009, JOR

Fig. 6.2 Timing of the effect of BMP-7 after impact injury in sheep. Growth factor application either immediately (**a**) or within 1 month (**b**) of traumatic chondral injury yields excellent cartilage repair on safranin O histology. Delayed growth factor application after 4 months (**c**) yields suboptimal cartilage repair, suggesting that there is a window for growth factor augmentation of cartilage repair shortly after chondral injury

6.3 Transforming Growth Factor-β

TGF- β is a cytokine secreted by many cell types; it plays crucial roles in cell proliferation, differentiation, development, apoptosis, tissue homeostasis, and the immune system. Signaling occurs primarily through the SMAD pathway, which involves a heterotetrameric complex that acts as a transcription factor via phosphorylation cascades. Among its three common isoforms, TGF-B1 has been mostly studied in chondrogenesis. TGF- β 1 has been shown to stimulate chondrocyte synthetic activity and to decrease the catabolic activity of interleukin (IL)-1 and tumor necrosis factor (TNF)- α (Lotz et al. 1995; Lires-Dean et al. 2008). TGF- β 1 helps to maintain chondrocyte characteristics during in vitro culture by promoting cell proliferation and extracellular matrix (ECM) protein synthesis and through inhibition of MMPs to protect normal morphology (Blaney Davidson et al. 2007). Furthermore, TGF-B1 stimulates chondrogenesis of synovial lining cells and of bone marrow-derived mesenchymal stem cells (Fan et al. 2010; Kurth et al. 2007). Studies have demonstrated significant enhancement of cartilage repair with TGF-B1 application in scaffolds applied to defects (Abe et al. 2003; Diao et al. 2009) and in human mesenchymal stem cells transfected with TGF- β 1 genes via an adenovirus (Oi et al. 2013; Lee et al. 2001).

As with BMP-7, cartilage regeneration efforts utilizing TGF- β have moved forward into human trials over the past few years. Springing from research by Noh and colleagues on the ability of genetically engineered chondrocytes virally transduced with TGF- β 1 (GEC-TGF- β 1) to repair articular cartilage defects in animals, TissueGene-C has emerged as a leading growth factor in human trials for knee OA (Noh et al. 2010). The safety and biologic activity of injectable GEC-TGF-\u00df1 were evaluated in a 2012 phase I trial by Ha and colleagues in patients with advanced knee OA (Ha et al. 2012). There were no severe adverse effects related to the GEC-TGF- β 1 treatment reported, and the most common adverse effect was effusion. Ten of twelve patients showed improvements in clinical scores at 6 months and improvements in range of motion and pain up to 1 year. Since then, two phase II trials have investigated the efficacy and outcomes of intra-articular injectable GEC-TGF-B1 for knee OA. Cherian and colleagues recently conducted a prospective, multicenter, double-blinded, placebo-controlled, randomized study of GEC-TGF-β1 in the knees of patients with grade 3 OA and showed improved responses on the International Knee Documentation Committee (IKDC) score and Visual Analogue Scale (VAS) at 1-year follow-up and decreased need for analgesics (Cherian et al. 2015). Similarly, Lee and colleagues recently published results of another phase II trial of GEC-TGF-β1 injections into the knees of patients with OA yielding improved IKDC and VAS pain scores at 6 months follow-up (Lee et al. 2015). Given such positive results this year, GEC-TGF-β1 appears to hold significant promise as an injection therapy for moderate knee OA. Further, longer follow-up and phase III trials are needed to better define appropriate indications and dosing regimens for patients with chondral disease.

6.4 Fibroblast Growth Factors

The fibroblast growth factor (FGF) family plays important roles in human embryonic development, cell growth, morphogenesis, tissue repair, tumor growth, and invasion. FGFs are heparin-binding proteins and interact with heparan sulfate proteoglycans on the cell surface for signal transduction (Friedl et al. 1997). In total, 22 members of the FGF family have been identified in humans. The FGF receptor family has four members: FGFR1, FGFR2, FGFR3, and FGFR4. Human articular chondrocytes express all four receptors but contain significantly higher concentrations of FGFR1 and FGFR3 compared with FGFR2 and FGFR4 (Yan et al. 2011). FGFs are important regulators of cartilage development and homeostasis (Ellman et al. 2008). Three members of FGF family, FGF-2, FGF-8, and FGF-18, have been investigated for their role in cartilage homeostasis.

The role of FGF-2 in the production of the ECM in cartilage is controversial. FGF-2 was initially suggested to stimulate a robust cartilage repair response (Henson et al. 2005). However, strong mitogenic effects of FGF-2 lead to clustering of chondrocytes and uneven production of ECM due to low levels of collagen type II (Ellman et al. 2008). Furthermore, FGF-2 has been shown to suppress aggrecan and type II collagen and to promote the expression of aggrecanase and TNF- α receptors (Ellman et al. 2008; Im et al. 2008). FGF-2 also inhibits the stimulatory effect of bone morphogenetic protein-7 (BMP-7) and insulin-like growth factor-1 (IGF-1) on chondrocyte proteoglycan synthesis (Loeser et al. 2003, 2005). Ultimately, these findings have precluded FGF-2 from further consideration as a biological treatment in OA (Ellman et al. 2013).

While FGF-2 is mostly known for its pro-catabolic responses in cartilage, FGF-18 has repeatedly been shown to exert strong anabolic effects in chondrocytes and chondroprogenitor cells, leading to enhanced chondrogenic cell differentiation and type II collagen production (Ellsworth et al. 2002; Moore et al. 2005; Bradley et al. 2015). FGF-18 signaling through FGFR3 promotes chondrocyte proliferation at early embryonic stages. When development is complete, signaling through the same receptor works to suppress chondrocyte proliferation and prevent hypertrophic differentiation (Liu et al. 2002, 2007). The ability of FGF-18 to promote chondrocyte proliferation and the production of type II collagen and proteoglycan has been shown in vitro using adenovirus-mediated transfer of FGF-18 into the pinnae of nude mice (Ellsworth et al. 2002). Similarly, Barr et al. demonstrated an anabolic in vitro effect of recombinant human FGF-18 (rhFGF-18) on damaged cartilage, as rhFGF-18 increased aggrecan synthesis and reduced collagen breakdown in response to damage (Barr et al. 2014). In vivo studies began in rats, with Moore and colleagues investigating the utility of FGF-18 injections in a rat meniscal tear model of OA (Moore et al. 2005). Intra-articular injection of FGF-18 provided substantial cartilage production and reduced cartilage degeneration in OA. In an ovine model of chondral defects, augmentation of microfracture surgery with intra-articular injection of rhFGF-18 improved quality and quantity of repair tissue in chondral defects (Power et al. 2014).



Fig. 6.3 A chronological timeline of advances in research of two growth factors with high potential in cartilage repair

Over the last decade, FGF-18 injection trials have commenced in humans (Fig. 6.3). Two phase II trials investigating the use of FGF-18 in patients with focal chondral defects were closed due to low enrollment. However, a randomized, double-blind, placebo-controlled, proof-of-concept trial recently investigated the efficacy and safety of recombinant human FGF-18 (sprifermin) in the treatment of symptomatic knee OA (Lohmander et al. 2014). There were no reports of local or systemic safety problems associated with any dose of sprifermin, and systemic levels of sprifermin were below detectable levels. Sprifermin was associated with statistically significant, dose-dependent reductions in loss of both total and lateral femorotibial cartilage thickness and volume on quantitative MRI, as well as reductions in radiographic joint space narrowing in the lateral femorotibial compartment. There was no significant relationship between treatment group and reduction in central medial femorotibial compartment cartilage thickness as measured by quantitative MRI. However, while all groups had improved WOMAC pain scores, patients receiving the 100 µg dose of sprifermin had significantly less improvement at 12 months compared with those receiving placebo. Despite this, positive imaging results and lack of toxicity in this phase I study suggest that FGF-18 may be a promising therapy for OA.

6.5 Connective Tissue Growth Factor

Connective tissue growth factor (CTGF, also known as CCN2) is an extracellular protein of the CCN family. Together, this group comprises extracellular matrix-associated heparin-binding proteins that play an important role in the regulation of

cellular proliferation, migration, adhesion, survival, differentiation, and synthesis of extracellular matrix proteins. Among these members, CTGF has been studied widely in chondrogenesis. CTGF plays a crucial role in the development of skeletal tissues, namely, by driving condensation of mesenchymal cells and regulating chondrocyte proliferation and differentiation (Nishida et al. 2000a; Song et al. 2007). CTGF promotes proliferation and cartilage matrix formation in the growth plate of the cartilage (Nakanishi et al. 2000) but has not been shown to cause hypertrophy or calcification in articular cartilage chondrocytes (Nishida et al. 2002).

In vivo studies exploring the use of CTGF in animal models of OA have emerged over the past decade. Such studies have further illustrated its role in promoting chondrocyte proliferation and differentiation during chondrogenesis, as CTGF knockout mice exhibited an expanded growth plate hypertrophic zone and impaired endochondral ossification (Ivkovic et al. 2003; Kawaki et al. 2008). Conversely, Tomita and colleagues demonstrated that overexpression of CTGF accelerated the process of endochondral ossification by promoting the proliferation and differentiation of growth plate chondrocytes (Tomita et al. 2013). Furthermore, cartilagespecific overexpression of CTGF in transgenic mouse studies slowed progression of age-related osteoarthritic changes in articular cartilage (Itoh et al. 2013). When applied to a full-thickness cartilage defect model, Nishida and colleagues showed that local administration of recombinant CTGF with gelatin hydrogel stimulated cartilage repair in a rat model (Nishida et al. 2004). Similarly, bone marrow mesenchymal stem cells transfected with CTGF recently provided hyaline-like cartilage regeneration similar to normal cartilage in a rabbit model of focal articular cartilage defects (Zhu et al. 2014). Such studies suggest a critical role for CTGF in both the protection and regeneration of articular cartilage. Though further testing is needed to clarify the safety and efficacy of this growth factor in cartilage disease, CTGF remains a promising therapeutic target for cartilage regeneration.

6.6 Insulin-Like Growth Factor-1

Insulin-like growth factor-1 (IGF-1) is a growth factor that is essential in embryonic tissue development and for growth and maintenance of mass throughout all stages of human life. Similar to BMP-7, IGF-1 has been shown to have robust anabolic and anti-catabolic effects. IGF-1 is required to maintain integrity of healthy articular cartilage, as rats with IGF-1 deficiency develop greater articular osteoarthritic cartilage degeneration in the knee than controls (Ekenstedt et al. 2006). The role of IGF-1 in healing of cartilage defects has best been demonstrated in a horse model, both by supplementing chondrocytes with IGF-1 (Fortier et al. 2002) and by genetically modifying chondrocytes with an IGF-1-encoded adenovirus (Goodrich et al. 2007). Unlike BMP-7, however, human chondrocytes have demonstrated a decreased response to IGF-1 with increasing age (Loeser et al. 2000, 2002) and with advanced OA (Schalkwijk et al. 1989). IGF-1 also appears to be more affective as a driver of synthetic function and less effective at preventing catabolism in advanced disease (Morales 2008). As such, there have not been any trials to date in humans with IGF-1 for the treatment of OA or focal chondral defects.

6.7 Platelet-Rich Plasma

While most studies to date have focused on individual growth factors, the use of combined autologous solutions of multiple growth factors, namely, in the form of platelet-rich plasma (PRP), has gained enormous popularity in the past decade. PRP is defined as plasma with a minimum twofold increase in platelet concentration above baseline levels or greater than 1.1×10^6 platelets/µL (Miller et al. 2007). Its use stems from the well-defined role of platelets in wound healing, initially through clot formation with fibrin to close wounds and ending with the release of factors involved in angiogenesis, inflammation, and the immune response (Nurden et al. 2008). Densely packed α -granules in platelets release a multitude of growth factors, including FGF, TGF-B, platelet-derived growth factor (PDGF), and vascular endothelial growth factor (VEGF), that modulate inflammation, attract local stem cells and fibroblasts, and trigger a regulated proliferative healing response (Smyth et al. 2009). In vitro studies of PRP on cartilage have yielded extremely positive results, as stem cells buffered in PRP had increased proliferation and chondrogenic differentiation (Mishra et al. 2009). Akeda and colleagues showed that porcine chondrocytes cultured in PRP displayed increased DNA content and proteoglycan and collagen type II synthesis (Akeda et al. 2006). Importantly, PRP did not affect the types of proteoglycans or collagen produced, suggesting that chondrocytes remain phenotypically stable when enriched with PRP. When compared to hyaluronic acid in co-culture with osteoarthritic chondrocytes, PRP-enriched samples resulted in a significant reduction in MMP expression, increased hyaluronan (HA) synthase expression by synoviocytes, and increased aggrecan production (Sundman et al. 2014). Aside from aiding in tissue healing, this data suggests that PRP may play a chondroprotective role as well (for details see Chap. 7).

With such overwhelmingly positive in vitro results, animal and human studies utilizing PRP have progressed rapidly. In rabbits, PRP-treated osteochondral defects showed greater cartilage regeneration and production of ECM than placebo (Sun et al. 2010). Early human knee injection trials with PRP showed improved pain and patient-reported outcome scores when compared with HA (Kon et al. 2010). In 2012, Sanchez and colleagues conducted a randomized, controlled trial comparing PRP to HA and achieved the same results at 6 months post-injection (Sanchez et al. 2012). In longer follow-up studies (12 months), symptomatic improvements in patients with knee OA have been also documented (Gobbi et al. 2012). Multiple randomized controlled trials have since shown benefits of PRP compared to other available therapies (Cerza et al. 2012). Furthermore, PRP has not been limited to the knee, as injections were shown to be effective in osteochondral lesions of the talus compared to HA (Mei-Dan et al. 2012). Unfortunately, latest studies have yielded mixed results, as a randomized controlled trial by Filardo and colleagues found no difference in outcomes between HA and PRP (Filardo et al. 2015). This finding is in agreement with an earlier study by the same group showing no superiority of PRP over HA (Filardo et al. 2012). However, a recent systematic review of PRP injections in the knee found that intra-articular injections are overall a viable therapy for patients with mild OA (Campbell et al. 2015). This group found that PRP injections carry a slightly higher risk of local adverse reactions after multiple injections and

work best in patients with early degenerative disease. Ultimately, more trials are needed to determine the proper patient cohort, dosing strategy, and injection frequency of PRP.

Conclusion

In summary, tremendous progress has been made in harnessing the potential of growth factors for cartilage repair, both in regard to halting deterioration following post-traumatic focal defects and in preserving cartilage from long-term degenerative changes. Of the individual growth factors, BMP-7, TGF- β , and FGF-18 appear to have the most short-term promise for future clinical studies. However, it has become increasingly clear that complete cartilage repair will not stem from the addition of a single growth factor, but rather a combination of growth factors (Fortier et al. 2011). This has been shown by the fact that BMP-7 produces better cartilage repair when applied in combination with TGF- β or IGF-1 than it does on its own (Loeser et al. 2003; Chubinskaya et al. 2007a) and by in vitro evidence that IGF-1, TGF- β , and FGF-2 regulate each other's gene expression and subsequent protein production (Shi 2009). Thus, it comes as no surprise that combinations of growth factors, such as those in platelet-rich plasma, may yield better and more sustainable clinical outcomes in patients with focal cartilage defects or OA, especially at early stages of disease.

Several unanswered questions remain in regard to the use of growth factors for cartilage repair. Aside from which growth factors produce optimal regeneration, it is unclear whether clinicians and researchers should strive to boost endogenous growth factor production or augment with recombinant, exogenous growth factors. Future short-term goals include obtaining a better understanding of the pathophysiology of cartilage degeneration so that growth factor therapy can be tailored to various stages of the healing process. Optimal doses and formulations must be determined in order to maximize clinical response and minimize side effects. Growth factors must be studied further in hostile, inflammatory environments to better understand their efficacy in disease states. This will likely underscore a difference in potential therapy for post-traumatic chondral defects versus therapy for chronic degenerative joint disease. Relationships between industry and academia must be fostered transparently, as previous clinical trials have come to a halt after suboptimal results. Ultimately, the solution will likely involve partnerships with regulatory agencies to move new technologies forward efficiently. Future clinical trials must be conducted with carefully selected patient cohorts. Lastly, costs of therapies will have to be reduced so that these may become financially feasible options for patients.

References

Abe T, Yamada H, Nakajima H, Kikuchi T, Takaishi H, Tadakuma T, Fujikawa K, Toyama Y (2003) Repair of full-thickness cartilage defects using liposomal transforming growth factorbeta1. J Orthop Sci 8(1):92–101. doi:10.1007/s007760300016

- Akeda K, An HS, Okuma M, Attawia M, Miyamoto K, Thonar EJ, Lenz ME, Sah RL, Masuda K (2006) Platelet-rich plasma stimulates porcine articular chondrocyte proliferation and matrix biosynthesis. Osteoarthritis Cartilage 14(12):1272–1280. doi:10.1016/j.joca.2006.05.008
- Badlani N, Oshima Y, Healey R, Coutts R, Amiel D (2009) Use of bone morphogenetic protein-7 as a treatment for osteoarthritis. Clin Orthop Relat Res 467(12):3221–3229. doi:10.1007/s11999-008-0569-9
- Barr L, Getgood A, Guehring H, Rushton N, Henson FM (2014) The effect of recombinant human fibroblast growth factor-18 on articular cartilage following single impact load. J Orthop Res 32(7):923–927. doi:10.1002/jor.22622
- Blaney Davidson EN, van der Kraan PM, van den Berg WB (2007) TGF-beta and osteoarthritis. Osteoarthritis Cartilage 15(6):597–604. doi:10.1016/j.joca.2007.02.005
- Bradley EW, Carpio LR, Newton AC, Westendorf JJ (2015) Deletion of the PH-domain and Leucine-rich repeat protein phosphatase 1 (Phlpp1) increases fibroblast growth factor (Fgf) 18 expression and promotes chondrocyte proliferation. J Biol Chem 290(26):16272–16280. doi:10.1074/jbc.M114.612937
- Campbell KA, Saltzman BM, Mascarenhas R, Khair MM, Verma NN, Bach BR Jr, Cole BJ (2015) Does intra-articular platelet-rich plasma injection provide clinically superior outcomes compared with other therapies in the treatment of knee Osteoarthritis? A systematic review of overlapping meta-analyses. Arthroscopy. doi:10.1016/j.arthro.2015.03.041
- Cerza F, Carni S, Carcangiu A, Di Vavo I, Schiavilla V, Pecora A, De Biasi G, Ciuffreda M (2012) Comparison between hyaluronic acid and platelet-rich plasma, intra-articular infiltration in the treatment of gonarthrosis. Am J Sports Med 40(12):2822–2827. doi:10.1177/0363546512461902
- Cherian JJ, Parvizi J, Bramlet D, Lee KH, Romness DW, Mont MA (2015) Preliminary results of a phase II randomized study to determine the efficacy and safety of genetically engineered allogeneic human chondrocytes expressing TGF-beta1 in patients with grade 3 chronic degenerative joint disease of the knee. Osteoarthritis Cartilage 23(12):2109–2118. doi:10.1016/j. joca.2015.06.019
- Chubinskaya S, Hakimiyan A, Pacione C, Yanke A, Rappoport L, Aigner T, Rueger DC, Loeser RF (2007a) Synergistic effect of IGF-1 and OP-1 on matrix formation by normal and OA chondrocytes cultured in alginate beads. Osteoarthritis Cartilage 15(4):421–430. doi:10.1016/j. joca.2006.10.004
- Chubinskaya S, Hurtig M, Rueger DC (2007b) OP-1/BMP-7 in cartilage repair. Int Orthop 31(6):773–781. doi:10.1007/s00264-007-0423-9
- Chubinskaya S, Segalite D, Pikovsky D, Hakimiyan AA, Rueger DC (2008) Effects induced by BMPS in cultures of human articular chondrocytes: comparative studies. Growth Factors 26(5):275–283. doi:10.1080/08977190802291733
- Chubinskaya S, Otten L, Soeder S, Borgia JA, Aigner T, Rueger DC, Loeser RF (2011) Regulation of chondrocyte gene expression by osteogenic protein-1. Arthritis Res Ther 13(2):R55. doi:10.1186/ar3300
- Diao H, Wang J, Shen C, Xia S, Guo T, Dong L, Zhang C, Chen J, Zhao J, Zhang J (2009) Improved cartilage regeneration utilizing mesenchymal stem cells in TGF-beta1 gene-activated scaffolds. Tissue Eng Part A 15(9):2687–2698. doi:10.1089/ten.TEA.2008.0621
- Ekenstedt KJ, Sonntag WE, Loeser RF, Lindgren BR, Carlson CS (2006) Effects of chronic growth hormone and insulin-like growth factor 1 deficiency on osteoarthritis severity in rat knee joints. Arthritis Rheum 54(12):3850–3858. doi:10.1002/art.22254
- Ellman MB, An HS, Muddasani P, Im HJ (2008) Biological impact of the fibroblast growth factor family on articular cartilage and intervertebral disc homeostasis. Gene 420(1):82–89. doi:10.1016/j.gene.2008.04.019
- Ellman MB, Yan D, Ahmadinia K, Chen D, An HS, Im HJ (2013) Fibroblast growth factor control of cartilage homeostasis. J Cell Biochem 114(4):735–742. doi:10.1002/jcb.24418
- Ellsworth JL, Berry J, Bukowski T, Claus J, Feldhaus A, Holderman S, Holdren MS, Lum KD, Moore EE, Raymond F, Ren H, Shea P, Sprecher C, Storey H, Thompson DL, Waggie K, Yao L, Fernandes RJ, Eyre DR, Hughes SD (2002) Fibroblast growth factor-18 is a trophic factor for mature chondrocytes and their progenitors. Osteoarthritis Cartilage 10(4):308–320. doi:10.1053/joca.2002.0514

- Elshaier AM, Hakimiyan AA, Rappoport L, Rueger DC, Chubinskaya S (2009) Effect of interleukin-1beta on osteogenic protein 1-induced signaling in adult human articular chondrocytes. Arthritis Rheum 60(1):143–154. doi:10.1002/art.24151
- Fan J, Gong Y, Ren L, Varshney RR, Cai D, Wang DA (2010) In vitro engineered cartilage using synovium-derived mesenchymal stem cells with injectable gellan hydrogels. Acta Biomater 6(3):1178–1185. doi:10.1016/j.actbio.2009.08.042
- Felson DT, Naimark A, Anderson J, Kazis L, Castelli W, Meenan RF (1987) The prevalence of knee osteoarthritis in the elderly. The Framingham Osteoarthritis Study. Arthritis Rheum 30(8):914–918
- Filardo G, Kon E, Di Martino A, Di Matteo B, Merli ML, Cenacchi A, Fornasari PM, Marcacci M (2012) Platelet-rich plasma vs hyaluronic acid to treat knee degenerative pathology: study design and preliminary results of a randomized controlled trial. BMC Musculoskelet Disord 13:229. doi:10.1186/1471-2474-13-229
- Filardo G, Di Matteo B, Di Martino A, Merli ML, Cenacchi A, Fornasari P, Marcacci M, Kon E (2015) Platelet-rich plasma intra-articular knee injections show no superiority versus viscosupplementation: a randomized controlled trial. Am J Sports Med 43(7):1575–1582. doi:10.1177/0363546515582027
- Fortier LA, Mohammed HO, Lust G, Nixon AJ (2002) Insulin-like growth factor-I enhances cellbased repair of articular cartilage. J Bone Joint Surg Br 84(2):276–288
- Fortier LA, Barker JU, Strauss EJ, McCarrel TM, Cole BJ (2011) The role of growth factors in cartilage repair. Clin Orthop Relat Res 469(10):2706–2715. doi:10.1007/s11999-011-1857-3
- Friedl A, Chang Z, Tierney A, Rapraeger AC (1997) Differential binding of fibroblast growth factor-2 and -7 to basement membrane heparan sulfate: comparison of normal and abnormal human tissues. Am J Pathol 150(4):1443–1455
- Giannoudis PV, Einhorn TA, Schmidmaier G, Marsh D (2008) The diamond concept--open questions. Injury 39(Suppl 2):S5–S8. doi:10.1016/s0020-1383(08)70010-x
- Gobbi A, Karnatzikos G, Mahajan V, Malchira S (2012) Platelet-rich plasma treatment in symptomatic patients with knee osteoarthritis: preliminary results in a group of active patients. Sports Health 4(2):162–172. doi:10.1177/1941738111431801
- Goldring MB, Tsuchimochi K, Ijiri K (2006) The control of chondrogenesis. J Cell Biochem 97(1):33–44. doi:10.1002/jcb.20652
- Goodrich LR, Hidaka C, Robbins PD, Evans CH, Nixon AJ (2007) Genetic modification of chondrocytes with insulin-like growth factor-1 enhances cartilage healing in an equine model. J Bone Joint Surg Br 89(5):672–685. doi:10.1302/0301-620x.89b5.18343
- Goumans MJ, Mummery C (2000) Functional analysis of the TGFbeta receptor/Smad pathway through gene ablation in mice. Int J Dev Biol 44(3):253–265
- Grgic M, Jelic M, Basic V, Basic N, Pecina M, Vukicevic S (1997) Regeneration of articular cartilage defects in rabbits by osteogenic protein-1 (bone morphogenetic protein-7). Acta Med Croatica 51(1):23–27
- Ha CW, Noh MJ, Choi KB, Lee KH (2012) Initial phase I safety of retrovirally transduced human chondrocytes expressing transforming growth factor-beta-1 in degenerative arthritis patients. Cytotherapy 14(2):247–256. doi:10.3109/14653249.2011.629645
- Hayashi M, Muneta T, Ju YJ, Mochizuki T, Sekiya I (2008) Weekly intra-articular injections of bone morphogenetic protein-7 inhibits osteoarthritis progression. Arthritis Res Ther 10(5):R118. doi:10.1186/ar2521
- Henson FM, Bowe EA, Davies ME (2005) Promotion of the intrinsic damage-repair response in articular cartilage by fibroblastic growth factor-2. Osteoarthritis Cartilage 13(6):537–544. doi:10.1016/j.joca.2005.02.007
- Hunter DJ, Pike MC, Jonas BL, Kissin E, Krop J, McAlindon T (2010) Phase 1 safety and tolerability study of BMP-7 in symptomatic knee osteoarthritis. BMC Musculoskelet Disord 11:232. doi:10.1186/1471-2474-11-232
- Hurtig M, Chubinskaya S, Dickey J, Rueger D (2009) BMP-7 protects against progression of cartilage degeneration after impact injury. J Orthop Res 27(5):602–611. doi:10.1002/jor. 20787

- Im HJ, Pacione C, Chubinskaya S, Van Wijnen AJ, Sun Y, Loeser RF (2003) Inhibitory effects of insulin-like growth factor-1 and osteogenic protein-1 on fibronectin fragment- and interleukinlbeta-stimulated matrix metalloproteinase-13 expression in human chondrocytes. J Biol Chem 278(28):25386–25394. doi:10.1074/jbc.M302048200
- Im HJ, Li X, Muddasani P, Kim GH, Davis F, Rangan J, Forsyth CB, Ellman M, Thonar EJ (2008) Basic fibroblast growth factor accelerates matrix degradation via a neuro-endocrine pathway in human adult articular chondrocytes. J Cell Physiol 215(2):452–463. doi:10.1002/jcp.21317
- Itoh S, Itoh F, Goumans MJ, Ten Dijke P (2000) Signaling of transforming growth factor-beta family members through Smad proteins. Eur J Biochem 267(24):6954–6967
- Itoh S, Hattori T, Tomita N, Aoyama E, Yutani Y, Yamashiro T, Takigawa M (2013) CCN family member 2/connective tissue growth factor (CCN2/CTGF) has anti-aging effects that protect articular cartilage from age-related degenerative changes. PLoS One 8(8):e71156. doi:10.1371/ journal.pone.0071156
- Ivkovic S, Yoon BS, Popoff SN, Safadi FF, Libuda DE, Stephenson RC, Daluiski A, Lyons KM (2003) Connective tissue growth factor coordinates chondrogenesis and angiogenesis during skeletal development. Development 130(12):2779–2791
- Jelic M, Pecina M, Haspl M, Kos J, Taylor K, Maticic D, McCartney J, Yin S, Rueger D, Vukicevic S (2001) Regeneration of articular cartilage chondral defects by osteogenic protein-1 (bone morphogenetic protein-7) in sheep. Growth Factors 19(2):101–113
- Jordan JM, Helmick CG, Renner JB, Luta G, Dragomir AD, Woodard J, Fang F, Schwartz TA, Abbate LM, Callahan LF, Kalsbeek WD, Hochberg MC (2007) Prevalence of knee symptoms and radiographic and symptomatic knee osteoarthritis in African Americans and Caucasians: the Johnston County Osteoarthritis Project. J Rheumatol 34(1):172–180
- Kawaki H, Kubota S, Suzuki A, Lazar N, Yamada T, Matsumura T, Ohgawara T, Maeda T, Perbal B, Lyons KM, Takigawa M (2008) Cooperative regulation of chondrocyte differentiation by CCN2 and CCN3 shown by a comprehensive analysis of the CCN family proteins in cartilage. J Bone Miner Res 23(11):1751–1764. doi:10.1359/jbmr.080615
- Kon E, Buda R, Filardo G, Di Martino A, Timoncini A, Cenacchi A, Fornasari PM, Giannini S, Marcacci M (2010) Platelet-rich plasma: intra-articular knee injections produced favorable results on degenerative cartilage lesions. Knee Surg Sports Traumatol Arthrosc 18(4):472–479. doi:10.1007/s00167-009-0940-8
- Kern D, Zlatkin MB, Dalinka MK (1988) Occupational and post-traumatic arthritis. Radiol Clin North Am 26(6):1349–1358
- Kuo AC, Rodrigo JJ, Reddi AH, Curtiss S, Grotkopp E, Chiu M (2006) Microfracture and bone morphogenetic protein 7 (BMP-7) synergistically stimulate articular cartilage repair. Osteoarthritis Cartilage 14(11):1126–1135. doi:10.1016/j.joca.2006.04.004
- Kurth T, Hedbom E, Shintani N, Sugimoto M, Chen FH, Haspl M, Martinovic S, Hunziker EB (2007) Chondrogenic potential of human synovial mesenchymal stem cells in alginate. Osteoarthritis Cartilage 15(10):1178–1189. doi:10.1016/j.joca.2007.03.015
- Lee KH, Song SU, Hwang TS, Yi Y, Oh IS, Lee JY, Choi KB, Choi MS, Kim SJ (2001) Regeneration of hyaline cartilage by cell-mediated gene therapy using transforming growth factor beta 1-producing fibroblasts. HumGene Ther 12(14):1805–1813. doi:10.1089/104303401750476294
- Lee MC, Ha CW, Elmallah RK, Cherian JJ, Cho JJ, Kim TW, Bin SI, Mont MA (2015) A placebocontrolled randomised trial to assess the effect of TGF-ss1-expressing chondrocytes in patients with arthritis of the knee. Bone Joint J 97-B(7):924–932. doi:10.1302/0301-620x.97b7.35852
- Lires-Dean M, Carames B, Cillero-Pastor B, Galdo F, Lopez-Armada MJ, Blanco FJ (2008) Antiapoptotic effect of transforming growth factor-beta1 on human articular chondrocytes: role of protein phosphatase 2A. Osteoarthritis Cartilage 16(11):1370–1378. doi:10.1016/j.joca.2008.04.001
- Liu Z, Xu J, Colvin JS, Ornitz DM (2002) Coordination of chondrogenesis and osteogenesis by fibroblast growth factor 18. Genes Dev 16(7):859–869. doi:10.1101/gad.965602
- Liu Z, Lavine KJ, Hung IH, Ornitz DM (2007) FGF18 is required for early chondrocyte proliferation, hypertrophy and vascular invasion of the growth plate. Dev Biol 302(1):80–91. doi:10.1016/j.ydbio.2006.08.071
- Loeser RF, Shanker G, Carlson CS, Gardin JF, Shelton BJ, Sonntag WE (2000) Reduction in the chondrocyte response to insulin-like growth factor 1 in aging and osteoarthritis: studies in a

non-human primate model of naturally occurring disease. Arthritis Rheum 43(9):2110–2120. doi:10.1002/1529-0131(20009)43:9<2110::aid-anr23>3.0.co;2-u

- Loeser RF, Carlson CS, Del Carlo M, Cole A (2002) Detection of nitrotyrosine in aging and osteoarthritic cartilage: correlation of oxidative damage with the presence of interleukin-1beta and with chondrocyte resistance to insulin-like growth factor 1. Arthritis Rheum 46(9):2349–2357. doi:10.1002/art.10496
- Loeser RF, Pacione CA, Chubinskaya S (2003) The combination of insulin-like growth factor 1 and osteogenic protein 1 promotes increased survival of and matrix synthesis by normal and osteoarthritic human articular chondrocytes. Arthritis Rheum 48(8):2188–2196. doi:10.1002/art.11209
- Loeser RF, Chubinskaya S, Pacione C, Im HJ (2005) Basic fibroblast growth factor inhibits the anabolic activity of insulin-like growth factor 1 and osteogenic protein 1 in adult human articular chondrocytes. Arthritis Rheum 52(12):3910–3917. doi:10.1002/art.21472
- Loeser RF, Gandhi U, Long DL, Yin W, Chubinskaya S (2014) Aging and oxidative stress reduce the response of human articular chondrocytes to insulin-like growth factor 1 and osteogenic protein 1. Arthritis Rheumatol 66(8):2201–2209. doi:10.1002/art.38641
- Lohmander LS, Hellot S, Dreher D, Krantz EF, Kruger DS, Guermazi A, Eckstein F (2014) Intraarticular sprifermin (recombinant human fibroblast growth factor 18) in knee osteoarthritis: a randomized, double-blind, placebo-controlled trial. Arthritis Rheumatol 66(7):1820– 1831. doi:10.1002/art.38614
- Lotz M, Rosen F, McCabe G, Quach J, Blanco F, Dudler J, Solan J, Goding J, Seegmiller JE, Terkeltaub R (1995) Interleukin 1 beta suppresses transforming growth factor-induced inorganic pyrophosphate (PPi) production and expression of the PPi-generating enzyme PC-1 in human chondrocytes. Proc Natl Acad Sci U S A 92(22):10364–10368
- Mason JM, Breitbart AS, Barcia M, Porti D, Pergolizzi RG, Grande DA (2000) Cartilage and bone regeneration using gene-enhanced tissue engineering. Clin Orthop Relat Res (379 Suppl):S171–S178
- Massague J, Chen YG (2000) Controlling TGF-beta signaling. Genes Dev 14(6):627-644
- Mei-Dan O, Carmont MR, Laver L, Mann G, Maffulli N, Nyska M (2012) Platelet-rich plasma or hyaluronate in the management of osteochondral lesions of the talus. Am J Sports Med 40(3):534–541. doi:10.1177/0363546511431238
- Miller Y, Bachowski G, Benjamin R, Eklund D, Hibbard A, Lightfoot T (2007) Practice guidelines for blood transfusion: a compilation from recent peer-reviewed literature, vol 56, 2nd edn. American Red Cross, Washington
- Mishra A, Tummala P, King A, Lee B, Kraus M, Tse V, Jacobs CR (2009) Buffered platelet-rich plasma enhances mesenchymal stem cell proliferation and chondrogenic differentiation. Tissue Eng Part C Methods 15(3):431–435. doi:10.1089/ten.tec.2008.0534
- Miyamoto C, Matsumoto T, Sakimura K, Shindo H (2007) Osteogenic protein-1 with transforming growth factor-beta1: potent inducer of chondrogenesis of synovial mesenchymal stem cells in vitro. J Orthop Sci 12(6):555–561. doi:10.1007/s00776-007-1176-4
- Moore EE, Bendele AM, Thompson DL, Littau A, Waggie KS, Reardon B, Ellsworth JL (2005) Fibroblast growth factor-18 stimulates chondrogenesis and cartilage repair in a rat model of injuryinduced osteoarthritis. Osteoarthritis Cartilage 13(7):623–631. doi:10.1016/j.joca.2005.03.003
- Morales TI (2008) The quantitative and functional relation between insulin-like growth factor-I (IGF) and IGF-binding proteins during human osteoarthritis. J Orthop Res 26(4):465–474. doi:10.1002/jor.20549
- Nakanishi T, Nishida T, Shimo T, Kobayashi K, Kubo T, Tamatani T, Tezuka K, Takigawa M (2000) Effects of CTGF/Hcs24, a product of a hypertrophic chondrocyte-specific gene, on the proliferation and differentiation of chondrocytes in culture. Endocrinology 141(1):264–273. doi:10.1210/endo.141.1.7267
- Nishida T, Nakanishi T, Asano M, Shimo T, Takigawa M (2000a) Effects of CTGF/Hcs24, a hypertrophic chondrocyte-specific gene product, on the proliferation and differentiation of osteoblastic cells in vitro. J Cell Physiol 184(2):197–206. doi:10.1002/1097-4652 (200008)184:2<197::aid-jcp7>3.0.co;2-r
- Nishida Y, Knudson CB, Eger W, Kuettner KE, Knudson W (2000b) Osteogenic protein 1 stimulates cells-associated matrix assembly by normal human articular chondrocytes: up-regulation

of hyaluronan synthase, CD44, and aggrecan. Arthritis Rheum 43(1):206–214. doi:10.1002/1529-0131(20001)43:1<206::aid-anr25>3.0.co;2-1

- Nishida T, Kubota S, Nakanishi T, Kuboki T, Yosimichi G, Kondo S, Takigawa M (2002) CTGF/ Hcs24, a hypertrophic chondrocyte-specific gene product, stimulates proliferation and differentiation, but not hypertrophy of cultured articular chondrocytes. J Cell Physiol 192(1):55–63. doi:10.1002/jcp.10113
- Nishida T, Kubota S, Kojima S, Kuboki T, Nakao K, Kushibiki T, Tabata Y, Takigawa M (2004) Regeneration of defects in articular cartilage in rat knee joints by CCN2 (connective tissue growth factor). J Bone Miner Res 19(8):1308–1319. doi:10.1359/jbmr.040322
- Noh MJ, Copeland RO, Yi Y, Choi KB, Meschter C, Hwang S, Lim CL, Yip V, Hyun JP, Lee HY, Lee KH (2010) Pre-clinical studies of retrovirally transduced human chondrocytes expressing transforming growth factor-beta-1 (TG-C). Cytotherapy 12(3):384–393. doi:10.3109/14653240903470639
- Nurden AT, Nurden P, Sanchez M, Andia I, Anitua E (2008) Platelets and wound healing. Front Biosci 13:3532–3548
- Pascual-Garrido C, Chubinskaya S (2015) Potential targets for pharmacologic therapies. In: Olson S, Guilak F (eds) Post-traumatic arthritis: pathogenesis, diagnosis, and management. Springer, pp 331–342
- Power J, Hernandez P, Guehring H, Getgood A, Henson F (2014) Intra-articular injection of rhFGF-18 improves the healing in microfracture treated chondral defects in an ovine model. J Orthop Res 32(5):669–676. doi:10.1002/jor.22580
- Qi BW, Yu AX, Zhu SB, Zhou M, Wu G (2013) Chitosan/poly(vinyl alcohol) hydrogel combined with Ad-hTGF-beta1 transfected mesenchymal stem cells to repair rabbit articular cartilage defects. Exp Biol Med (Maywood) 238(1):23–30. doi:10.1258/ebm.2012.012223
- Roos H, Adalberth T, Dahlberg L, Lohmander LS (1995) Osteoarthritis of the knee after injury to the anterior cruciate ligament or meniscus: the influence of time and age. Osteoarthritis Cartilage 3(4):261–267
- Sanchez M, Fiz N, Azofra J, Usabiaga J, Aduriz Recalde E, Garcia Gutierrez A, Albillos J, Garate R, Aguirre JJ, Padilla S, Orive G, Anitua E (2012) A randomized clinical trial evaluating plasma rich in growth factors (PRGF-Endoret) versus hyaluronic acid in the short-term treatment of symptomatic knee osteoarthritis. Arthroscopy 28(8):1070–1078. doi:10.1016/j. arthro.2012.05.011
- Schalkwijk J, Joosten LA, van den Berg WB, van de Putte LB (1989) Chondrocyte nonresponsiveness to insulin-like growth factor 1 in experimental arthritis. Arthritis Rheum 32(7):894–900
- Shen B, Wei A, Whittaker S, Williams LA, Tao H, Ma DD, Diwan AD (2010) The role of BMP-7 in chondrogenic and osteogenic differentiation of human bone marrow multipotent mesenchymal stromal cells in vitro. J Cell Biochem 109(2):406–416. doi:10.1002/jcb.22412
- Shi S, Mercer S, Eckert GJ, Trippel SB. Growth factor regulation of growth factors in articular chondrocytes. J Biol Chem 2009;284:6697–6704
- Shimmin A, Young D, O'Leary S, Shih MS, Rueger DC, Walsh WR (2003) Growth factor augmentation of an ovine mosaicplasty model. Trans ICRS 4th Symposium, no. 16, Toronto
- Smyth SS, McEver RP, Weyrich AS, Morrell CN, Hoffman MR, Arepally GM, French PA, Dauerman HL, Becker RC (2009) Platelet functions beyond hemostasis. J Thromb Haemost 7(11):1759–1766. doi:10.1111/j.1538-7836.2009.03586.x
- Song JJ, Aswad R, Kanaan RA, Rico MC, Owen TA, Barbe MF, Safadi FF, Popoff SN (2007) Connective tissue growth factor (CTGF) acts as a downstream mediator of TGF-beta1 to induce mesenchymal cell condensation. J Cell Physiol 210(2):398–410. doi:10.1002/jcp.20850
- Sun Y, Feng Y, Zhang CQ, Chen SB, Cheng XG (2010) The regenerative effect of platelet-rich plasma on healing in large osteochondral defects. Int Orthop 34(4):589–597. doi:10.1007/ s00264-009-0793-2
- Sundman EA, Cole BJ, Karas V, Della Valle C, Tetreault MW, Mohammed HO, Fortier LA (2014) The anti-inflammatory and matrix restorative mechanisms of platelet-rich plasma in osteoarthritis. Am J Sports Med 42(1):35–41. doi:10.1177/0363546513507766

- Tomita N, Hattori T, Itoh S, Aoyama E, Yao M, Yamashiro T, Takigawa M (2013) Cartilagespecific over-expression of CCN family member 2/connective tissue growth factor (CCN2/ CTGF) stimulates insulin-like growth factor expression and bone growth. PLoS One 8(3):e59226. doi:10.1371/journal.pone.0059226
- Vinall RL, Lo SH, Reddi AH (2002) Regulation of articular chondrocyte phenotype by bone morphogenetic protein 7, interleukin 1, and cellular context is dependent on the cytoskeleton. Exp Cell Res 272(1):32–44. doi:10.1006/excr.2001.5395
- Yan D, Chen D, Cool SM, van Wijnen AJ, Mikecz K, Murphy G, Im HJ (2011) Fibroblast growth factor receptor 1 is principally responsible for fibroblast growth factor 2-induced catabolic activities in human articular chondrocytes. Arthritis Res Ther 13(4):R130. doi:10.1186/ar3441
- Zhu S, Zhang B, Man C, Ma Y, Liu X, Hu J (2014) Combined effects of connective tissue growth factor-modified bone marrow-derived mesenchymal stem cells and NaOH-treated PLGA scaffolds on the repair of articular cartilage defect in rabbits. Cell Transplant 23(6):715–727. doi:10.3727/096368913x669770

The Role of Platelet-Rich Plasma in Cartilage Repair

Elizaveta Kon, Giuseppe Filardo, Berardo Di Matteo, and Maurilio Marcacci

Abstract

The use of biological agents to positively modulate articular homeostasis and provide a stimulus to tissue healing is a growing field in orthopedics. Among the available strategies, platelet-rich plasma (PRP), an autologous blood-derived concentrate rich in growth factors, is currently the most exploited biological approach for conservative management (simple intra-articular injections) and as an augmentation during surgical procedures. It has been applied both to treat osteoarthritis and chondral/osteochondral lesions in different joints, with the primary aim of providing symptomatic relief and functional recovery, and also to induce a positive modulation of the entire articular microenvironment. In this chapter, we analyze the clinical evidence available on the role of PRP to treat cartilage pathology, focusing in particular on the data coming from randomized controlled trials.

PRP is a novel technology, and, based on the available evidence, there is an overall support on the safety of this biological approach but still no conclusive

G. Filardo

E. Kon • B. Di Matteo (🖂) • M. Marcacci

III Orthopaedic Division, Knee Joint Reconstruction Center, Humanitas Clinical Institute, Rozzano, Milan, Italy e-mail: elisaveta.kon@gmail.com; berardo.dimatteo@gmail.com; m.marcacci@biomec.ior.it

Nano-Biotechnology Lab, Rizzoli Orthopaedic Institute, Bologna, Italy e-mail: g.filardo@biomec.ior.it

[©] Springer International Publishing AG 2017 S. Grässel, A. Aszodi (eds.), *Cartilage*, DOI 10.1007/978-3-319-53316-2_7

evidence about its real efficacy in the management of cartilage pathology, especially when applied during surgical procedures. Looking at conservative application, PRP proved to be better than placebo but not clearly superior to more traditional approaches such as viscosupplementation.

The improvement in this field will pass through the understanding of the interaction dynamics between PRP and the etiopathogenetic/regenerative cartilage processes, as well as the identification of the best PRP features and application modalities, together with the identification of the patient and lesion type that may benefit from this treatment.

7.1 Platelet-Rich Plasma: Why Using It in Cartilage Pathology?

A healthy joint requires a well-controlled balance between molecular signals regulating anabolic and catabolic activities. This balance is determined both at the level of single cells and also at the whole tissue architecture, consisting of complex interactions among all the different articular tissues such as the cartilage, bone, synovium, ligaments, tendons, and menisci (Lories 2008). Several factors may come into play to impair the maintenance of joint homeostasis, and this process leads to the development of chondral or osteochondral lesions and, in case of chronicity, to the onset of osteoarthritis (OA) (Heijink et al. 2012; Hunter and Felson 2006), which is a degenerative disorder of the entire joint characterized by loss of articular cartilage, subchondral bone remodeling, osteophyte formation, synovial hyperplasia, changes in synovial fluid, fibrosis of joint capsule, and changes in menisci and ligaments (Lotz and Carames 2011).

Currently, several treatment options have been proposed to mitigate the symptoms related to cartilage lesions and/or OA and to improve joint function including changes in lifestyle, regular exercises, specific rehabilitation protocols, dietary supplements, "conventional" pharmacologic agents (such as NSAIDs), and also minimally invasive treatments consisting in intra-articular injections, the most common being corticosteroids and hyaluronic acid (Reginster et al. 2015; Cutolo et al. 2015; Jacobs et al. 2014). However, none of these treatments is fully effective, and the outcome is never clearly predictable: furthermore, the majority of these conservative options are not able to act on the inner mechanisms responsible of the tissue damage, and, therefore, they have minimal or no role in modifying the course of the disease. In recent years, research has focused on the possibility of modulating overall joint tissue homeostasis through the application of biological agents which could promote healing minimizing the risks and adverse events related to the use of traditional "on-the-shelf" products (Filardo et al. 2016; Kon et al. 2012).

In this scenario, the most exploited strategy is platelet-rich plasma (PRP), an autologous blood derivative containing a higher platelet concentration than whole blood. Platelets have been documented to act as a reservoir of many autologous growth factors (GFs), which serve as the rational for their concentration and the production of PRP: a safe and easily available method to have bioactive molecules delivered into the lesion site, with the aim of enhancing the healing process and driving it toward a better quality of tissue repair with improved biomechanical properties. More in detail, activation of platelets contained in PRP induces the release of several GFs that are stored in a-granules, such as platelet-derived growth factor (PDGF), transforming growth factor (TGF)-ß, insulin-like growth factor (IGF), vascular endothelial growth factor (VEGF), fibroblast growth factor (FGF), epidermal growth factor (EGF), and many other bioactive and anti-inflammatory molecules as well (Boswell et al. 2012; Xie et al. 2014) (Fig. 7.1). These agents, released at the site of injury in high concentrations, may promote tissue healing by stimulating cell proliferation, chemotaxis, cell differentiation, and angiogenesis and also by modulating inflammatory and catabolic molecules involved in the cartilage degenerative processes (Braun et al. 2014).

PRP can be easily prepared from venous blood even in an outpatient setting, and it can be delivered via a simple intra-articular injection or during surgical procedure as an augmentation. This appealing treatment option has encountered a large success among clinicians and has been increasingly used in the last decade, with several industries introducing their "proprietary" kits for PRP production. Despite the widespread of this biological treatment in orthopedics, clinical results are sometimes contradictory with no clear treatment indications, due to low-level clinical studies and the lack of understanding of the mechanism of action of this blood derivative (Kon et al. 2013).



Fig. 7.1 Role and action of the most important platelet-derived growth factors in cartilage healing

7.2 Review of Current Clinical Evidence

7.2.1 Knee

7.2.1.1 Conservative Management

Among the many studies dealing with PRP injective intra-articular application, just a few randomized controlled trials (RCTs) have been published. Sanchez et al. (2012a) investigated the efficacy of single-spinning leukocyte-free PRP compared to HA in 153 patients evaluated up to 6 months of follow-up. The only aspect where a superiority of PRP was found was the percentage of responders (patients with at least 50% of pain reduction), which was significantly higher in the PRP group. Besides this finding, the study did not show that PRP in moderate/severe OA was more effective than HA. Similar considerations were made by Filardo et al. (2015), according to the results of their randomized double-blind trial including 192 patients evaluated up to 12 months and comparing PRP and HA: no statistical intergroup difference was reported at any follow-up evaluation. Conversely, Cerza et al. (2012) treated 120 patients by either autologous conditioned plasma (ACP, a lowconcentrated PRP without leukocytes) or HA. Surprisingly, the ACP group showed a significantly better performance than HA in all groups of treatment, including patients affected by grade III knee OA. Furthermore, the clinical gap between treatments increased over time in favor of ACP. Finally, a recent randomized trial by Patel et al. (2013) was the first to test PRP versus saline. Seventy-eight patients affected by Kellgren grade I-III OA were included and treated bilaterally with one injection of PRP, two injections of PRP (3 weeks apart), or one injection of saline. Despite the low number of patients included, a significant difference was observed between PRP and saline solution in terms of clinical outcome. Interestingly, no difference was reported among patients who received one or two PRP injections. Clear superiority of PRP compared to saline was also reported in the RCT recently published by Smith (2016) on 30 patients followed up to 12 months: the biological approach outperformed viscosupplementation at any follow-up evaluation.

Only one RCT (Forogh et al. 2016) investigated the comparison between PRP and corticosteroid: 41 patients were treated by either a single injection of PRP or corticosteroids. Even in this case, both at 2 and 6 months' follow-up, PRP provided better results than corticosteroids.

7.2.1.2 Augmentation in Knee Surgery

Dhollander et al. (2011) treated five symptomatic osteochondral defects of the patella with microfracturing followed by the application of a collagen I/III scaffold membrane. PRP was administered beneath the membrane at the interface with the microfractured subchondral bone. The final follow-up was 24 months after the surgical procedure. Clinical results after 2 years were satisfactory both in terms of pain relief and functional improvement, and MRI evaluation showed good quality of the repair tissue. A further study by Siclari et al. (2012) showed the efficacy of a polyglycolic acid/hyaluronan scaffold immersed in PRP for treating full-thickness chondral defects of the knee: 52 patients were treated arthroscopically by perforations

and scaffold implantation. At 1-year follow-up, a significant clinical improvement was observed in all KOOS subcategories. Osteochondral lesions were treated by Buda et al. (2010), who used an innovative arthroscopic one-stage approach based on autologous bone marrow-derived mesenchymal stem cells (BMDCs), PRP, and, alternately, porcine collagen powder or hyaluronic acid (HA) membrane to create a scaffold: 20 patients were treated and followed up for 24 months with IKDC and KOOS scores combined with MRI analysis. Besides the significant improvement in clinical scores, interesting correlations were found: combined surgery slowed down recovery although, at final evaluation, similar results were obtained with respect to those of patients without combined procedures; hyperintense MRI signal of repair tissue was correlated with poorer clinical results. In general, these preliminary reports suggested good results, but the low scientific level of these papers and even more the concomitant application of different treatments hinder the possibility to understand the real role of PRP.

7.2.2 Ankle

7.2.2.1 Conservative Management

Conservative application was tested in a prospective study of Mei-Dan et al. (2012) who compared the efficacy of hyaluronic acid (HA) and PRP in 30 patients (15 per group) affected by talar osteochondral lesions not responsive to other previous conservative management. Patients were allocated to receive three-weekly intraarticular injections of HA (2 ml each) or PRP (2 ml each) and were evaluated up to 28 weeks of follow-up. Investigators used AHFS, AOFAS, and VAS for pain, stiffness, and function scores. Results were statistically significant, and PRP proved to be more effective in controlling pain and reestablishing function.

7.2.2.2 Augmentation in Ankle Surgery

Surgical application of PRP in talar osteochondral lesions has been exploited by Giannini et al. in two different studies (Giannini et al. 2009, 2010). They described the first clinical application of an innovative arthroscopic one-stage technique involving autologous mesenchymal stem cells (MSCs), PRP, and, alternately, porcine collagen powder or HA membrane. The procedure consisted of harvesting bone-marrow derived cells from the posterior iliac crest of the patients through a traditional marrow needle. A total of 60 ml bone marrow aspirate was collected and immediately put into a cell separator-concentrator to obtain a final 6 ml of concentrated MSCs. At this point, a collagen powder or the hyaluronic acid membrane could be used. In case of the former, 2 ml of MSCs concentrate were added to 1 g of collagen powder and 1 ml of platelet-rich fibrin gel (obtained the day before surgery). In case of HA membrane, it was cut to match the size of the talar osteochondral lesion and then covered with 2 ml of MSCs concentrate and 1 ml of platelet-rich fibrin gel. The total procedure was conducted via ankle arthroscopy, and, after the preparation of the lesion, the biological composite was located onto the defect through a cannula, using a probe to obtain the best possible fitting.

The first clinical trial (Giannini et al. 2009) involved 48 patients (average age = 28.5 years) affected by focal lesions (average size = 2.1 cm^2) evaluated at 6, 12, 18, and 24 months of follow-up through the AOFAS score. A significant increase of this parameter was already registered 6 months after the surgical procedure, and this result was later confirmed up to the final follow-up. The rate of return to high impact sports activity was satisfying, with more than 75% of the patients back at the 11 months follow-up. Investigators found a correlation between clinical outcome and lesion size, with lower results for defects >3 cm², and it was also shown that previous surgery negatively affected the outcome. On the other hand, no influence was observed for lesion depth, neither difference was seen according to the scaffold used (collagen powder or HA membrane). Five second-look arthroscopies were performed at 1-year follow-up: in two cases, biopsies were taken revealing, after histological and immunohistological analysis, the presence of new cartilage tissue with various degrees of tissue remodeling toward the hyaline type. The overall findings made possible to assert that this novel approach could stimulate tissue regeneration with interesting clinical efficacy, with results even comparable to the ones of autologous chondrocyte implantation (ACI) but without the double surgical time and the inherent stress for the patient. To this particular aspect, the same authors dedicated a further study (Giannini et al. 2010) focusing on the comparison of the MSCs + PRP + scaffold technique with open and arthroscopic ACI. Eighty-one patients were included in this analysis, 10 of which treated with open ACI, 46 with arthroscopic ACI, and 25 with bone marrow-derived mesenchymal cells (BMDCs) "one-step" technique. Clinical results were compared up to 3 years of follow-up. AOFAS was the instrument adopted for clinical evaluation together with radiographical analysis. The clinical improvement in each subgroup was significant, and no intergroup difference was observed, thus confirming the possibility of matching the effectiveness of chondrocyte transplantation through a single-step procedure. X-rays showed no sign of osteoarthritis progression; MRI revealed a good rate of defect filling and integration of the newly regenerated tissue within the surrounding tissue. Another aspect worth of consideration is the economical issue: in fact, authors pointed out that their one-step regenerative technique costs less than half of the traditional arthroscopic ACI.

7.2.3 Hip

7.2.3.1 Conservative Management

Few studies have been published on this particular topic. The first one, authored by Battaglia et al. (2011), reported the results of PRP ultrasound-guided injective treatment in 20 patients affected by hip OA (Kellgren-Lawrence Score from I to III): three intra-articular injections 2 weeks apart were performed with patients followed up for 1 year. The clinical outcome was positive, but a worsening occurred 3 months the final evaluation, thus confirming the time-dependent effect of PRP. Similar results were confirmed in a later randomized study by the same authors (Battaglia et al. 2013), where PRP was compared to HA. One hundred patients were divided

into two treatment groups, receiving three intra-articular injections: a comparable clinical benefit was documented in both groups, without significant difference, and the same trend of worsening results over time was registered. Sanchez et al. (2012b) treated 40 patients affected by OA with 3-weekly ultrasound-guided injections of PRP. Evaluation was carried out for 6 months using the WOMAC, Harris, and VAS scores for pain. Satisfactory results were reported with a significant reduction in pain level at the first evaluation after 6 weeks, which was confirmed at the final 6 months' follow-up. Functional recovery was encouraging as evaluated through a specific subscale of the WOMAC score. However, 11 out of 40 patients did not have any beneficial effect after injective treatment: in these cases, a metal resurfacing was required. Recently, a RCT was published by Dallari et al. (2016) who randomized 111 patients in three different treatment groups: one to receive 3-weekly HA injections, one to receive 3-weekly PRP injections, and the last to receive 3-weekly PRP+HA injections. They found that PRP-treated patients were the best responding group at 12 months evaluation in terms of functional recovery and pain control, without any significant contribution from the addition of HA to the platelet concentrate.

7.3 Discussion

PRP technology is facing a remarkable widespread in everydays' clinical practice, both as an augmentation during surgery and as a conservative approach for treating disparate musculoskeletal conditions (Vannini et al. 2015; Di Matteo et al. 2015). For what concerns the current understanding on the potential and feasibility of applying PRP in the management of cartilage pathology, looking at the surgical application, it is not possible to draw definite conclusions about the efficacy of this treatment. Based on the current evidence, it is very difficult to identify how much PRP might contribute to determine the clinical outcome with respect to the surgical treatment performed alone, and more comparative studies aimed at assessing the specific role of PRP are needed. Furthermore, in many cases, PRP is administered together with other biological augmentation methods, such as mesenchymal stem cells or bioengineered scaffolds, so it is even more difficult to determine the contribution of PRP as such. The studies currently available are just case series treating disparate conditions in biomechanically very different joints, so the overall evidence is very limited and should be considered with caution. Moreover, follow-up evaluation is mainly at short term, and further studies are needed to determine the persistence of the reported good clinical outcome. In the near future, PRP will be more and more widely used in cartilage regenerative techniques, but, despite being safe, according to the present evidence, there is still no recommendation for using PRP as an augmentation during surgical procedures (Perdisa et al. 2014).

Considering PRP for conservative management, many preliminary studies have already confirmed the safety of PRP and also suggested promising clinical outcome, but also a gradual worsening over time. Two studies comparing PRP and saline for the treatment of knee cartilage pathology have been published (Patel et al. 2013; Smith 2016), and in both of them, it appeared clearly that the biological solution provided significantly better outcome. However, the real challenge comes when taking into account the randomized trials comparing PRP to viscosupplementation, which is currently the most adopted injective treatment worldwide. Actually it should be pointed out that each trial employed a different type of PRP. The study authored by Filardo et al. (2015), which had the longest follow-up evaluation of patients treated (1 year), reports no overall difference between PRP and HA in terms of clinical outcome and objective measurements at any follow-up evaluation. Therefore, in the case of more advanced signs of OA, PRP does not seem to be superior to viscosupplementation, a conclusion that was also reached in the study by Sanchez et al. (2012a), who were able to demonstrate just a better percentage of "responders" (patients with at least 50% reduction in pain) in PRP group compared to viscosupplementation. Surprisingly, Cerza et al. (2012) reported significantly better results for ACP even in grade III OA: worthy of consideration is the fact that they used different blood-derived products with evaluation limited up to 6 months of follow-up.

All the randomized trials deal with the application of PRP in patients affected by different stages of the disease, from chondropathy to severe OA. Therefore, no conclusions can be drawn about the possibility of applying this approach to a specific phase of cartilage degenerative pathology: subgroup analysis does not allow, in any of the trials published, sufficient statistical strength to provide a real clinical indication. What emerges can be considered just a "suggestion" to avoid the indiscriminate use of PRP, which seems to offer clinical benefits but cannot vet be considered a first-line treatment for knee cartilage pathology (Di Matteo et al. 2016). The situation is not clearer when looking at different joints, due both to the paucity of literature and the low scientific quality of studies published. Just one RCT has been published investigating the potential of PRP in treating osteochondral defects of the talar dome (Mei-Dan et al. 2012), but the small number of patients included and the follow-up topped at 7 weeks are major limitations. Hip OA has been the subject of four studies, two of them (Battaglia et al. 2013; Sanchez et al. 2012b) characterized by a low number of patients included (about 40); the most recent one included more than 100 patients (Dallari et al. 2016), divided into three treatment groups, to explore the efficacy of PRP against HA and also the potential of the combination of PRP+HA. The best performance, at 1-year evaluation, was achieved by the PRP group, and no contribution was due to the addition of HA to the blood derivative: in any case, this study is not sufficient to fully endorse the application of PRP in hip OA, also considering the controversial results reported by previous smaller trials. At present, the main problem related to PRP is the immense inter-product variability (Tschon et al. 2011). Cellularity is maybe the most debated aspect: leukocytes, monocytes, macrophages, and other cells are present in variable proportions according to the production method PRP was adopted from. In particular, the possible negative results of leukocytes have been investigated by in vitro studies that showed deleterious effects in terms of inflammation, cell migration, and matrix molecule formation/degradation (Assirelli et al. 2015; McCarrel et al. 2012). However, despite these findings, up to now, no clinical trial has shown the superiority of leukocyte-poor PRP with respect to leukocyte-rich formulation, thus further underlining the complexity of the in vivo environment and the pathogenic mechanisms of cartilage disease that can hardly be mirrored during in vitro trials. Beyond cellularity, several other variables have to be considered, such as the preparation procedures, activation methods, storage modalities, application protocols, and many other aspects that might not be of secondary importance for determining PRP properties and clinical efficacy. The number of variables to take into account is so high that it is likely to hypothesize that certain PRP formulations may be not ideal for the treatment of cartilage pathology with respect to others, and therefore research should move toward in identifying the best PRP features that could promote cartilage healing and joint homeostasis. At the current state of art, we are far from reaching this goal, and clinicians should remember that using an autologous blood-derived product is not, in any case, a guarantee of safety and efficacy. The number of names and acronyms encountered searching for studies on this biological treatment approach, such as PRP, PRGF, ACP, PL, etc., clearly represents the complexity of this field and explains the difficulties in the literature analysis, study comparison, and understanding of some contradictory results. There have been many attempts in the past to classify PRP products (Dohan Ehrenfest et al. 2009; Mautner et al. 2015), but the biological nature of this autologous approach does not help standardization, and, at present, there is no useful classification to be adopted in clinical practice. Even identifying macro-categories (Dohan Ehrenfest et al. 2010) has more theoretical value than a practical utility: the range of products that can be obtained is tremendously wide, and so the potential clinical usage, that it is difficult to put order in this complex and rapidly evolving field. However, new insights into the importance of technical and biological variables are now emerging (Salamanna et al. 2015; Marmotti et al. 2015) and will probably help to optimize platelet concentrates for this clinical application in the future. Until a new generation of products specifically conceived for intra-articular use is developed, physicians should be aware of the current options and their potential and limitations.

Conclusion

PRP is a novel technology, and the literature on cartilage treatment can rely only on a few high-level trials: there is an overall support on the safety of this biological approach but inconclusive evidence about its real efficacy in the management of cartilage pathology, especially when compared to more traditional treatment options. The improvement in this field will pass through the understanding of the mechanism of interaction between PRP and the etiopathogenetic/regenerative cartilage processes, as well as the identification of the best PRP features and application modalities, together with the identification of the patient and lesion type that may benefit from this treatment. Until further high-level studies will confirm the indications, results, and limitations suggested by the available literature, PRP should be considered as a second-line treatment and performed within controlled studies, rather than applied indiscriminately in the clinical practice.

References

- Assirelli E, Filardo G, Mariani E, Kon E, Roffi A, Vaccaro F, Marcacci M, Facchini A, Pulsatelli L (2015) Effect of two different preparations of platelet-rich plasma on synoviocytes. Knee Surg Sports Traumatol Arthrosc 23(9):2690–2703
- Battaglia M, Guaraldi F, Vannini F, Buscio T, Buda R, Galletti S, Giannini S (2011) Platelet-rich plasma (PRP) intra-articular ultrasound-guided injections as a possible treatment for hip osteoarthritis: a pilot study. Clin Exp Rheumatol 29(4):754. Epub 2011 Sep 1
- Battaglia M, Guaraldi F, Vannini F, Rossi G, Timoncini A, Buda R, Giannini S (2013) Efficacy of ultrasound-guided intra-articular injections of platelet-rich plasma versus hyaluronic acid for hip osteoarthritis. Orthopedics 36(12):e1501–e1508
- Boswell SG, Cole BJ, Sundman EA, Karas V, Fortier LA (2012) Platelet-rich plasma: a milieu of bioactive factors. Arthroscopy 28(3):429–439
- Braun HJ, Kim HJ, Chu CR, Dragoo JL (2014) The effect of platelet-rich plasma formulations and blood products on human synoviocytes: implications for intra-articular injury and therapy. Am J Sports Med 42(5):1204–1210
- Buda R, Vannini F, Cavallo M, Grigolo B, Cenacchi A, Giannini S (2010) Osteochondral lesions of the knee: a new one-step repair technique with bone-marrow-derived cells. J Bone Joint Surg Am 2:2–11
- Cerza F, Carni S, Carcangiu A, Di Vavo I, Schiavilla V, Pecora A, De Biasi G, Ciuffreda M (2012) Comparison between hyaluronic acid and platelet-rich plasma, intra-articular infiltration in the treatment of gonarthrosis. Am J Sports Med 40(12):2822–2827
- Cutolo M, Berenbaum F, Hochberg M, Punzi L, Reginster JY (2015) Commentary on recent therapeutic guidelines for osteoarthritis. Semin Arthritis Rheum 44(6):611–617
- Dallari D, Stagni C, Rani N, Sabbioni G, Pelotti P, Torricelli P, Tschon M, Giavaresi G (2016) Ultrasound-guided injection of platelet-rich plasma and hyaluronic acid, separately and in combination, for hip osteoarthritis: a randomized controlled study. Am J Sports Med 44(3):664–671
- Dhollander AA, De Neve F, Almqvist KF, Verdonk R, Lambrecht S, Elewaut D, Verbruggen G, Verdonk PC (2011) Autologous matrix-induced chondrogenesis combined with platelet-rich plasma gel: technical description and a five pilot patients report. Knee Surg Sports Traumatol Arthrosc 19(4):536–542
- Di Matteo B, Filardo G, Kon E, Marcacci M (2015) Platelet-rich plasma: evidence for the treatment of patellar and Achilles tendinopathy--a systematic review. Musculoskelet Surg 99(1):1–9
- Di Matteo B, Kon E, Filardo G (2016) Intra-articular platelet-rich plasma for the treatment of osteoarthritis. Ann Transl Med 4(3):2305–5839
- Dohan Ehrenfest DM, Rasmusson L, Albrektsson T (2009) Classification of platelet concentrates: from pure platelet-rich plasma (P-PRP) to leucocyte- and platelet-rich fibrin (L-PRF). Trends Biotechnol 27(3):158–167
- Dohan Ehrenfest DM, Bielecki T, Del Corso M, Inchingolo F, Sammartino G (2010) Shedding light in the controversial terminology for platelet-rich products: platelet-rich plasma (PRP), platelet-rich fibrin (PRF), platelet-leukocyte gel (PLG), preparation rich in growth factors (PRGF), classification and commercialism. J Biomed Mater Res A 95(4):1280–1282. doi:10.1002/jbm.a.32894. Epub 2010 Oct 5
- Filardo G, Di Matteo B, Di Martino A, Merli ML, Cenacchi A, Fornasari P, Marcacci M, Kon E (2015) Platelet-rich plasma intra-articular knee injections show no superiority versus viscosupplementation: a randomized controlled trial. Am J Sports Med 43(7):1575–1582
- Filardo G, Kon E, Longo UG, Madry H, Marchettini P, Marmotti A, Van Assche D, Zanon G, Peretti GM (2016) Non-surgical treatments for the management of early osteoarthritis. Knee Surg Sports Traumatol Arthrosc 24(6):1775–1785
- Forogh B, Mianehsaz E, Shoaee S, Ahadi T, Raissi GR, Sajadi S (2016) Effect of single injection of platelet-rich plasma in comparison with corticosteroid on knee osteoarthritis: a double-blind randomized clinical trial. J Sports Med Phys Fitness 56(7-8):901–908
- Giannini S, Buda R, Vannini F, Cavallo M, Grigolo B (2009) One-step bone marrow-derived cell transplantation in talar osteochondral lesions. Clin Orthop Relat Res 467(12):3307–3320
- Giannini S, Buda R, Cavallo M, Ruffilli A, Cenacchi A, Cavallo C, Vannini F (2010) Cartilage repair evolution in post-traumatic osteochondral lesions of the talus: from open field autologous chondrocyte to bone-marrow-derived cells transplantation. Injury 41(11):1196–1203
- Heijink A, Gomoll AH, Madry H, Drobnic M, Filardo G, Espregueira-Mendes J, Van Dijk CN (2012) Biomechanical considerations in the pathogenesis of osteoarthritis of the knee. Knee Surg Sports Traumatol Arthrosc 20(3):423–435
- Hunter DJ, Felson DT (2006) Osteoarthritis. BMJ 332(7542):639-642
- Jacobs JJ, Jevsevar DS, Brown GA (2014 Jun) Cummins DS AAOS Osteoarthritis Guideline: transparency and credibility. Arthroscopy 30(6):656–658. doi:10.1016/j.arthro.2014.03.002
- Kon E, Filardo G, Drobnic M, Madry H, Jelic M, van Dijk N, Della Villa S (2012) Non-surgical management of early knee osteoarthritis. Knee Surg Sports Traumatol Arthrosc 20(3):436–449
- Kon E, Filardo G, Di Matteo B, Marcacci M (2013) PRP for the treatment of cartilage pathology. Open Orthop J 7:120–128
- Lories RJ (2008) Joint homeostasis, restoration, and remodeling in osteoarthritis. Best Pract Res Clin Rheumatol 22(2):209–220
- Lotz MK, Carames B (2011) Autophagy and cartilage homeostasis mechanisms in joint health, aging and OA. Nat Rev Rheumatol 7(10):579–587
- Marmotti A, Rossi R, Castoldi F, Roveda E, Michielon G, Peretti GM (2015) PRP and articular cartilage: a clinical update. Biomed Res Int 542502(10):5
- Mautner K, Malanga GA, Smith J, Shiple B, Ibrahim V, Sampson S, Bowen JE (2015) A call for a standard classification system for future biologic research: the rationale for new PRP nomenclature. PM R 7(4 Suppl):005
- McCarrel TM, Minas T, Fortier LA (2012) Optimization of leukocyte concentration in platelet-rich plasma for the treatment of tendinopathy. J Bone Joint Surg Am 94(19):1–8
- Mei-Dan O, Carmont MR, Laver L, Mann G, Maffulli N, Nyska M (2012) Platelet-rich plasma or hyaluronate in the management of osteochondral lesions of the talus. Am J Sports Med 40(3):534–541
- Patel S, Dhillon MS, Aggarwal S, Marwaha N, Jain A (2013) Treatment with platelet-rich plasma is more effective than placebo for knee osteoarthritis: a prospective, double-blind, randomized trial. Am J Sports Med 41(2):356–364
- Perdisa F, Filardo G, Di Matteo B, Marcacci M, Kon E (2014) Platelet rich plasma: a valid augmentation for cartilage scaffolds? A systematic review. Histol Histopathol 29(7):805–814
- Reginster JY, Cooper C, Hochberg M, Pelletier JP, Rizzoli R, Kanis J, Abadie E, Maheu E, Brandi ML, Devogelaer JP, Branco J, Herrero-Beaumont G, D'Hooghe P, Bruyere O (2015) Comments on the discordant recommendations for the use of symptomatic slow-acting drugs in knee osteoarthritis. Curr Med Res Opin 31(5):1041–1045
- Salamanna F, Veronesi F, Maglio M, Della Bella E, Sartori M, Fini M (2015) New and emerging strategies in platelet-rich plasma application in musculoskeletal regenerative procedures: general overview on still open questions and outlook. Biomed Res Int 846045(10):5
- Sanchez M, Fiz N, Azofra J, Usabiaga J, Aduriz Recalde E, Garcia Gutierrez A, Albillos J, Garate R, Aguirre JJ, Padilla S, Orive G, Anitua E (2012a) A randomized clinical trial evaluating plasma rich in growth factors (PRGF-Endoret) versus hyaluronic acid in the short-term treatment of symptomatic knee osteoarthritis. Arthroscopy 28(8):1070–1078
- Sanchez M, Guadilla J, Fiz N, Andia I (2012b) Ultrasound-guided platelet-rich plasma injections for the treatment of osteoarthritis of the hip. Rheumatology 51(1):144–150
- Siclari A, Mascaro G, Gentili C, Cancedda R, Boux E (2012) A cell-free scaffold-based cartilage repair provides improved function hyaline-like repair at one year. Clin Orthop Relat Res 470(3):910–919
- Smith PA (2016) Intra-articular autologous conditioned plasma injections provide safe and efficacious treatment for knee osteoarthritis: an FDA-sanctioned, randomized, double-blind, placebo-controlled clinical trial. Am J Sports Med 44(4):884–891

- Tschon M, Fini M, Giardino R, Filardo G, Dallari D, Torricelli P, Martini L, Giavaresi G, Kon E, Maltarello MC, Nicolini A, Carpi A (2011) Lights and shadows concerning platelet products for musculoskeletal regeneration. Front Biosci 3:96–107
- Vannini F, Di Matteo B, Filardo G (2015) Platelet-rich plasma to treat ankle cartilage pathology from translational potential to clinical evidence: a systematic review. J Exp Orthop 2(1):015–0019
- Xie X, Zhang C, Tuan RS (2014) Biology of platelet-rich plasma and its clinical application in cartilage repair. Arthritis Res Ther 16(1):204

Subchondral Bone in Articular Cartilage Regeneration

8

Arndt F. Schilling

Abstract

Subchondral sclerosis is one of the hallmark findings of osteoarthritis (OA) and has long been discussed as one of its causes. Indeed, the changes in the subchondral bone often precede cartilage destruction in the development of the disease. Integration of the so far published data including in vitro, in vivo, and mathematical work suggests a critical role for this tissue in nutrition and oxygen supply to the articular cartilage, which may become even more critical in energy demanding processes of healing and regeneration.

Indeed, the success of current predictive diagnostics like specialized MRI techniques and scintigraphy as well as successful regenerative clinical therapies like microfracturing, AMIC, or NAMIC can be better explained if the subchondral bone is taken into the account as supply route for the cartilage.

Consequently, subchondral bone has to be included into the diagnostic and therapeutic concepts aiming to regenerate lost or damaged cartilage for advanced diagnosis and treatment of OA.

8.1 The Multi-tissue Human Joint

The mammalian locomotion system has evolved for movement at very low energy costs (Pontzer and Kamilar 2009). In the human bipedal locomotion, relatively high forces have to be transmitted from one moving bone to the next. The hip joint is already loaded with three times bodyweight when walking (Bergmann et al. 2016), and much higher forces (>9× bodyweight) are seen in sports (McNitt-Gray et al.

A.F. Schilling

Klinik für Unfallchirurgie, Orthopädie und Plastische Chirurgie, Universitätsmedizin Göttingen, Robert-Koch-Str. 40, 37075 Göttingen, Germany e-mail: arndt.schilling@med.uni-goettingen.de

[©] Springer International Publishing AG 2017

S. Grässel, A. Aszodi (eds.), Cartilage, DOI 10.1007/978-3-319-53316-2_8

1993). A multilayered tissue structure that is constantly adapted to its current use has evolved to meet these demanding requirements. Far away from the joint, the diaphyses of the long bones resemble tubes with a strong cortical ring, which encases the embedded bone marrow, reflecting their primary loading in the direction of the long axis. Closer to the joints starting in the metaphyses, the major muscles insert, and the pattern of loading changes to multidirectional, which is reflected by a change in the bone structure to trabecular bone with less cortical casing. On top of the metaphysis resides the epiphysis. It develops from a secondary ossification center and consists of trabecular bone, blending into calcified cartilage with its cover consisting of hyaline cartilage (Fig. 8.1).

Between the calcified cartilage and the uncalcified cartilage is the so-called tidemark, a zone of distinct glycoprotein profile marking the zone of mineralization (Lyons et al. 2005). The combination of calcified cartilage and underlying trabecular bone is referred to as "subchondral bone" (Fig. 8.1). Interestingly, impulsive forces that act on the joint during movement are mainly attenuated in this bone and not in the cartilage, although the bone is roughly ten times stiffer (Radin and Paul 1970). The main function of articular cartilage is therefore reducing friction. The excellent tribological properties of hyaline cartilage make it possible to move the articulating bones relative to each other while transducing the high loads between them. This allows energy efficient movements with multiple degrees of freedom while preserving the cartilage (Chen et al. 2013).



Fig. 8.1 Histological section of the osteochondral area

8.2 Cartilage Injury and Lack of Regeneration

Injury to or degradation of the articular cartilage consequently leads to breakdown of this sophisticated, concerted locomotion system. Loss of cartilage results in increased friction between the articulating bones which leads to inflammation and thereby loss of more cartilage completing the vicious circle (Glyn-Jones et al. 2015). The patients experience loss of mobility and pain. There is a general consensus that cartilage injuries usually do not heal. Surprisingly, it has been shown that chondrocytes have the ability to generate new cartilage tissue if the environment is permitting this (Hoenig et al. 2011, 2013). Furthermore, the development of osteophytes with full cartilage cover in joint disease shows a general ability of forming articular cartilage de novo (Gelse et al. 2003). So why is there usually no regeneration of cartilage defects in vivo? A possible road to the answer for this question may lie in the observation that tissue healing is associated with a high demand for energy that needs to be satisfied. Indeed, tissues with a rich blood supply (e.g., liver, oral cavity) heal well, while chronic wounds are usually found in patients with impaired blood supply to the wounded areas (peripheral vascular diseases, diabetics, smokers, burn patients) (Varricchi et al. 2015; Wu et al. 2007).

8.3 Nutrition of Cartilage: Synovial Nutrition (SyN) vs. Subchondral Nutrition (SuN)

Healthy articular cartilage does not contain blood vessels. It consists only of chondrocytes embedded in extracellular matrix produced by these chondrocytes (Hoenig et al. 2013). Consequently, nutrition and oxygen for the metabolism of the cells inside the cartilage must be maintained by diffusion, and the diffusing nutrients have to reach the cartilage somehow. Possible routes for this traffic of nutrients would be from the synovia, supplied by the vessels into the joint capsule (synovial nutrition: SyN), or through the subchondral bone supplied by the epiphyseal plexus (subchondral nutrition SuN) or both (Ingelmark and Saaf 1948) (Figs. 8.2a,b). The notion that cartilage may be supplied by the synovial fluid alone (SyN) is mainly based on experiments with young and adult rabbit joints (Hodge and McKibbin 1969). These experiments clearly show that in young rabbits, a route through the subchondral bone exists which is however completely closed in the adult animals. This lead the authors to the conclusion that nutrition of the cartilage in adult animals can only be mediated through the synovial fluid.

In humans however, there is intraosseous and extraosseous blood supply to the distal femoral condyles (Reddy and Frederick 1998). Holmdahl and Ingelmark described two types of channels connecting the medullary cavity through the sub-chondral bone with the cartilage: (1) "ampulla-like" with a diameter of ca. 40 μ m and (2) "canal-like" or "dendritic" with a diameter of around 15 μ m (Holmdahl and Ingelmark 1950). For human joints, it was demonstrated that a fluorescent dye instilled into the marrow space can be found in the cartilage after 16 h and completely penetrates the cartilage after 48 h (Greenwald and Haynes 1969). In a



Fig. 8.2 (a) Schematic representation of possible routes of nutrition (b) and disposal (c) for the cartilage; *SyN* synovial nutrition, *SuN* subchondral nutrition, *SuD* subchondral disposal, *SyD* synovial disposal; (d) SuN-SyD Hypothesis

modeling study, Zhou et al. showed that a favorable oxygen concentration around 5% throughout the cartilage can only be maintained by diffusion, if the lower part of cartilage is supplied by the bone route (Zhou et al. 2004, 2008). Indeed, gadolinium (Gd(DTPA)2) penetrated into cartilage from the articular surface, when injected intra-articularly, and from both sides when injected intravenously. Furthermore, intravenous injection showed a shorter overall penetration time (Bashir et al. 1997), suggesting a more efficient supply through vessels of the subchondral bone (SuN). The vascularity of the subchondral bone was found to be age dependent and location dependent (Lane et al. 1977); more vessels are present in the more loaded areas of the articular surfaces, which suggests a better supply of the cells that experience a higher load assuming subchondral nutrition (SuN). The density of cells in the cartilage however increases with distance from the subchondral bone (Stockwell 1971). If the cells are viewed separately, this could be interpreted as an indication that the nutrition at the surface is higher and therefore there are more cells in this area. The cell density of a tissue is determined by relative ratios of cells and ECM assuming that either a high cell density means that there are many cells or that there is little ECM. However,

in taking the production of the extracellular matrix (ECM) by the cells into account, a different picture emerges: the expression of collagen II by chondrocytes rises with rising oxygen tension, nearly doubling from 0% oxygen to 20% oxygen (Grimshaw and Mason 2001). If we assume a SuN route, the oxygen tension would be high close to the subchondral bone; thus, the cells would produce more collagen II in this area and therefore stand further apart leading to lower cell density at the subchondral bone border. With rising distance from the subchondral bone, oxygen tension would drop, eventually leading to hypoxia and decrease of collagen II production, which would lead to downregulation of matrix synthesis and therefore indirectly increase the cell density at the joint surface. This is also in accordance with a higher collagen II content in smaller vertebral disks (Boubriak et al. 2013). Interestingly, while in different species the thickness of articular cartilage in the knee ranges over a factor of nearly $50 \times (0.05 - 2.3 \text{ mm})$, the number of cells over a square millimeter of subchondral bone is surprisingly constant $(25,500 \pm 8800 \text{ cells/mm}^2)$, suggesting that a given surface area of subchondral bone only has the ability to supply a given amount of cells. If O₂ would be transported by the synovial fluid (SyN), there would be no obvious reason for such a relationship. Furthermore, more synovial fluid should increase the oxygen tension in the joint. Surprisingly, however, the amount of synovial fluid in the joint is inversely related with its oxygen tension (Richman et al. 1981). This observation of lower levels of oxygen in joints with more synovial fluid can hardly be explained with changes in influx. Increased influx of freely diffusing fluid from the circulation should bring in more oxygen, not less. There are two possible ways to explain these findings. First, patients with higher volumes of synovial fluid may have an even higher metabolism of the cells in the joint (e.g., due to inflammatory processes), with higher counts of leucocytes in the joint. There was however no correlation between white blood cell counts and oxygen consumption in these experiments (Richman et al. 1981). Second, a diminished disposal of synovial fluid may be responsible.

8.4 Waste Disposal in the Joint: Subchondral Disposal (SuD) vs. Synovial Disposal (SyD)

In this case, the increase in volume would be due to longer clearance time bringing the synovial fluid in contact with the oxygen consuming cells for a longer time thereby explaining why pO_2 is lower in patients with more fluid in the joint. As the cells in the joint have a metabolism (although a slow one), the waste products have to be disposed of. Indeed, the published data suggests a weak positive correlation between the amount of synovial fluid and CO_2 waste, supporting a role of waste disposal from the joint and synovial efflux (Richman et al. 1981). In healthy joints, there is efficient clearance of the synovial fluid with half-lives of intra-articularly injected drugs in the range of hours (Owen et al. 1994). This clearance is dependent on particle size and health status of the joint (Pradal et al. 2016). For clearance routes again, there are the same two possible options conceivable: through the subchondral bone (subchondral disposal (SuD)) or through the synovial capsule (synovial disposal (SyD)) (Fig. 8.2c). Ingelmark and Saaf already showed in 1948 that micrometer-sized particles injected into the joint space can be found after several loading cycles of the joint in the synovial membrane but not in the cartilage (Ingelmark and Saaf 1948). However, when they injected the particles into the bone marrow, they could find them in different zones of the cartilage, dependent on the movement of the joint, suggesting a transport mechanism from the subchondral bone through the cartilage into the synovia. It is hardly conceivable how other waste products should diffuse against this flow. Further favoring a possible role in waste disposal, the human synovium has a relatively large surface area (277 cm² in the knee) with capillaries that form delicate anastomosing loops and a general histological appearance resembling rather mesenterial tissue than pleura (Davies 1946).

Taken together, while there is currently a common belief in a combined role for synovial nutrition and waste disposal through dialysis of serum in the joint capsule (SyN-SyD), the published data allow to hypothesize a SuN-SyD nutrition/disposal pattern: flow of fluid and nutrients in healthy joints from the epiphyseal plexus, through the cartilage, and waste disposal by resorption through the large surface of the mesentery-like synovial capsule (Fig. 8.2d).

8.5 Changes of Subchondral Bone in Age and Joint Disease

If we assume a SuN-SyD nutrition/disposal, pathological changes in the subchondral bone and/or synovial capsule should lead to distinct changes in the joints leading to the same outcome: degradation of the cartilage. Synovitis is only present in 50% of patients with early OA but in nearly all patients with late-stage OA (Haywood et al. 2003). So, in half of the patients, the disease starts either in the subchondral bone or directly in the cartilage. The resulting cartilage degradation and ensuing inflammation then probably lead to synovitis in the process. Although there are a number of pharmaceutical approaches to counteract inflammation in joint disease, so far this approach was only effective against pain, but did not reverse the disease (for review see Philp et al. 2016). A possible reason for this might be lack of stratification of these patients, as probably only 50% of patients with leading inflammation would benefit from the treatment and the combination with patients without this indication would presumably generate impaired statistical power leading to nonsignificant results.

Changes in subchondral bone as a cause for rather than an effect of cartilage degradation have been discussed at least since 1970 (Radin et al. 1972; Radin and Paul 1970). In healthy joints, the thickness of the calcified cartilage decreases with age from 200 μ m at the age of 20 to 100 μ m at the age of 90, while the number of tidemarks increases from one to three in the same time frame (Lane and Bullough 1980). A common finding of joint injury and OA is sometimes a painful "bone bruise" originating from a localized increase in water content of a trabecular region detectable in MRI (Fig. 8.3).

It is commonly explained as intratrabecular bleeding of micro-injuries of the subchondral bone. Considering a Sun-SyD pattern, this phenomenon could also be

Fig. 8.3 Bone bruise



a posttraumatic congestion of the micro-channels that supply the SuN route, leading to buildup of fluid before the block. This would also explain the associated pain as induced by increased pressure in the affected bone. Indeed, the penetration of fluorescent dye from the marrow significantly reduces in a specimen with arthritic changes of the joint when compared to normal controls (Greenwald and Haynes 1969). Subchondral sclerosis, a thickening of the subchondral zone which is a hallmark feature of osteoarthritis, would have a similar effect. Indeed, there is an inverse correlation between OA and osteoporosis (Bergink et al. 2005). Microarray gene expression analysis of osteoarthritic bone suggests altered bone remodeling through WNT, TGF-beta, and BMP signaling (Hopwood et al. 2005, 2007) leading to increased bone formation. The bone tissue mineralization in the subchondral bone is decreased in OA, most probably reflecting this increased buildup of new bone (Cox et al. 2012). Furthermore, in osteoarthritic men, serum RANKL levels drop with age (Findlay et al. 2008) possibly leading to decreased bone resorption in the subchondral bone.

8.6 Regeneration of Cartilage and the Role of Subchondral Bone

If indeed the subchondral bone plays an important role in the pathophysiology of cartilage degeneration, it may also play a role in cartilage regeneration. Currently, different treatment options came up, some of which influence the bone and therefore presumably also subchondral bone.

8.6.1 Pharmacotherapy of Bone Turnover

Subchondral bone turnover is increased 20-fold in OA (Bailey et al. 2004) which can be visualized by MRI/99^{mTc}-DPD-SPECT/CT (Maas et al. 2015). The progression of joint space narrowing can be predicted by scintigraphy of the subchondral bone (Dieppe et al. 1993). Blocking of bone turnover by inhibition of osteoclasts

can prevent the progression of arthritic changes in rats (Hayami et al. 2004). There are a number of available drugs that can influence this pathway in clinical trials (Karsdal et al. 2008), but so far the results mainly point to a positive effect on osteoarthritic pain (Varenna et al. 2015; Davis et al. 2013). This may in part be due to the slow remodeling of the bone. Laslett et al. report a first positive trend of bisphosphonate treatment on joint space narrowing after 4 years of treatment (Laslett et al. 2014). So it may take time until positive effects of inhibition of remodeling become visible.

8.6.2 Changing the Biomechanics Leading to Unloading of the Subchondral Bone

Lifestyle Modification Weight loss in obese patients can improve the symptoms of osteoarthritis (Gudbergsen et al. 2012). It can be hypothesized that weight loss and the subsequent decrease of biomechanical forces in the joints may lead to changes in the subchondral bone. As weight loss in these patients is associated with a multitude of beneficial effects and there is no data on subchondral bone in these patients available, this is only speculation.

Surgery There are different surgical techniques for changing the biomechanics of the joint in OA. It has been shown that open-wedge osteotomy changes the subchondral bone structure (Ziegler et al. 2015). The change of the alignment of the joint through wedge osteotomy or reorientation of the acetabulum has been shown effective in the treatment of different forms of OA (Spahn et al. 2013; Hartig-Andreasen et al. 2012). Furthermore, open-wedge osteotomy (Kesemenli et al. 2013) has also been shown to resolve painful bone marrow edema, supporting the Sun-SyD hypothesis.

8.6.3 Reopening the SuN Pathway

The most commonly applied surgical procedure in early OA is microfracturing of the subchondral bone also called bone marrow stimulation technique (BST) (Min et al. 2013) (for further details, see also Chap. 5). It is sometimes combined with the use of biomaterials (AMIC, NAMIC) (Benthien and Behrens 2015; Anders et al. 2013). The conceptual framework underlying this procedure assumes that by opening the barrier to the bone marrow, this allows mesenchymal stem cells to repopulate the cartilage and repair the tissue (Min et al. 2013; Shapiro et al. 1993). While this procedure has very high success rate in a first-line treatment, it does not seem to perform as well in a salvage situation (Truong et al. 2014) although the amount of recruited stem cells stays the same. Differences in subchondral bone repair following first-line and salvage procedure may explain these clinical findings, if reopening of the SuN route is assumed as underlying mechanism. The scar tissue that is formed in the first-line procedure would need a different microfracture

technique than the original subchondral bone. This is similar to scar surgery, where the results of scar incision are worse than the results of scar excision (Davids et al. 2016).

There are at least two recent reports of cartilage regeneration after complete cartilage loss. First, using a combination of high tibial wedge osteotomy together with subchondral drilling and stem cell therapy, it was possible to regenerate cartilage from grade 4 bone-on-bone lesions (Saw et al. 2015). Second, even more promising, temporary surgical joint distraction alone leads to symptomatic and structural improvement in end-stage knee osteoarthritis (Wiegant et al. 2013) which was comparable to the results of open-wedge osteotomy (van der Woude et al. 2016). Both cases can be explained by structural optimization of the subchondral bone allowing SuD. However, this hypothesis will have to be further tested in the coming years.

Conclusion

Regeneration of lost cartilage is possible and is clinically performed worldwide. A prerequisite for eventual cartilage healing seems to be a favorable environment especially of the subchondral bone, but also of the synovial capsule. Many findings can be explained, if a physiologic flow of nutrients and oxygen from the subchondral bone through the cartilage into the synovial capsule is hypothesized. Closer inspection of the physiology and pathology of the delicate subchondral structures and their connection with the overlying cartilage may lead to better individualized treatments for patients with cartilage defects.

Acknowledgments AFS wants to thank Dr. Wolfgang Schilling for critical discussion. This work is in part supported by a grant of the German research foundation to AFS (FOR 2407; Schi 857/9-1).

References

- Anders S, Volz M, Frick H, Gellissen J (2013) A randomized, controlled trial comparing autologous matrix-induced chondrogenesis (AMIC(R)) to microfracture: analysis of 1- and 2-year follow-up data of 2 centers. Open Orthop J 7:133–143. doi:10.2174/1874325001307010133
- Bailey AJ, Mansell JP, Sims TJ, Banse X (2004) Biochemical and mechanical properties of subchondral bone in osteoarthritis. Biorheology 41(3-4):349–358
- Bashir A, Gray ML, Boutin RD, Burstein D (1997) Glycosaminoglycan in articular cartilage: in vivo assessment with delayed Gd(DTPA)(2-)-enhanced MR imaging. Radiology 205(2):551– 558. doi:10.1148/radiology.205.2.9356644
- Benthien JP, Behrens P (2015) Nanofractured autologous matrix induced chondrogenesis (NAMIC(c))--further development of collagen membrane aided chondrogenesis combined with subchondral needling: a technical note. Knee 22(5):411–415. doi:10.1016/j. knee.2015.06.010
- Bergink AP, Uitterlinden AG, Van Leeuwen JP, Hofman A, Verhaar JA, Pols HA (2005) Bone mineral density and vertebral fracture history are associated with incident and progressive radiographic knee osteoarthritis in elderly men and women: the Rotterdam Study. Bone 37(4):446–456. doi:10.1016/j.bone.2005.05.001

- Bergmann G, Bender A, Dymke J, Duda G, Damm P (2016) Standardized loads acting in hip implants. PLoS One 11(5):e0155612. doi:10.1371/journal.pone.0155612
- Boubriak OA, Watson N, Sivan SS, Stubbens N, Urban JP (2013) Factors regulating viable cell density in the intervertebral disc: blood supply in relation to disc height. J Anat 222(3):341– 348. doi:10.1111/joa.12022
- Chen C, Tambe DT, Deng L, Yang L (2013) Biomechanical properties and mechanobiology of the articular chondrocyte. Am J Physiol Cell Physiol 305(12):C1202–C1208. doi:10.1152/ajpcell.00242.2013
- Cox LG, van Donkelaar CC, van Rietbergen B, Emans PJ, Ito K (2012) Decreased bone tissue mineralization can partly explain subchondral sclerosis observed in osteoarthritis. Bone 50(5):1152–1161. doi:10.1016/j.bone.2012.01.024
- Davids JR, Diaz K, Leba TB, Adams S, Westberry DE, Bagley AM (2016) Outcomes of cutaneous scar revision during surgical implant removal in children with cerebral palsy. J Bone Joint Surg Am 98(16):1351–1358. doi:10.2106/jbjs.15.01418
- Davies DV (1946) Synovial membrane and synovial fluid of joints. Lancet 2(6432):815-819
- Davis AJ, Smith TO, Hing CB, Sofat N (2013) Are bisphosphonates effective in the treatment of osteoarthritis pain? A meta-analysis and systematic review. PLoS One 8(9):e72714. doi:10.1371/journal.pone.0072714
- Dieppe P, Cushnaghan J, Young P, Kirwan J (1993) Prediction of the progression of joint space narrowing in osteoarthritis of the knee by bone scintigraphy. Ann Rheum Dis 52(8):557–563
- Findlay D, Chehade M, Tsangari H, Neale S, Hay S, Hopwood B, Pannach S, O'Loughlin P, Fazzalari N (2008) Circulating RANKL is inversely related to RANKL mRNA levels in bone in osteoarthritic males. Arthritis Res Ther 10(1):R2. doi:10.1186/ar2348
- Gelse K, Soder S, Eger W, Diemtar T, Aigner T (2003) Osteophyte development--molecular characterization of differentiation stages. Osteoarthritis Cartilage 11(2):141–148
- Glyn-Jones S, Palmer AJ, Agricola R, Price AJ, Vincent TL, Weinans H, Carr AJ (2015) Osteoarthritis. Lancet 386(9991):376–387. doi:10.1016/s0140-6736(14)60802-3
- Greenwald AS, Haynes DW (1969) A pathway for nutrients from the medullary cavity to the articular cartilage of the human femoral head. J Bone Joint Surg Br 51(4):747–753
- Grimshaw MJ, Mason RM (2001) Modulation of bovine articular chondrocyte gene expression in vitro by oxygen tension. Osteoarthritis Cartilage 9(4):357–364. doi:10.1053/joca.2000.0396
- Gudbergsen H, Boesen M, Lohmander LS, Christensen R, Henriksen M, Bartels EM, Christensen P, Rindel L, Aaboe J, Danneskiold-Samsoe B, Riecke BF, Bliddal H (2012) Weight loss is effective for symptomatic relief in obese subjects with knee osteoarthritis independently of joint damage severity assessed by high-field MRI and radiography. Osteoarthritis Cartilage 20(6):495–502. doi:10.1016/j.joca.2012.02.639
- Hartig-Andreasen C, Troelsen A, Thillemann TM, Soballe K (2012) What factors predict failure 4 to 12 years after periacetabular osteotomy? Clin Orthop Relat Res 470(11):2978–2987. doi:10.1007/s11999-012-2386-4
- Hayami T, Pickarski M, Wesolowski GA, McLane J, Bone A, Destefano J, Rodan GA, Duong LT (2004) The role of subchondral bone remodeling in osteoarthritis: reduction of cartilage degeneration and prevention of osteophyte formation by alendronate in the rat anterior cruciate ligament transection model. Arthritis Rheum 50(4):1193–1206. doi:10.1002/art.20124
- Haywood L, McWilliams DF, Pearson CI, Gill SE, Ganesan A, Wilson D, Walsh DA (2003) Inflammation and angiogenesis in osteoarthritis. Arthritis Rheum 48(8):2173–2177. doi:10.1002/art.11094
- Hodge JA, McKibbin B (1969) The nutrition of mature and immature cartilage in rabbits. An autoradiographic study. J Bone Joint Surg Br 51(1):140–147
- Hoenig E, Winkler T, Mielke G, Paetzold H, Schuettler D, Goepfert C, Machens HG, Morlock MM, Schilling AF (2011) High amplitude direct compressive strain enhances mechanical properties of scaffold-free tissue-engineered cartilage. Tissue Eng Part A 17(9-10):1401–1411. doi:10.1089/ten.TEA.2010.0395
- Hoenig E, Leicht U, Winkler T, Mielke G, Beck K, Peters F, Schilling AF, Morlock MM (2013) Mechanical properties of native and tissue-engineered cartilage depend on carrier permeabil-

ity: a bioreactor study. Tissue Eng Part A 19(13-14):1534–1542. doi:10.1089/ten. TEA.2012.0538

- Holmdahl DE, Ingelmark BE (1950) The contact between the articular cartilage and the medullary cavities of the bone. Acta Orthop Scand 20(2):156–165
- Hopwood B, Gronthos S, Kuliwaba JS, Robey PG, Findlay DM, Fazzalari NL (2005) Identification of differentially expressed genes between osteoarthritic and normal trabecular bone from the intertrochanteric region of the proximal femur using cDNA microarray analysis. Bone 36(4):635–644. doi:10.1016/j.bone.2005.02.003
- Hopwood B, Tsykin A, Findlay DM, Fazzalari NL (2007) Microarray gene expression profiling of osteoarthritic bone suggests altered bone remodelling, WNT and transforming growth factorbeta/bone morphogenic protein signalling. Arthritis Res Ther 9(5):R100. doi:10.1186/ar2301
- Ingelmark BE, Saaf J (1948) Not available. Acta Orthop Scand 17(3-4):303-357
- Karsdal MA, Leeming DJ, Dam EB, Henriksen K, Alexandersen P, Pastoureau P, Altman RD, Christiansen C (2008) Should subchondral bone turnover be targeted when treating osteoarthritis? Osteoarthritis Cartilage 16(6):638–646. doi:10.1016/j.joca.2008.01.014
- Kesemenli CC, Memisoglu K, Muezzinoglu US, Akansel G (2013) Treatment for painful bone marrow edema by open wedge tibial osteotomy. Eur J Orthop Surg Traumatol 23(7):825–829. doi:10.1007/s00590-012-1077-x
- Lane LB, Bullough PG (1980) Age-related changes in the thickness of the calcified zone and the number of tidemarks in adult human articular cartilage. J Bone Joint Surg Br 62(3):372–375
- Lane LB, Villacin A, Bullough PG (1977) The vascularity and remodelling of subchondrial bone and calcified cartilage in adult human femoral and humeral heads. An age- and stress-related phenomenon. J Bone Joint Surg Br 59(3):272–278
- Laslett LL, Kingsbury SR, Hensor EM, Bowes MA, Conaghan PG (2014) Effect of bisphosphonate use in patients with symptomatic and radiographic knee osteoarthritis: data from the Osteoarthritis Initiative. Ann Rheum Dis 73(5):824–830. doi:10.1136/ annrheumdis-2012-202989
- Lyons TJ, Stoddart RW, McClure SF, McClure J (2005) The tidemark of the chondro-osseous junction of the normal human knee joint. J Mol Histol 36(3):207–215. doi:10.1007/ s10735-005-3283-x
- Maas O, Joseph GB, Sommer G, Wild D, Kretzschmar M (2015) Association between cartilage degeneration and subchondral bone remodeling in patients with knee osteoarthritis comparing MRI and (99m)Tc-DPD-SPECT/CT. Osteoarthritis Cartilage 23(10):1713–1720. doi:10.1016/j. joca.2015.05.014
- McNitt-Gray JL, Yokoi T, Millward C (1993) Landing strategy adjustments made by female gymnasts in response to drop height and mat composition. J Appl Biomech 9(3):173–190. doi:10.1123/jab.9.3.173
- Min BH, Choi WH, Lee YS, Park SR, Choi BH, Kim YJ, Jin LH, Yoon JH (2013) Effect of different bone marrow stimulation techniques (BSTs) on MSCs mobilization. J Orthop Res 31(11):1814–1819. doi:10.1002/jor.22380
- Owen SG, Francis HW, Roberts MS (1994) Disappearance kinetics of solutes from synovial fluid after intra-articular injection. Br J Clin Pharmacol 38(4):349–355
- Philp AM, Davis ET, Jones SW (2016) Developing anti-inflammatory therapeutics for patients with osteoarthritis. Rheumatology (Oxford). doi:10.1093/rheumatology/kew278
- Pontzer H, Kamilar JM (2009) Great ranging associated with greater reproductive investment in mammals. Proc Natl Acad Sci U S A 106(1):192–196. doi:10.1073/pnas.0806105106
- Pradal J, Maudens P, Gabay C, Seemayer CA, Jordan O, Allemann E (2016) Effect of particle size on the biodistribution of nano- and microparticles following intra-articular injection in mice. Int J Pharm 498(1-2):119–129. doi:10.1016/j.ijpharm.2015.12.015
- Radin EL, Paul IL (1970) Does cartilage compliance reduce skeletal impact loads? The relative force-attenuating properties of articular cartilage, synovial fluid, periarticular soft tissues and bone. Arthritis Rheum 13(2):139–144
- Radin EL, Paul IL, Rose RM (1972) Role of mechanical factors in pathogenesis of primary osteoarthritis. Lancet 1(7749):519–522

- Reddy AS, Frederick RW (1998) Evaluation of the intraosseous and extraosseous blood supply to the distal femoral condyles. Am J Sports Med 26(3):415–419
- Richman AI, Su EY, Ho G Jr (1981) Reciprocal relationship of synovial fluid volume and oxygen tension. Arthritis Rheum 24(5):701–705
- Saw KY, Anz A, Jee CS, Ng RC, Mohtarrudin N, Ragavanaidu K (2015) High tibial osteotomy in combination with chondrogenesis after stem cell therapy: a histologic report of 8 cases. Arthroscopy 31(10):1909–1920. doi:10.1016/j.arthro.2015.03.038
- Shapiro F, Koide S, Glimcher MJ (1993) Cell origin and differentiation in the repair of fullthickness defects of articular cartilage. J Bone Joint Surg Am 75(4):532–553
- Spahn G, Hofmann GO, von Engelhardt LV, Li M, Neubauer H, Klinger HM (2013) The impact of a high tibial valgus osteotomy and unicondylar medial arthroplasty on the treatment for knee osteoarthritis: a meta-analysis. Knee Surg Sports Traumatol Arthrosc 21(1):96–112. doi:10.1007/s00167-011-1751-2
- Stockwell RA (1971) The interrelationship of cell density and cartilage thickness in mammalian articular cartilage. J Anat 109(Pt 3):411–421
- Truong MD, Chung JY, Kim YJ, Jin LH, Kim BJ, Choi BH, Min BH (2014) Histomorphochemical comparison of microfracture as a first-line and a salvage procedure: is microfracture still a viable option for knee cartilage repair in a salvage situation? J Orthop Res 32(6):802–810. doi:10.1002/jor.22592
- Varenna M, Zucchi F, Failoni S, Becciolini A, Berruto M (2015) Intravenous neridronate in the treatment of acute painful knee osteoarthritis: a randomized controlled study. Rheumatology (Oxford) 54(10):1826–1832. doi:10.1093/rheumatology/kev123
- Varricchi G, Granata F, Loffredo S, Genovese A, Marone G (2015) Angiogenesis and lymphangiogenesis in inflammatory skin disorders. J Am Acad Dermatol 73(1):144–153. doi:10.1016/j. jaad.2015.03.041
- Wiegant K, van Roermund PM, Intema F, Cotofana S, Eckstein F, Mastbergen SC, Lafeber FP (2013) Sustained clinical and structural benefit after joint distraction in the treatment of severe knee osteoarthritis. Osteoarthritis Cartilage 21(11):1660–1667. doi:10.1016/j.joca.2013.08.006
- van der Woude JA, Wiegant K, van Heerwaarden RJ, Spruijt S, van Roermund PM, Custers RJ, Mastbergen SC, Lafeber FP (2016) Knee joint distraction compared with high tibial osteotomy: a randomized controlled trial. Knee Surg Sports Traumatol Arthrosc. doi:10.1007/ s00167-016-4131-0
- Wu SC, Driver VR, Wrobel JS, Armstrong DG (2007) Foot ulcers in the diabetic patient, prevention and treatment. Vasc Health Risk Manag 3(1):65–76
- Zhou S, Cui Z, Urban JP (2004) Factors influencing the oxygen concentration gradient from the synovial surface of articular cartilage to the cartilage-bone interface: a modeling study. Arthritis Rheum 50(12):3915–3924. doi:10.1002/art.20675
- Zhou S, Cui Z, Urban JP (2008) Nutrient gradients in engineered cartilage: metabolic kinetics measurement and mass transfer modeling. Biotechnol Bioeng 101(2):408–421. doi:10.1002/bit.21887
- Ziegler R, Goebel L, Seidel R, Cucchiarini M, Pape D, Madry H (2015) Effect of open wedge high tibial osteotomy on the lateral tibiofemoral compartment in sheep. Part III: analysis of the microstructure of the subchondral bone and correlations with the articular cartilage and meniscus. Knee Surg Sports Traumatol Arthrosc 23(9):2704–2714. doi:10.1007/ s00167-014-3134-y

Gene Transfer Strategies for Articular Cartilage Repair

9

Magali Cucchiarini

Abstract

Gene therapy protocols are well suited to deliver genes coding for therapeutic factors over time in a spatially defined manner within sites of cartilage injury resulting from acute trauma or during osteoarthritis. The focus of this chapter is to examine the benefits of gene therapy to improve cartilage repair in such lesions, based on promising experimental and clinical evidence in relevant models in vivo using growth, transcription, and signalling factors capable of stimulating the chondrogenic and chondro-reparative processes locally. A continuous, combined effort between scientists and orthopaedic surgeons may allow to bring gene therapy from encouraging data at the bench to a successful, safe translation in the broadly affected human population.

9.1 Introduction

Articular cartilage defects like after trauma or during osteoarthritis (OA) have a limited capacity for self-repair. The idea of applying gene transfer strategies to enhance cartilage repair by local application of therapeutic (growth, transcription, signalling) factors originates from the possibility to extend therapeutic transgene expression and subsequent effects over time compared with the injection of recombinant agents showing relatively short half-lives. Increasing, promising experimental data have shown the benefits of providing such therapeutic sequences using various gene transfer systems in relevant in vitro, in situ, and in vivo models of focal

M. Cucchiarini

Center of Experimental Orthopaedics, Saarland University Medical Center, Kirrbergerstr. Bldg 37, 66421 Homburg, Saar, Germany e-mail: mmcucchiarini@hotmail.com

[©] Springer International Publishing AG 2017

S. Grässel, A. Aszodi (eds.), Cartilage, DOI 10.1007/978-3-319-53316-2_9

and OA cartilage lesions via direct (cell-free) and indirect (cell-associated) procedures, allowing to rejuvenate the affected cells and to improve the repair of the damaged tissues. The availability of clinically suited vectors and of workable strategies may allow in a close future to provide adapted, effective novel options to treat cartilage injuries in patients.

9.2 Articular Cartilage Repair

Adult hyaline cartilage, the tissue that affords a smooth gliding of articulating surfaces and protects the subchondral bone against mechanical stress in the joint, has a limited intrinsic ability to heal in the absence of vascularization that may provide chondro-regenerative cells such as progenitor cell populations in response to injury (Heijink et al. 2012; Pape et al. 2010). The articular cartilage is formed by chondrocytes embedded in a self-produced, complex extracellular matrix in different layers, with a calcified basal layer towards the underlying subchondral bone. The cartilage matrix contains proteoglycans bound to 70–80% water, collagen fibrils (mostly type II but also type VI, IX, XI, and XIV collagen), and other molecules such as the cartilage oligomeric matrix protein (COMP), link protein, decorin, fibromodulin, fibronectin, and tenascin. Cartilage lesions can be circumscribed (focal defects) or generalized (osteoarthritis, OA, a chronic, inflammatory/catabolic whole degenerative joint disorder with a gradual degradation of the cartilage, subchondral bone, synovium, ligaments, tendons, and muscles) (Madry et al. 2012; Heijink et al. 2012; Loeser et al. 2012).

The repair of cartilage is considered as the production of a tissue that shares structural similarities with the hyaline articular cartilage. Regeneration, however, is a complete reproduction of the hyaline cartilage. Natural repair of chondral defects occurs by migration of cells from the synovial membrane but leading to a tissue that does not integrate with the surrounding cartilage and to larger defects. Osteochondral defects are filled with a clot from the bone marrow containing chondrogenically and osteogenically competent bone marrow-derived mesenchymal stem cells (BM-MSCs), but again a fibrocartilaginous repair tissue is produced with early signs of OA. OA cartilage has also restricted repair capabilities leading to an irreversible degradation of the cartilage and a remodelling of the osteochondral unit.

Current interventions for chondral defects include marrow stimulation (subchondral drilling, microfracture, and abrasion arthroplasty), and the transplantation of autologous chondrocytes with or without supportive matrix (autologous chondrocyte implantation, i.e. ACI) and those for osteochondral defects is based on the implantation of uninjured osteochondral cylinders and of subchondral bone grafts combined with ACI (Madry et al. 2011a) (for details, see Chap. 5). OA treatments are either conservative (non-pharmacological and pharmacological options using non-steroidal anti-inflammatory drugs, opioid analgesics, or intraarticular corticosteroid or hyaluronic acid (HA) injections) or surgical like by osteotomy (Madry et al. 2011a). As none of these procedures are capable of fully managing any of these lesions, novel, effective options are needed to improve cartilage repair, like those afforded by gene therapy that may need to take into account the differences between the nature, size, number, and location of the injury and the stage of the disease (Madry and Cucchiarini 2013, 2015; Frisch et al. 2015b; Cucchiarini et al. 2014; Madry et al. 2011b; Cucchiarini and Madry 2005).

9.3 Gene Therapy: Principles and Experimental Approaches for Cartilage Repair

Gene therapy aims at treating human disorders by applying gene transfer techniques in patients in vivo. Foreign genes may penetrate the target cell to reach the nucleus where they stay extrachromosomal (more or less stable episomal forms) or become integrated within the host genome in specific or unspecific locations (stable forms) as features inherent to the class and biology of the vector employed. Targeting by gene transfer usually affects a high number of cells that are sufficient for the production of the transgene product and for consequent therapeutic applications. The best characterized vectors available for gene therapy protocols mostly include nonviral constructs and approaches and viral vectors (Table 9.1) (Cucchiarini and Madry 2005; Madry and Cucchiarini 2013, 2015; Frisch et al. 2015b; Madry et al. 2011b).

| Types | | Advantages | Limitations | Integration |
|------------------|-----------------------|--|---|--------------------|
| Nonviral vectors | | Not infectious Not toxic Easy to produce Large capacity | • Low efficiency • Short-term expression | _ |
| Viral vectors | Adenovirus | • High efficiency | Replication competence Toxic Immunogenic Short-term expression | - |
| | Retro- /Lentivirus | High efficiencyLong-term expression | Replication competence Insertional mutagenesis | Yes |
| | HSV | High efficiencyLarge capacity | Cytotoxic Short-term expression | - |
| | rAAV | High efficiency Long-term expression Low immunogenic | Difficult to produce Size limitation | Mostly episomal |

 Table 9.1
 Common vectors available for gene therapy

Abbreviations: HSV Herpes simplex virus, rAAV recombinant adeno-associated virus

9.3.1 Gene Transfer Vectors

9.3.1.1 Nonviral Systems

Transfection is the transfer of foreign genes via nonviral vectors (cationic lipids and liposomes, polymers, polyamines, polyethylenimines, nanoparticles). These systems are large, easy to generate, and safe (lack of acquiring replication competence like for viral vectors, no immunogenicity), allowing for repeated administration. Yet, they display a relatively low efficacy compared with viral vectors. Also, as mostly episomal forms, they commonly promote short-term transgene expression. These vectors are thus rather employed in indirect gene transfer protocols by implantation of ex vivo modified cells in the recipient.

9.3.1.2 Viral Vectors

Transduction is the term characterizing viral-based gene transfer. Viral vectors employ their natural entry pathways in the cells. Adenoviral vectors have a high gene transfer efficacy, enabling direct approaches in vivo, but they are highly immunogenic and remain episomal, restricting transgene expression to only about 1–2 weeks. Retroviruses instead can integrate into the host genome, allowing for the maintenance of the transgene over time. Still, integration is associated with a risk of insertional mutagenesis and of tumour gene activation. In addition, these vectors transduce only dividing cells (progenitor cells but not differentiated chondrocytes and bone cells) at a restricted host range, making them more adapted for indirect, ex vivo approaches, further allowing to increasing their otherwise low gene transfer efficacy. Lentiviral vectors that derive from the Human immunodeficiency virus (HIV) instead can integrate in nondividing cells at higher efficiency, but concerns remain due to the potential for insertional mutagenesis and to the problem of introducing material carrying HIV sequences. Large Herpes simplex virus (HSV) vectors can target nondividing cells, but still induce cytotoxic responses and allow only for very transient transgene expression. Small recombinant adeno-associated viral (rAAV) vectors have various advantages as they are safe and low immunogenic in the complete absence of viral gene sequences. This is particularly important as cartilage lesions are not life-threatening problems. They can transduce both dividing and nondividing cells and are mostly kept as very stable episomes, allowing for long-term transgene expression.

9.4 Strategies to Improve Cartilage Repair

Target cells relevant for cartilage repair permissive to gene transfer include articular chondrocytes to repopulate the injured cartilage, bone cells (osteoblasts/osteocytes to reconstruct the subchondral bone), synoviocytes (as a source of therapeutic factors), other affected cells (meniscal fibrochondrocytes, tendon/ligament cells, muscle cells), and progenitor cells (BM-MSCs that may commit towards mesodermal cell lineages

as isolated/expanded suspensions or directly in marrow aspirates/concentrates) (Arai et al. 2000; Baragi et al. 1995; Cucchiarini et al. 2014; Doherty et al. 1998; Gouze et al. 2002; Hildebrand et al. 1999; Madry et al. 2004a; Madry and Trippel 2000; Mason et al. 2000; Nita et al. 1996; Orth et al. 2014; Rey-Rico et al. 2015a; Stender et al. 2007). These cells may be directly genetically modified within the lesions by direct gene transfer vector administration or first isolated and/or expanded for indirect gene transfer ex vivo prior to reimplantation. This may offer to employ other cell types such as MSCs from other tissues like the adipose tissue, synovium, periosteum, and perichondrium (Cucchiarini and Madry 2005; Madry et al. 2011b). Various approaches have been developed experimentally to deliver therapeutic gene sequences in sites of articular cartilage injury (Fig. 9.1), among which administration of a therapeutic



Fig. 9.1 Experimental approaches for the delivery of therapeutic gene sequences in sites of articular cartilage injury. (1) Preparation of a therapeutic composition with gene vectors, cells (suspension or as concentrates), and a biocompatible matrix. The composition may be provided (2) intraarticularly or (3) via arthrotomy in either focal defects or in OA cartilage

compound intraarticularly or via arthrotomy using either the gene vector itself or genetically modified cells as a suspension or using a biocompatible matrix (Cucchiarini and Madry 2005, 2014b; Johnstone et al. 2013; Madry and Cucchiarini 2013, 2015; Cucchiarini et al. 2014; Frisch et al. 2015b; Madry et al. 2011b).

9.4.1 Gene Therapy for Cartilage Repair: Evidence In Vitro

Pathways that might be targeted to improve cartilage repair by gene therapy include the activation of cell proliferation and survival, the stimulation of anabolic responses, and the prevention of inflammation and tissue degradation via single or combined approaches.

9.4.1.1 Activation of Cell Proliferation and Survival

These processes may be stimulated by gene transfer of mitogenic growth factors (insulin-like growth factor I (IGF-I), basic fibroblast growth factor (FGF-2), bone morphogenetic proteins (BMPs), transforming growth factor beta (TGF- β)) (Madry et al. 2001, 2004b; Cucchiarini et al. 2009, 2011; Weimer et al. 2012; Shi et al. 2013; Neumann et al. 2013; Venkatesan et al. 2013; Frisch et al. 2014a, b, 2015a), telomerase (hTERT) (Piera-Velazquez et al. 2002), inhibitors of apoptosis (bcl-2) (Surendran et al. 2006), or of the heat shock protein 70 (HSP70) (Grossin et al. 2006). Most remarkably, delivery and overexpression of the key IGF-I and TGF- β factors via potent rAAV-mediated gene transfer allowed for a stable reproduction of the cell proliferative indices in human OA articular cartilage explant cultures in situ, reaching levels and patterns typical of normal cartilage over an extended period of 90 days when the cells are embedded in their natural matrix (Weimer et al. 2012; Venkatesan et al. 2013; Venkatesan et al. 2012; Venkatesan et al. 2013).

9.4.1.2 Stimulation of Anabolic Responses

Production of matrix components may be enhanced by application of sequences for the matrix molecules themselves or of the enzymes that synthesize them (Venkatesan et al. 2004), of pro-anabolic growth and signalling factors (IGF-I, FGF-2, BMPs, TGF- β , parathyroid hormone-related peptide (PTHrP), Indian hedgehog (Ihh)) (Smith et al. 2000; Nixon et al. 2000; Shuler et al. 2000; Mi et al. 2000; Madry et al. 2001; Brower-Toland et al. 2001; Palmer et al. 2005; Cucchiarini et al. 2005; Ulrich-Vinther et al. 2005; Wang et al. 2011; Weimer et al. 2012; Steinert et al. 2012; Neumann et al. 2013; Venkatesan et al. 2013; Frisch et al. 2014a, b, 2015a), or of tissue-specific transcription factors (sex-determining region Y-type high mobility group box—SOX family, zinc-finger protein 145 (ZNF145)) (Li et al. 2004; Cucchiarini et al. 2007; Liu et al. 2011; Rey-Rico et al. 2015a; Shi et al. 2015).

Among these many factors, here again administration of IGF-I and TGF- β sequences via rAAV vectors may be best suited in light of reports showing close-tonormal, long-term reconstruction of the extracellular matrix in human OA articular cartilage for at least 90 days in situ (Weimer et al. 2012; Venkatesan et al. 2013), together with the use of the cartilage-specific SOX9 transcription factor that also exhibits highly effective chondro-regenerative activities that are essential for cartilage repair (Cucchiarini et al. 2007; Rey-Rico et al. 2015a).

9.4.1.3 Prevention of Inflammation and Tissue Degradation

Reduction of catabolic processes may be achieved with inhibitors of matrixdegrading enzymes (Kafienah et al. 2003) and of pro-inflammatory cytokines (interleukin-1 receptor antagonist—IL-1Ra) (Baragi et al. 1995; Roessler et al. 1995; Glass et al. 2014). Glass et al. recently provided interesting evidence that combining gene therapy with functional tissue engineering provided powerful tools to produce cartilage with immunomodulatory properties via scaffold-mediated IL-1Ra gene transfer in MSCs, permitting chondrogenesis upon pathologic activation by IL-1 (Glass et al. 2014). Prevention of osteophyte formation in OA may be also considered like by application of antagonists of the TGF- β /BMP pathway (latencyassociated peptide—mLAP-1, Smads) (Scharstuhl et al. 2003) or of an IL-1Ra formulation (Fernandes et al. 1999).

9.4.1.4 Combined Approaches

Multifactorial approaches have been attempted by cotransfer of activators of proliferative/anabolic processes (IGF-I/FGF-2/BMPs/SOX) (Ikeda et al. 2004; Cucchiarini et al. 2009; Orth et al. 2011; Shi et al. 2012, 2013) or of activators of anabolic/proliferative pathways with inhibitors of catabolism (IGF-I/FGF-2/ IL-1Ra) (Nixon et al. 2005; Haupt et al. 2005; Chen et al. 2010). Approaches that aim both at counteracting deleterious inflammation while restoring the altered metabolic balance might be desirable to address the spectrum of pathomechanisms triggered in sites of cartilage injury. Specifically, several groups showed that cells cotransduced with IGF-I/IL-1Ra via adenoviral vectors were more potent to reverse IL-1-mediated proteoglycan depletion in cartilage compared with single gene treatments (Nixon et al. 2005; Haupt et al. 2005; Chen et al. 2010).

9.4.2 Gene Therapy for Cartilage Repair: Evidence In Vivo

Both direct and indirect gene transfer strategies have been tested in animal models of cartilage injury to provide therapeutic genes for enhanced cartilage repair. Direct strategies are less invasive, but necessitate that the vectors are adapted to effectively reach the target cells within their dense matrix. In this case, the small rAAV vectors are probably the most adequate gene vehicles to achieve this goal. Indirect strategies might be more desirable to repopulate tissue lesions, having the additional advantages of introducing modified cells rather than free vector particles while permitting extensive control of the cells prior to reimplantation and allowing the use of biocompatible scaffolds.

9.4.2.1 Direct Gene Transfer Strategies

Such approaches have been tested to enhance cartilage repair in experimental models of focal defects (Cucchiarini et al. 2005, 2013; Morisset et al. 2007; Cucchiarini and Madry 2014a; Griffin et al. 2015) and of OA (Fernandes et al. 1999; Frisbie et al. 2002; Grossin et al. 2006; Chen et al. 2008, 2010; Hsieh et al. 2009, 2010; Shen et al. 2011; Oh et al. 2012; Santangelo et al. 2012; Zhang et al. 2015) using

sequences for growth factors (IGF-I, FGF-2, TGF-β) (Cucchiarini et al. 2005; Chen et al. 2010; Cucchiarini and Madry 2014a; Griffin et al. 2015; Zhang et al. 2015), IL-1Ra (Fernandes et al. 1999; Frisbie et al. 2002; Chen et al. 2010; Santangelo et al. 2012; Zhang et al. 2015), SOX9 (Cucchiarini et al. 2013), HSP70 (Grossin et al. 2006), silencers of NF-kappaBp65 (Chen et al. 2008), inhibitors of inflammatory pain processes (pro-opiomelanocortin, POMC) (Shen et al. 2011), antagonists of the canonical Wnt pathway (Dickkopf1, Dkk-1) (Oh et al. 2012), kallistatin or angiogenic inhibitors (thrombospondin-1, TSP-1) (Hsieh et al. 2009, 2010), or combined approaches (IGF-I/IL-1Ra, TGF-\beta/IL-1Ra) (Morisset et al. 2007; Zhang et al. 2015), allowing for improved cartilage repair in these different systems. Most significantly, treatment of experimental focal (osteochondral) defects by direct application of an rAAV vector carrying the potent SOX9 transcription factor improved the processes of cartilage repair in rabbit knee joints for a stable period of 16 weeks without detrimental effects (Cucchiarini et al. 2013). Regarding OA, therapeutic success has been reported by various groups based on the application of IL-1Racoding nonviral and adenoviral vectors both in OA guinea pigs, rabbits, and horses, promoting a net reduction in the disease severity together with improvements in cartilage preservation (Fernandes et al. 1999; Frisbie et al. 2002; Santangelo et al. 2012; Zhang et al. 2015).

9.4.2.2 Indirect Gene Transfer Strategies

Administration of genetically modified cells has been also reported to treat experimental models of focal defects (Mason et al. 2000; Lee et al. 2001; Hidaka et al. 2003; Georgi et al. 1992; Madry et al. 2005, 2013; Guo et al. 2006; Kaul et al. 2006; Kuroda et al. 2006; Goodrich et al. 2007; Evans et al. 2009; Vogt et al. 2009; Che et al. 2010; Ivkovic et al. 2010; Noh et al. 2010; Liu et al. 2011; Orth et al. 2011; Ortved et al. 2015; Sieker et al. 2015) and of OA (Bandara et al. 1993; Zhang et al. 2004; Matsumoto et al. 2009) using genetic modification of chondrocytes (Hidaka et al. 2003; Madry et al. 2005, 2013; Kaul et al. 2006; Goodrich et al. 2007; Che et al. 2010; Noh et al. 2010; Orth et al. 2011; Ortved et al. 2015), synoviocytes (Bandara et al. 1993; Zhang et al. 2004), various progenitor cells as suspensions or as marrow aspirates/concentrates (Mason et al. 2000; Katayama et al. 2004; Guo et al. 2006; Kuroda et al. 2006; Vogt et al. 2009; Matsumoto et al. 2009; Ivkovic et al. 2010; Liu et al. 2011; Sieker et al. 2015), and tissue grafts (Evans et al. 2009) to overexpress growth factors (IGF-I, FGF-2, BMPs, TGF-β) (Mason et al. 2000; Lee et al. 2001; Hidaka et al. 2003; Madry et al. 2005, 2013; Guo et al. 2006; Kaul et al. 2006; Kuroda et al. 2006; Goodrich et al. 2007; Evans et al. 2009; Vogt et al. 2009; Che et al. 2010; Ivkovic et al. 2010; Noh et al. 2010; Ortved et al. 2015), IL-1Ra (Bandara et al. 1993; Zhang et al. 2004), SOX9 (Liu et al. 2011), Ihh (Sieker et al. 2015), the cartilage-derived morphogenetic protein 1 (CDMP-1) (Katayama et al. 2004), IGF-I/FGF-2 (Orth et al. 2011), IL-1Ra/IL-10 (Zhang et al. 2004), or BMP-4/sFlt-1 (an antagonist of the vascular endothelial growth factor, VEGF) (Matsumoto et al. 2009), allowing for improved cartilage repair in these various models. Significant advances have been made in experimental cartilage repair when implanting IGF-I-expressing vectors in articular chondrocytes or BMP-2-modified

tissues (fat, muscle, fibrin clots) in focal defects in rabbits and horses using nonviral, adenoviral, retroviral, and rAAV vectors (Goodrich et al. 2007; Evans et al. 2009; Vogt et al. 2009; Madry et al. 2013; Ortved et al. 2015), leading to improved tissue healing. In OA models, mostly synovial cells overexpressing IL-1Ra via retroviral vectors have been provided to prevent cartilage breakdown in OA rabbits (Bandara et al. 1993; Zhang et al. 2004), while injection of muscle-derived MSCs producing BMP-4/sFlt1 in OA rats led to durable cartilage repair (Matsumoto et al. 2009).

9.5 Gene Therapy: Current Advances in Clinical Trials for Cartilage Repair—Perspectives

Compared with the large body of available experimental data, relatively few trials have been initiated to treat patients with focal cartilage lesions and OA, reflecting the difficulty to translate research ideas and visions into clinical and commercial reality (Evans et al. 2012, 2013; Cucchiarini et al. 2014; Bara et al. 2015). While no trials are ongoing for focal defects, phase I and II clinical trials have been recently published to treat OA patients by injecting retrovirally modified chondrocytes to produce TGF- β 1 (Ha et al. 2012, 2015; Lee et al. 2015; Cherian et al. 2015), showing a trend towards efficacy with improvements in pain, function, and physical ability. Yet, in light of multiple reports showing adverse effects of TGF- β in experimental models in vivo (synovial inflammation and fibrosis, osteophyte formation) (van Beuningen et al. 2013), other approaches might be necessary to improve the safety of the outcomes and prevent such detrimental, undesirable effects in patients.

Remarkably, rAAV vectors have emerged as well-suited systems for clinical applications as these vectors are highly effective in direct, less invasive protocols compared with retroviral vectors and considered as safe delivery compounds, receiving market authorization by the European Union Medicines Agency's Committee for Medicinal Products for Human Use (CHMP) for the treatment of lipoprotein lipase (LPL) deficiency (Glybera[®]) (Yla-Herttuala 2012; Buning 2013). Interestingly, a phase I study by direct administration of an rAAV IL-1Ra construct has been initiated to treat OA patients (Evans et al. 2013). It remains to be seen whether such a trial will allow for successful outcomes as a major concern to the clinical use of this vector class is the significance of pre-existing and induced immune responses against the viral capsid proteins of AAV, with a clear impact on clinical efficacy and safety (Ferreira et al. 2014a, b). While an rAAV-based gene therapy might be provided in conditions of immunosuppression to improve efficacy (Ferreira et al. 2014b), the safety of the immunosuppressed patients remains a concern as OA is a nonlethal disease. A newly developed, highly potent approach to circumvent these issues is to provide rAAV using effective controlled release strategies by delivery of the vectors from adapted biomaterials as a means to mask the immunogenic capsid epitopes in the host without impairing the efficacy of gene transfer and target cell modification (Rey-Rico et al. 2015b, c; Rey-Rico and Cucchiarini 2015; Diaz-Rodriguez et al. 2015). Tissue-engineering strategies may also prove beneficial to improve the repair of cartilage lesions and thus address the challenge of providing functional, adapted cartilage replacement therapies as largely tested in preclinical settings in relevant models (Johnstone et al. 2013; Cucchiarini and Madry 2014b; Madry and Cucchiarini 2014). Eventually, as complete regeneration of an original hyaline cartilage has not been reported thus far in preclinical models, further research is clearly needed to identify the most effective genes or combinations of genes for therapy, suggesting that a better understanding of the joint pathologies is essential by continuous efforts between orthopaedic surgeons, scientists, and regulatory organizations to advance the current approaches and trials in patients (Cucchiarini et al. 2014).

References

- Arai Y, Kubo T, Fushiki S, Mazda O, Nakai H, Iwaki Y, Imanishi J, Hirasawa Y (2000) Gene delivery to human chondrocytes by an adeno associated virus vector. J Rheumatol 27(4):979–982
- Bakker AC, van de Loo FA, van Beuningen HM, Sime P, van Lent PL, van der Kraan PM, Richards CD, van den Berg WB (2001) Overexpression of active TGF-beta-1 in the murine knee joint: evidence for synovial-layer-dependent chondro-osteophyte formation. Osteoarthritis Cartilage 9(2):128–136. doi:10.1053/joca.2000.0368
- Bandara G, Mueller GM, Galea-Lauri J, Tindal MH, Georgescu HI, Suchanek MK, Hung GL, Glorioso JC, Robbins PD, Evans CH (1993) Intraarticular expression of biologically active interleukin 1-receptor-antagonist protein by ex vivo gene transfer. Proc Natl Acad Sci U S A 90(22):10764–10768
- Bara JJ, Herrmann M, Evans CH, Miclau T, Ratcliffe A, Richards RG (2015) Improving translation success of cell-based therapies in orthopaedics. J Orthop Res. doi:10.1002/jor.23055
- Baragi VM, Renkiewicz RR, Jordan H, Bonadio J, Hartman JW, Roessler BJ (1995) Transplantation of transduced chondrocytes protects articular cartilage from interleukin 1-induced extracellular matrix degradation. J Clin Invest 96(5):2454–2460. doi:10.1172/JCI118303
- van Beuningen HM, Glansbeek HL, van der Kraan PM, van den Berg WB (1998) Differential effects of local application of BMP-2 or TGF-beta 1 on both articular cartilage composition and osteophyte formation. Osteoarthritis Cartilage 6(5):306–317. doi:10.1053/joca.1998.0129
- Blaney Davidson EN, Vitters EL, van Beuningen HM, van de Loo FA, van den Berg WB, van der Kraan PM (2007) Resemblance of osteophytes in experimental osteoarthritis to transforming growth factor beta-induced osteophytes: limited role of bone morphogenetic protein in early osteoarthritic osteophyte formation. Arthritis Rheum 56(12):4065–4073. doi:10.1002/art.23034
- Brower-Toland BD, Saxer RA, Goodrich LR, Mi Z, Robbins PD, Evans CH, Nixon AJ (2001) Direct adenovirus-mediated insulin-like growth factor I gene transfer enhances transplant chondrocyte function. Hum Gene Ther 12(2):117–129. doi:10.1089/104303401750061186
- Buning H (2013) Gene therapy enters the pharma market: the short story of a long journey. EMBO Mol Med 5(1):1–3. doi:10.1002/emmm.201202291
- Che JH, Zhang ZR, Li GZ, Tan WH, Bai XD, Qu FJ (2010) Application of tissue-engineered cartilage with BMP-7 gene to repair knee joint cartilage injury in rabbits. Knee Surg Sports Traumatol Arthrosc 18(4):496–503. doi:10.1007/s00167-009-0962-2
- Chen LX, Lin L, Wang HJ, Wei XL, Fu X, Zhang JY, Yu CL (2008) Suppression of early experimental osteoarthritis by in vivo delivery of the adenoviral vector-mediated NF-kappaBp65specific siRNA. Osteoarthritis Cartilage 16(2):174–184. doi:10.1016/j.joca.2007.06.006
- Chen B, Qin J, Wang H, Magdalou J, Chen L (2010) Effects of adenovirus-mediated bFGF, IL-1Ra and IGF-1 gene transfer on human osteoarthritic chondrocytes and osteoarthritis in rabbits. Exp Mol Med 42(10):684–695. doi:10.3858/emm.2010.42.10.067

- Cherian JJ, Parvizi J, Bramlet D, Lee KH, Romness DW, Mont MA (2015) Preliminary results of a phase II randomized study to determine the efficacy and safety of genetically engineered allogeneic human chondrocytes expressing TGF-beta1 in patients with grade 3 chronic degenerative joint disease of the knee. Osteoarthritis Cartilage 23(12):2109–2118. doi:10.1016/j. joca.2015.06.019
- Cucchiarini M, Madry H (2005) Gene therapy for cartilage defects. J Gene Med 7(12):1495–1509. doi:10.1002/jgm.824
- Cucchiarini M, Madry H (2014a) Overexpression of human IGF-I via direct rAAV-mediated gene transfer improves the early repair of articular cartilage defects in vivo. Gene Ther 21(9):811– 819. doi:10.1038/gt.2014.58
- Cucchiarini M, Madry H (2014b) Use of tissue engineering strategies to repair joint tissues in osteoarthritis: viral gene transfer approaches. Curr Rheumatol Rep 16(10):449. doi:10.1007/s11926-014-0449-0
- Cucchiarini M, Madry H, Ma C, Thurn T, Zurakowski D, Menger MD, Kohn D, Trippel SB, Terwilliger EF (2005) Improved tissue repair in articular cartilage defects in vivo by rAAVmediated overexpression of human fibroblast growth factor 2. Mol Ther 12(2):229–238. doi:10.1016/j.ymthe.2005.03.012
- Cucchiarini M, Thurn T, Weimer A, Kohn D, Terwilliger EF, Madry H (2007) Restoration of the extracellular matrix in human osteoarthritic articular cartilage by overexpression of the transcription factor SOX9. Arthritis Rheum 56(1):158–167. doi:10.1002/art.22299
- Cucchiarini M, Terwilliger EF, Kohn D, Madry H (2009) Remodelling of human osteoarthritic cartilage by FGF-2, alone or combined with Sox9 via rAAV gene transfer. J Cell Mol Med 13(8B):2476–2488. doi:10.1111/j.1582-4934.2008.00474.x
- Cucchiarini M, Ekici M, Schetting S, Kohn D, Madry H (2011) Metabolic activities and chondrogenic differentiation of human mesenchymal stem cells following recombinant adenoassociated virus-mediated gene transfer and overexpression of fibroblast growth factor 2. Tissue Eng Part A 17(15-16):1921–1933. doi:10.1089/ten.TEA.2011.0018
- Cucchiarini M, Orth P, Madry H (2013) Direct rAAV SOX9 administration for durable articular cartilage repair with delayed terminal differentiation and hypertrophy in vivo. J Mol Med 91(5):625–636. doi:10.1007/s00109-012-0978-9
- Cucchiarini M, Madry H, Guilak F, Saris DB, Stoddart MJ, Koon Wong M, Roughley P (2014) A vision on the future of articular cartilage repair. Eur Cell Mater 27:12–16
- Diaz-Rodriguez P, Rey-Rico A, Madry H, Landin M, Cucchiarini M (2015) Effective genetic modification and differentiation of hMSCs upon controlled release of rAAV vectors using alginate/ poloxamer composite systems. Int J Pharm. doi:10.1016/j.ijpharm.2015.11.008
- Doherty PJ, Zhang H, Tremblay L, Manolopoulos V, Marshall KW (1998) Resurfacing of articular cartilage explants with genetically-modified human chondrocytes in vitro. Osteoarthritis Cartilage 6(3):153–159. doi:10.1053/joca.1998.0107
- Evans CH, Liu FJ, Glatt V, Hoyland JA, Kirker-Head C, Walsh A, Betz O, Wells JW, Betz V, Porter RM, Saad FA, Gerstenfeld LC, Einhorn TA, Harris MB, Vrahas MS (2009) Use of genetically modified muscle and fat grafts to repair defects in bone and cartilage. Eur Cell Mater 18:96–111
- Evans CH, Ghivizzani SC, Robbins PD (2012) Orthopedic gene therapy--lost in translation? J Cell Physiol 227(2):416–420. doi:10.1002/jcp.23031
- Evans CH, Ghivizzani SC, Robbins PD (2013) Arthritis gene therapy and its tortuous path into the clinic. Transl Res 161(4):205–216. doi:10.1016/j.trsl.2013.01.002
- Fernandes J, Tardif G, Martel-Pelletier J, Lascau-Coman V, Dupuis M, Moldovan F, Sheppard M, Krishnan BR, Pelletier JP (1999) In vivo transfer of interleukin-1 receptor antagonist gene in osteoarthritic rabbit knee joints: prevention of osteoarthritis progression. Am J Pathol 154(4):1159–1169. doi:10.1016/S0002-9440(10)65368-0
- Ferreira V, Petry H, Salmon F (2014a) Immune responses to AAV-vectors, the glybera example from bench to bedside. Front Immunol 5:82. doi:10.3389/fimmu.2014.00082
- Ferreira V, Twisk J, Kwikkers K, Aronica E, Brisson D, Methot J, Petry H, Gaudet D (2014b) Immune responses to intramuscular administration of alipogene tiparvovec (AAV1-

LPL(S447X)) in a phase II clinical trial of lipoprotein lipase deficiency gene therapy. Hum Gene Ther 25(3):180–188. doi:10.1089/hum.2013.169

- Frisbie DD, Ghivizzani SC, Robbins PD, Evans CH, McIlwraith CW (2002) Treatment of experimental equine osteoarthritis by in vivo delivery of the equine interleukin-1 receptor antagonist gene. Gene Ther 9(1):12–20. doi:10.1038/sj.gt.3301608
- Frisch J, Venkatesan JK, Rey-Rico A, Schmitt G, Madry H, Cucchiarini M (2014a) Determination of the chondrogenic differentiation processes in human bone marrow-derived mesenchymal stem cells genetically modified to overexpress transforming growth factor-beta via recombinant adeno-associated viral vectors. Hum Gene Ther 25(12):1050–1060. doi:10.1089/ hum.2014.091
- Frisch J, Venkatesan JK, Rey-Rico A, Schmitt G, Madry H, Cucchiarini M (2014b) Influence of insulin-like growth factor I overexpression via recombinant adeno-associated vector gene transfer upon the biological activities and differentiation potential of human bone marrowderived mesenchymal stem cells. Stem Cell Res Ther 5(4):103. doi:10.1186/scrt491
- Frisch J, Rey-Rico A, Venkatesan JK, Schmitt G, Madry H, Cucchiarini M (2015a) Chondrogenic differentiation processes in human bone marrow aspirates upon rAAV-mediated gene transfer and overexpression of the insulin-like growth factor I. Tissue Eng Part A 21(17-18):2460– 2471. doi:10.1089/ten.TEA.2014.0679
- Frisch J, Venkatesan JK, Rey-Rico A, Madry H, Cucchiarini M (2015b) Current progress in stem cell-based gene therapy for articular cartilage repair. Curr Stem Cell Res Ther 10(2):121–131
- Georgi P, Emrich D, Heidenreich P, Moser E, Reiners C, Schicha H (1992) Radioiodine therapy of differentiated thyroid cancer. Recommendations of the Therapy Study Group of the German Society of Nuclear Medicine. Nuklearmedizin 31(4):151–153
- Glass KA, Link JM, Brunger JM, Moutos FT, Gersbach CA, Guilak F (2014) Tissue-engineered cartilage with inducible and tunable immunomodulatory properties. Biomaterials 35(22):5921– 5931. doi:10.1016/j.biomaterials.2014.03.073
- Goodrich LR, Hidaka C, Robbins PD, Evans CH, Nixon AJ (2007) Genetic modification of chondrocytes with insulin-like growth factor-1 enhances cartilage healing in an equine model. J Bone Joint Surg 89(5):672–685. doi:10.1302/0301-620X.89B5.18343
- Gouze E, Pawliuk R, Pilapil C, Gouze JN, Fleet C, Palmer GD, Evans CH, Leboulch P, Ghivizzani SC (2002) In vivo gene delivery to synovium by lentiviral vectors. Mol Ther 5(4):397–404. doi:10.1006/mthe.2002.0562
- Griffin DJ, Ortved KF, Nixon AJ, Bonassar LJ (2015) Mechanical properties and structure-function relationships in articular cartilage repaired using IGF-I gene-enhanced chondrocytes. J Orthop Res. doi:10.1002/jor.23038
- Grossin L, Cournil-Henrionnet C, Pinzano A, Gaborit N, Dumas D, Etienne S, Stoltz JF, Terlain B, Netter P, Mir LM, Gillet P (2006) Gene transfer with HSP 70 in rat chondrocytes confers cytoprotection in vitro and during experimental osteoarthritis. FASEB J 20(1):65–75. doi:10.1096/ fj.04-2889com
- Guo X, Zheng Q, Yang S, Shao Z, Yuan Q, Pan Z, Tang S, Liu K, Quan D (2006) Repair of full-thickness articular cartilage defects by cultured mesenchymal stem cells transfected with the transforming growth factor beta1 gene. Biomed Mater 1(4):206–215. doi:10.1088/ 1748-6041/1/4/006
- Ha CW, Noh MJ, Choi KB, Lee KH (2012) Initial phase I safety of retrovirally transduced human chondrocytes expressing transforming growth factor-beta-1 in degenerative arthritis patients. Cytotherapy 14(2):247–256. doi:10.3109/14653249.2011.629645
- Ha CW, Cho JJ, Elmallah RK, Cherian JJ, Kim TW, Lee MC, Mont MA (2015) A multicenter, single-blind, phase IIa clinical trial to evaluate the efficacy and safety of a cell-mediated gene therapy in degenerative knee arthritis patients. Hum Gene Ther Clin Dev 26(2):125–130. doi:10.1089/humc.2014.145
- Haupt JL, Frisbie DD, McIlwraith CW, Robbins PD, Ghivizzani S, Evans CH, Nixon AJ (2005) Dual transduction of insulin-like growth factor-I and interleukin-1 receptor antagonist protein controls cartilage degradation in an osteoarthritic culture model. J Orthop Res 23(1):118–126. doi:10.1016/j.orthres.2004.06.020

- Heijink A, Gomoll AH, Madry H, Drobnic M, Filardo G, Espregueira-Mendes J, Van Dijk CN (2012) Biomechanical considerations in the pathogenesis of osteoarthritis of the knee. Knee Surg Sports Traumatol Arthrosc 20(3):423–435. doi:10.1007/s00167-011-1818-0
- Hidaka C, Goodrich LR, Chen CT, Warren RF, Crystal RG, Nixon AJ (2003) Acceleration of cartilage repair by genetically modified chondrocytes over expressing bone morphogenetic protein-7. J Orthop Res 21(4):573–583. doi:10.1016/S0736-0266(02)00264-4
- Hildebrand KA, Deie M, Allen CR, Smith DW, Georgescu HI, Evans CH, Robbins PD, Woo SL (1999) Early expression of marker genes in the rabbit medial collateral and anterior cruciate ligaments: the use of different viral vectors and the effects of injury. J Orthop Res 17(1):37–42. doi:10.1002/jor.1100170107
- Hsieh JL, Shen PC, Shiau AL, Jou IM, Lee CH, Teo ML, Wang CR, Chao J, Chao L, Wu CL (2009) Adenovirus-mediated kallistatin gene transfer ameliorates disease progression in a rat model of osteoarthritis induced by anterior cruciate ligament transection. Hum Gene Ther 20(2):147– 158. doi:10.1089/hum.2008.096
- Hsieh JL, Shen PC, Shiau AL, Jou IM, Lee CH, Wang CR, Teo ML, Wu CL (2010) Intraarticular gene transfer of thrombospondin-1 suppresses the disease progression of experimental osteoarthritis. J Orthop Res 28(10):1300–1306. doi:10.1002/jor.21134
- Ikeda T, Kamekura S, Mabuchi A, Kou I, Seki S, Takato T, Nakamura K, Kawaguchi H, Ikegawa S, Chung UI (2004) The combination of SOX5, SOX6, and SOX9 (the SOX trio) provides signals sufficient for induction of permanent cartilage. Arthritis Rheum 50(11):3561–3573. doi:10.1002/art.20611
- Ivkovic A, Pascher A, Hudetz D, Maticic D, Jelic M, Dickinson S, Loparic M, Haspl M, Windhager R, Pecina M (2010) Articular cartilage repair by genetically modified bone marrow aspirate in sheep. Gene Ther 17(6):779–789. doi:10.1038/gt.2010.16
- Johnstone B, Alini M, Cucchiarini M, Dodge GR, Eglin D, Guilak F, Madry H, Mata A, Mauck RL, Semino CE, Stoddart MJ (2013) Tissue engineering for articular cartilage repair--the state of the art. Eur Cell Mater 25:248–267
- Kafienah W, Al-Fayez F, Hollander AP, Barker MD (2003) Inhibition of cartilage degradation: a combined tissue engineering and gene therapy approach. Arthritis Rheum 48(3):709–718. doi:10.1002/art.10842
- Katayama R, Wakitani S, Tsumaki N, Morita Y, Matsushita I, Gejo R, Kimura T (2004) Repair of articular cartilage defects in rabbits using CDMP1 gene-transfected autologous mesenchymal cells derived from bone marrow. Rheumatology 43(8):980–985. doi:10.1093/rheumatology/keh240
- Kaul G, Cucchiarini M, Arntzen D, Zurakowski D, Menger MD, Kohn D, Trippel SB, Madry H (2006) Local stimulation of articular cartilage repair by transplantation of encapsulated chondrocytes overexpressing human fibroblast growth factor 2 (FGF-2) in vivo. J Gene Med 8(1):100–111. doi:10.1002/jgm.819
- Kuroda R, Usas A, Kubo S, Corsi K, Peng H, Rose T, Cummins J, Fu FH, Huard J (2006) Cartilage repair using bone morphogenetic protein 4 and muscle-derived stem cells. Arthritis Rheum 54(2):433–442. doi:10.1002/art.21632
- Lee KH, Song SU, Hwang TS, Yi Y, Oh IS, Lee JY, Choi KB, Choi MS, Kim SJ (2001) Regeneration of hyaline cartilage by cell-mediated gene therapy using transforming growth factor beta 1-producing fibroblasts. Hum Gene Ther 12(14):1805–1813. doi:10.1089/104303401750476294
- Lee MC, Ha CW, Elmallah RK, Cherian JJ, Cho JJ, Kim TW, Bin SI, Mont MA (2015) A placebocontrolled randomised trial to assess the effect of TGF-ss1-expressing chondrocytes in patients with arthritis of the knee. Bone Joint J 97-B(7):924–932. doi:10.1302/0301-620X.97B7.35852
- Li Y, Tew SR, Russell AM, Gonzalez KR, Hardingham TE, Hawkins RE (2004) Transduction of passaged human articular chondrocytes with adenoviral, retroviral, and lentiviral vectors and the effects of enhanced expression of SOX9. Tissue Eng 10(3-4):575–584. doi:10.1089/ 107632704323061933
- Liu TM, Guo XM, Tan HS, Hui JH, Lim B, Lee EH (2011) Zinc-finger protein 145, acting as an upstream regulator of SOX9, improves the differentiation potential of human mesenchymal stem cells for cartilage regeneration and repair. Arthritis Rheum 63(9):2711–2720. doi:10.1002/art.30430

- Loeser RF, Goldring SR, Scanzello CR, Goldring MB (2012) Osteoarthritis: a disease of the joint as an organ. Arthritis Rheum 64(6):1697–1707. doi:10.1002/art.34453
- Madry H, Cucchiarini M (2013) Advances and challenges in gene-based approaches for osteoarthritis. J Gene Med 15(10):343–355. doi:10.1002/jgm.2741
- Madry H, Cucchiarini M (2014) Tissue-engineering strategies to repair joint tissue in osteoarthritis: nonviral gene-transfer approaches. Curr Rheumatol Rep 16(10):450. doi:10.1007/ s11926-014-0450-7
- Madry H, Cucchiarini M (2015) Gene therapy for human osteoarthritis: principles and clinical translation. Expert Opin Biol Ther. doi:10.1517/14712598.2016.1124084
- Madry H, Trippel SB (2000) Efficient lipid-mediated gene transfer to articular chondrocytes. Gene Ther 7(4):286–291. doi:10.1038/sj.gt.3301086
- Madry H, Zurakowski D, Trippel SB (2001) Overexpression of human insulin-like growth factor-I promotes new tissue formation in an ex vivo model of articular chondrocyte transplantation. Gene Ther 8(19):1443–1449. doi:10.1038/sj.gt.3301535
- Madry H, Cucchiarini M, Kaul G, Kohn D, Terwilliger EF, Trippel SB (2004a) Menisci are efficiently transduced by recombinant adeno-associated virus vectors in vitro and in vivo. Am J Sports Med 32(8):1860–1865
- Madry H, Emkey G, Zurakowski D, Trippel SB (2004b) Overexpression of human fibroblast growth factor 2 stimulates cell proliferation in an ex vivo model of articular chondrocyte transplantation. J Gene Med 6(2):238–245. doi:10.1002/jgm.488
- Madry H, Kaul G, Cucchiarini M, Stein U, Zurakowski D, Remberger K, Menger MD, Kohn D, Trippel SB (2005) Enhanced repair of articular cartilage defects in vivo by transplanted chondrocytes overexpressing insulin-like growth factor I (IGF-I). Gene Ther 12(15):1171–1179. doi:10.1038/sj.gt.3302515
- Madry H, Grun UW, Knutsen G (2011a) Cartilage repair and joint preservation: medical and surgical treatment options. Dtsch Arztebl Int 108(40):669–677. doi:10.3238/arztebl.2011.0669
- Madry H, Orth P, Cucchiarini M (2011b) Gene therapy for cartilage repair. Cartilage 2(3):201– 225. doi:10.1177/1947603510392914
- Madry H, Luyten FP, Facchini A (2012) Biological aspects of early osteoarthritis. Knee Surg Sports Traumatol Arthrosc 20(3):407–422. doi:10.1007/s00167-011-1705-8
- Madry H, Kaul G, Zurakowski D, Vunjak-Novakovic G, Cucchiarini M (2013) Cartilage constructs engineered from chondrocytes overexpressing IGF-I improve the repair of osteochondral defects in a rabbit model. Eur Cell Mater 25:229–247
- Mason JM, Breitbart AS, Barcia M, Porti D, Pergolizzi RG, Grande DA (2000) Cartilage and bone regeneration using gene-enhanced tissue engineering. Clin Orthop Relat Res (379 Suppl):S171–S178
- Matsumoto T, Cooper GM, Gharaibeh B, Meszaros LB, Li G, Usas A, Fu FH, Huard J (2009) Cartilage repair in a rat model of osteoarthritis through intraarticular transplantation of musclederived stem cells expressing bone morphogenetic protein 4 and soluble Flt-1. Arthritis Rheum 60(5):1390–1405. doi:10.1002/art.24443
- Mi Z, Ghivizzani SC, Lechman ER, Jaffurs D, Glorioso JC, Evans CH, Robbins PD (2000) Adenovirus-mediated gene transfer of insulin-like growth factor 1 stimulates proteoglycan synthesis in rabbit joints. Arthritis Rheum 43(11):2563–2570. doi:10.1002/1529-0131(200011)43: 11<2563::AID-ANR25>3.0.CO;2-8
- Mi Z, Ghivizzani SC, Lechman E, Glorioso JC, Evans CH, Robbins PD (2003) Adverse effects of adenovirus-mediated gene transfer of human transforming growth factor beta 1 into rabbit knees. Arthritis Res Ther 5(3):R132–R139
- Morisset S, Frisbie DD, Robbins PD, Nixon AJ, McIlwraith CW (2007) IL-1ra/IGF-1 gene therapy modulates repair of microfractured chondral defects. Clin Orthop Relat Res 462:221–228. doi:10.1097/BLO.0b013e3180dca05f
- Neumann AJ, Alini M, Archer CW, Stoddart MJ (2013) Chondrogenesis of human bone marrowderived mesenchymal stem cells is modulated by complex mechanical stimulation and adenoviral-mediated overexpression of bone morphogenetic protein 2. Tissue Eng Part A 19(11-12):1285–1294. doi:10.1089/ten.TEA.2012.0411

- Nita I, Ghivizzani SC, Galea-Lauri J, Bandara G, Georgescu HI, Robbins PD, Evans CH (1996) Direct gene delivery to synovium. An evaluation of potential vectors in vitro and in vivo. Arthritis Rheum 39(5):820–828
- Nixon AJ, Brower-Toland BD, Bent SJ, Saxer RA, Wilke MJ, Robbins PD, Evans CH (2000) Insulinlike growth factor-I gene therapy applications for cartilage repair. Clin Orthop Relat Res (379 Suppl):S201–S213
- Nixon AJ, Haupt JL, Frisbie DD, Morisset SS, McIlwraith CW, Robbins PD, Evans CH, Ghivizzani S (2005) Gene-mediated restoration of cartilage matrix by combination insulin-like growth factor-I/interleukin-1 receptor antagonist therapy. Gene Ther 12(2):177–186. doi:10.1038/sj. gt.3302396
- Noh MJ, Copeland RO, Yi Y, Choi KB, Meschter C, Hwang S, Lim CL, Yip V, Hyun JP, Lee HY, Lee KH (2010) Pre-clinical studies of retrovirally transduced human chondrocytes expressing transforming growth factor-beta-1 (TG-C). Cytotherapy 12(3):384–393. doi: 10.3109/14653240903470639
- Oh H, Chun CH, Chun JS (2012) Dkk-1 expression in chondrocytes inhibits experimental osteoarthritic cartilage destruction in mice. Arthritis Rheum 64(8):2568–2578. doi:10.1002/ art.34481
- Orth P, Kaul G, Cucchiarini M, Zurakowski D, Menger MD, Kohn D, Madry H (2011) Transplanted articular chondrocytes co-overexpressing IGF-I and FGF-2 stimulate cartilage repair in vivo. Knee Surg Sports Traumatol Arthrosc 19(12):2119–2130. doi:10.1007/s00167-011-1448-6
- Orth P, Rey-Rico A, Venkatesan JK, Madry H, Cucchiarini M (2014) Current perspectives in stem cell research for knee cartilage repair. Stem Cells Cloning 7:1–17. doi:10.2147/SCCAA. S42880
- Ortved KF, Begum L, Mohammed HO, Nixon AJ (2015) Implantation of rAAV5-IGF-I transduced autologous chondrocytes improves cartilage repair in full-thickness defects in the equine model. Mol Ther 23(2):363–373. doi:10.1038/mt.2014.198
- Palmer GD, Steinert A, Pascher A, Gouze E, Gouze JN, Betz O, Johnstone B, Evans CH, Ghivizzani SC (2005) Gene-induced chondrogenesis of primary mesenchymal stem cells in vitro. Mol Ther 12(2):219–228. doi:10.1016/j.ymthe.2005.03.024
- Pape D, Filardo G, Kon E, van Dijk CN, Madry H (2010) Disease-specific clinical problems associated with the subchondral bone. Knee Surg Sports Traumatol Arthrosc 18(4):448–462. doi:10.1007/s00167-010-1052-1
- Piera-Velazquez S, Jimenez SA, Stokes D (2002) Increased life span of human osteoarthritic chondrocytes by exogenous expression of telomerase. Arthritis Rheum 46(3):683–693. doi:10.1002/ art.10116
- Remst DF, Blaney Davidson EN, Vitters EL, Blom AB, Stoop R, Snabel JM, Bank RA, van den Berg WB, van der Kraan PM (2013) Osteoarthritis-related fibrosis is associated with both elevated pyridinoline cross-link formation and lysyl hydroxylase 2b expression. Osteoarthritis Cartilage 21(1):157–164. doi:10.1016/j.joca.2012.10.002
- Rey-Rico A, Cucchiarini M (2015) Controlled release strategies for rAAV-mediated gene delivery. Acta Biomater. doi:10.1016/j.actbio.2015.10.015
- Rey-Rico A, Frisch J, Venkatesan JK, Schmitt G, Madry H, Cucchiarini M (2015a) Determination of effective rAAV-mediated gene transfer conditions to support chondrogenic differentiation processes in human primary bone marrow aspirates. Gene Ther 22(1):50–57. doi:10.1038/ gt.2014.90
- Rey-Rico A, Venkatesan JK, Frisch J, Rial-Hermida I, Schmitt G, Concheiro A, Madry H, Alvarez-Lorenzo C, Cucchiarini M (2015b) PEO-PPO-PEO micelles as effective rAAV-mediated gene delivery systems to target human mesenchymal stem cells without altering their differentiation potency. Acta Biomater 27:42–52. doi:10.1016/j.actbio.2015.08.046
- Rey-Rico A, Venkatesan JK, Frisch J, Schmitt G, Monge-Marcet A, Lopez-Chicon P, Mata A, Semino C, Madry H, Cucchiarini M (2015c) Effective and durable genetic modification of human mesenchymal stem cells via controlled release of rAAV vectors from self-assembling peptide hydrogels with a maintained differentiation potency. Acta Biomater 18:118–127. doi:10.1016/j.actbio.2015.02.013

- Roessler BJ, Hartman JW, Vallance DK, Latta JM, Janich SL, Davidson BL (1995) Inhibition of interleukin-1-induced effects in synoviocytes transduced with the human IL-1 receptor antagonist cDNA using an adenoviral vector. Hum Gene Ther 6(3):307–316. doi:10.1089/ hum.1995.6.3-307
- Santangelo KS, Nuovo GJ, Bertone AL (2012) In vivo reduction or blockade of interleukin-1beta in primary osteoarthritis influences expression of mediators implicated in pathogenesis. Osteoarthritis Cartilage 20(12):1610–1618. doi:10.1016/j.joca.2012.08.011
- Scharstuhl A, Diepens R, Lensen J, Vitters E, van Beuningen H, van der Kraan P, van den Berg W (2003) Adenoviral overexpression of Smad-7 and Smad-6 differentially regulates TGF-betamediated chondrocyte proliferation and proteoglycan synthesis. Osteoarthritis Cartilage 11(11):773–782
- Shen PC, Shiau AL, Jou IM, Lee CH, Tai MH, Juan HY, Lin PR, Liu GS, Wu CL, Hsieh JL (2011) Inhibition of cartilage damage by pro-opiomelanocortin prohormone overexpression in a rat model of osteoarthritis. Exp Biol Med 236(3):334–340. doi:10.1258/ebm.2010.010319
- Shi S, Mercer S, Eckert GJ, Trippel SB (2012) Regulation of articular chondrocyte aggrecan and collagen gene expression by multiple growth factor gene transfer. J Orthop Res 30(7):1026– 1031. doi:10.1002/jor.22036
- Shi S, Mercer S, Eckert GJ, Trippel SB (2013) Growth factor transgenes interactively regulate articular chondrocytes. J Cell Biochem 114(4):908–919. doi:10.1002/jcb.24430
- Shi S, Wang C, Acton AJ, Eckert GJ, Trippel SB (2015) Role of sox9 in growth factor regulation of articular chondrocytes. J Cell Biochem 116(7):1391–1400. doi:10.1002/jcb.25099
- Shuler FD, Georgescu HI, Niyibizi C, Studer RK, Mi Z, Johnstone B, Robbins RD, Evans CH (2000) Increased matrix synthesis following adenoviral transfer of a transforming growth factor beta1 gene into articular chondrocytes. J Orthop Res 18(4):585–592. doi:10.1002/ jor.1100180411
- Sieker JT, Kunz M, Weissenberger M, Gilbert F, Frey S, Rudert M, Steinert AF (2015) Direct bone morphogenetic protein 2 and Indian hedgehog gene transfer for articular cartilage repair using bone marrow coagulates. Osteoarthritis Cartilage 23(3):433–442. doi:10.1016/j. joca.2014.11.008
- Smith P, Shuler FD, Georgescu HI, Ghivizzani SC, Johnstone B, Niyibizi C, Robbins PD, Evans CH (2000) Genetic enhancement of matrix synthesis by articular chondrocytes: comparison of different growth factor genes in the presence and absence of interleukin-1. Arthritis Rheum 43(5):1156–1164. doi:10.1002/1529-0131(200005)43:5<1156::AID-ANR26>3.0.CO;2-M
- Steinert AF, Weissenberger M, Kunz M, Gilbert F, Ghivizzani SC, Gobel S, Jakob F, Noth U, Rudert M (2012) Indian hedgehog gene transfer is a chondrogenic inducer of human mesenchymal stem cells. Arthritis Res Ther 14(4):R168. doi:10.1186/ar3921
- Stender S, Murphy M, O'Brien T, Stengaard C, Ulrich-Vinther M, Soballe K, Barry F (2007) Adeno-associated viral vector transduction of human mesenchymal stem cells. Eur Cell Mater 13:93–99. discussion 99
- Surendran S, Kim SH, Jee BK, Ahn SH, Gopinathan P, Han CW (2006) Anti-apoptotic Bcl-2 gene transfection of human articular chondrocytes protects against nitric oxide-induced apoptosis. J Bone Joint Surg 88(12):1660–1665. doi:10.1302/0301-620X.88B12.17717
- Ulrich-Vinther M, Stengaard C, Schwarz EM, Goldring MB, Soballe K (2005) Adeno-associated vector mediated gene transfer of transforming growth factor-beta1 to normal and osteoarthritic human chondrocytes stimulates cartilage anabolism. Eur Cell Mater 10:40–50
- Venkatesan N, Barre L, Benani A, Netter P, Magdalou J, Fournel-Gigleux S, Ouzzine M (2004) Stimulation of proteoglycan synthesis by glucuronosyltransferase-I gene delivery: a strategy to promote cartilage repair. Proc Natl Acad Sci U S A 101(52):18087–18092. doi:10.1073/ pnas.0404504102
- Venkatesan JK, Rey-Rico A, Schmitt G, Wezel A, Madry H, Cucchiarini M (2013) rAAV-mediated overexpression of TGF-beta stably restructures human osteoarthritic articular cartilage in situ. J Transl Med 11:211. doi:10.1186/1479-5876-11-211
- Vogt S, Wexel G, Tischer T, Schillinger U, Ueblacker P, Wagner B, Hensler D, Wilisch J, Geis C, Wubbenhorst D, Aigner J, Gerg M, Kruger A, Salzmann GM, Martinek V, Anton M, Plank C,

Imhoff AB, Gansbacher B (2009) The influence of the stable expression of BMP2 in fibrin clots on the remodelling and repair of osteochondral defects. Biomaterials 30(12):2385–2392. doi:10.1016/j.biomaterials.2009.01.016

- Wang D, Taboas JM, Tuan RS (2011) PTHrP overexpression partially inhibits a mechanical straininduced arthritic phenotype in chondrocytes. Osteoarthritis Cartilage 19(2):213–221. doi:10.1016/j.joca.2010.11.003
- Weimer A, Madry H, Venkatesan JK, Schmitt G, Frisch J, Wezel A, Jung J, Kohn D, Terwilliger EF, Trippel SB, Cucchiarini M (2012) Benefits of recombinant adeno-associated virus (rAAV)mediated insulinlike growth factor I (IGF-I) overexpression for the long-term reconstruction of human osteoarthritic cartilage by modulation of the IGF-I axis. Mol Med 18:346–358. doi:10.2119/molmed.2011.00371
- Yla-Herttuala S (2012) Endgame: glybera finally recommended for approval as the first gene therapy drug in the European union. Mol Ther 20(10):1831–1832. doi:10.1038/mt.2012.194
- Zhang X, Mao Z, Yu C (2004) Suppression of early experimental osteoarthritis by gene transfer of interleukin-1 receptor antagonist and interleukin-10. J Orthop Res 22(4):742–750. doi:10.1016/j.orthres.2003.12.007
- Zhang P, Zhong ZH, Yu HT, Liu B (2015) Exogenous expression of IL-1Ra and TGF-beta1 promotes in vivo repair in experimental rabbit osteoarthritis. Scand J Rheumatol 44(5):404–411. doi:10.3109/03009742.2015.1009942

Tissue Engineering Strategies for Cartilage Repair

10

Holger Jahr

Abstract

Cartilage repair addresses several facets of the diversity in our present approaches to regenerate articulating surfaces. The following chapter now summarizes several promising options to engineer articular cartilage-like constructs, ranging from applying biological factors and/or mechanical, magnetic, or even electrical stimuli. The paradigm of cartilage tissue engineering classically comprises three pillars: cells, scaffolds, and signals. As cell sources for cartilage repair are addressed by other chapters in this volume, the next pages will focus on the two remaining pillars: first, due to their importance for the subsequent tissue engineering path, scaffold-free and scaffold-based applications are distinguished. Although most classical techniques in the field are scaffold based, relative more attention is now paid to emerging scaffold-free methods as articular cartilage repair constructs. Only proper tissue organization will permit long-term functional durability, and mimicking tissue growth without artificial support structures holds a lot of potential. While the extracellular matrix is an integral aspect of the tissue properties, it also impedes the integration of the repair construct into the surrounding host tissue. Several approaches to tackle this dilemma are depicted. The importance to develop bioreactors is also emphasized as they are inevitable for the reproducible application of sophisticated mechanobiological stimulation regimes. In this context, the contribution of selected growth factors is described. Towards the end of the present chapter, the importance of integrating multiple of these parameters into multimodal concepts for achieving phenotypic stability of the engineered cartilagelike constructs is addressed.

H. Jahr

RWTH University Hospital Aachen, Department of Orthopaedic Surgery, Pauwelsstraße 30, 52074 Aachen, Germany e-mail: hjahr@ukaachen.de

[©] Springer International Publishing AG 2017

S. Grässel, A. Aszodi (eds.), Cartilage, DOI 10.1007/978-3-319-53316-2_10

10.1 Introduction

Of the three types of cartilage (i.e. elastic, fibro-, and hyaline), this chapter will focus on hyaline articular cartilage due to its socioeconomic relevance and its importance in withstanding mechanical loads.

Chondrocytes reside in a complex, multiphasic extracellular matrix (ECM) described by three principal phases: a solid phase, a fluid phase, and an ion phase (Mow et al. 1999). Their interaction results in a complex mechano-electrochemical environment in which chondrocytes are exposed to multiple biophysical stimuli including mechanical forces (i.e. tension, compression, shear), fluid flow, hydrostatic and osmotic pressure gradients, and electrokinetic events (Brady et al. 2015b). The mechanism by which these cells convert mechanical signals into biochemical responses is called mechanotransduction (Ingber 2006), and it is well accepted that chondrocytes directly respond to mechanical forces (Wong and Carter 2003; Brady et al. 2015b).

Our current knowledge of these biophysical cues can be readily applied to engineer neocartilage. However, hyaline cartilage remains one of the most challenging tissues to replace, owing to its mechanical strength at articulating surfaces. Limited biomechanical properties and poor post-implantation integration with the host are major challenges in cartilage tissue engineering (TE). To this end, cartilage TE aims at delivering ex vivo generated constructs that are biologically mature and mechanically functional from day one to potentially decrease the recovery time for the patient.

In this chapter, general strategies for cartilage TE will be covered. While most commonly used adult stem cells for cartilage TE may be MSCs derived from bone marrow (BMSC), adipose (ADSC), and synovium (SDSC), appropriate cell sources for cartilage TE are dealt with in other chapters in this volume (details see Chaps. 2–4). While scaffold-based techniques are only briefly described, emphasis is put on recent scaffold-free approaches. The importance of tissue structure and bioactive compounds is briefly elaborated. Different currently reported bioreactor concepts to realize cartilage TE are discussed as is the role of mechanobiological aspects. Towards the end, other physicochemical factors are revealed as are issues with their integration and phenotypic stability upon implantation.

10.2 Medical Need for Articular Cartilage Tissue Engineering

Degeneration of articular hyaline cartilage may occur due to trauma and metabolic or mechanical deficits leading to osteoarthritis. Although current options for cartilage repair are reasonably effective to alleviate pain, they have their respective limitations (Huey et al. 2012). A major clinical need thus exists for cartilage repair and regeneration, but identifying optimal cell types and robust pretreatment conditions remains challenging. Currently, pre-culture under chondrogenic conditions seems necessary to maintain a long-term chondrocytic phenotype.

Thus, effects of growth factors (GFs) known to play key roles in cartilage development, either alone or in combination, have been investigated for maintenance of the chondrogenic phenotype and for promoting cartilage formation in vitro.

10.3 General Strategies for Cartilage Tissue Engineering

Besides a plethora of biological factors, mechanical, magnetic, and electrical stimuli have been used to promote proliferation, differentiation, and maturation of chondrocytes within established dose parameters or "biological windows". While cells are crucial for tissue regeneration as they are responsible for synthesizing the ECM molecules that reshape the tissue structure and confer mechanical properties, for deeper insights, the reader is referred to more specialized chapters in this volume.

Articular cartilage withstands compressive, tensile, and shear loads as part of its function, and it is also lubricious, resulting in a tissue with very low coefficient of friction. Cartilage's biochemical content, primarily water, collagen, and proteogly-cans (PGs), and the specific organization of these molecules within the ECM allow for appropriate load distribution and transmission. One of the major design criteria in cartilage TE is therefore the creation of neocartilage with biomechanical properties that will withstand the demanding mechanical environment in vivo. Proper organization of the neocartilage is anticipated to allow for long-term functionality and durability.

10.4 Scaffolds for Cartilage Tissue Engineering

10.4.1 Scaffold-Based Applications

The choice of biomaterials and scaffold design is an important consideration for successful cartilage TE. Briefly, natural materials, like agarose, alginate, hyaluronic acid, fibrin, collagen derivatives (e.g. gelatine), or de-cellularized matrices, are being used (Raghunath et al. 2007). Though being generally attractive, disadvantages like inferior mechanical strength, antigenicity, and potential risk of disease transfer as well as rapid and variable host-specific degradation times have to be considered.

As the encapsulation forces of hydrogels, i.e. hydrophilic polymers with high water content and elastic nonadhesive properties, encourage cells to assume chondrocytic morphologies, they have been frequently applied in cartilage TE. However, their swelling from water uptake and inferior mechanical properties is a drawback.

With synthetic materials, on the other hand, scaffold properties are much easier to control and modify. The currently used product portfolio encompasses polyhydroxy acids (e.g. PLLA, PGA, PCL), poly(ethylene)glycol, or elastomeric polyurethanes (Raghunath et al. 2007; Camarero-Espinosa et al. 2016). Most, if not all, synthetic biopolymers lack bioactive molecules that promote cell attachment, proliferation, and differentiation. By combining materials, synthetic biopolymers can potentially be tailored to provide the bulk properties, degradation profiles, and structures, while natural materials could provide bioactive molecules needed to activate desired signalling pathways. An exponentially increasing number of natural, synthetic, or composite polymeric scaffold materials with or without bioactive coatings or structural surface treatment exist. As these cannot be reviewed in the present chapter, the reader is referred to excellent overview articles (Darling 2013; Camarero-Espinosa et al. 2016).

10.4.2 Scaffold-Free Techniques

The traditional scaffold-based TE paradigm consists of cells, signals, and scaffolds (Fig. 10.1), but recent scaffold-free approaches consisting of just cells and signals can also be attractive. To make a clinically relevant scaffold-free tissue, TE



Fig. 10.1 Traditional cartilage TE paradigm. Cells, signals, and scaffolds are classical pillars of the three main cartilage regenerative strategies: (1) scaffold-based chemotaxis of host cells, (2) cell-seeded (biomimetic) scaffolds in combination with diverse signals, and (3) cell-free techniques. Recently, scaffold-free methods gained special attention. MSCs and chondrocytes (or combinations of both) appear to be best described cell sources. Neocartilage can be cultured ex vivo under different extracellular (i.e. biological, biophysical, mechanobiological) stimulation protocols for which bioreactor systems may be required. Modified after Makris et al. (2015). *AMIC* autologous matrix-induced chondrogenesis, *ECM* extracellular matrix, *DCM* de-cellularized ECM, *iPSC* induced pluripotent stem cells, *MACI* matrix-assisted autologous chondrocyte implantation, *MSC* mesenchymal stem cells, *pO2* partial oxygen pressure, *pH* potentia hydrogenii, *PRP* platelet-rich plasma, *SRM* synergistic regenerative medicine

considerations must include cell sourcing, stimulation of tissue-specific ECM production, and tissue organization. As scaffold-free approaches lack the exogenous material of scaffold-based methods, the resulting engineered cartilages (neocartilages) unfortunately commonly require large cell numbers. On the other hand, by mirroring cartilage development, only cell-secreted ECM contributes to neotissue properties. As cartilage is heterogeneous and anisotropic, achieving proper tissue morphology and organization is crucial for success (DuRaine et al. 2015).

This may be addressed by moulding or in vitro confining, assembling tissues as building blocks, and promoting their fusion to form higher-order structures. Applying mechanical stimulation to mature, the ECM may create functional anisotropy in these constructs (MacBarb et al. 2013). DuRaine and colleagues recently reviewed three promising scaffold-free TE principles: cell sheet engineering, aggregate engineering, and self-assembling process (DuRaine et al. 2015).

Cell sheets are cohesive monolayers and well known for their application as skin transplants from keratinocytes. This idea can be applied to chondrocytes, too, with subsequent draping, layering, or rolling to increase the application spectrum. While a phase III clinical trial using sheets of expanded juvenile allogeneic chondrocytes has been started (DuRaine et al. 2015), the technique has its limitations: chondrocytes are known to dedifferentiate during monolayer culture and progress towards a fibroblastic phenotype (Darling et al. 2009). In addition, harvesting and production of thicker, multicellular tissues may require substantial handling, too.

Aggregate constructs (also referred to as pellet cultures) commonly form by applying a rotational force to cells in suspension (Furukawa et al. 2003). This method is used to (re)differentiate cells or to form cartilaginous microtissue. The phenotype of chondrocyte aggregates is similar to that of native cartilage and thus thought to mirror cell aggregation and matrix production during cartilage development.

Recently, pellet culture of human MSCs in a demineralized bone matrix was used to produce engineered articular cartilage with a physiologically relevant compressive Young's modulus of about 800 kPa and an equilibrium friction coefficient of ~0.28 (Bhumiratana et al. 2014). Limitations of this technique are rather uncontrolled and nonhomogeneous shapes (Gigout et al. 2009) and potentially compromised core cell viability with subsequent loss of cell-type homogeneity. Also here, a phase III clinical trial was started in Europe, and 1-year follow-up results after treatment of full-thickness patella-femoral or femoral-condylar defects are promising (Fickert et al. 2012).

Self-assembling differs from aggregate formation in the way the tissue forms and the properties of the resulting ECM. This technique is applicable to different cell sources and the ECM composition, and its organization in the neocartilage appears similar to that of native cartilage (DuRaine et al. 2015). The adaptability and reproducibility of self-assembling processes make them highly promising to produce tissues with physiologically relevant properties. While successfully producing functional neocartilage, drawbacks are that (1) cells must be amenable to producing large amounts of ECM and (2) survive minimal cell-substrate interactions as the ECM accumulates during the initial phase of the self-assembling process. In
addition, self-assembling requires huge amounts of cells (Huey et al. 2012) which, in turn, currently requires monolayer expansion with subsequent redifferentiation. Also here, construct size of the engineered tissue is limited by diffusion (Leddy et al. 2004). However, self-assembling recapitulates developmental processes while allowing to control tissue geometry to create constructs with biologically reminiscent properties (Ofek et al. 2008).

10.5 Mimicking Tissue Structure and Properties

Articular cartilage withstands compressive, tensile, and shear loads and also has a very low coefficient of friction. Cartilage's biochemical content, primarily water, collagen, and PGs, and the organization of these molecules allow for appropriate load distribution and transmission. Therefore, one of the major design criteria in cartilage TE is the creation of neocartilage with biomechanical properties that will withstand the demanding mechanical environment in vivo.

Reported mechanical properties of natural cartilage include an aggregate modulus (0.1–2 MPa), hydraulic permeability ($10^{-16} - 10^{15} \text{ m}^4/\text{Ns}$), compressive Young's modulus (0.24–0.85 MPa), Poisson's ratio (0.06–0.3), tensile equilibrium modulus (5–12 MPa), tensile Young's modulus (5–25 MPa), tensile strength (0.8–25 MPa), equilibrium shear modulus (0.05–0.4 MPa), complex shear modulus (0.2–2.5 MPa), and a shear loss angle (10° – 15°) (Little et al. 2011). Its coefficient of friction ranges from 0.03 to 0.06 (DuRaine et al. 2009).

Therefore, proper tissue organization is believed to permit long-term functionality and durability of the neocartilage.

10.6 Bioactive Approaches

10.6.1 Growth Factors and Mechanical Stimuli

Improving the tensile properties of neocartilage has been one of the major challenges in cartilage TE (Makris et al. 2014b). Among the different GFs used in articular cartilage engineering, transforming growth factor (TGF)- β may be most prominently improving tensile properties, with 30 ng/ml continuously applied TGF- β 1 increasing tensile properties by twofold as a result of increased collagen synthesis (Elder and Athanasiou 2009b; Mauck et al. 2003).

TGF- β 1 is also able to enhance the stimulatory effect of dynamic, unconfined compression (i.e. 0–10% strain, 1 Hz) on chondrocytes seeded in agarose hydrogels by increasing collagen production by approximately twofold (Mauck et al. 2003). TGF- β 1 or insulin-like growth factor (IGF)-1 resulted in a three- to fourfold increase in the aggregate modulus for both GFs, as well as in collagen and glycosaminogly-can (GAG) synthesis, respectively (Mauck et al. 2003). TGF- β 3 release from poly(lactide-co-caprolactone) (PLCL) scaffolds increased the compressive properties of the engineered cartilage gradually from ~300 kPa at 4 weeks to ~400 kPa at

8 weeks and to ~550 kPa at 12 weeks (Kim et al. 2015). It is, however, noteworthy that, in general, changes in biomechanical properties do not scale linearly with changes in biochemical components. This suggests an important role for interactions between matrix components and matrix organization in determining biomechanical tissue properties. While, in cartilage TE, using anabolic GFs in combination with mechanical cues is still a promising approach to enhance the compressive properties of neocartilage, improvements in tensile properties due to these stimulation regimes are usually substantially lower than values derived from native cartilage (Elder and Athanasiou 2008).

As a result, biophysical agents have been investigated to further improve the tensile properties of neocartilage. As an example, chondroitinase ABC (c-ABC) cleaves chondroitin and dermatan sulphate and depletes GAGs (Natoli et al. 2009b; Asanbaeva et al. 2007), which upon recovery during subsequent in vitro culture restore the compressive mechanical properties (Natoli et al. 2009a, b). Counterintuitively, persistent effects of c-ABC treatment with enhanced collagen production and subsequently increased tensile properties were observed (Natoli et al. 2009a), and multiple c-ABC applications had an additive effect (Natoli et al. 2009a; O'Connell et al. 2014) of, most likely, biophysical nature (Responte et al. 2012). At this point, additional studies are required to elucidate the role of c-ABC in restoring tensile properties in engineered cartilage.

10.6.2 Gene Therapy

An increasing number of potentially therapeutic genes have emerged as tools in cartilage TE. These genes may not only be administered to cells as soluble factors but also through ex vivo or in vivo approaches using viral and non-viral vectors. The latter would allow for sophisticated and potentially better controllable, inducible longitudinal expression. Gene transfer may further occur through gene-activated matrices, thereby employing state-of-the-art scaffolds. Although gene therapy has converged with cartilage TE in recent years, genetic approaches bear certain risks which have to be well balanced against their potential benefits. For a recent comprehensive overview, please see elsewhere (Hu 2014).

10.7 Bioreactors

Most TE is performed ex vivo, while some promising in vivo bioreactor concepts have also been reported (Stevens et al. 2005), at least for the bone. In cartilage TE, bioreactors have been mainly used for three applications: (1) cultivation/proliferation of (suspended) cells prior to seeding into scaffolds, (2) maintaining vitality of cells upon seeding on/into porous scaffolds, and (3) achieving specific physical stimulation (Gelinsky et al. 2015; Martin et al. 2004; Hansmann et al. 2013). It is difficult to manually expand cell under good manufacturing practice (GMP) conditions, a prerequisite for clinical translation, and expansion bioreactors are

cost-efficient and safe alternatives (King and Miller 2007). Large-scale production of MSCs has been recently reviewed (Jung et al. 2012), with microcarrier seeding technologies reported by Chen et al. (2013). Physical stimulation in bioreactors can be "passive" or "active", where passive could be the stiffness of the matrix the cells adhere to (Rehfeldt et al. 2007) and active may refer to mechanical, electrical, or magnetic stimulation. MSC differentiation, for example, is strongly influenced by microenvironmental stiffness and mechanical cues (Wang and Chen 2013) alike. Anyhow, giving a concise overview of the available bioreactor options is difficult due to the rapidly developing field, and a good overview has recently been published (Jin et al. 2015). Bioreactor systems play an important role in TE, as they enable reproducible and controlled changes in specific environmental factors. They can provide technical means to perform controlled studies to understand specific biological, chemical, or physical effects. To date the goals and expectations of bioreactor development have been fulfilled only to some extent, as bioreactor design in TE is very complex and still at an early stage of development.

10.7.1 Mechanobiological Aspects

While several other mechanical tissue properties of cartilage are rather well described, studies also reporting tensile or shear properties are surprisingly hard to find.

Tensile strains, inherent in articular cartilage, are a major contributor to the mechanical functionality of this tissue. "Biaxial" tension, or stretch, has been applied through thin, cell-seeded membranes in either radial or circumferential direction (Fan and Waldman 2010). Here, not only the magnitude of stain but also its nature and the differentiation state of the chondrocytes do matter (Das et al. 2008b). Although being biologically relevant, several technical challenges restrict the use of Flexcell-like systems (Jansen et al. 2004, 2006; Garvin et al. 2003), and tension bioreactors will thus not be further discussed.

Shear is a naturally occurring physiological condition of mechanical stimulation in a functional joint. Three general categories of "high-shear" bioreactors (i.e. contact shear, fluid shear, and perfusion shear) have been investigated in cartilage TE. As yet, employing *contact shear* or sliding shear in cartilage TE has been reported in only a few instances, and the interested reader is therefore referred to the work by Wang and colleagues (Wang et al. 2013, 2014) or a recent excellent overview (Darling 2013). Although some reports for articular cartilage exist, *fluid shear* appears typically rather to be applied in vascular TE (Darling 2013), where much work has gone into optimizing cell seeding of scaffolds in, e.g. spinner flasks. Both positive and negative results are reported, depending on the level of shear applied. *Perfusion shear* can be simply applied by flowing media steadily through a chamber containing cells or cell-seeded scaffolds. Of note, direct perfusion bioreactors can be employed to align cells in the direction of flow (Pazzano et al. 2000), but even very low shear levels (0.092 Pa or 0.92 dyne/cm²) can adversely affect cells (Goodwin et al. 1993) and chondrocytes appear to be particularly sensitive to turbulences (Darling 2013). *Cyclic compression* and *hydrostatic pressure* are the most common mechanical stimulation techniques in cartilage TE (Gelinsky et al. 2015). Rather broad regimes ranging from 0.0001 to 3 Hz and compressive stresses ranging from 0.1 to 24 MPa and between 0.1 and 25% of strain have been described (Freyria et al. 2005).

Hydrostatic pressure (HP), similar to direct compression, is a stimulus that improves the compressive and biochemical properties of engineered cartilage (Elder and Athanasiou 2009a). Combined with TGF- β 3 in a chondrogenesis model using MSCs, intermittent HP of 10 MPa at 1 Hz stimulated mRNA expression of SOX-9, collagen II, and aggrecan (Miyanishi et al. 2006). Combined with bone morphogenic protein (BMP)-2 and IGF-1, 10 MPa static HP in neocartilage, engineered from articular chondrocytes, increased aggregate and tensile modulus values by 17 and 30%, respectively, were reported (Elder and Athanasiou 2008). The same HP regimen also improved aggregate and tensile moduli when applied with TGF- β 1. Since HP can be used before engineered cartilage develops robust mechanical characteristics, HP can be applied earlier than direct compression during TE of articular cartilage.

Microgravity High-shear perfusion can successfully stimulate cartilage matrix production, but the resulting tissue is often of a fibrous nature. In contrast, slower fluid flow seems to have a stimulatory effect on ECM synthesis while preserving the chondrocyte phenotype. Dynamic laminar flow in rotating (wall) bioreactors provides efficient oxygen supply and presents an attractive tool for TE studies (Darling 2013). The major difference between rotating bioreactors and past perfusion systems is the reduction in shear force. Stress exerted on a construct rotating at 19 rpm was calculated to approximately 0.15 Pa (Freed and Vunjak-Novakovic 1995), which is 300 times higher than that measured on microbeads but also significantly lower than that in fluid flow bioreactors (Darling 2013). The mass transfer enhancement by the rotation seems crucial to the success of this bioreactor type in cartilage TE (Marlovits et al. 2003a, b). Considering the scaffold-bioreactor compatibility, physical, mechanical, and material characteristics of the used scaffold material logically exclude certain types of bioreactors, while rotating bioreactors are generally saved to use for a wider range of applications. A potential limitation of this type of bioreactor is the random motion of the scaffolds in the vessel.

Mechanotransduction through the cytoskeleton may begin with integrinmediated adhesions that transmit forces from the ECM to cytoskeletal filaments (Wang et al. 2009). Wang et al. postulated that cadherins may have a role in mechanotransduction due to their close interaction with both integrins and kinase receptors. Along this line, disturbances of microtubule organization further prevented the stimulatory effect of HP loading on PG synthesis (Jortikka et al. 2000). Specifically, microtubule depolymerization induced by nocodazole, an anti-polymerization agent, inhibited the usually observed increase in PG synthesis upon applying 5 MPa cyclic hydrostatic pressure at 0.5 Hz (Jortikka et al. 2000).

It should also be noted that even sophisticated bioreactor systems rarely apply only a single type of mechanical stimulus. From a biological point of view, this may more closely approximate the native environment as articular cartilage likely experiences multiple types of loading at a time but also obstructs drawing hard conclusions about the primary stimulating force. While it is generally accepted that mechanotransduction pathways are involved in articular cartilage homeostasis, the exact role of mechanosensors in this process is yet to be determined. Current knowledge indicates the beginning of an exciting era where modulating these pathways would potentially result in significantly improved properties of engineered cartilage.

10.7.2 Physicochemical and Other Factors

While traditionally being studied rather in bone TE, electrical and magnetic stimulation protocols found their way into cartilage TE, too. Recent clinical benefits in relieving pain have been reviewed by Zeng and colleagues (Zeng et al. 2015). A hypothesis is that electromagnetic pulses may control inflammation and stimulate anabolic activities. Our knowledge of electromechanical signals in articular cartilage was recently excellently reviewed by Brady et al. (2015a, b), also describing the effect of biophysical, magnetic, and electrical stimulation regimes in cartilage TE.

pH Control of chondrocyte pH (pH(i)) determines ECM metabolism, and several transporters have been implicated in this process, which further revealed cartilage zone dependency in situ (Simpkin et al. 2007; Hall et al. 1996; Browning and Wilkins 2004). Next to increasing the concentrations of cations and increasing extracellular osmolarity, daily joint loading also reduces extracellular pH in cartilage (Wilkins et al. 2000), further pointing to an important regulatory function of pH in controlling cartilage homeostasis. Bioreactors are well suited to control several microenvironmental parameters simultaneously and were employed to show that even subtle fluctuations in extracellular pH (and oxygen tension) influence chondrocyte metabolism and marker expression on mRNA and protein level (Das et al. 2008a, 2010b).

Hypoxia In most tissues, "hypoxia" is a non-physiological stage and might inhibit robust matrix synthesis and cell proliferation. In contrast, low oxygen tension is a more physiological environment not only to BMSCs (Das et al. 2010a) but certainly also to adult cartilage-derived cells. In cartilage, chondrocytes appear adapted to this *physoxia* (i.e. physiologically low oxygen pressure). However, higher oxygen tensions, promoting chondrocyte proliferation, may be used to stimulate cellular expansion when seeding scaffolds (Schrobback et al. 2012). Once sufficient cells have populated the scaffold, chondrogenic differentiation can be promoted, or the chondrocytic phenotype be stabilized, by switching to lower oxygen tensions (Das et al. 2008a, 2010b; Heywood et al. 2010; Malda et al. 2004; Saini and Wick 2004). To this end, a high adaptability of specialized bioreactors in a standardized production process seems a promising strategy to engineer cartilage under reduced oxygen tension (Portner et al. 2009). In multiple studies, chondrocytic gene expressions of

collagen II and aggrecan were elevated in 5% oxygen compared to 20% oxygen, whereas expression of collagens I and X was suppressed (Das et al. 2010b; Schrobback et al. 2012) and TGF- β signalling was shown to participate in preserving the chondrocyte phenotype under hypoxia (Das et al. 2015). It should be noted that even for chondrocytes, excessively hypoxic environments may compromise glycosaminoglycan (GAG) synthesis (Murphy and Sambanis 2001). Next to a direct role of partial oxygen pressures in cartilage TE, manipulating nuclear factor (ery-throid-derived 2)-like 2 activity and thus redox homeostasis may also hold promises for cartilage TE (Jahr 2015).

Osmolarity Tonicity is a measure of the effective osmotic pressure gradient, and chondrocytes are cells under such a constant pressure in situ (Urban 1994). Chronic exposure to anisotonic conditions through cyclic changes in extracellular osmolarity during daily joint loading subjects chondrocytes to swelling or shrinking, respectively. Intriguingly, Leijten and coworkers identified BMP and WNT signalling antagonists, like frizzled-related protein (FRZB), as natural inhibitors of hypertrophic differentiation (Leijten et al. 2013). In their study, effects of intermittent cyclical loading (0.5 MPa, 10 N) at 0.33 Hz and increasing medium osmolarity to 380 mOsm/kg had a very similar effect on FrzB mRNA abundances (Leijten et al. 2013). Sub-physiological osmolarities (i.e. plasma level osmolarities around 280 mOsm/kg) rather appear to encourage proliferation of chondrocytes while inhibiting ECM production (Xu et al. 2010). In contrast, hyper-osmolarity stabilizes SOX-9 mRNAs (Tew et al. 2009). This for chondrocytes rather physiological osmolarity is able to improve chondrocyte marker expression in human chondrocytes (van der Windt et al. 2010a). Under such conditions, pharmacological intervention, e.g. calcineurin inhibition, can selectively improve anabolic chondrocyte markers while suppressing catabolic ones in vitro (van der Windt et al. 2012). While osmolarity also seems to be an easy means to modulate in vitro chondrogenic differentiation capacity of progenitor cells (Caron et al. 2013b), calcineurin inhibitors may promote chondrogenic marker expression through stimulating the TGF-β pathway (van der Windt et al. 2010b). Of note, using an agonist of the transient receptor potential cation channel subfamily V member (TRPV)4 in combination with hyperosmotic stimulation synergistically improved compressive and tensile stiffness of engineered neocartilage (Lee et al. 2014). Recently, it was further shown that osmolarity directly beneficially influences chondrocyte repair after injury of human articular cartilage (Huang et al. 2015) by, among others, protecting against apoptosis (Amin et al. 2008). This may thus not only have implications for cartilage TE but for cartilage repair surgery, too (Eltawil et al. 2015). The emerging "channelome" of chondrocytes identified several candidates that potentially mediate osmotic stress response and volume changes in these cells (Barrett-Jolley et al. 2010) and paved the road for future improvements in this direction.

As the field of TE matures, more sophisticated bioreactors, enabling largely automated, non-supervised, and feedback-regulated parameter controls, will become available. Co-applying compressive and frictional forces, for instance, seems a promising future strategy (Stoddart et al. 2006; Shahin and Doran 2015).

Despite the progress in engineering biologically functional cartilage tissue in vitro with the aid of signalling molecules and bioreactors, challenges remain for its successful clinical translation. Issues often arise after the transplantation of engineered tissue in vivo, including phenotypic instability and poor integration.

10.8 Phenotypic Stability

The phenotypic instability of engineered cartilage is currently still a clinical problem. Undesired cell phenotypes potentially lead to the formation of tissues with deficient biological and biomechanical functions, often resulting from fibrous cartilage formation. In stem cell-based treatments, in vitro chondrogenesis often results in hypertrophic differentiation with increased type X collagen expression in vivo, which may cause mineralization of the engineered construct (Vinardell et al. 2012). To suppress hypertrophic differentiation, several molecules have been explored; in addition to parathyroid hormone (PTH) and parathyroid hormone-related peptide (PTHrP), BMP-7 may inhibit collagen X expression in human MSCs while further inducing chondrogenic differentiation (Caron et al. 2013a). As also osteoarthritic chondrocytes often express collagen X (von der Mark et al. 1992) and this phenotype may be induced by several cytokines, GFs, or ECM degradation products (Pitsillides and Beier 2011), cell-based TE constructs are at risk facing a similar fate upon implantation in an inflamed environment. Understanding corresponding signalling pathways will be pivotal towards overcoming the challenge of phenotypic instability of engineered cartilage.

10.9 Integration

Finally, failure of integration between engineered tissue and surrounding native cartilage continues to be a fundamental problem in the field. Cartilage lesions can also extend into the subchondral bone, making cartilage to cartilage, cartilage to bone, and bone to bone integration processes relevant for the clinical translation of cartilage implants. Several factors can influence the ability of the repair tissue to integrate with the native cartilage, such as cell death at the wound edge, the phenotype of the cells in the implanted tissue, and donor age (Khan et al. 2008). In addition, the degree of maturation of engineered constructs affects integration (Obradovic et al. 2001). Traditionally, cartilage TE aimed at stimulating the production of collagens and GAGs at levels similar to the native tissue. Ironically, the ECM can prevent adhesion and diffusion of cells and matrix proteins (Hunziker and Kapfinger 1998; Rice et al. 2008) and thus impede the engineered tissue from integrating (Rice et al. 2008). Disrupting certain matrix molecules via enzymatic treatment can enhance integration; for example, collagenase and hyaluronidase applied to the wound site increased chondrocyte density and improved cartilage integration (van de Breevaart Bravenboer et al. 2004). Furthermore, β -aminopropionitrile (BAPN) may enhanced integration or "priming" the implant with exogenous lysyl oxidase homolog (LOXL)2

(McGowan and Sah 2005; Makris et al. 2014a). As mentioned above, using c-ABC or trypsin was shown to be promising, too (Obradovic et al. 2001; Hunziker and Kapfinger 1998). These enzymes can further be combined with anabolic factors to counteract their potential catabolic effects (Responte et al. 2012) or with factors disrupting ECM formation, like IGF-1, BAPN, and para-nitrophenyl- β -D-xyloside (Bastiaansen-Jenniskens et al. 2009). Although the presence of matrix components, such as GAG and collagens, in engineered cartilage is necessary to withstand stresses in vivo, only their temporal absence may allow for robust integration.

Conclusions

Current articular cartilage TE is mainly inspired by phenomena occurring during cartilage development and homeostasis. Most recent advances in cartilage biofunctionality arose from our understanding of these aspects. Still, engineering articular cartilage with native biochemical and biomechanical properties is challenging. Existing bioreactors designed for mechanical stimulation mostly do not enable non-invasive analysis of the engineered construct. Integration of imaging modalities allowing non-invasive assessment of load-induced tissue changes is currently being developed and may apply for cartilage TE in the future (Nebelung et al. 2016). The design of multimodal high-throughput stimulation platforms might further help speeding up this research (Moraes et al. 2010). Applying sequential stimuli of different natures synergistically, rather than trying to combine them all in one bioreactor design (Brady et al. 2015a), should also be considered. However, since the physiological conditions mimicked in bioreactors have not yet been shown to result in cartilage repair in vivo, even the act of mimicking these conditions in vitro may be questioned.

Normally, moderate joint loading serves to maintain cartilage homeostasis, but tissue composition, histomorphology, and structural properties continually change during lifetime. To better understand which load-induced biophysical changes are the most important in cartilage, research should attempt to define the cellular pathways that are responsible for generating an appropriate biosynthetic response.

Adequate biophysical stimulation seems to enhance the synthesis of GFs through natural pathways and could offer an alternative cheap, simple, and flexible way to deliver cytokines (Balint et al. 2013). Interestingly, this seems to hold for a broad spectrum of stimuli like mechanical, osmotic, and electrical cues.

The responsiveness of chondrocytes to a vast number of experimental perturbations was rewarding for experimentalists, but a bane to achieving consensus in the field (Wong and Carter 2003). Standardization is urgently needed. Currently, material properties of the engineered cartilage replacement constructs are reported in less than 30% of cases (Lujan et al. 2011). The cartilage tissue engineer should further be aware of site-specific mechanical properties and should rigorously confirm and report relevant mechanical values. Altogether, the achievement of cartilage TE strategies that fulfil all required conditions will still need significant work to realize suitable next-generation articular cartilage repair constructs.

Acknowledgements Funding from the START-Program (Grant No. 691513) of the Faculty of Medicine, RWTH Aachen, the Umbrella Research Cooperation (Grant No. 700116), and the D-Board Consortium, which has received funding from the European Union's Seventh Framework Programme for Research, Technological Development, and Demonstration under Grant Agreement No. 305815, is acknowledged.

References

- Amin AK, Huntley JS, Bush PG, Simpson AH, Hall AC (2008) Osmolarity influences chondrocyte death in wounded articular cartilage. J Bone Joint Surg Am 90(7):1531–1542. doi:10.2106/ JBJS.G.00857
- Asanbaeva A, Masuda K, Thonar EJ, Klisch SM, Sah RL (2007) Mechanisms of cartilage growth: modulation of balance between proteoglycan and collagen in vitro using chondroitinase ABC. Arthritis Rheum 56(1):188–198. doi:10.1002/art.22298
- Balint R, Cassidy NJ, Cartmell SH (2013) Electrical stimulation: a novel tool for tissue engineering. Tissue Eng Part B Rev 19(1):48–57. doi:10.1089/ten.TEB.2012.0183
- Barrett-Jolley R, Lewis R, Fallman R, Mobasheri A (2010) The emerging chondrocyte channelome. Front Physiol 1:135. doi:10.3389/fphys.2010.00135
- Bastiaansen-Jenniskens YM, Koevoet W, Feijt C, Bos PK, Verhaar JA, Van Osch GJ, DeGroot J (2009) Proteoglycan production is required in initial stages of new cartilage matrix formation but inhibits integrative cartilage repair. J Tissue Eng Regen Med 3(2):117–123. doi:10.1002/term.147
- Bhumiratana S, Eton RE, Oungoulian SR, Wan LQ, Ateshian GA, Vunjak-Novakovic G (2014) Large, stratified, and mechanically functional human cartilage grown in vitro by mesenchymal condensation. Proc Natl Acad Sci U S A 111(19):6940–6945. doi:10.1073/pnas.1324050111
- Brady MA, Waldman SD, Ethier CR (2015a) The application of multiple biophysical cues to engineer functional neocartilage for treatment of osteoarthritis. Part I: cellular response. Tissue Eng Part B Rev 21(1):1–19. doi:10.1089/ten.TEB.2013.0757
- Brady MA, Waldman SD, Ethier CR (2015b) The application of multiple biophysical cues to engineer functional neocartilage for treatment of osteoarthritis. Part II: signal transduction. Tissue Eng Part B Rev 21(1):20–33. doi:10.1089/ten.TEB.2013.0760
- van de Breevaart Bravenboer J, In der Maur CD, Bos PK, Feenstra L, Verhaar JA, Weinans H, van Osch GJ (2004) Improved cartilage integration and interfacial strength after enzymatic treatment in a cartilage transplantation model. Arthritis Res Ther 6(5):R469–R476. doi:10.1186/ ar1216
- Browning JA, Wilkins RJ (2004) Mechanisms contributing to intracellular pH homeostasis in an immortalised human chondrocyte cell line. Comp Biochem Physiol A Mol Integr Physiol 137(2):409–418. doi:10.1016/j.cbpb.2003.10.020
- Camarero-Espinosa S, Rothen-Rutishauser B, Foster EJ, Weder C (2016) Articular cartilage: from formation to tissue engineering. Biomater Sci. doi:10.1039/c6bm00068a
- Caron MM, Emans PJ, Cremers A, Surtel DA, Coolsen MM, van Rhijn LW, Welting TJ (2013a) Hypertrophic differentiation during chondrogenic differentiation of progenitor cells is stimulated by BMP-2 but suppressed by BMP-7. Osteoarthritis Cartilage 21(4):604–613. doi:10.1016/j.joca.2013.01.009
- Caron MM, van der Windt AE, Emans PJ, van Rhijn LW, Jahr H, Welting TJ (2013b) Osmolarity determines the in vitro chondrogenic differentiation capacity of progenitor cells via nuclear factor of activated T-cells 5. Bone 53(1):94–102. doi:10.1016/j.bone.2012.11.032
- Chen AK, Reuveny S, Oh SK (2013) Application of human mesenchymal and pluripotent stem cell microcarrier cultures in cellular therapy: achievements and future direction. Biotechnol Adv 31(7):1032–1046. doi:10.1016/j.biotechadv.2013.03.006

Darling EM (2013) Articular cartilage, 1st edn. CRC Press, Taylor & Francis Group, Boca Raton

- Darling EM, Pritchett PE, Evans BA, Superfine R, Zauscher S, Guilak F (2009) Mechanical properties and gene expression of chondrocytes on micropatterned substrates following dedifferentiation in monolayer. Cell Mol Bioeng 2(3):395–404
- Das R, Kreukniet M, Oostra J, van Osch G, Weinans H, Jahr H (2008a) Control of oxygen tension and pH in a bioreactor for cartilage tissue engineering. Biomed Mater Eng 18(4-5):279–282
- Das RH, Jahr H, Verhaar JA, van der Linden JC, van Osch GJ, Weinans H (2008b) In vitro expansion affects the response of chondrocytes to mechanical stimulation. Osteoarthritis Cartilage 16(3):385–391. doi:10.1016/j.joca.2007.07.014
- Das R, Jahr H, van Osch GJ, Farrell E (2010a) The role of hypoxia in bone marrow-derived mesenchymal stem cells: considerations for regenerative medicine approaches. Tissue Eng Part B Rev 16(2):159–168. doi:10.1089/ten.TEB.2009.0296
- Das RH, van Osch GJ, Kreukniet M, Oostra J, Weinans H, Jahr H (2010b) Effects of individual control of pH and hypoxia in chondrocyte culture. J Orthop Res 28(4):537–545. doi:10.1002/ jor.20994
- Das R, Timur UT, Edip S, Haak E, Wruck C, Weinans H, Jahr H (2015) TGF-beta2 is involved in the preservation of the chondrocyte phenotype under hypoxic conditions. Ann Anat 198:1–10. doi:10.1016/j.aanat.2014.11.003
- DuRaine G, Neu CP, Chan SM, Komvopoulos K, June RK, Reddi AH (2009) Regulation of the friction coefficient of articular cartilage by TGF-beta1 and IL-1beta. J Orthop Res 27(2):249– 256. doi:10.1002/jor.20713
- DuRaine GD, Brown WE, Hu JC, Athanasiou KA (2015) Emergence of scaffold-free approaches for tissue engineering musculoskeletal cartilages. Ann Biomed Eng 43(3):543–554. doi: 10.1007/s10439-014-1161-y
- Elder BD, Athanasiou KA (2008) Synergistic and additive effects of hydrostatic pressure and growth factors on tissue formation. PLoS One 3(6):e2341. doi:10.1371/journal.pone.0002341
- Elder BD, Athanasiou KA (2009a) Effects of temporal hydrostatic pressure on tissue-engineered bovine articular cartilage constructs. Tissue Eng Part A 15(5):1151–1158. doi:10.1089/ten. tea.2008.0200
- Elder BD, Athanasiou KA (2009b) Systematic assessment of growth factor treatment on biochemical and biomechanical properties of engineered articular cartilage constructs. Osteoarthritis Cartilage 17(1):114–123. doi:10.1016/j.joca.2008.05.006
- Eltawil NM, Howie SE, Simpson AH, Amin AK, Hall AC (2015) The use of hyperosmotic saline for chondroprotection: implications for orthopaedic surgery and cartilage repair. Osteoarthritis Cartilage 23 (3):469–477. doi:10.1016/j.joca.2014.12.004. pii: S1063-4584(14)01374-0
- Fan JC, Waldman SD (2010) The effect of intermittent static biaxial tensile strains on tissue engineered cartilage. Ann Biomed Eng 38(4):1672–1682. doi:10.1007/s10439-010-9917-5
- Fickert S, Gerwien P, Helmert B, Schattenberg T, Weckbach S, Kaszkin-Bettag M, Lehmann L (2012) One-year clinical and radiological results of a prospective, investigator-initiated trial examining a novel, purely autologous 3-dimensional autologous chondrocyte transplantation product in the knee. Cartilage 3(1):27–42. doi:10.1177/1947603511417616
- Freed LE, Vunjak-Novakovic G (1995) Cultivation of cell-polymer tissue constructs in simulated microgravity. Biotechnol Bioeng 46(4):306–313. doi:10.1002/bit.260460403
- Freyria AM, Yang Y, Chajra H, Rousseau CF, Ronziere MC, Herbage D, El Haj AJ (2005) Optimization of dynamic culture conditions: effects on biosynthetic activities of chondrocytes grown in collagen sponges. Tissue Eng 11(5-6):674–684. doi:10.1089/ten.2005.11.674
- Furukawa KS, Suenaga H, Toita K, Numata A, Tanaka J, Ushida T, Sakai Y, Tateishi T (2003) Rapid and large-scale formation of chondrocyte aggregates by rotational culture. Cell Transplant 12(5):475–479
- Garvin J, Qi J, Maloney M, Banes AJ (2003) Novel system for engineering bioartificial tendons andapplicationofmechanicalload.TissueEng9(5):967–979.doi:10.1089/107632703322495619
- Gelinsky M, Bernhardt A, Milan F (2015) Bioreactors in tissue engineering: advances in stem cell culture and three-dimensional tissue constructs. Eng Life Sci 15:6670–6677. doi:10.1002/elsc.201400216

- Gigout A, Buschmann MD, Jolicoeur M (2009) Chondrocytes cultured in stirred suspension with serum-free medium containing pluronic-68 aggregate and proliferate while maintaining their differentiated phenotype. Tissue Eng Part A 15(8):2237–2248. doi:10.1089/ten. tea.2008.0256
- Goodwin TJ, Prewett TL, Wolf DA, Spaulding GF (1993) Reduced shear stress: a major component in the ability of mammalian tissues to form three-dimensional assemblies in simulated microgravity. J Cell Biochem 51(3):301–311. doi:10.1002/jcb.240510309
- Hall AC, Horwitz ER, Wilkins RJ (1996) The cellular physiology of articular cartilage. Exp Physiol 81(3):535–545
- Hansmann J, Groeber F, Kahlig A, Kleinhans C, Walles H (2013) Bioreactors in tissue engineering – principles, applications and commercial constraints. Biotechnol J 8(3):298–307. doi: 10.1002/biot.201200162
- Heywood HK, Knight MM, Lee DA (2010) Both superficial and deep zone articular chondrocyte subpopulations exhibit the Crabtree effect but have different basal oxygen consumption rates. J Cell Physiol 223(3):630–639. doi:10.1002/jcp.22061
- Hu YC (2014) Gene therapy for cartilage tissue engineering. In: Hu YC (ed) Gene therapy for cartilage and bone tissue engineering. Springer, Berlin, pp 55–89. doi:10.1007/978-3-642-53923-7
- Huang Y, Zhang Y, Ding X, Liu S, Sun T (2015) Osmolarity influences chondrocyte repair after injury in human articular cartilage. J Orthop Surg Res 10:19. doi:10.1186/ s13018-015-0158-z
- Huey DJ, Hu JC, Athanasiou KA (2012) Unlike bone, cartilage regeneration remains elusive. Science 338(6109):917–921. doi:10.1126/science.1222454
- Hunziker EB, Kapfinger E (1998) Removal of proteoglycans from the surface of defects in articular cartilage transiently enhances coverage by repair cells. J Bone Joint Surg Br 80(1):144–150
- Ingber DE (2006) Cellular mechanotransduction: putting all the pieces together again. FASEB J 20(7):811–827. doi:10.1096/fj.05-5424rev
- Jahr H (2015) HDACi and Nrf2: not from alpha to omega but from acetylation to OA. Arthritis Res Ther 17(1):381. doi:10.1186/s13075-015-0885-x
- Jansen JH, Weyts FA, Westbroek I, Jahr H, Chiba H, Pols HA, Verhaar JA, van Leeuwen JP, Weinans H (2004) Stretch-induced phosphorylation of ERK1/2 depends on differentiation stage of osteoblasts. J Cell Biochem 93(3):542–551. doi:10.1002/jcb.20162
- Jansen JH, Jahr H, Verhaar JA, Pols HA, Chiba H, Weinans H, van Leeuwen JP (2006) Stretch-induced modulation of matrix metalloproteinases in mineralizing osteoblasts via extracellular signal-regulated kinase-1/2. J Orthop Res 24(7):1480–1488. doi:10.1002/jor.20186
- Jeon JE, Schrobback K, Hutmacher DW, Klein TJ (2012) Dynamic compression improves biosynthesis of human zonal chondrocytes from osteoarthritis patients. Osteoarthritis Cartilage 20(8):906–915. doi:10.1016/j.joca.2012.04.019
- Jin G, Yang GH, Kim G (2015) Tissue engineering bioreactor systems for applying physical and electrical stimulations to cells. J Biomed Mater Res B Appl Biomater 103(4):935–948. doi:10.1002/jbm.b.33268
- Jortikka MO, Parkkinen JJ, Inkinen RI, Karner J, Jarvelainen HT, Nelimarkka LO, Tammi MI, Lammi MJ (2000) The role of microtubules in the regulation of proteoglycan synthesis in chondrocytes under hydrostatic pressure. Arch Biochem Biophys 374(2):172–180. doi:10.1006/ abbi.1999.1543
- Jung S, Panchalingam KM, Wuerth RD, Rosenberg L, Behie LA (2012) Large-scale production of human mesenchymal stem cells for clinical applications. Biotechnol Appl Biochem 59(2):106– 120. doi:10.1002/bab.1006
- Khan IM, Gilbert SJ, Singhrao SK, Duance VC, Archer CW (2008) Cartilage integration: evaluation of the reasons for failure of integration during cartilage repair. A review. Eur Cell Mater 16:26–39
- Kim SH, Kim SH, Jung Y (2015) TGF-beta3 encapsulated PLCL scaffold by a supercritical CO2-HFIP co-solvent system for cartilage tissue engineering. J Control Release 206:101–107. doi:10.1016/j.jconrel.2015.03.026
- King JA, Miller WM (2007) Bioreactor development for stem cell expansion and controlled differentiation. Curr Opin Chem Biol 11(4):394–398. doi:10.1016/j.cbpa.2007.05.034

- Leddy HA, Awad HA, Guilak F (2004) Molecular diffusion in tissue-engineered cartilage constructs: effects of scaffold material, time, and culture conditions. J Biomed Mater Res B Appl Biomater 70(2):397–406. doi:10.1002/jbm.b.30053
- Lee JK, Gegg CA, Hu JC, Kass PH, Athanasiou KA (2014) Promoting increased mechanical properties of tissue engineered neocartilage via the application of hyperosmolarity and 4alpha-phorbol 12,13-didecanoate (4alphaPDD). J Biomech 47(15):3712–3718. doi:10.1016/j. jbiomech.2014.09.018
- Leijten JC, Bos SD, Landman EB, Georgi N, Jahr H, Meulenbelt I, Post JN, van Blitterswijk CA, Karperien M (2013) GREM1, FRZB and DKK1 mRNA levels correlate with osteoarthritis and are regulated by osteoarthritis-associated factors. Arthritis Res Ther 15(5):R126. doi:10.1186/ar4306
- Little CJ, Bawolin NK, Chen X (2011) Mechanical properties of natural cartilage and tissueengineered constructs. Tissue Eng Part B Rev 17(4):213–227
- Lujan TJ, Wirtz KM, Bahney CS, Madey SM, Johnstone B, Bottlang M (2011) A novel bioreactor for the dynamic stimulation and mechanical evaluation of multiple tissue-engineered constructs. Tissue Eng Part C Methods 17(3):367–374. doi:10.1089/ten.TEC.2010.0381
- MacBarb RF, Chen AL, Hu JC, Athanasiou KA (2013) Engineering functional anisotropy in fibrocartilage neotissues. Biomaterials 34(38):9980–9989. doi:10.1016/j.biomaterials.2013.09.026
- Makris EA, MacBarb RF, Paschos NK, Hu JC, Athanasiou KA (2014a) Combined use of chondroitinase-ABC, TGF-beta1, and collagen crosslinking agent lysyl oxidase to engineer functional neotissues for fibrocartilage repair. Biomaterials 35(25):6787–6796. doi:10.1016/j. biomaterials.2014.04.083
- Makris EA, Responte DJ, Paschos NK, Hu JC, Athanasiou KA (2014b) Developing functional musculoskeletal tissues through hypoxia and lysyl oxidase-induced collagen cross-linking. Proc Natl Acad Sci U S A 111(45):E4832–E4841. doi:10.1073/pnas.1414271111
- Makris EA, Gomoll AH, Malizos KN, Hu JC, Athanasiou KA (2015) Repair and tissue engineering techniques for articular cartilage. Nat Rev Rheumatol 11(1):21–34. doi:10.1038/ nrrheum.2014.157
- Malda J, van Blitterswijk CA, van Geffen M, Martens DE, Tramper J, Riesle J (2004) Low oxygen tension stimulates the redifferentiation of dedifferentiated adult human nasal chondrocytes. Osteoarthritis Cartilage 12(4):306–313. doi:10.1016/j.joca.2003.12.001
- Marlovits S, Tichy B, Truppe M, Gruber D, Schlegel W (2003a) Collagen expression in tissue engineered cartilage of aged human articular chondrocytes in a rotating bioreactor. Int J Artif Organs 26(4):319–330
- Marlovits S, Tichy B, Truppe M, Gruber D, Vecsei V (2003b) Chondrogenesis of aged human articular cartilage in a scaffold-free bioreactor. Tissue Eng 9(6):1215–1226. doi:10.1089/ 10763270360728125
- Martin I, Wendt D, Heberer M (2004) The role of bioreactors in tissue engineering. Trends Biotechnol 22(2):80–86. doi:10.1016/j.tibtech.2003.12.001
- Mauck RL, Nicoll SB, Seyhan SL, Ateshian GA, Hung CT (2003) Synergistic action of growth factors and dynamic loading for articular cartilage tissue engineering. Tissue Eng 9(4):597– 611. doi:10.1089/107632703768247304
- McGowan KB, Sah RL (2005) Treatment of cartilage with beta-aminopropionitrile accelerates subsequent collagen maturation and modulates integrative repair. J Orthop Res 23(3):594–601. doi:10.1016/j.orthres.2004.02.015
- Miyanishi K, Trindade MC, Lindsey DP, Beaupre GS, Carter DR, Goodman SB, Schurman DJ, Smith RL (2006) Effects of hydrostatic pressure and transforming growth factor-beta 3 on adult human mesenchymal stem cell chondrogenesis in vitro. Tissue Eng 12(6):1419–1428. doi:10.1089/ten.2006.12.1419
- Moraes C, Chen JH, Sun Y, Simmons CA (2010) Microfabricated arrays for high-throughput screening of cellular response to cyclic substrate deformation. Lab Chip 10(2):227–234. doi:10.1039/b914460a
- Mow VC, Wang CC, Hung CT (1999) The extracellular matrix, interstitial fluid and ions as a mechanical signal transducer in articular cartilage. Osteoarthritis Cartilage 7(1):41–58
- Murphy CL, Sambanis A (2001) Effect of oxygen tension and alginate encapsulation on restoration of the differentiated phenotype of passaged chondrocytes. Tissue Eng 7(6):791–803. doi:10.1089/107632701753337735

- Natoli RM, Responte DJ, Lu BY, Athanasiou KA (2009a) Effects of multiple chondroitinase ABC applications on tissue engineered articular cartilage. J Orthop Res 27(7):949–956. doi:10.1002/jor.20821
- Natoli RM, Revell CM, Athanasiou KA (2009b) Chondroitinase ABC treatment results in greater tensile properties of self-assembled tissue-engineered articular cartilage. Tissue Eng Part A 15(10):3119–3128. doi:10.1089/ten.TEA.2008.0478
- Nebelung S, Brill N, Muller F, Tingart M, Pufe T, Merhof D, Schmitt R, Jahr H, Truhn D (2016) Towards optical coherence tomography-based elastographic evaluation of human cartilage. J Mech Behav Biomed Mater 56:106–119. doi:10.1016/j.jmbbm.2015.11.025
- Obradovic B, Martin I, Padera RF, Treppo S, Freed LE, Vunjak-Novakovic G (2001) Integration of engineered cartilage. J Orthop Res 19(6):1089–1097. doi:10.1016/S0736-0266(01)00030-4
- O'Connell GD, Nims RJ, Green J, Cigan AD, Ateshian GA, Hung CT (2014) Time and dosedependent effects of chondroitinase ABC on growth of engineered cartilage. Eur Cell Mater 27:312–320
- Ofek G, Revell CM, Hu JC, Allison DD, Grande-Allen KJ, Athanasiou KA (2008) Matrix development in self-assembly of articular cartilage. PLoS One 3(7):e2795. doi:10.1371/journal. pone.0002795
- Pazzano D, Mercier KA, Moran JM, Fong SS, DiBiasio DD, Rulfs JX, Kohles SS, Bonassar LJ (2000) Comparison of chondrogensis in static and perfused bioreactor culture. Biotechnol Prog 16(5):893–896. doi:10.1021/bp000082v
- Pitsillides AA, Beier F (2011) Cartilage biology in osteoarthritis--lessons from developmental biology. Nat Rev Rheumatol 7(11):654–663
- Portner R, Goepfert C, Wiegandt K, Janssen R, Ilinich E, Paetzold H, Eisenbarth E, Morlock M (2009) Technical strategies to improve tissue engineering of cartilage-carrier-constructs. Adv Biochem Eng Biotechnol 112:145–181. doi:10.1007/978-3-540-69357-4_7
- Raghunath J, Rollo J, Sales KM, Butler PE, Seifalian AM (2007) Biomaterials and scaffold design: key to tissue-engineering cartilage. Biotechnol Appl Biochem 46(Pt 2):73–84. doi:10.1042/ BA20060134
- Rehfeldt F, Engler AJ, Eckhardt A, Ahmed F, Discher DE (2007) Cell responses to the mechanochemical microenvironment--implications for regenerative medicine and drug delivery. Adv Drug Deliv Rev 59(13):1329–1339. doi:10.1016/j.addr.2007.08.007
- Responte DJ, Arzi B, Natoli RM, Hu JC, Athanasiou KA (2012) Mechanisms underlying the synergistic enhancement of self-assembled neocartilage treated with chondroitinase-ABC and TGF-beta1. Biomaterials 33(11):3187–3194. doi:10.1016/j.biomaterials.2012.01.028
- Rice MA, Homier PM, Waters KR, Anseth KS (2008) Effects of directed gel degradation and collagenase digestion on the integration of neocartilage produced by chondrocytes encapsulated in hydrogel carriers. J Tissue Eng Regen Med 2(7):418–429. doi:10.1002/term.113
- Saini S, Wick TM (2003) Concentric cylinder bioreactor for production of tissue engineered cartilage: effect of seeding density and hydrodynamic loading on construct development. Biotechnol Prog 19(2):510–521. doi:10.1021/bp0256519
- Shahin K, Doran PM (2015) Shear and compression bioreactor for cartilage synthesis. Methods Mol Biol 1340:221–233. doi:10.1007/978-1-4939-2938-2_16
- Simpkin VL, Murray DH, Hall AP, Hall AC (2007) Bicarbonate-dependent pH(i) regulation by chondrocytes within the superficial zone of bovine articular cartilage. J Cell Physiol 212(3):600–609. doi:10.1002/jcp.21054
- Stevens MM, Marini RP, Schaefer D, Aronson J, Langer R, Shastri VP (2005) In vivo engineering of organs: the bone bioreactor. Proc Natl Acad Sci U S A 102(32):11450–11455. doi:10.1073/ pnas.0504705102
- Stoddart MJ, Ettinger L, Hauselmann HJ (2006) Enhanced matrix synthesis in de novo, scaffold free cartilage-like tissue subjected to compression and shear. Biotechnol Bioeng 95(6):1043– 1051. doi:10.1002/bit.21052
- Tew SR, Peffers MJ, McKay TR, Lowe ET, Khan WS, Hardingham TE, Clegg PD (2009) Hyperosmolarity regulates SOX9 mRNA posttranscriptionally in human articular chondrocytes. Am J Physiol Cell Physiol 297(4):C898–C906
- Urban JP (1994) The chondrocyte: a cell under pressure. Br J Rheumatol 33(10):901-908

- Vinardell T, Sheehy EJ, Buckley CT, Kelly DJ (2012) A comparison of the functionality and in vivo phenotypic stability of cartilaginous tissues engineered from different stem cell sources. Tissue Eng Part A 18(11-12):1161–1170. doi:10.1089/ten.TEA.2011.0544
- Wang YK, Chen CS (2013) Cell adhesion and mechanical stimulation in the regulation of mesenchymal stem cell differentiation. J Cell Mol Med 17(7):823–832. doi:10.1111/jcmm.12061
- Wang N, Tytell JD, Ingber DE (2009) Mechanotransduction at a distance: mechanically coupling the extracellular matrix with the nucleus. Nat Rev Mol Cell Biol 10(1):75–82. doi:10.1038/ nrm2594
- Wang N, Grad S, Stoddart MJ, Niemeyer P, Sudkamp NP, Pestka J, Alini M, Chen J, Salzmann GM (2013) Bioreactor-induced chondrocyte maturation is dependent on cell passage and onset of loading. Cartilage 4(2):165–176. doi:10.1177/1947603512471345
- Wang N, Grad S, Stoddart MJ, Niemeyer P, Reising K, Schmal H, Sudkamp NP, Alini M, Salzmann GM (2014) Particulate cartilage under bioreactor-induced compression and shear. Int Orthop 38(5):1105–1111. doi:10.1007/s00264-013-2194-9
- Wilkins RJ, Browning JA, Urban JP (2000) Chondrocyte regulation by mechanical load. Biorheology 37(1-2):67–74
- van der Windt AE, Haak E, Das RH, Kops N, Welting TJ, Caron MM, van Til NP, Verhaar JA, Weinans H, Jahr H (2010a) Physiological tonicity improves human chondrogenic marker expression through nuclear factor of activated T-cells 5 in vitro. Arthritis Res Ther 12(3):R100. doi:10.1186/ar3031
- van der Windt AE, Jahr H, Farrell E, Verhaar JA, Weinans H, van Osch GJ (2010b) Calcineurin inhibitors promote chondrogenic marker expression of dedifferentiated human adult chondrocytes via stimulation of endogenous TGFbeta1 production. Tissue Eng Part A 16(1):1–10
- van der Windt AE, Haak E, Kops N, Verhaar JA, Weinans H, Jahr H (2012) Inhibiting calcineurin activity under physiologic tonicity elevates anabolic but suppresses catabolic chondrocyte markers. Arthritis Rheum 64(6):1929–1939. doi:10.1002/art.34369
- von der Mark K, Kirsch T, Nerlich A, Kuss A, Weseloh G, Gluckert K, Stoss H (1992) Type X collagen synthesis in human osteoarthritic cartilage. Indication of chondrocyte hypertrophy. Arthritis Rheum 35(7):806–811
- Wong M, Carter DR (2003) Articular cartilage functional histomorphology and mechanobiology: a research perspective. Bone 33(1):1–13
- Xu X, Urban JP, Tirlapur UK, Cui Z (2010) Osmolarity effects on bovine articular chondrocytes during three-dimensional culture in alginate beads. Osteoarthritis Cartilage 18(3):433–439. doi:10.1016/j.joca.2009.10.003
- Zeng C, Li H, Yang T, Deng ZH, Yang Y, Zhang Y, Lei GH (2015) Electrical stimulation for pain relief in knee osteoarthritis: systematic review and network meta-analysis. Osteoarthritis Cartilage 23(2):189–202. doi:10.1016/j.joca.2014.11.014

Animal Models in Cartilage Repair

11

Lars Goebel, Liang Gao, and Henning Madry

Abstract

Animal models play an important role to test novel experimental strategies and reconstructive surgical treatments of focal articular cartilage defects. Such animal models need to reflect the different appearances and aetiologies of cartilage defects, e.g. caused by trauma or osteoarthritis. Depth of articular cartilage defects plays an important role. Full-thickness chondral defects do not extend into the subchondral bone, while osteochondral defects penetrate the cement line and extend to the subchondral bone, thereby changing its structural integrity. Mice, rats, rabbits, goat, sheep, minipigs and horses are representing good models, bridging the gap between in vitro studies and clinical experiments in human. Each of them has benefits and limitations. Evaluation of cartilage repair may be performed using a large variety of methods, among which non-destructive evaluations and histological scoring, the latter being considered as the gold standard. As the available reconstructive surgical approaches for articular cartilage repair become increasingly complex, precise animal models to test and to translate new surgical techniques into appropriate clinical treatments are required.

e-mail: Lars.Goebel@uniklinikum-saarland.de; henning.madry@uniklinikum-saarland.de

L. Gao

L. Goebel • H. Madry (🖂)

Center of Experimental Orthopaedics, Saarland University Medical Center and Saarland University, 66421 Homburg, Saar, Germany

Cartilage Net of the Greater Region, Homburg, Saar, Germany

Department of Orthopaedic Surgery, Saarland University Medical Center and Saarland University, 66421 Homburg, Saar, Germany

Center of Experimental Orthopaedics, Saarland University Medical Center and Saarland University, 66421 Homburg, Saar, Germany

Cartilage Net of the Greater Region, Homburg, Saar, Germany e-mail: lianggao22@gmail.com

[©] Springer International Publishing AG 2017

S. Grässel, A. Aszodi (eds.), Cartilage, DOI 10.1007/978-3-319-53316-2_11

11.1 Introduction

Animal models play an important role to test novel experimental strategies and reconstructive surgical treatments of focal articular cartilage defects (Hunziker 2009; Blaney Davidson et al. 2014). Such experimental strategies include, but are not limited to, cell-, scaffold- and gene-based approaches (Cucchiarini et al. 2014). Reconstructive surgical treatments include marrow stimulation techniques, autologous or allogeneic chondrocyte implantation (ACI), osteochondral transplants and the refixation of chondral or osteochondral fragments, among others (Hunziker et al. 2015). When such novel treatments are introduced into the clinical practice or when established techniques are refined, studies in animal models are indispensable to closing the gap between in vitro experiments and the clinical reality in a human joint. Moreover, clinical data on the macroscopic and histological aspects and the biochemical composition of the cartilaginous repair tissue are nearly never available, mainly due to ethical concerns of performing a second-look operation with the sole purpose of evaluating articular cartilage repair.

This chapter will focus on the different animal models reflecting focal cartilage defects. It will first give an overview of the different aspects that need to be addressed when choosing the right animal model. Next, the benefits and limitations of each major animal model currently used in articular cartilage research are given. Finally, a discussion of the evaluation of articular cartilage repair is included.

11.2 Classification of Articular Cartilage Defects

Articular cartilage defects can be clinically classified based on their nature and their depth.

11.2.1 Nature of Articular Cartilage Defects

Animal models for cartilage repair need first to reflect the different appearances and aetiologies of cartilage defects, which may be caused by trauma, osteochondritis dissecans but also osteoarthritis (OA) and osteonecrosis (Madry et al. 2010). It is of key importance to understand the difference between focal, non-OA cartilage defects and the often ill-defined OA lesions. Focal defects are usually surrounded by a normal adjacent cartilage. In contrast, OA lesions are often larger in size, may affect the entire joint surface and are of different depths (Madry et al. 2011; Pritzker et al. 2006). OA models differ from focal articular cartilage defect models (Little and Zaki 2012), as the method for OA induction needs to reflect the clinical entity which caused the disease and therefore has to be carefully chosen (Cook et al. 2010; Gerwin et al. 2010; Glasson et al. 2010; Kraus et al. 2010; Laverty et al. 2010; Little et al. 2010; McIlwraith et al. 2010). Moreover, the time of therapeutic intervention has to be selected following or during OA induction. As the Osteoarthritis Research Society International (OARSI) devoted an entire issue of their journal *Osteoarthritis and Cartilage* for highly detailed recommendations for animal models to study osteoarthritis, such models are not discussed within the context of this chapter.

11.2.2 Articular Cartilage Repair and Regeneration

Articular cartilage repair is different from articular cartilage regeneration, the latter referring to a complete restoration to its original architecture, indistinguishable from normal hyaline articular cartilage. Damaged articular cartilage, for example, resulting from a traumatic event, does not regenerate (Hunziker 2002). Often, lesions are termed "critical size", referring to the fact that they encounter a limited repair process if they are larger than a (species-specific) certain defect size, although in most adult animals even smaller lesions do not regenerate (Hunziker 2009).

11.2.3 Depth of Articular Cartilage Defects

When performing studies on articular cartilage repair, a focus should be laid on the entire osteochondral unit which consists of the articular cartilage and the subchondral bone. The articular cartilage may be separated in the superficial, intermediate and deep layer of the hyaline cartilage, which is separated by the tidemark from the calcified cartilage. The cement line is considered as border to the subchondral bone, which itself is discriminated into the subchondral bone plate and the subarticular spongiosa (Fig. 11.1) (Madry et al. 2010). Accordingly, defects may be graded as partial- or full-thickness chondral or as osteochondral defects (Fig. 11.2).

A full-thickness chondral defect does, by definition, not extend into the subchondral bone. It ends at the intersection of the calcified cartilage with the subchondral bone plate without penetrating the cement line (Frisbie et al. 2006b; Hunziker 1999b; Madry et al. 2010). Coherently, a partial-thickness articular cartilage defect does not affect all cartilage layers. The spontaneous repair of these chondral defects is very limited, as it depends on cells which migrate from the synovial membrane. However, here the integrity of the subchondral bone remains untouched.

In contrast, osteochondral defects penetrate the cement line and extend to the subchondral bone alternating the structural integrity of the subchondral bone plate and spongiosa. Nevertheless, the spontaneous repair of these defects is usually much more effective, as they access bone marrow with cells from a pluripotent progenitor pool migrating into the defect (Hunziker and Rosenberg 1996).

Fig. 11.1 The osteochondral unit is composed of the articular cartilage and the subchondral bone. The hyaline articular cartilage can be separated in a superficial, intermediate and deep layer, separated from the calcified cartilage by the tidemark. The cement line is the border to the subchondral bone, which itself is categorized into the subchondral bone plate and the subarticular spongiosa



Articular cartilage

Calcified cartilage

Subchondral bone plate

Subchondral spongiosa

Fig. 11.2 Classification of articular cartilage defects. Partial- and full-thickness chondral defects involve only the articular cartilage layers. In contrast to chondral defects, osteochondral defects penetrate the subchondral bone plate and often reach into the subarticular spongiosa. Modified from (Madry et al. 2015), with permission Chondral defect, partial thickness



Chondral defect, full thickness



Osteochondral defect



11.3 Common Requirements for Animal Models of Focal Articular Cartilage Repair

Common requirements for animal models of focal articular cartilage repair are diverse:

- 1. The model needs to reflect the precise clinical nature of the defect (e.g. a traumatic lesion).
- 2. The defect size and depth should also be comparable to the human situation.
- 3. The articular cartilage defect should not regenerate, but merely repair at different levels. This begs for a good knowledge of the age of the animals, including knowledge of the time of skeletal maturity and of growth place fusion (when applicable).
- 4. The results must be applicable to the specific clinical situation.
- 5. The biomechanical properties of the joint cartilage should be close to the human knee cartilage, as these are species specific (Simon 1971) and show different mechanical properties (Athanasiou et al. 1991).

- 6. Not only the cartilage thickness is important but also its microstructure such as cartilage cellularity (Aigner et al. 2010; Poole et al. 2010) and the thickness of the subchondral bone plate in relation to the articular cartilage thickness (Chevrier et al. 2015).
- 7. The size of the animal joint needs to be comparable to the human (knee) joint.
- 8. The usual quadruped pattern, joint range of motion and its resting position need to be similar to the bipedal locomotion of humans, as they result in different biomechanics (Rudert et al. 2000).
- 9. The reconstructive surgery for articular cartilage repair must be technically feasible and reflect the clinical situation.
- 10. The rehabilitation of the selected limb may need to be as close as possible to the clinical regimen in patients, including a phase of protected partial weight-bearing.

Taking these considerations into account, all models differ in many key structural and functional aspects compared to the human situation (Poole et al. 2010). Therefore, no animal model exists that perfectly replicates the human knee joint as a gold standard (Aigner et al. 2010).

11.4 Topographic Considerations of Defect Location

Articular cartilage defects may occur in any human joint. However, most clinical studies on cartilage repair have been performed in the knee joint, followed by the ankle and hip joints. This fact is also reflected in animal models, as most studies have been performed in the knee (stifle) joint, comparably less in the ankle joints and only a few in the hip joints.

Within a joint, topographic location of a defect also is of importance. Clinical work indicates that articular cartilage defects are often located in the femoral condyles and the trochlear groove (Hjelle et al. 2002; Curl et al. 1997). Interestingly, chondral lesions in the human femoral condyles do heal better than when located in the trochlea (Kreuz et al. 2006b). In most animal models, the medial and also lateral femoral condyles, the trochlear groove and the patella are the joint regions most often used to introduce articular cartilage defects. However, it is long since known that the histological (Armstrong et al. 1995), biomechanical (Appleyard et al. 2003) and biochemical (Little et al. 1996) characteristics of articular cartilage show a discrepancy between such separate topographical regions of the knee joint. Of note, these divergences do also have the potential to bias comparative analyses of experimental cartilage repair. It has been shown in a large animal model that the topographic location also dictates the structural patterns and biochemical composition of the repair tissue in the sheep model. For instance, cartilage repair in the sheep model is improved when the defect has been created in the trochlea and not in the condyle. Here, the repair pattern of the sheep trochlea may

be more reflective of the human femoral condyle (Orth et al. 2013c). Thus, the different repair characteristics of cartilage defects at different anatomical sites need to be respected, and defects in such regions need to be assessed independently.

11.5 Defect Geometry and Surgical Technique of Defect Creation

Deciding on the geometry of the articular cartilage defect is important when contemplating about its standardized evaluation. Articular cartilage defects in animal models are usually created applying either circular or rectangular patterns. Circular defects can be outlined using commercial or custom-made punches, e.g. dermal biopsy punches. Rectangular defects are outlined using custom-made punches rather than a scalpel blade to allow for a standardized volume. The articular cartilage within these defects is meticulously removed using curettes or spoons.

Particular attention has to be paid to a meticulous surgical technique when creating chondral defect models. Here, the defect must not extend into the subchondral bone (Frisbie et al. 2006a, b; Drobnic et al. 2010), as this would allow for an influx of marrow elements that may interfere with the different spontaneous repair patterns of a purely chondral lesion (Shapiro et al. 1993). Spontaneous repair of chondral defects is always incomplete, as only some cells originating from the synovial membrane do migrate into the defect over time (Hunziker and Rosenberg 1996).

In contrast, an access to the subchondral bone marrow is granted, e.g. by performing a marrow stimulation at the bottom of the defects using subchondral drilling (Smillie 1957; Pridie 1959), microfracture (Steadman et al. 2001) or abrasion arthroplasty (Johnson 2001); pluripotent progenitor cells from the subchondral bone marrow region below the defect migrate into the lesion, differentiate into chondrocytes and form a repair tissue (Frisbie et al. 1999, 2003; Shapiro et al. 1993).

11.6 Animal Models of Focal Articular Cartilage Repair

Mice, rats, rabbits, goat, sheep, minipigs and horses are representing good models to induce articular cartilage defects (Hunziker 1999a, b, 2000, 2009). These animal models bridge the gap between in vitro studies and clinical experiments in human. Depending on the scope of the study, different animal models are more suitable for specific research questions. In general, joint size and cartilage thickness correspond to the animal size, while articular conditions comparable to human joints may only be found in large animal models (Ahern et al. 2009; Osterhoff et al. 2011). The major considerations of focal articular cartilage defect models are summarized in Table 11.1.

| Factor | Comments |
|----------------------------------|--|
| Cartilage thickness | Cartilage thickness usually increases with the size of the animals. Depends on the anatomic location within the joint |
| Subchondral bone plate thickness | Subchondral bone plate thickness is not always reflective of the size of the animals. Minipigs, for example, have a thin subchondral bone plate, while sheep have a thick subchondral bone plate |
| Age of animals | Adult animals are preferred as juvenile animals have a higher degree of spontaneous repair |
| Defect size | Can be given as the area of the defect and placed in relation with the condylar width |
| Defect depth | Needs to be adapted to the osteochondral anatomy to reflect the desired defect type, e.g. chondral or osteochondral |
| Defect anatomy | Circular or rectangular patterns are commonly used |
| Defect location | Topographic differences within a joint exist for cartilage thickness, biochemical composition and repair potential |
| Knee resting position | Differs among animals, often lack of full extension as in humans |
| Gait patterns | Differs among animals, the sheep/goat/horse usually considered to best resemble the situation in humans |

Table 11.1 Considerations for focal articular cartilage defect models

Modified from Madry et al. (2015) with permission

11.6.1 Mice

Mice are suitable to serve as proof-of-concept models. Here, large animal numbers are possible, and usage of animals from identical strains allows for very homogenous study populations. Additionally, for specific research questions, athymic, transgenic or knockout strains as well as mice with a compromised immune system are available. These mice develop, for example, osteoarthritis early and are ideal candidates for allogeneic or xenogenic in vivo studies of novel cartilage repair protocols. For example, Matsuoka et al. described an osteochondral cartilage repair mouse model (Matsuoka et al. 2015). Like other rodents, mice are in general inexpensive to house and breed, while the development and maintenance of specific knockout mice may be more challenging and expensive.

Mice joints are small and the articular cartilage layer is thin, only consisting of a few cell layers. This makes it more or less impossible to study the effects of solid scaffolds for articular cartilage repair in partial- or full-thickness cartilage defects. In contrast to humans, their growth plates remain open, increasing the intrinsic healing potential of cartilage repair. For outcome assessment besides macroscopic, histological and biochemical evaluation, high-field magnetic resonance imaging (μ MRI) or micro-computed tomography (μ CT) can be considered (Mak et al. 2015), but biomechanical testing may also be performed.

11.6.2 Rats

Compared to mice, rat joints are much larger, while the cartilage layer still remains very thin. In the rat knee joint, chondral or even osteochondral defects with a diameter up to 3 mm may be created in the femoral condyle or the trochlea. The critical size of a defect is unknown. Choi and coworkers described in a rat model the intraarticular application of transforming growth factor β 1 in modular chitosan hydrogels for the treatment of articular cartilage defects (Choi et al. 2015). Also, xenogenic cells may be implanted in (osteo-)chondral defects to examine their repair potential in vivo (Pagnotto et al. 2007). Rats are utilized for in vivo testing of new solid and hydrogel scaffolds. Similar to mice, they are economical to house but face analogous problems, such as improved intrinsic repair and persisting open growth plates. The outcome assessment recommendations are the same as for mice, but the translational potential remains very limited.

11.6.3 Rabbits

The lapine model is commonly used in articular cartilage research (Laverty et al. 2010). Rabbits serve to bridge the gap between small and larger animal models and require relatively simple husbandry. The condyles and the trochlea are large enough to create osteo- or chondral defects with a maximal size of 3–4 mm. Distal femoral growth plates close at 6–8 months of age (Kaweblum et al. 1994; Hunziker et al. 2007). In rabbits, the degree of knee flexion is much higher compared with humans, resulting in higher forces in the trochlea than in humans. The relatively good endogenous healing potential has led to the assumption that cartilage defects in the rabbit do regenerate. However, even osteochondral defects of 2 mm in diameter in adult rabbits only repair and not regenerate. As in the rat, there is a cartilaginous patella located proximally from the bony patella.

11.6.4 Dogs

Accumulating ethical concerns for the continuous use of dogs in articular cartilage research have been arisen in the past years. Of note, however, dogs suffer from very similar diseases as humans, e.g. osteochondritis dissecans or osteoarthritis, and have a similarly poor intrinsic healing potential for cartilage damages, mimicking more closely the situation in human than other animal models. While defect sizes up to 12 mm are described in the literature, most defects have a diameter of 4 mm (Ahern et al. 2009). Cartilage thickness is thinner compared to humans but allows creation of partial-thickness cartilage defects as well. Canines reach skeletal maturity between 12 and 24 months.

In spite of the ethical concerns, some specific points may advocate the use of dogs for (osteo-) chondral repair studies: The stifle joint is amendable to arthroscopic examination, if longitudinal observations or in vivo studies are warranted.

Additionally, dogs may be trained to perform specific rehabilitation programmes or use a treadmill and also tolerate braces or slings, for example, if partial weightbearing is necessary. For canine, beside standardized cartilage evaluation protocols, pain level, quality of life or gait analyses may be performed.

11.6.5 Goat

The articular cartilage of goat has a poor intrinsic capability to heal. Jackson et al. reported the critical size of an osteochondral defect in the medial femoral condyle of the stifle joint to be 6 mm in diameter (Jackson et al. 2001). Compared to smaller animal models, the joint size is larger, and the ratio between articular cartilage and subchondral bone structures is comparable to humans, while in total the thickness of the articular cartilage layer remains thinner with respect to humans (Jackson et al. 2001). In caprines, growth plates close after 36–48 months. Studies of partial-and full-thickness chondral, as well as osteochondral, defects are possible. Also, new resurfacing implants or scaffolds have been tested in goats. Custers et al. described the use of a metallic implant (Custers et al. 2009), and Kon et al. recently reported their results of a novel aragonite-hyaluronate biphasic scaffold for critical size osteochondral defects (Kon et al. 2015).

Goats are easy to house and, compared to other large animal models, relatively inexpensive. Here, if longitudinal assessments are warranted, arthroscopic examinations are possible, too, while complex rehabilitation programmes and partial weightbearing are difficult to achieve.

11.6.6 Sheep

The sheep is an excellent large animal model for translational research of articular cartilage repair (Orth et al. 2015a; Eldracher et al. 2014; Goebel et al. 2015). Osterhoff et al. described common anatomical characteristics of human and ovine knee and concluded that the ovine stifle may be considered as scaled down model of a human knee (Osterhoff et al. 2011). Sheep are also simple to house, and the stifle joint can be assessed easily. Schinhan et al. described 7 mm cartilage defects to be critical to develop osteoarthritis in stable joints (Schinhan et al. 2012). Comparable to humans, the intrinsic repair potential is very limited. Chondral and osteochondral defect repair in the ovine trochlea is better compared with the medial condyle (Orth et al. 2013c), which is, however, contrary to the situation in human where defects of the medial femoral condyle perform in general better. Through a medial mini-arthrotomy, good exposure of the ovine femoral condyle and distal trochlea to create (osteo-) chondral defects, as well as assessment of the anterior cruciate ligament or medial meniscus, is possible. Here, outmost surgical caution must be taken to avoid a post-operative patellar luxation (Orth and Madry 2013). To evaluate the effect of pressure on cartilage repair, also performing of opening and closing wedge high tibial osteotomies is possible to mimic the clinical situation of patient with malalignment and cartilage pathology (Pape and Madry 2013).

Among gait analyses, macroscopic, histological, biochemical and biomechanical analyses, as well as evaluations using μ MRI or μ CT (Goebel et al. 2014; Orth et al. 2012a) or radiographs, can be performed. Arthroscopic follow-up surgery in longitudinal experimental settings is possible, too. Also, in this animal model, complex rehabilitation protocols are not practicable.

11.6.7 Minipigs

The stifle joint of the pig is closely similar to human condition in joint size, weightbearing requirements and cartilage thickness (Jiang et al. 2007). However, conventional large size swines are difficult to handle, being sometimes aggressive and housing requirements are challenging. Here, minipigs offer some advantages while they are considerably expensive compared to goats or sheep. They become skeletally mature between 16 and 24 months.

The stifle joint of minipigs, notable smaller than the human knee, allows for (osteo-)chondral defect creation with a diameter of 6–8 mm or even larger either in the femoral condyles or the trochlear groove, while the thickness of articular cartilage and the subchondral bone plate is relatively thinner. Similar to human, the cartilage of adult minipigs also has limited capability for self-repair. The bone apposition rate, trabecular thickness, and collagen fibre arrangement in articular cartilage of the Göttingen minipigs are similar to those in humans (Kaab et al. 1998). Studies for both chondral and osteochondral defects can be performed with minipigs. Gotterbarm et al. proved the limited capability for endogenous repair of chondral and osteochondral defects and validated the utility of the Göttingen minipigs for articular cartilage repair research (Gotterbarm et al. 2008). Christensen et al. evaluated multiple cartilage repair strategies in the Göttingen minipig model and proved a clinical consistent repair process following different treatments in minipigs (Christensen et al. 2015). In an osteochondral defect repair study, the cartilage regeneration potential of human umbilical cord blood-derived mesenchymal stem cells with a hyaluronic acid hydrogel composite has been proven at 12 weeks postoperatively (Ha et al. 2015). These studies support the feasibility and practicability of the minipig model for cartilage repair studies with partial-thickness, fullthickness, chondral defects and osteochondral defects.

11.6.8 Horses

The horse is the largest animal model available for cartilage research. Large facilities for housing, special technical skill and equipment are required to manage horse studies. Furthermore, this animal model is expensive to purchase and ethical concerns may arise here as well. Compared with other animal models, the large joint dimension, articular cartilage and subchondral bone thickness and fully extended, upright stifle joints during gait reflect closely the human knee anatomy. Also, horses suffer from similar joint diseases as humans, e.g. osteochondritis dissecans, focal cartilage lesions or osteoarthritis, and, due to horse racing environment, cartilage surgery techniques are established in this translational animal model (Nixon et al. 1999).

Similar to humans, horse articular cartilage shows little intrinsic capability for self-repair. Surgically created defects averagely measure between 6 and 20 mm, while a size above 9 mm is considered critical. The cartilage thickness allows for the production of partial- or full-thickness chondral or osteochondral defects. It also permits cartilage repair studies in defects approximately the same size and depth of lesions seen in humans (McIlwraith et al. 2011). Thus, the horse model highly benefits the preclinical evaluation of the efficacy of novel technologies. For example, Frisbie et al. examined chondroprogenitor cells derived from autologous and allogeneic articular cartilage to treat articular cartilage defects in horse (Frisbie et al. 2015). The relative long life and athletic property of the horse enable it to evaluate cartilage repair strategies in chronic defects. Moreover, the availability of postoperative exercises allows for evaluation of repair under various rehabilitation protocols.

Due to the body weight and joint loading biomechanics of the horse, the cartilage will be subjected to greater loading than in humans, and post-operative protected weight-bearing is commonly difficult to ensure (Murray et al. 2001). To avoid the overloading of the treated defects, the relatively unloaded lateral femoral trochlea has been frequently recommended for creation of cartilage defects in the horse model. The tibiotalar joint of horses is anatomically equivalent to the ankle joint of humans (McIlwraith et al. 2011). Specific rehabilitation programmes can be implemented in training, and therefore horses are suitable for studies requiring specific rehabilitation protocols. Second-look arthroscopies may be performed, if longitudinal data is warranted. Also, gait analyses may be performed. The suggested assessment methods of cartilage repair remains similar to other large animal models.

11.7 Evaluation of Articular Cartilage Repair

A large variety of methods may be applied for the evaluation of osteochondral repair (Orth et al. 2015b). After euthanization, high-resolution photographs of the articular cartilage defects should be taken under standardized conditions allowing for a non-destructive macroscopic evaluation of cartilage repair (Goebel et al. 2012). As next step, μ MRI (Goebel et al. 2014) or μ CT (Orth et al. 2012a; Eldracher et al. 2014) may be advocated to non-destructively and accurately evaluate structures of the entire osteochondral unit, e.g. articular cartilage and repair tissue, or microstructural indices of the subchondral bone. The histological analysis is considered the gold standard when experimental osteochondral repair needs to be graded (Getgood et al. 2014). However, it involves decalcification and the irreversible destruction of the samples, and, unfortunately, only single layers can be evaluated, while with μ MRI or μ CT, the entire sample can be assessed. For the histological analysis, several elementary and comprehensive scoring systems are validated and suit well to quantify the structure of cartilaginous repair tissue (Orth et al. 2012b). Of note, when

evaluating osteochondral repair, only the Sellers and the Pineda score allow for an assessment of the osteochondral junction to the subchondral bone (Sellers et al. 1997; Pineda et al. 1992). Also immunohistochemical examinations for collagens types I or II are mandatory. Similarly, an evaluation by polarized light microscopy which is uncomplicated and reveals the pattern of collagen fibril orientation has to be included (Madry et al. 2013). Biochemical (Kiss et al. 2014) and molecular biological examinations of the repair tissue (Cucchiarini and Madry 2014) may complete the analysis of the osteochondral samples.

11.8 Outlook and Conclusions

The simultaneous evaluation of the osteochondral unit in the context of articular cartilage repair has been a focus in the recent years (Chen et al. 2009, 2011; Hoemann et al. 2012; Madry 2010; Goebel et al. 2012; Orth et al. 2012a, 2013a, b, 2015a). Specific changes of the subchondral bone such as the upward migration of the subchondral bone plate, intralesional osteophytes, subchondral bone cysts and a generalized impairment of the osseous microarchitecture below the defect may affect articular cartilage repair both in animal models and in patients (Niemeyer et al. 2015; Bert 2015; Sansone et al. 2015; Orth et al. 2013a; Cole et al. 2011; Vasiliadis et al. 2010; Brown et al. 2004; Kreuz et al. 2006a; Mithoefer et al. 2005; Saris et al. 2009; Dhollander et al. 2011; Henderson and La Valette 2005).

Yet, the ever increasing complexity of available reconstructive surgical approaches for articular cartilage repair requires precise animal models to test and to translate new surgical techniques into appropriate clinical treatments. They will also provide better insights into the basic science of articular cartilage repair (Madry et al. 2014).

References

- Ahern BJ, Parvizi J, Boston R, Schaer TP (2009) Preclinical animal models in single site cartilage defect testing: a systematic review. Osteoarthritis Cartilage 17(6):705–713. doi:10.1016/j. joca.2008.11.008
- Aigner T, Cook JL, Gerwin N, Glasson SS, Laverty S, Little CB, McIlwraith W, Kraus VB (2010) Histopathology atlas of animal model systems – overview of guiding principles. Osteoarthritis Cartilage 18(Suppl 3):S2–S6. doi:10.1016/j.joca.2010.07.013
- Appleyard RC, Burkhardt D, Ghosh P, Read R, Cake M, Swain MV, Murrell GA (2003) Topographical analysis of the structural, biochemical and dynamic biomechanical properties of cartilage in an ovine model of osteoarthritis. Osteoarthritis Cartilage 11(1):65–77
- Armstrong SJ, Read RA, Price R (1995) Topographical variation within the articular cartilage and subchondral bone of the normal ovine knee joint: a histological approach. Osteoarthritis Cartilage 3(1):25–33
- Athanasiou KA, Rosenwasser MP, Buckwalter JA, Malinin TI, Mow VC (1991) Interspecies comparisons of in situ intrinsic mechanical properties of distal femoral cartilage. J Orthop Res 9(3):330–340. doi:10.1002/jor.1100090304
- Bert JM (2015) Abandoning microfracture of the knee: has the time come? Arthroscopy 31(3):501– 505. doi:10.1016/j.arthro.2014.12.018

- Blaney Davidson EN, van de Loo FA, van den Berg WB, van der Kraan PM (2014) How to build an inducible cartilage-specific transgenic mouse. Arthritis Res Ther 16(3):210. doi:10.1186/ ar4573
- Brown WE, Potter HG, Marx RG, Wickiewicz TL, Warren RF (2004) Magnetic resonance imaging appearance of cartilage repair in the knee. Clin Orthop Relat Res (422):214–223. pii:00003086-200405000-00036
- Chen H, Sun J, Hoemann CD, Lascau-Coman V, Ouyang W, McKee MD, Shive MS, Buschmann MD (2009) Drilling and microfracture lead to different bone structure and necrosis during bone-marrow stimulation for cartilage repair. J Orthop Res 27(11):1432–1438. doi:10.1002/ jor.20905
- Chen H, Chevrier A, Hoemann CD, Sun J, Ouyang W, Buschmann MD (2011) Characterization of subchondral bone repair for marrow-stimulated chondral defects and its relationship to articular cartilage resurfacing. Am J Sports Med 39(8):1731–1740. doi:10.1177/0363546511403282
- Chevrier A, Kouao AS, Picard G, Hurtig MB, Buschmann MD (2015) Interspecies comparison of subchondral bone properties important for cartilage repair. J Orthop Res 33(1):63–70. doi:10.1002/jor.22740
- Choi B, Kim S, Fan J, Kowalski T, Petrigliano F, Evseenko D, Lee M (2015) Covalently conjugated transforming growth factor-beta1 in modular chitosan hydrogels for the effective treatment of articular cartilage defects. Biomater Sci 3(5):742–752. doi:10.1039/c4bm00431k
- Christensen BB, Foldager CB, Olesen ML, Vingtoft L, Rölfing JHD, Ringgaard S, Lind M (2015) Experimental articular cartilage repair in the Göttingen minipig: the influence of multiple defects per knee. J Exp Orthop 2:13. doi:10.1186/s40634-015-0031-3
- Cole BJ, Farr J, Winalski CS, Hosea T, Richmond J, Mandelbaum B, De Deyne PG (2011) Outcomes after a single-stage procedure for cell-based cartilage repair: a prospective clinical safety trial with 2-year follow-up. Am J Sports Med 39(6):1170–1179. doi:10.1177/0363546511399382
- Cook JL, Kuroki K, Visco D, Pelletier JP, Schulz L, Lafeber FP (2010) The OARSI histopathology initiative – recommendations for histological assessments of osteoarthritis in the dog. Osteoarthritis Cartilage 18(Suppl 3):S66–S79. doi:10.1016/j.joca.2010.04.017
- Cucchiarini M, Madry H (2014) Overexpression of human IGF-I via direct rAAV-mediated gene transfer improves the early repair of articular cartilage defects in vivo. Gene Ther 21(9):811– 819. doi:10.1038/gt.2014.58
- Cucchiarini M, Madry H, Guilak F, Saris DB, Stoddart MJ, Koon Wong M, Roughley P (2014) A vision on the future of articular cartilage repair. Eur Cell Mater 27:12–16
- Curl WW, Krome J, Gordon ES, Rushing J, Smith BP, Poehling GG (1997) Cartilage injuries: a review of 31,516 knee arthroscopies. Arthroscopy 13(4):456–460
- Custers RJ, Dhert WJ, Saris DB, Verbout AJ, van Rijen MH, Mastbergen SC, Lafeber FP, Creemers LB (2009) Cartilage degeneration in the goat knee caused by treating localized cartilage defects with metal implants. Osteoarthritis Cartilage 23:23
- Dhollander AA, De Neve F, Almqvist KF, Verdonk R, Lambrecht S, Elewaut D, Verbruggen G, Verdonk PC (2011) Autologous matrix-induced chondrogenesis combined with platelet-rich plasma gel: technical description and a five pilot patients report. Knee Surg Sports Traumatol Arthrosc 19(4):536–542. doi:10.1007/s00167-010-1337-4
- Drobnic M, Radosavljevic D, Cor A, Brittberg M, Strazar K (2010) Debridement of cartilage lesions before autologous chondrocyte implantation by open or transarthroscopic techniques: a comparative study using post-mortem materials. J Bone Joint Surg Br 92(4):602–608. doi:10.1302/0301-620X.92B3.22558
- Eldracher M, Orth P, Cucchiarini M, Pape D, Madry H (2014) Small subchondral drill holes improve marrow stimulation of articular cartilage defects. Am J Sports Med 42(11):2741– 2750. doi:10.1177/0363546514547029
- Frisbie DD, Trotter GW, Powers BE, Rodkey WG, Steadman JR, Howard RD, Park RD, McIlwraith CW (1999) Arthroscopic subchondral bone plate microfracture technique augments healing of large chondral defects in the radial carpal bone and medial femoral condyle of horses. Vet Surg 28(4):242–255

- Frisbie DD, Oxford JT, Southwood L, Trotter GW, Rodkey WG, Steadman JR, Goodnight JL, McIlwraith CW (2003) Early events in cartilage repair after subchondral bone microfracture. Clin Orthop Relat Res 407:215–227
- Frisbie DD, Cross MW, McIlwraith CW (2006a) A comparative study of articular cartilage thickness in the stifle of animal species used in human pre-clinical studies compared to articular cartilage thickness in the human knee. Vet Comp Orthop Traumatol 19(3):142–146
- Frisbie DD, Morisset S, Ho CP, Rodkey WG, Steadman JR, McIlwraith CW (2006b) Effects of calcified cartilage on healing of chondral defects treated with microfracture in horses. Am J Sports Med 34(11):1824–1831. Epub 2006 Jul 1810
- Frisbie DD, McCarthy HE, Archer CW, Barrett MF, McIlwraith CW (2015) Evaluation of articular cartilage progenitor cells for the repair of articular defects in an equine model. J Bone Joint Surg Am 97(6):484–493. doi:10.2106/JBJS.N.00404
- Gerwin N, Bendele AM, Glasson S, Carlson CS (2010) The OARSI histopathology initiative recommendations for histological assessments of osteoarthritis in the rat. Osteoarthritis Cartilage 18(Suppl 3):S24–S34. doi:10.1016/j.joca.2010.05.030
- Getgood A, Henson F, Skelton C, Brooks R, Guehring H, Fortier L, Rushton N (2014) Osteochondral tissue engineering using a biphasic collagen/GAG scaffold containing rhFGF18 or BMP-7 in an ovine model. J Exp Orthop 1(13). doi:10.1186/s40634-014-0013-x
- Glasson SS, Chambers MG, Van Den Berg WB, Little CB (2010) The OARSI histopathology initiative – recommendations for histological assessments of osteoarthritis in the mouse. Osteoarthritis Cartilage 18(Suppl 3):S17–S23. doi:10.1016/j.joca.2010.05.025
- Goebel L, Orth P, Muller A, Zurakowski D, Bucker A, Cucchiarini M, Pape D, Madry H (2012) Experimental scoring systems for macroscopic articular cartilage repair correlate with the MOCART score assessed by a high-field MRI at 9.4 T – comparative evaluation of five macroscopic scoring systems in a large animal cartilage defect model. Osteoarthritis Cartilage 20(9):1046–1055. doi:10.1016/j.joca.2012.05.010
- Goebel L, Zurakowski D, Muller A, Pape D, Cucchiarini M, Madry H (2014) 2D and 3D MOCART scoring systems assessed by 9.4 T high-field MRI correlate with elementary and complex histological scoring systems in a translational model of osteochondral repair. Osteoarthritis Cartilage 22(10):1386–1395. doi:10.1016/j.joca.2014.05.027
- Goebel L, Muller A, Bucker A, Madry H (2015) High resolution MRI imaging at 9.4 Tesla of the osteochondral unit in a translational model of articular cartilage repair. BMC Musculoskelet Disord 16(1):91. doi:10.1186/s12891-015-0543-0
- Gotterbarm T, Breusch SJ, Schneider U, Jung M (2008) The minipig model for experimental chondral and osteochondral defect repair in tissue engineering: retrospective analysis of 180 defects. Lab Anim 42(1):71–82. doi:10.1258/la.2007.06029e
- Ha CW, Park YB, Chung JY, Park YG (2015) Cartilage repair using composites of human umbilical cord blood-derived mesenchymal stem cells and hyaluronic acid hydrogel in a minipig model. Stem Cells Transl Med 4(9):1044–1051. doi:10.5966/sctm.2014-0264
- Henderson IJ, La Valette DP (2005) Subchondral bone overgrowth in the presence of full-thickness cartilage defects in the knee. Knee 12(6):435–440. doi:10.1016/j.knee.2005.04.003
- Hjelle K, Solheim E, Strand T, Muri R, Brittberg M (2002) Articular cartilage defects in 1,000 knee arthroscopies. Arthroscopy 18(7):730–734
- Hoemann CD, Lafantaisie-Favreau CH, Lascau-Coman V, Chen G, Guzman-Morales J (2012) The cartilage-bone interface. J Knee Surg 25(2):85–97
- Hunziker EB (1999a) Articular cartilage repair: are the intrinsic biological constraints undermining this process insuperable? Osteoarthritis Cartilage 7(1):15–28
- Hunziker EB (1999b) Biologic repair of articular cartilage. Defect models in experimental animals and matrix requirements. Clin Orthop (367 Suppl):S135–S146
- Hunziker EB (2000) Articular cartilage repair: problems and perspectives. Biorheology 37(1-2):163-164
- Hunziker EB (2002) Articular cartilage repair: basic science and clinical progress. A review of the current status and prospects. Osteoarthritis Cartilage 10(6):432–463. doi:10.1053/ joca.2002.0801

- Hunziker EB (2009) The elusive path to cartilage regeneration. Adv Mater 21(32-33):3419–3424. doi:10.1002/adma.200801957
- Hunziker EB, Rosenberg LC (1996) Repair of partial-thickness defects in articular cartilage: cell recruitment from the synovial membrane. J Bone Joint Surg Am 78(5):721–733
- Hunziker EB, Kapfinger E, Geiss J (2007) The structural architecture of adult mammalian articular cartilage evolves by a synchronized process of tissue resorption and neoformation during postnatal development. Osteoarthritis Cartilage 15(4):403–413. doi:10.1016/j.joca.2006.09.010
- Hunziker EB, Lippuner K, Keel MJ, Shintani N (2015) An educational review of cartilage repair: precepts & practice--myths & misconceptions--progress & prospects. Osteoarthritis Cartilage 23(3):334–350. doi:10.1016/j.joca.2014.12.011
- Jackson DW, Lalor PA, Aberman HM, Simon TM (2001) Spontaneous repair of full-thickness defects of articular cartilage in a goat model. A preliminary study. J Bone Joint Surg Am 83-A(1):53–64
- Jiang CC, Chiang H, Liao CJ, Lin YJ, Kuo TF, Shieh CS, Huang YY, Tuan RS (2007) Repair of porcine articular cartilage defect with a biphasic osteochondral composite. J Orthop Res 25(10):1277–1290. doi:10.1002/jor.20442
- Johnson LL (2001) Arthroscopic abrasion arthroplasty: a review. Clin Orthop (391 Suppl):S306–S317
- Kaab MJ, Gwynn IA, Notzli HP (1998) Collagen fibre arrangement in the tibial plateau articular cartilage of man and other mammalian species. J Anat 193(Pt 1):23–34
- Kaweblum M, Aguilar MC, Blancas E, Kaweblum J, Lehman WB, Grant AD, Strongwater AM (1994) Histological and radiographic determination of the age of physeal closure of the distal femur, proximal tibia, and proximal fibula of the New Zealand white rabbit. J Orthop Res 12(5):747–749
- Kiss A, Cucchiarini M, Menger MD, Kohn D, Hannig M, Madry H (2014) Enamel matrix derivative inhibits proteoglycan production and articular cartilage repair, delays the restoration of the subchondral bone and induces changes of the synovial membrane in a lapine osteochondral defect model in vivo. J Tissue Eng Regen Med 8(1):41–49. doi:10.1002/term.1495
- Kon E, Filardo G, Shani J, Altschuler N, Levy A, Zaslav K, Eisman JE, Robinson D (2015) Osteochondral regeneration with a novel aragonite-hyaluronate biphasic scaffold: up to 12-month follow-up study in a goat model. J Orthop Surg Res 10:81. doi:10.1186/ s13018-015-0211-y
- Kraus VB, Huebner JL, DeGroot J, Bendele A (2010) The OARSI histopathology initiative recommendations for histological assessments of osteoarthritis in the guinea pig. Osteoarthritis Cartilage 18(Suppl 3):S35–S52. doi:10.1016/j.joca.2010.04.015
- Kreuz PC, Erggelet C, Steinwachs MR, Krause SJ, Lahm A, Niemeyer P, Ghanem N, Uhl M, Sudkamp N (2006a) Is microfracture of chondral defects in the knee associated with different results in patients aged 40 years or younger? Arthroscopy 22(11):1180–1186
- Kreuz PC, Steinwachs MR, Erggelet C, Krause SJ, Konrad G, Uhl M, Sudkamp N (2006b) Results after microfracture of full-thickness chondral defects in different compartments in the knee. Osteoarthritis Cartilage 14(11):1119–1125 Epub 2006 Jul 11
- Laverty S, Girard CA, Williams JM, Hunziker EB, Pritzker KP (2010) The OARSI histopathology initiative – recommendations for histological assessments of osteoarthritis in the rabbit. Osteoarthritis Cartilage 18(Suppl 3):S53–S65. doi:10.1016/j.joca.2010.05.029
- Little CB, Zaki S (2012) What constitutes an "animal model of osteoarthritis"--the need for consensus? Osteoarthritis Cartilage 20(4):261–267. doi:10.1016/j.joca.2012.01.017
- Little CB, Ghosh P, Bellenger CR (1996) Topographic variation in biglycan and decorin synthesis by articular cartilage in the early stages of osteoarthritis: an experimental study in sheep. J Orthop Res 14(3):433–444. doi:10.1002/jor.1100140314
- Little CB, Smith MM, Cake MA, Read RA, Murphy MJ, Barry FP (2010) The OARSI histopathology initiative – recommendations for histological assessments of osteoarthritis in sheep and goats. Osteoarthritis Cartilage 18(Suppl 3):S80–S92. doi:10.1016/j.joca.2010.04.016
- Madry H (2010) The subchondral bone: a new frontier in articular cartilage repair. Knee Surg Sports Traumatol Arthrosc 18(4):417–418. doi:10.1007/s00167-010-1071-y

- Madry H, van Dijk CN, Mueller-Gerbl M (2010) The basic science of the subchondral bone. Knee Surg Sports Traumatol Arthrosc 18(4):419–433. doi:10.1007/s00167-010-1054-z
- Madry H, Grun UW, Knutsen G (2011) Cartilage repair and joint preservation: medical and surgical treatment options. Dtsch Arztebl Int 108(40):669–677. doi:10.3238/arztebl.2011.0669
- Madry H, Kaul G, Zurakowski D, Vunjak-Novakovic G, Cucchiarini M (2013) Cartilage constructs engineered from chondrocytes overexpressing IGF-I improve the repair of osteochondral defects in a rabbit model. Eur Cell Mater 25:229–247
- Madry H, Alini M, Stoddart MJ, Evans C, Miclau T, Steiner S (2014) Barriers and strategies for the clinical translation of advanced orthopaedic tissue engineering protocols. Eur Cell Mater 27:17–21. discussion 21
- Madry H, Ochi M, Cucchiarini M, Pape D, Seil R (2015) Large animal models in experimental knee sports surgery: focus on clinical translation. J Exp Orthop 2:9. doi:10.1186/ s40634-015-0025-1
- Mak J, Leonard C, Foniok T, Rushforth D, Dunn JF, Krawetz R (2015) Evaluating endogenous repair of focal cartilage defects in C57BL/6 and MRL/MpJ mice using 9.4T magnetic resonance imaging: a pilot study. Magn Reson Imaging 33(5):690–694. doi:10.1016/j. mri.2015.01.001
- Matsuoka M, Onodera T, Sasazawa F, Momma D, Baba R, Hontani K, Iwasaki N (2015) An articular cartilage repair model in common C57Bl/6 mice. Tissue Eng Part C Methods 21(8):767–772. doi:10.1089/ten.TEC.2014.0440
- McIlwraith CW, Frisbie DD, Kawcak CE, Fuller CJ, Hurtig M, Cruz A (2010) The OARSI histopathology initiative – recommendations for histological assessments of osteoarthritis in the horse. Osteoarthritis Cartilage 18(Suppl 3):S93–105. doi:10.1016/j.joca.2010.05.031
- McIlwraith CW, Fortier LA, Frisbie DD, Nixon AJ (2011) Equine models of articular cartilage repair. Cartilage 2(4):317–326. doi:10.1177/1947603511406531
- Mithoefer K, Williams RJ 3rd, Warren RF, Potter HG, Spock CR, Jones EC, Wickiewicz TL, Marx RG (2005) The microfracture technique for the treatment of articular cartilage lesions in the knee. A prospective cohort study. J Bone Joint Surg Am 87(9):1911–1920
- Murray RC, Vedi S, Birch HL, Lakhani KH, Goodship AE (2001) Subchondral bone thickness, hardness and remodelling are influenced by short-term exercise in a site-specific manner. J Orthop Res 19(6):1035–1042. doi:10.1016/S0736-0266(01)00027-4
- Niemeyer P, Uhl M, Salzmann GM, Morscheid YP, Sudkamp NP, Madry H (2015) Evaluation and analysis of graft hypertrophy by means of arthroscopy, biochemical MRI and osteochondral biopsies in a patient following autologous chondrocyte implantation for treatment of a fullthickness-cartilage defect of the knee. Arch Orthop Trauma Surg. doi:10.1007/ s00402-015-2194-x
- Nixon AJ, Fortier LA, Williams J, Mohammed H (1999) Enhanced repair of extensive articular defects by insulin-like growth factor-I-laden fibrin composites. J Orthop Res 17(4):475–487
- Orth P, Madry H (2013) A low morbidity surgical approach to the sheep femoral trochlea. BMC Musculoskelet Disord 14:5. doi:10.1186/1471-2474-14-5
- Orth P, Goebel L, Wolfram U, Ong MF, Graber S, Kohn D, Cucchiarini M, Ignatius A, Pape D, Madry H (2012a) Effect of subchondral drilling on the microarchitecture of subchondral bone: analysis in a large animal model at 6 months. Am J Sports Med 40(4):828–836. doi:10.1177/0363546511430376
- Orth P, Zurakowski D, Wincheringer D, Madry H (2012b) Reliability, reproducibility, and validation of five major histological scoring systems for experimental articular cartilage repair in the rabbit model. Tissue Eng Part C Methods 18(5):329–339. doi:10.1089/ten.TEC.2011.0462
- Orth P, Cucchiarini M, Kohn D, Madry H (2013a) Alterations of the subchondral bone in osteochondral repair--translational data and clinical evidence. Eur Cell Mater 25:299–316. discussion 314-296
- Orth P, Cucchiarini M, Zurakowski D, Menger MD, Kohn DM, Madry H (2013b) Parathyroid hormone [1-34] improves articular cartilage surface architecture and integration and subchondral bone reconstitution in osteochondral defects in vivo. Osteoarthritis Cartilage 21(4):614– 624. doi:10.1016/j.joca.2013.01.008

- Orth P, Meyer HL, Goebel L, Eldracher M, Ong MF, Cucchiarini M, Madry H (2013c) Improved repair of chondral and osteochondral defects in the ovine trochlea compared with the medial condyle. J Orthop Res 31(11):1772–1779. doi:10.1002/jor.22418
- Orth P, Duffner J, Zurakowski D, Cucchiarini M, Madry H (2015a) Small diameter awls improve articular cartilage repair following microfracture treatment in a translational animal model. Am J Sports Med. doi:10.1177/0363546514547029
- Orth P, Peifer C, Goebel L, Cucchiarini M, Madry H (2015b) Comprehensive analysis of translational osteochondral repair: Focus on the histological assessment. Prog Histochem Cytochem 50(3):19–36. doi:10.1016/j.proghi.2015.10.001
- Osterhoff G, Loffler S, Steinke H, Feja C, Josten C, Hepp P (2011) Comparative anatomical measurements of osseous structures in the ovine and human knee. Knee 18(2):98–103. doi:10.1016/j. knee.2010.02.001
- Pagnotto MR, Wang Z, Karpie JC, Ferretti M, Xiao X, Chu CR (2007) Adeno-associated viral gene transfer of transforming growth factor-beta1 to human mesenchymal stem cells improves cartilage repair. Gene Ther 14(10):804–813. Epub 2007 Mar 2008
- Pape D, Madry H (2013) The preclinical sheep model of high tibial osteotomy relating basic science to the clinics: standards, techniques and pitfalls. Knee Surg Sports Traumatol Arthrosc 21(1):228–236. doi:10.1007/s00167-012-2135-y
- Pineda S, Pollack A, Stevenson S, Goldberg V, Caplan A (1992) A semiquantitative scale for histologic grading of articular cartilage repair. Acta Anat (Basel) 143(4):335–340
- Poole R, Blake S, Buschmann M, Goldring S, Laverty S, Lockwood S, Matyas J, McDougall J, Pritzker K, Rudolphi K, van den Berg W, Yaksh T (2010) Recommendations for the use of preclinical models in the study and treatment of osteoarthritis. Osteoarthritis Cartilage 18(Suppl 3):S10–S16. doi:10.1016/j.joca.2010.05.027
- Pridie KH (1959) A method of resurfacing osteoarthritic knee joints. Proc Br Orthop Assoc J Bone Joint Surg Br 41:618–619
- Pritzker KP, Gay S, Jimenez SA, Ostergaard K, Pelletier JP, Revell PA, Salter D, van den Berg WB (2006) Osteoarthritis cartilage histopathology: grading and staging. Osteoarthritis Cartilage 14(1):13–29. doi:10.1016/j.joca.2005.07.014
- Rudert M, Moller HD, Schulze M, Wirth CJ (2000) Tissue engineering for therapy of osteochondral cartilage lesions. Zentralbl Chir 125(6):509–515
- Sansone V, de Girolamo L, Pascale W, Melato M, Pascale V (2015) Long-term results of abrasion arthroplasty for full-thickness cartilage lesions of the medial femoral condyle. Arthroscopy 31(3):396–403. doi:10.1016/j.arthro.2014.10.007
- Saris DB, Vanlauwe J, Victor J, Almqvist KF, Verdonk R, Bellemans J, Luyten FP (2009) Treatment of symptomatic cartilage defects of the knee: characterized chondrocyte implantation results in better clinical outcome at 36 months in a randomized trial compared to microfracture. Am J Sports Med 37(Suppl 1):10S–19S. doi:10.1177/0363546509350694
- Schinhan M, Gruber M, Vavken P, Dorotka R, Samouh L, Chiari C, Gruebl-Barabas R, Nehrer S (2012) Critical-size defect induces unicompartmental osteoarthritis in a stable ovine knee. J Orthop Res 30(2):214–220. doi:10.1002/jor.21521
- Sellers RS, Peluso D, Morris EA (1997) The effect of recombinant human bone morphogenetic protein-2 (rhBMP-2) on the healing of full-thickness defects of articular cartilage. J Bone Joint Surg Am 79(10):1452–1463
- Shapiro F, Koide S, Glimcher MJ (1993) Cell origin and differentiation in the repair of fullthickness defects of articular cartilage. J Bone Joint Surg Am 75(4):532–553
- Simon WH (1971) Scale effects in animal joints. II Thickness and elasticity in the deformability of articular cartilage. Arthritis Rheum 14(4):493–502
- Smillie IS (1957) Treatment of osteochondritis dissecans. J Bone Joint Surg Br 39(2):248-260
- Steadman JR, Rodkey WG, Rodrigo JJ (2001) Microfracture: surgical technique and rehabilitation to treat chondral defects. Clin Orthop (391 Suppl):S362–S369
- Vasiliadis HS, Danielson B, Ljungberg M, McKeon B, Lindahl A, Peterson L (2010) Autologous chondrocyte implantation in cartilage lesions of the knee: long-term evaluation with magnetic resonance imaging and delayed gadolinium-enhanced magnetic resonance imaging technique. Am J Sports Med 38(5):943–949. doi:10.1177/0363546509358266